

STRATHCLYDE BUSINESS SCHOOL DEPARTMENT OF MARKETING

Framing Practices, Spaces and Socialities of Consumption in Vintage

Katherine M. Duffy Submitted for the Degree of Doctor of Philosophy July 2014

Word Count: 78,949

Declaration and Copyright

This thesis is the result of the author's original research. It has been composed by the author and has not been previously submitted for examination, which has led to the award of a degree. The copyright of this thesis belongs to the author under the terms of the United Kingdom Copyright Acts as qualified by University of Strathclyde Regulation 3.50. Due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.

Signed: Katherine M. Duffy

Date: 22nd July 2014

Publications and Conference Papers

Some parts of this thesis have been published and presented at academic conferences during the development of this research:

Duffy K., Hewer P. (2013) The Vintagescape as Embodied and Practiced Space, *Advances in Consumer Research*. Volume 40.

Duffy K., Hewer P., Wilson J. (2012) Granny Would be Proud: On the Labours of Doing Vintage, Practices and Emergent Socialities, *Advances in Consumer Research*. Volume 39.

Conference papers:

Duffy K. (2014) 'Audience Required' - The Marketplace Performance of Vintage, *GAMMA Global Fashion Management Conference*. London, UK.

Duffy K. (2012a) The Vintage Fashion Turn, *Fashioning Management Workshop*. Seminar Paper, London College of Fashion, UK.

Duffy K. (2012b) Practices of Vintage and Emergent Nostalgias, *Academy of Marketing - Doctoral Colloquium*. Conference Paper, University of Southampton, UK.

Duffy K. (2012c) Framing Value in the Vintage Marketplace, *2nd Interdisciplinary Market Studies Workshop*. Seminar Paper, UCD Dublin, Ireland.

Duffy K. (2012d) Performing Vintage, Poster Presentation, 7th Consumer Culture Theory Conference. Saïd Business School, Oxford University, UK.

Acknowledgements

Firstly I would like to thank my supervisory team Dr Paul Hewer and Dr Juliette Wilson for their continued support, motivation and words of encouragement throughout this study – Paul for your continued intellectual guidance and Juliette for all your support. I thank you both for your guidance, insight and overall understanding.

Thank you also to my friends who I made during this process. My PhD friends at the University of Strathclyde have helped, supported and advised me throughout this undertaking.

For my family and friends, especially my parents as without their understanding, encouragement and unwavering support, I could not have completed this process.

Finally I would like to thank my participants who were so generous with their time, stories and continual cups of tea.

Katherine Duffy

July 2014

Abstract

This thesis explores how practices, spaces and socialities intersect in the consumption of vintage. The study is based within Consumer Culture Theory (CCT) with a particular focus on studies that consider alternative spaces and practices of consumption. Vintage has become a widespread movement in the UK that encompasses production, marketing and consumption, but it has largely been omitted from the consumer research discussions to date. As a site of enquiry this ethnographic study focuses within the vintage 'pop-up markets' of Glasgow, UK and considers the vintage scene as an embodied set of social practices and values.

This thesis addresses four research objectives detailing the 'doings' of vintage from a practice perspective and extends the CCT understanding of the 'circuit of practice' to be spatially bound. The markets exist as marginal sites of consumption and were found to be active spaces that were an amalgamation of market objects, actors and values that encoded particular practices.

The findings of this study contribute to CCT in demonstrating how objects, practices and knowledge intersected in the marketspace in the performance of the alternative. This research contributes to an understanding of the consumer as an active practitioner in the marketspace and the aesthetic labour this necessitates. Further it finds the vintage tribe as a collective consumption site of learning and apprenticeship. The findings advance consumer research understandings of space with the pop-up markets as a practiced space of 'the familiar'. The final contribution frames the *'fluidity of value'* in vintage as liminal and transient in practice. The study concludes by emphasising the importance of a turn to a practice perspective methodologically within the CCT landscape and the development of future research opportunities that probe market labour, formation and creativity in practice.

Contents

Declaration and Copyrighti
Publications and Conference Papersii
Acknowledgements iii
Abstract iv
List of Figuresxi
List of Tablesxi
Chapter One: Introduction to the Research
1.0 Introduction
1.1 Theoretical Tradition of the Research1
1.2 Research Context
1.2.1 Vintage in Contemporary Culture
1.2.2 Historical Perspective
1.2.3 Vintage in the UK
1.3 Research Aim and Contribution9
1.4 Research Objectives and Methodology9
1.5 Outline of Thesis
Chapter Two: The Consumption Landscape 12
2.0 Introduction

2.0 Introduction	12
2.1 Defining Consumption	12
2.1.1 The History of Consumption	12
2.1.2 The Consumer Research Turn	14
2.2 The Development of Consumer Culture Theory (CCT)	15
2.2.1 CCT background	15
2.2.2 Narrowing of the CCT Focus	18
2.3 The Construct of the Consumer	19
2.4 Consumption Communities	22
2.4.1 Subcultures of Consumption	23

2.4.2 Brand Communities	26
2.4.3 Tribes	28
2.5 Alternative Marketplaces within Consumer Research	33
2.6 The Alternative in Vintage	36
2.7 Fashion Context	43
2.7.1 The Fashion Object	46
2.7.2 Fashion Subcultures	50
2.8 Consumption and Materiality	53
2.8.1 The Potential of Objects	56
2.9 Conclusions	59
2.9.1 Positioning this research and Key Literature Gaps	59

Chapter Three: Practices and Value
3.0 Introduction
3.1. Defining Practice Theory
3.2 Developing Practice Theory
3.3 Consumer Research Moves to Practice
3.3.1 Implications for Consumption Studies
3.3.2 The Consumer as the Agent of Practice
3.4 Practice Perspectives
3.4.1 Practice and the Material
3.4.2 Practice and Space
3.4.3 Practice as Markets
3.4.4 Practice from an Organisational Lens
3.5 Practices of the Collective
3.6 Value in Practice
3.6.1 Marketing and Value
3.6.2 CCT and Value
3.6.3 Disposal Practices and Value
3.6.4 Contemporary Considerations of Value within CCT 102
3.7 Limitations of Practice Theory
3.8 Conclusion
3.8.1 Positioning This Research and Key Literature Gaps 104

3.8.2 Applying Practice Theory	106
Chapter Four: Methodology	109
4.0 Introduction	109
4.1 Research Aim	110
4.2 Research Strategy	
4.3 Research Philosophy	111
4.3.1 Ontological Position	112
4.3.2 Interpretivism within Consumer Research	113
4.3.3 Consumer Culture Theory and Ontological Positioning	117
4.4 Epistemology	123
4.4.1 Ethnography	123
4.4.2 Criticisms of Ethnography	128
4.5 The Practice Theory Lens	131
4.6 Research Objectives	136
4.7 Data Collection	136
4.7.1 The Research Context	138
4.7.2 The Wider Context - Glasgow	138
4.7.3 Glasgow's Vintage Scene	138
4.7.4 Sites of Enquiry	139
4.8 Theoretical Sampling	141
4.8.2 Observation and Field Notes	144
4.8.3 Interviewing	148
4.8.4 Photographic Data	151
4.8.5 Researcher Diary and Reflexivity	153
4.9 Overview of Data Analysis	157
4.10 Methodological Limitations	159
4.11 Conclusion	160
Chapter 5: Findings – The Practice, The Space and The Socialities	
5.1 Introduction	
5.2 The Practice	163
5.2.1 The Practice: The Role of the Practitioner	163

5.2.2 The Practice: Conceptualising Vintage	171
5.2.3 The Practice: The Knowledgeable Practitioner	180
5.2.4 The Practice: The Materiality	192
5.2.5 Conclusion of 'Practice'	199
5.3 The Space	201
5.3.1 Introduction	201
5.3.2 The Space: The Market Setting	201
5.3.3 The Space: The Marketspace Performance	215
5.3.4 The Space: Performing Politics	219
5.3.5 The Space: Inherent Instability of Vintage	224
5.3.6 Conclusion of 'Space'	229
5.4 The Socialities	233
5.4.1 Introduction	233
5.4.2 Socialities: The Tribal Nature of the Collective	234
5.4.3 Socialities: Collective Hierarchy	238
5.4.4 Socialities: Collective Creativity	246
5.4.5 Socialities: Sociality of Exchange	250
5.4.6 Socialities: Virtual Community Spaces	259
5.4.7 Conclusion of 'Socialities'	263

268
268
269
271
272
274
277
280
283
283
283
284
285

6.5 Conclusion	285
References	
Appendix i - Sample Depth interview	

List of Figures

Figure 1 - Topshop Oxford St: the Vintage Floor	7
Figure 2 - Alice's store: photograph	176
Figure 3 - Judy's Vintage Fair 'Oh Happy Days' stall	188
Figure 4 – The Balcony setting of Granny Would be Proud market	203
Figure 5 - Judy's Vintage Fair: 'Practice of browsing'	206
Figure 6 - Judy's Vintage Fair - 'Practice of Organising' in the marketspace	207
Figure 7 - Granny Would be Proud: 'Practice of display'	207
Figure 8 - Stall table being dressed in Granny Would be Proud: 'Annie's bijoux box'	210
Figure 9 - Little Birds Market 'Practices of disjumbling'	213
Figure 10 - Granny Would be Proud 'Practices of disjumbling'	213
Figure 11 - Judy's Vintage Market: 'Practices of disjumbling'	214
Figure 12 - Michael 'The Eyedresser' at Granny Would be Proud	217
Figure 13 - Facebook screenshot of 'Granny Would be Proud' market page	261
Figure 14 - Practice bound by space	273

List of Tables

Table 1 – Role of Consumer	78
Table 2 – 'Summary of Research Approaches'	112
Table 3 - Types of qualitative inquiry	114
Table 4 - Overview of analytic framework for interpretive consumer research	117
Table 5 - Characteristics of Ethnographic Studies	124
Table 6 - The role of ethnographic research within consumer research	126
Table 7 – Overview of the ethnography and data collection	137
Table 8 – Overview of in-depth interview respondents	142
Table 9 – Recording with field notes	145

Chapter One: Introduction to the Research

1.0 Introduction

This thesis is primarily concerned with gaining an understanding into the complex practices that consumers undertake in the marketplace to frame vintage and materialise it in practice. This research contributes to consumer research and to existing theories of alternative consumption spaces within Consumer Culture Theory (CCT) by exploring the practices, socialities and spaces of vintage consumption. The vintage collective will be explored through the use of practice theory that allows for the practice of vintage consumption to be placed at the centre of the research and for the development of a more active understanding of consumption. This chapter provides an introduction to the thesis and an overview of the progression of the discussion to follow. It firstly outlines the core theoretical tradition to which it contributes, Consumer Culture Theory (CCT) as outlined by Arnould and Thompson (2005, 2007) before moving on to establish the context for the research study. This section briefly defines the research aims, objectives and methodology employed in the study. Finally a brief summary of each of the chapters of the thesis is detailed.

1.1 Theoretical Tradition of the Research

Arnould and Thompson's (2005) seminal article in the *Journal of Consumer Research (JCR)*, proposes a theoretical domain, Consumer Culture Theory (CCT) which highlights the ways through which collectives and individuals employ images and objects in their everyday consumption practices to construct identity and to make sense of the contemporary consumption landscape. This thesis is underpinned by consumer research which has focused on *"the imbricated layers of cultural meaning* *that structure consumer actions in a given social context*" (Thompson and Troester, 2002: 550) and through this CCT focus, this study will consider the theoretical tradition of Consumer Culture Theory (CCT) and specifically "*the dynamic relationships between consumer actions, the marketplace and cultural meanings*"

(Arnould and Thompson, 2005: 868). Coming from the CCT approach in which markets are seen as social and cultural constructs, the marketplace is unpacked in this thesis with a focus on the collective effort needed in performing this market (Arnould and Thompson, 2005; Geiger et al., 2012). Ideas of community pervade CCT literature: from experience communities (Arnould and Price, 1993), subcultures of consumption (Schouten and McAlexander, 1995), brand communities (Muniz and O'Guinn, 2001), to virtual communities (Kozinets, 1999), these conceptions of community within the consumption context show consumers joining together to assert their unique preferences (Carroll and Ahuvia, 2006). Further within the CCT domain, value is considered a social and market activity and values are considered as multiple, contested and dynamic (Kjellberg and Helgesson, 2010). This research seeks to explore the vintage scene as an embodied set of social practices and values.

1.2 Research Context

The research context of this thesis is the vintage marketplace and the purpose of this section is to define this as an appropriate domain with which to meet the research aim. It begins by considering the contemporary cultural interest in vintage before moving on to consider the practice of vintage as an under-researched consumption domain particularly within the realms of Consumer Culture Theory. It considers that vintage is an appropriate context to explore consumer marketplace practices as it crosses the material culture, fashion theory and consumer culture boundaries.

1.2.1 Vintage in Contemporary Culture

More than a fashion culture, the vintage collective is seen to function as a microcommunity that has flourished all over the UK in the past five years. Awareness and interest for vintage clothing has increased dramatically since the early 1990's. This has been attributed to media attention (Sex and the City, Mad Men, Kirstie's Homemade Home, The Great Gatsby) and celebrity endorsement from a proliferation of female celebrities (Kate Moss, Sienna Miller, Alexa Chung). Modern vintage is held in the media as being in stark contrast to the perceived homogenised culture of mainstream fashion. Vintage in this respect has become a movement that encompasses production, marketing and consumption practices. The term has become overtly used and widely exposed as a practice in the marketplace. Jo-ann Fortune (2012) writing for The Guardian questions: "Is vintage clothing passé? Bargains are rare, traders are struggling and clothes from the 90s are considered period pieces." 'Vintage' semantically is no longer seen as the domain of the wine/guitar/car collector or jumble sale aficionado. The term vintage no longer pertains simply to these specialist industries, it has become popularised to the extent that it is now used to define any "old" goods, from jewellery to cars to homeware. The terminology has been adopted by the fashion world where it is used to define "a rare and authentic piece that represents the style of a particular couturier or era" (Gerval, 2008). An alternative definition, which seems to have gained acceptance in the marketplace, identifies clothing as vintage "when it is produced in the period between the 1920s and the 1980s" (Gerval, 2008). Clothing originating from before the 1920s is classified as antique, whilst clothing produced after the 1980s is not

considered to be vintage as yet, with the most recent produced clothing being labelled 'modern' or 'contemporary fashion' (Cornett, 2010). On eBay this potential ambiguity is rectified by categorisation of 'clothing pre-1980s' in the category of *Vintage Clothing and Accessories*' to separate out vintage pieces. The shift in semantics in describing this phenomenon of second-hand, retro, vintage, pre-loved, the Americanism of thrift¹, can be seen as representative of the momentum of this popular trend in the marketplace and representative of the constant search for new forms of value which had originally triggered the researcher's curiosity.

1.2.2 Historical Perspective

The origins of the vintage clothing trend stretch back to the mid-1960s when young people both in the UK and the USA began accessorising their outfits with antique clothes and military uniforms bought in antique markets and army surplus stores. The only notable historic influence on clothing prior to the 1960s was the Teddy Boys of the 1950s with their Edwardian style jackets. The boutique trends of the 1960s brought a change in attitude to dressing especially for the younger generations. This lack of understanding of the youth market in the high street led Mary Quant to design, make and sell her own clothes from her boutique Bazaar that she opened in 1955 on London's King's Road. Quant was seen as the pioneer of youth fashions in

¹ Merriam-Webster defines thrift as "a careful use of money so that it is not wasted" and thrift stores as "selling second-hand articles and especially clothes and is often run for charitable purposes." Thrifting is an American saying that is used in the media and contemporary culture and describes "when one visits several different thrift shops, second-hand shops, and vintage clothing stores in the hopes of buying several items of cheap and unusual clothing and other items. One usually does this with friends." (Urban Dictionary)

the 60s but it was boutiques such as BIBA and Granny Takes a Trip², both run by their designer owners/proprietors, who had first sold 'second-hand' clothing alongside their new collections. The vintage trend eased by the early 70s with the exception of some small subcultures such as Rockers and Teddy Boys, but rose to popularity again in the late 70s with the Mod revival. As the Mod revival became more purist in direction, its followers adopted the authentic 1960s Mod look and began to wear vintage 60s youth clothing to create an entire look in the same way as the Rockers and Teddy Boys had been doing before them.

Modern day vintage has become a global industry, from Decades in Los Angeles, Rellik in London to the Saint-Ouen markets of Paris. Walking down any high street in the UK, vintage as a branding and advertising strategy has been adopted by mainstream clothing retailers such as Topshop, Urban Outfitters, New Look to homewares in Cath Kidston, Laura Ashley and John Lewis with high-end fashion stores also embracing the heritage aspect of the trend such as Ralph Lauren, Burberry and Mulberry (Fashion houses, such as Louis Vuitton for its 2010-2011 winter collection started reproducing and reinterpreting vintage clothing or, such as Yves Saint Laurent in 2009, re-using vintage fabrics. These styles are generally referred to as *"vintage style"* or *"vintage inspired"*). One of the main outlets of vintage has been the Internet, which has made 'vintaging' possible regardless of location and opened

² The Granny Takes A Trip boutique opened in 1966 on the King's Road, London and was run by Victorian clothing collector Sheila Cohen, her boyfriend, Nigel Waymouth and his friend, John Pearse. The store originally was a place to sell antique clothing but it continued to evolve with their own new, unique designs and collaborations. Their clothes became a feature of Pink Floyd shows throughout the late 60s and early 1970s. The store became world renowned as part of "London the swinging city" (TIME Magazine) and had many famous patrons (Vintage Fashion Guild).

up prospective markets for sellers around the world. The internet has moved beyond eBay as the main vintage facilitator, with major retailers such as ASOS capitalising on the trend with their online market platform: tag line: *'The place where anyone, anywhere in the world, can sell fashion, to anyone, anywhere in the world. Boutiques are small businesses selling their own label, other brands or vintage collections, directly to you. And there are thousands of individuals just like you, selling their wardrobes to fund their next fashion fix*" (ASOS Marketplace). Upmarket, high-end fashions are also not excluded from vintage with specialist sites such as Vestiaire Collective stocking quality, used designer goods. With vintage wedding wear, millinery, jewellery and craft, vintage seems to be everywhere.

Within the retail setting, hierarchies of vintage consumption are created as a result of the expanded popularity of second-hand clothing. For example the Topshop Oxford Circus flagship store (London, UK) has created a vintage space in the basement floor of their premises – with pre-selected goods, the familiarity of Topshop merchandising, well lit fitting rooms and the ability to mix contemporary and vintage goods in a mediated, familiar setting, vintage is integrated into the heart of high street fashion and as a practice, made accessible. Figure 1 - Topshop Oxford St: the Vintage Floor, taken by researcher on June 24th 2012



Within this normal high street setting, the boundaries of old and new are easily permeated by consumers, the margins for errors of taste greatly reduced and vintage by association is depicted as the 'epitome of cool' (Cochrane, 2010) – in this mixture of contemporary style with the ability to mix in suitable vintage pieces. This familiar setting also serves to mitigate any potential feelings of anxiety by consumers towards the often unreliable and unpredictable vintage searching process. This example serves to illustrate the proliferation of vintage in the UK, yet as a concept it exists without strict definition.

1.2.3 Vintage in the UK

The discussion below illustrates that vintage, once a niche interest has become a more mainstream consumption choice in the UK. In line with the popularity of vintage that has swept the UK, niche vintage festivals and markets arrived on the scene. With a familiar format to popular UK music festivals, in 2010, the Hemmingway design team put on the largest vintage festival of its kind in the UK -

the events gained attention on a national scale and have since developed into annual events in slightly different formats each year (Vintage at Goodwood, Vintage at Southbank Centre, Vintage by Hemmingway, Vintage at London Fashion Weekend, Vintage 2013 and 2014 at the Merchant City Festival Glasgow). Similarly, The Twinwood Festival originally started in 2002, with a vintage lifestyle focus becoming more prominent in 2007. The festivals bring vintage to life and focus on the experience of vintage over the festival period. Within the vintage context, this thesis focuses on the 'pop-up' market format that replicates the festival atmosphere and transient model, but in the city centre. The 'pop-up' format has been used for events, markets, restaurants, galleries, clubs and ranges in size. As a model it has historical roots with artists subverting abandoned spaces (Cochrane, 2010). In contemporary use, the term refers to a space that allows a unique environment to be constructed for a limited time period. This format can be traced back to the retail model of Comme des Garcons who in 2004^3 set up a temporary retail outlet in a disused building in Berlin. Even with its counter-cultural roots, the format has now found mainstream success and continues to be utilised by big retail brands such as Innocent, eBay, Harrods and destinations such as Boxpark in London (Cochrane, 2010).

³ Comme des Garcons opened a guerrilla pop-up retail space in 2004 for one year in Berlin. It was a space that was not heavily designed or merchandised but a 'raw space:' "*The whole concept gives consumers something that can be perceived as exclusive, discovery-driven and get it while it lasts*" (Trend Watching.com, 2004). The trend of temporary retail spaces has continued to develop since this report in 2004 and has grown into what the retail industry named "*instant commerce*" (Trend Watching.com, 2009).

1.3 Research Aim and Contribution

Within the research context and theoretical origins presented above, this thesis aims to gain an understanding of how the practices, spaces and socialites intersect in the consumption of vintage. As detailed above, using the context of the 'pop-up' vintage market this study explores the practices, socialities and spaces of consumption. The findings of this thesis will contribute towards the body of work upon practice theory and alternative spaces of consumption within the Consumer Culture Theory tradition.

1.4 Research Objectives and Methodology

To achieve this aim stated above, four research objectives emerged and are developed throughout the study to guide and orientate the research direction.

The research objectives that structure the study are:

- 1. What practices are used in the marketspace to materialise vintage?
- 2. How do practices, objects and knowledge intersect in the marketspace?
- 3. Does vintage create a shared consumption space?
- 4. How is value ascribed and framed in vintage?

Using an in-depth ethnographic research design focuses upon consumer practices and allows for a rich data set to be captured over the two years of the research study, the findings from which are discussed in detail in the findings section in Chapter Five. Further detailed discussion of the justification for the primary research approach is presented in more depth in Chapter Four.

1.5 Outline of Thesis

This thesis is presented in six chapters. Having introduced the theoretical context, tradition and aims in this introductory chapter, Chapter Two begins the review of the consumer research tradition before more deeply focusing in on the CCT domain and previous studies around constructs of community, alternative marketplaces and the fashion theory understandings of vintage.

Chapter Three reviews the concept of practice theory, its origins and relationship to consumption studies. It focuses on the recent shift within CCT to a more practice orientated approach. The practices of value are also considered and the extant literature discussed.

Chapter Four discusses the methodological approach of this study and discusses in detail the ethnographic approach that underpins this research. It further presents the decisions made in the research design, considering the data collection, sampling and analysis.

Chapter Five presents the findings of the data collected in this study. Chapter Five focuses on the construct of the practitioner and the necessary skill needed in the marketplace as both consumer and seller. It also considers the materiality needed to practice and this discussion centres on the object of practice. It further discusses the collective of the marketplace and considers the spatial elements of practice. It discusses the theoretical implications and contributions of this thesis.

Chapter Six considers again the research aim and synthesises the contributions of this study. It discusses the implications of the findings and acknowledges the limitations

of the study before making recommendations for possible future research and concluding with reflections of the researcher.

Chapter Two: The Consumption Landscape

2.0 Introduction

This chapter introduces the theoretical context of the research. This chapter begins with an overview of the emergence of consumer culture as a research area within the broader discipline of consumer behaviour. The debates surrounding the consumer culture paradigm are highlighted. This is followed by a discussion of the consolidation of the Consumer Culture Theory (CCT) literature and the key themes that emerge. Building on the argument that the individual and their relationship with other consumers lie at the centre of consumer research (Holt, 2005; Holt and Thompson, 2004), this thesis explores the collective in their consumption practices in the vintage marketplace. Finally summaries of the key literature gaps that are addressed by this study are outlined and conclusions follow.

2.1 Defining Consumption

2.1.1 The History of Consumption

Consumption has a long and complex history. The term has been used throughout history to suggest both utilisation and a destruction of items. This pluralistic term has both negative and positive connotations. Aldrige (2003) suggests that it was originally a pejorative term that was synonymous with decaying diseases. This negative image has persisted throughout pre-industrial consumer history. Veblen's (1899) theory of *'conspicuous consumption'* proposes that consumerism; with the production of unnecessary goods and services was a result of material values generated by the capitalistic marketplace. Since the beginning of the 20th century consumption as we have come to understand it, developed in Western economies. Veblen (1899) identifies that the industrialisation of the late nineteenth century brought freedoms to the upper and middle classes. The following Fordism of the 1920s transforms this selective freedom to a mass phenomenon (Gabriel and Lang, 1995). Customers are altered into consumers in this era with the advent of branded goods, advertising and the more informal buyer-seller relationship (Strasser, 1989). The shift to 'post-Fordist' marks the start of a global marketplace. Gabriel and Lang (1995) propose that this heralds the emergence of post-modernity through which advertising and consumer discourses come to the foreground. Further they suggest this shifts focus from the collective material gain, to the individual. Tadajewski (2006) reinforces this and suggests that the way in which the consumer was researched also shifts and a greater focus on the experience of the consumer emerges and the notion of a consumer culture develops.

Consumption was traditionally held in stark contrast to production. In this view consumption was often considered only as a motivating force for continued production (Firat, 1999: 287). This modernist approach to consumption (Firat and Venkatesh, 1995) places consumption in opposition to production. In this respect, individuals were held as opportunity creators for production through their consumption. Firat and Dholakia (1998) consider that 'worth' was formulated from production and opportunities for sustained production. This functional view of consumption views production as *"the engine of society"* and that everything else is *"subservient or a consequence"* (Firat and Venkatesh, 1995: 242). Consumption, in this way was perceived as a 'using up' of resources and a 'waste' producer. The individual object was considered 'material' and 'real' as it satisfied consumer needs

and subsequently created value. This viewed the object's value as simply an ability to meet and satisfy consumer's real and material needs. In considering the role of consumers, Firat (1999: 290) asserts that this view rids them of their "*ability to produce life modes and experiences that break out of the vicious cycle that generates and regenerates the conditions that enchant lives.*" This modernist approach to consumption views consumption as a negative force that is the result of Western capitalist expansion.

2.1.2 The Consumer Research Turn

Consumption was seen as an important part of contemporary culture and an important means for sense making. It was a relatively neglected field in the 1980s as contemporary consumption theory was pluralistic, multiple and challenged the accepted traditions. Two of the most prolific academics within consumer research expressed in the late eighties a need for a focusing of the discipline: Holbrook (1987) describes a *"crisis of identity"* in consumer research in the late eighties and posed the ontological question *"What is consumer research?"* Belk concurs, *"My own vision is one of consumer behaviour as a discipline unto itself"* (1986: 423). Holbrook further argues that consumer research studies consumption (in all its various facets, including its potential breakdown) and should stand alone as a separate discipline. Holbrook (1987) argues that consumer research should embrace almost all-human activity from the view of consummation. He proposes that almost everything that humans do involved consumption:

"People get up in the morning, start consuming the moment their toes touch the carpet, allocate their time to various consumption activities throughout the day, and

continue consuming until they finally drift off to sleep at night, after which they confine their consumption mostly to dreams, pyjamas, and bed linens..." (Holbrook, 1985:146).

This highlights the importance of the act of consumption as it is seen to extend far beyond the mere act of market exchange. Holt (1995) further called for research that describes the variety of ways through which people consume to gain an understanding of how these differences alter across groups and how these groups are structured differently (1995:1). Holt's (1995) work built on Simmel's (1950) early work and began to frame consumption practices. Simmel's (1950) work has defined consumption as a type of social action whereby people use consumption objects in a variety of ways. This basic consumer action was termed *'consumption practices'*. Holt (1995) built on this language to develop a typology of consumption objects. He proposes that consumption should be categorised into: *consumption as experience, as integration, classification and as a play*. Holt's typology suggests that consumption has multiple forms and was not restricted to the object - as different consumers may consume it differently. The motivations and consumer interactions during these practices became a key focus of consumption theory.

2.2 The Development of Consumer Culture Theory (CCT)

2.2.1 CCT background

In Arnould and Thompson's (2005) seminal article in the *Journal of Consumer Research (JCR)*, they built upon Holt's typology and propose through their mapping of Consumer Culture Theory (CCT), a theoretical domain was created to bring together the main focus of these consumer researchers. Consumer Culture Theory (CCT) answers these calls for research that focuses more on *"the messy contextual details of consumer life"* (Holt, 1997:344). Their consumer culture highlights the ways through which collectives and individuals employed images and objects in their everyday consumption practices to construct identity and make sense of the contemporary consumption landscape.

Arnould and Thompson's (2005) seminal article in the JCR provided an amalgamation of twenty years of consumer research and explores how the CCT approach was firmly anchored in this tradition. Grounded in the work of early Baudrillard and his semiotic field (1968), Bourdieu (1984, 1985), De Certeau (1984a), Jameson (1984, 1985) Douglas and Isherwood (1980) and McCracken's (1986, 1988a) movement of meaning, Arnould and Thompson (2005) outline four main categories within consumer research that CCT focuses on: the socio-cultural, experiential, symbolic and ideological aspects of consumption rather than the production, consumption and exchange traditional approach. They illustrate the interrelatedness and theoretical linkages between these categories. Influenced by the work of Geertz (1973), Arnould and Thompson highlight the fluidity, fragmentation and complexity of consumer action. CCT they propose was based on the shared theoretical outlook that placed an importance on cultural complexity and "the dynamic relationships between consumer action, the marketplace and cultural meanings" (Arnould and Thompson, 2005: 868). Although not a unified theory with clear theoretical boundaries, the research stream encompasses similar theoretical interests and questions. CCT researchers share a common outlook towards cultural multiplicity and complexity, with a focus on: "the relationship between lived culture and social resources. CCT conceptualizes culture as the very fabric of experience,

meaning, and action" (Arnould and Thompson, 2005: 869). According to Arnould and Thompson (2005) the CCT research stream should cover studies of multiple domains: firstly consumption as a practice of ideological reproduction and resistance, secondly consumer identity projects, thirdly the socio-historic patterning of consumption and finally the study of marketplace cultures:

"CCT is an interdisciplinary research tradition that has advanced knowledge about consumer culture (in all its heterogeneous manifestations) and generated empirically grounded findings and theoretical innovations that are relevant to a broad constituency in the base social science disciplines, public policy arena and managerial sectors" (Arnould and Thompson, 2005: 869).

This thesis focuses on a contribution to two of the domains as outlined by Arnould and Thompson (2005): *marketplace cultures and the socio-historic patterning of consumption.* Firstly, marketplace cultures which focuses on consumers as culture producers. Secondly, the domain of the socio-historic patterning of consumption reflects Bourdieu's (1984, 1985) ideas of social capital through which individual choices are governed by cultural capital. With this focus of the enactment of vintage in the marketplace, Arnould and Thompson's socio-historic domain are pertinent in the discussion of how social positions and roles were enacted by consumers in the marketplace (Goulding and Saren, 2009). Arnould and Thompson (2005) propose that contributions to these domains would aid the placement and understanding of the researcher in the field. The boundaries between the domains are permeable and not fixed, this in turn suggests that the research could overlap domains and does not need to be rigidly fixed in simply one. This plurality is held as an important condition for the continued development of knowledge. Arnould and Thompson (2005)

acknowledge this fluidity of approach within CCT research that there is no single consumer culture theory as CCT acknowledges numerous theoretical perspectives and methodological approaches. Arnould and Thompson further acknowledge this plurality in 2007 in *Research in Consumer Behaviour* in which they explain that with hindsight a more suitable name would have been 'Consumer Culture Theoretics':

"... This label would have better represented the theoretical, ontological and epistemological heterogeneity that has contributed to this research stream; an eclecticism that is itself a unique by-product of the contested and rather circuitous way that 'alternative' research perspectives gain a foothold in the psychologically dominated field of consumer research" (Arnould and Thompson, 2007: 7).

They highlight that their 2005 synthesis of CCT was intended to demonstrate the major trajectories underpinning the new paradigm, rather than to create rigidity around thematic approaches. Sherry and Fischer (2008) further conclude that CCT is now categorised as a distinct form of consumer research. More recent descriptions of CCT have called it an *'epistemic community,'* and posit that CCT is more than a label; it is *''an act of power''* (Moisander et al., cited in Sherry and Fischer, 2008:9). Moisander et al. (2008) (cited in Sherry and Fischer, 2008:9) propose that the CCT framing pulls from diverse strands of intellectual traditions and positions researchers as situated within *'existing methodological customs'* and *'theoretical ways of thinking.'*

2.2.2 Narrowing of the CCT Focus

From the mid-1980s through the 1990s, within consumer research attention was given to the role of consumption in identity formation (Arnould and Wilk, 1984;

Belk 1988, 1992; McCracken 1986, 1988a). Research depicts the implicit relationship between identity and consumption, with consumption directly translating into identity and consumption being capable of illuminating identity. Belk (1988) proposes that essentially consumers are what they consume in his 'Extended Self' discussion. In the much cited work, 'Possessions and The Extended Self,' Belk (1988) shows that possessions assist in self-perception and in turn become part of the consumer body, both symbolically and figuratively. This close link between the self and material possessions was an important consideration in how consumers utilise objects to express their personal characteristics. The constructs of 'the self' and 'not self' however were not new concerns, Sartre (1956) discusses these relationships amongst 'having, doing and being' and the synthesis of 'having and being'. However within Consumer Culture Theory, Belk's (1988) formulation of the extended-self became the dominant means of conceptualising how consumption objects are interwoven into concepts of the self. Arnould and Price's (1993) work adds to this discussion in their demonstration of how certain consumption processes function as an important aspect of identity construction and communication of self to others both individual and communal. Indeed, drawing from their extensive review Arnould and Thompson (2005) argue that the self that is forged with market-generated materials is still coherent even though often diversified and fragmented.

2.3 The Construct of the Consumer

Consumption is underpinned by the idea that consumer behaviour is social, complex, often irrational and unpredictable (Venkatesh, 1992). Within the area of Consumer Culture Theory consumer researchers conceptualise consumers as *"interpretive agents rather than passive dupes"* (Arnould and Thompson, 2005). They explore the

ways through which consumers: "exert agency and pursue identity goals through a dialogue (both practical and narrative) with the cultural frames imposed by dominant ideologies" (Arnould and Thompson, 2007: 10). In this regard these ideologies are associated with either the market-generated ideology, such as advertising messages, or the ideologies that relates to institutional forces such as political, social or cultural. In this domain consumers are treated as "active agents" that make sense of these ideologies and construct critical responses (Firat and Venkatesh, 1995). Firat and Venkatesh (1995) expose how deeply anchored the field is in 'modernist ideals' and through the application of postmodernist insights generated new contributions to the field (Moisander et al., in Sherry and Fischer, 2008:13). The description of the postmodern consumer by Firat and Venkatesh (1995) as fragmented and de-centred received growing acceptance within the field of consumer research (Goulding, 2003). De-centering of the subject also refers to the postmodernist view that the human subject is not a self-knowing, independent agent but historically and socially constructed (Firat and Venkatesh, 1995). Firat and Venkatesh (1995) highlight that the postmodern condition of reversal of production and consumption is related to the postmodern consumer: as each act of consumption is also an act of production, consumers not only consume but at the same time also produce symbols and meanings into the world (Firat and Venkatesh, 1995). Whilst Firat and Venkatesh (1995) demonstrate how contemporary consumers are freed from having or seeking a centre, this did not mean that the self could not be coherent.

The construct of the consumer has continued to develop in CCT research with the focus on the active role of consumers and consumer interaction brought to the foreground in consumer research. Following on from the trend in consumer research

that focuses on the blurring of consumption and production in the market systems (Penaloza, 2000; Goulding and Saren, 2007; Thompson and Tian, 2008) the active role of consumers becomes critical. Through 'prosumption' (Ritzer and Jurgenson, 2010), 'working consumers' (Cova and Dalli, 2009) or Campbell's (2005) 'craft consumption' consumers are no longer said to be simply consuming when they buy a product, but through their participation in the construction of consumption objects, they have extended the set of spaces in which consumption can be said to occur. Campbell's (2005) 'craft consumer' could be used to highlight this: the consumer who transforms objects into new entities. He emphasises that the same objects are used in a variety of differing social, cultural and practical settings. Campbell (2005) further notes that concepts such as 'possession,' 'personalisation,' 'customisation' and 'appropriation' overlook the transformative processes and practices of consumption. In considering how consumers affect marketplace dynamics, Gielser's (2008) perspective of marketplace dynamics outlines the concept of 'marketplace drama' to explain the events that unfold when consumers engage in free music downloading in an attempt to gain access to artistic work. This context illuminates the market changes and dynamics whereby a compromise between past practices by marketers and changes originating from consumers are achieved. This focus on the spaces of consumption and consumer action has been developed further around the domain of the Internet and the creation of shared desires, experiences and identities for consumers. Through the lens of consumption and brands, these platforms act as a mode for interaction and integration and bringing people together (Hewer and Hamilton, 2010b). These platforms also allow for more creativity to take place between consumers and as such consumers adopt the role of active culture creators.

This understanding of the active consumer is developed further in Chapter Three with the advent of the practitioner. In this thesis, a focus on how the consumer actively participates in a practical manner and alters the marketplace is key (Shove and Pantzar, 2005) and the vintage context provides an opportunity to move the discussion of production and consumption forward through the practice theory lens.

2.4 Consumption Communities

CCT research has focused on the consumer as the individual actor but also on the complex construct of the consumption community. This section considers the long, rich history surrounding the construct of 'community' within interpretive consumer research. Within CCT, the proposition that human beings are culture bearers and as such are active culture producers has been widely explored on the basis that the formation of social relations is significant for understanding both consumers and culture. According to Firat and Venkatesh (1995) the dominance of the market has caused consumers to fragment, become more isolated and to form more groupings (Firat and Venkatesh, 1995). One concept employed in the literature is that of 'communitas' (Arnould and Price, 1993; Turner, 1995; Goulding et al., 2004a) and within the consumer research literature this social formation has developed around subcultures (Brownlie et al., 2007), tribes (Maffesoli, 2007; Cova and Cova, 2002; Cova et al., 2007) and brand communities (Schouten and McAlexander, 1995; Muniz and O'Guinn, 2001) or as Schouten et al., (2007) term 'mosaics of microcultures'. The situated nature of consumption practices has been sought to be understood within the domain of consumer research in studies of Mountain Men (Belk and Costa, 1998), farmer's markets (McGrath et al., 1993), rave cultures (Goulding, Shankar and Elliott, 2001, 2002), 'Goths' (Goulding, Saren and Follett, 2004a),

trekkies (Kozinets, 2001) and bikers (Schouten and McAlexander, 1995). Collectives whether online or offline are increasingly being formed around particular brands, activities or products of consumption which unite members. In understanding consumption as a way of doing and performing community, a focus on the constitution of the collective and the social bonds is necessary (Muniz and O'Guinn, 2001). With regards to communal consumption, Arnould and Price (1993) highlight that community is developed through emotions of linkage, belonging and group devotion. Penaloza and Venkatesh (2006) refer to this as *'the agentic practices'*, which are crucial for co-creation, value and use. Further Muniz and O'Guinn (2001) note how the consumption activity can be placed centrally in the creation of collective belonging. The work on subcultures, brand communities and tribes is considered in the following sections and the practices around which the social bond is created and maintained.

2.4.1 Subcultures of Consumption

The early work of the Birmingham Centre for Contemporary Cultural Studies (CCCS) (Hebdige, 1979; Hall and Jefferson, 1993; Gelder and Thornton, 1997) strongly influenced the studies of communities of consumption. In employing the term 'subcultures' the CCCS explores the collective forms of subversive behaviour amongst cohorts of the British working-class youth. These subcultural groups were seen to consume distinctive styles that separated them out from the mainstream traditional class based identities. As a form of consumer resistance, the CCCS studies depict these individual practices becoming group orientated away from the mainstream and establishing a certain style based on these opposition consumption practices. Further research was anchored in the exploration of the symbolism

attached to consumer products such as clothing, cosmetics, automobiles and other outward symbols of membership of subcultures (Kellner, 1995: Hebdige, 1979, 1997: Frith, 1996; Willis, 1978, 1990).

Within the CCT mapping, the consumption collective concept is outlined as Arnould and Thompson (2005: 874) assert that subcultures of consumption are the communities in which individuals: "define their symbolic boundaries through an ongoing opposition to dominant lifestyle norms and mainstream consumer sensibilities." This juxtaposition to the mainstream can be seen in early theories of subcultures which highlight communities that develop strong interpersonal bonds. ritualised modes of expression, and beliefs that preclude other social affiliations in order to augment dominant institutions such as family, schooling, and class issues (Goulding et al., 2002). This built on the work of Schouten and McAlexander (1995) who introduce the term 'subcultures of consumption' as a way to characterise the individual and collective organising structure. Within this they describe a clearly defined hierarchical structure with clearly organised systems of formal and informal membership along with a set of beliefs, practices and rituals. These are similar to the shared rituals described by Cova and Cova (2002) in his neo-tribes but differences are evident. Within Schouten and McAlexander's (1995) subcultures of consumption, the hierarchical social structures define the subcultural groupings and status of the individual. They describe that the collective is characterised by the cocentric social structures that are governed by the commitment to the group: 'hard core' members display a "commitment and ideology that is full time and enduring" (Schouten and McAlexander, 1995: 48). These members act as opinion leaders to the 'soft core' members who are less committed to the group and are less involved than
the 'hard core' members. Lastly they describe the 'pretenders' who demonstrate a great interest in the group but only ever 'delved superficially' into the unique ethos of the group. They function as an audience and support system to the more soft-core and hard-core members. Members could be seen to adopt active roles of co-creation (Holt, 2004; Gummeson, 2008; Lusch et al., 2010) within the subculture, which is of particular relevance to marketers. Brownlie et al. (2007:113) highlights that subcultures are differentiated by "*clear hierarchical social structures that may identify the status of individual members*." In this type of sociality, membership is defined by the adherence to the group norms and values and maintenance of the group ideology "*membership is seen to be static, one mask being permanently worn, in that distinct dress codes and a specific stable way of life permeates everyday activities*" (Brownlie et al., 2007:113). Adding to this Schouten et al., (2007: 74) develop the argument and suggest that a subculture of consumption functions on shared consumption values and decisions with concerns over authenticity.

Within consumer research, consumers have been shown to enjoy membership of multiple fragmented groups in which identity is constructed and discarded as necessary (Firat and Venkatesh, 1995). Unlike sociological studies of subcultures, the consumer research perspective had shown that consumption communities are self-selecting and are held together by shared experiences and consumption preferences (Schouten and McAlexander, 1995; Kozinets, 2001). Goulding and Saren's (2007) work on the gothic subculture illustrates: *"subcultures have often been conceptualised as the catalyst for counter-hegemonic strategies of resistance (Kellner, 1995); however they are also cultures of consumption."* This articulates that although these groups are keen to demonstrate their difference from the

mainstream, they must participate and rely on marketplace materials to fulfil this resistance.

Subcultures of consumption have continued to be a focus of scholarly attention within the CCT domain Askegaard's (2010: 362) work depicts that contemporary consumer groups are situated in socio-historical contexts that display a "complex interplay between commercial interests (commercial) media, creative agents and consumer subcultures." However, some authors have questioned subcultural theory as few subcultural communities appear to present with the subcultural characteristics of cohesion, dedication and resistance and scholars have sought different theoretical description through the development of the brand communities focus and tribal theory (Elliot and Davies, 2006; Beverland and Farrelly, 2010; Canniford, 2011b). In attempting to theorise new forms of community, the turn to structured social relationships around a brand is detailed further in the following section.

2.4.2 Brand Communities

Through drawing distinctions between a ranges of different social groups centred on consumption, the concept of brand communities has received considerable attention in consumer research (McAlexander and Schouten 1998; Muniz and O'Guinn 2001; Schau et al., 2009). A brand community is: "*a specialised, non-geographically bound community, based on a structured set of social relationships among admirers of a brand*" (Muniz and O'Guinn 2001:4). Research has shown that brand communities are complex entities with their own rituals, traditions, values and behaviour codes. In particular, brand communities generate shared rituals and ways of thinking and traditions, as well as a sense of moral responsibility towards other

members, and religious zeal towards the focal brand (Muniz and O'Guinn, 2001; Muniz and Schau, 2005). Schouten and McAlexander (1995) in their seminal paper on brand communities detail a set of consumers and their relationship to Harley-Davidson. They highlight that consumers form brand relationships through interaction with other consumers, along with the brand. Consumers develop a relationship with the brand, which in turn impacts on their consumer-to-consumer relationship. Muniz and O'Guinn's (2001) study examines three brand communities *(Ford Bronco trucks, Macintosh computers and Saab cars)* they demonstrate that members of these communities gain an important part of their brand consumption experience through their membership to these communities. Furthermore, they found that members achieved an aspect of personal identity from this participation in the community.

The underpinnings of CCT have guided the inquiries of holistic understanding of social phenomena in highly contextualised settings. The notion of the brand community has claimed centrality in research on consumer collectives in real-life consumption contexts (Celsi et al., 1993; Schouten and McAlexander, 1995; McAlexander et al., 2002; Algesheimer et al., 2005) as well as in online research (e.g. Nelson and Otnes, 2005; Kozinets, 1997, 2002, 2006). Overall, brand communities have been characterised as consumer collectives that share norms of behaviour related to the brand discourse and the institutionalised practices that arise from them. Commonly, the brand discourse has been seen as the overarching driver of consumer interaction (e.g. Cova and Cova, 2002; Kozinets, 2002), and value has been conceptualised as arising from all practices that pertain and emerge from the social interactions within the community (Schau et al., 2009). This interest has also

shifted onto online communities and their social consumption practices (Kozinets, 2002; Muniz and Schau, 2005; Cova and Pace, 2006). This focus on online communities became brand orientated as technology progressed the capacity for participating in online communities. Some netnographic studies have also shown interest in the negotiation (and constant conflict) between producers and consumers (Giesler, 2008) and raised their dynamic and complex nature into the forefront (McAlexander et al., 2002). There has been a shift away from a singular brand discourse to a view of them functioning as organic meeting places that are focused around practices rather than a brand (Goulding et al., 2013).

The brand community similarly to a subculture is representative of a cohesive and dedicated community but rather than the resistance of the subculture that leads to a sense of communal belonging, the brand is central to the social structure (Canniford, 2011b). Recent research has found that many consumption communities do not orientate their socialisation around a single brand, rather it is more loosely associated with a collection of brands, products, activities and services: Fournier and Lee (2009) cite examples of the lifestyle rather than the specific brand affiliation. In this manner the brand community approach begins to be less useful and a tribal approach to consumption becomes more relevant.

2.4.3 Tribes

Going back to Arnould and Thompson's (2005) outline of the CCT domains, *marketplace cultures*, focuses on the exploration of the postmodern construct of *'neo-tribes'* (Maffesoli, 1996; Cova et al., 2007) and 'the move away from individualism'. This phenomenon is developed in the literature with a focus on the

creation of these new types of social bonds. Maffesoli's (1988) 'The Time of the Tribes' outlines his construction of tribes as including producers and not only consumers, describing them in a variety of ways: 'small community group[s]', 'networks of solidarity' and 'youth groups, affinity associations, small-scale industrial enterprises' (Maffesoli, 1988 cited in Cova et al., 2007: 72-75). Maffesoli (2007) discusses the shift in traditional structures in society has resulted in individuals seeking their own means of belonging and that they are increasingly doing this through shared consumption practices, discussed above by Schouten and McAlexander (1995) as 'brand communities.' Maffesoli (1996) posits that 'neotribes' are distinct because of their ephemeral nature; they do not have any permanent membership other than through the duration of the rituals. It is also possible to belong to more than one neo-tribe at the same time: membership is in this way temporary and constantly in motion. Maffesoli (1996: 76) suggests "it is less a question of belonging to a gang, a family or a community, than of switching from one group to another," this sentiment is reiterated by Cova and Cova (2001:71) who claims they could be so ephemeral to be "just a feeling, a fancy, a fantasy." Much of the work within consumer research has focused upon the increased individualism of modern society (Bauman, 1988). However within post-modern literature Cova and Cova (2002) discuss the formation of new communities, Tribes. Post-modernity they suggest is less about the individual's identity and more focused on a search for social links based on shared passions and interests (Cova and Cova, 2002). These links are asserted to be through consumption practices and the formation of social structures. For Cova and Cova (2002: 67):

"These neo-tribes are inherently unstable, small scale, affectual and not fixed by any of the established parameters of modern society; instead they can be held together through shared emotions, styles of life, new moral beliefs and consumption practices."

Consumption is argued to be crucial to tribes and is central to gaining an understanding into the collective nature of the tribe (Mitchell and Imrie, 2011). Cova and Cova's (2002) tribes are socially and demographically diverse but joined together through shared consumption passions. Cova and Cova (2002) propose that not all tribe members interact with the collective in the same manner. Similarly to the subculture engagement of Schouten and McAlexander (1995:48) they suggest four categories of engagement: a 'member', a 'participant', a 'practitioner' and a 'sympathiser' (Cova and Cova, 2002). These consumer-to-consumer relationships are self-formed groups that held meaning and relevance for the consumers within them. Bauman (1992) considers neo-tribes as existing around a group allegiance that is displayed through symbolism. This symbolic consumption acts as a platform to create a social link that expresses self-identity (Cova and Cova, 2002). With regards to marketing to consumer tribes, it could be argued that the social influences are the most important factor on the individual's consumption activity (Bagozzi, 2000). Tribalism suggests that social engagement is based on consumption activity and that groups, whether in virtual or real worlds, are increasingly formed around brands, products or activities of consumption which unites members through 'linking value' (Cova and Cova, 2001: 69). With tribes there is an important focus on reappropriation in the acts of being, gathering and experiencing together as a group.

This is in line with the ongoing work on tribes by Goulding, et al. (2013) in which tribes could be seen to express values of unity, quality and disseminating knowledge:

"Tribes need to feel unique as averse to mainstream; they need a story, myth or narrative around which they can unite" (Goulding et al., 2013: 830).

In this respect, CCT has often emphasised that consumers are not passive recipients of marketer created meaning, but are active participants in the construction of what the particular brand or product would mean in their own lifestyle group (Carroll and Ahuvia, 2006). Tribal theory observes consumers as utilising a variety of brands that are threaded together in an ongoing practice of 'bricolage' and innovation (Cova et al., 2007). It is orientated around temporal and emergent elements, such as with Goulding et al.'s (2009) rave collectives: community is temporarily constructed through the inter-dependent co-presence of multiple consumption resources (Cova and Cova, 2002). Tribal communities are constructed on a more ad-hoc basis and more open-access resources than brand communities (Cova et al., 2007; Goulding et al., 2009; Kozinets, 2002; Visconti et al., 2010). Consumer tribes differ from subcultures in that their connections are much more narrowly focused with similar beliefs and values that set them apart from the dominant culture (Schiffman et al., 2008). Further a brand community is established specifically around a particular product or brand (de Burgh-Woodman and Brace-Govan, 2007; Brownlie et al., 2007). Kozinets and Handelman (2004) demonstrate that consumer tribes in contrast may diminish brand equity by their consumer activism and placing themselves in opposition to the mainstream consumer. The use of tribal theory locates the consumer as part of a collection of diffusely empowered consumers (Shankar et al., 2006) who interact, perform and practice through select market offerings. Tribal

theory presents a vital theoretical alternative to illuminating the ways in which consumers combine resources and practices in hybrid networks to allow them to socially interact for transient periods (Canniford, 2011b: 69).

Shared consumption collectives have been argued to be a search by the post-modern consumer to create social links with other individuals (Simmons, 2008; Cova and Salle, 2008). Brownlie and Hewer argue: "Consumption practices are the stuff of sociality. They are the stuff of belonging" (Brownlie and Hewer, 2011: 249). Over the past twenty years, consumer researchers have explored a variety of consumptionorientated collectives ranging from extreme sports groups (Celsi et al., 1993), surfers (Canniford and Shankar 2007, 2013), anti-brand festivals (Kozinets, 2002), skating tribes (Cova and Cova, 2002), brand cults (Belk and Tumbat, 2005), brand communities (Muniz and O'Guinn, 2001), and consumption subcultures (Schouten and McAlexander, 1995). Within this field of new social formations there are a number of related theoretical groupings that have been explored: tribes, brand communities and subcultures. From the literature it has been shown that the concept of consumer tribes offers an understanding to describe communities that do not fit within the subcultural or brand community dialogues. Tribes have been shown to be networking platforms that allow for knowledge to be created and shared, with a focus on play, communication and innovation as central to participation (Canniford, 2011b). The ad-hoc nature of this tribal understanding is an important consideration for this thesis in exploring the construct of the vintage collective as a grouping that is formed on shared consumption interests and practices, as such the tribal theory lens will inform this study.

2.5 Alternative Marketplaces within Consumer Research

Consumer researchers have not been slow to highlight the value of a turn to alternative marketplaces (Arsel and Thompson, 2011), 'utopian marketplaces' (Maclaran and Brown, 2005), second-hand markets (Bardhi, 2003; Bardhi and Arnould, 2003; McRobbie, 1988; Roux and Korchia, 2006), or of collecting (Belk, 1991a). This notion that more ad-hoc, alternative marketplaces are worthy of discussion was expressed early on by Davis (1966) who suggests that the antiquity of the institution should not preclude an assessment of its contemporary manifestations (Davis, 1966). There is also a strong basis for this focus on the alternative rooted in the seminal work of the 'Consumer Behaviour Odyssey' (Belk et al., 1991c), which is founded originally on the notion of seeking out alternative spaces of consumption to explore a sense of the past (Belk, 1991c). One of the first pieces of work on the importance of alternative marketplaces was Maisel's (1974) study. He deemed the flea market an 'action scene'. With a focus on pricing, vendor and consumer search, dynamics of the bargaining process and "market myths," Maisel's ethnographic study highlights the playful nature of the flea market. Sherry's (1990a, 1990b) research built on Maisel's work exploring the socio-cultural significance of the flea market. He describes it as an alternative marketing system that could be thought of as complimenting more commonplace marketplaces. He asserts that this marketplace has never been treated with the merit of a marketplace worthy of examination: "The flea market is regarded more as a curiosity or throwback than as an object of serious inquiry" (Sherry, 1990a: 13). In previous consumer research the focus tends to be on the economic dimension and is reductionist in nature, Sherry's work moves this focus forward. Sherry's (1990a) seminal exploration of the Midwestern American

flea market demonstrates the social embedding of consumption and the immediacy of the purchase experience. From a material perspective, the observation that anything may become a collectible (Belk et al., 1988) is displayed in the studies of flea markets. Further Belk, Sherry and Wallendorf's (1988) study focuses on a single venue but utilises a wide range of naturalistic methods to enable them to provide a more holistic view of the flea market around the themes of: freedom versus rules, boundaries versus transitions, and competition versus cooperation, sacred versus profane and bargaining versus buying. This focus on alternative marketplaces has continued in the more recent literature. Bardhi's (2003a) study on thrift shopping presents this practice as a money saving activity through which shoppers derive recreational benefits - she describes that consumers gain pleasure not from finding the gem, but from finding the gem "at just 99 cents" (Bardhi, 2003a: 375). Further Guiot and Roux's (2010) ongoing work on second-hand describes that the growth in the second-hand market is attributed to the decline in the purchasing power of the middle classes from the 1980s (Williams and Paddock, 2003). They suggest that in both Europe and the US a trend occurs through which the reselling, recovery and recycling of clothes has become acceptable in the marketplace (Guiot and Roux, 2010). Roux and Guiot (2008) define the practice of second-hand consumption as: "the acquisition of used objects through often specific modes and places of exchange" (2008: 66). They highlight hedonic and economic motives as the main drivers for this consumption mode (Roux and Guiot, 2008). In considering the hedonic motives they argue that recreational aspects were at the heart of the secondhand consumption experience, such as the social contact with friendly and passionate sales people, the entertainment aspects of the shopping process and the "serendipity

ensuing from the unexpected encounter with certain objects" (Roux and Guiot, 2008:67). These dimensions they suggest would be motivational drivers to the purchaseintentions of second-hand and vintage purchases alike.

Yet, vintage consumption could also be suggested to be related to collecting, a practice defined by Belk (1995a) as: "the process of actively, selectively and passionately acquiring and possessing things removed from ordinary use and perceived as part of a set of non-identical objects or experiences" (1995a: 479). The notion of collecting and the ties to identity and the practice of these judgements around the concept of 'collectable' are an important consideration in the second-hand marketplace. In considering the material world, collections are defined as groupings of objects that share one or more similar characteristic or affiliation. In considering objects as active agents of identity that have the potential to form social bonds, the interpretation of collecting as a form of an active process around possessions, has stimulated much discussion (Belk, 1988; Thompson, 1976). Belk explains: "collecting things and displaying things should flourish among individuals in a consumer culture in which consumer goods have become the central focus of our dreams and desires" (Belk, 1995a: 139). From a traditional economic perspective, possessions are used solely for utilitarian purposes or as status symbols, in contrast from a material point of view, possessions convey and construct identity and to socially identify with others and add to the biographical narrative of individual's lives. According to Belk's (1995a: 65) work "possessions may be perceived as symbolically obsolete" however in contrast it is through this obsolescence that objects may appeal even more as collectable. Belk (1995a) defines collecting as the

process of actively, selectively and passionately acquiring and possessing things

removed from ordinary use as part of a set of non-identical objects or experiences. This definition suggests that the 'collector' must have his or her own selection criteria. With regards to the vintage marketplace the notion of collecting and the centrality of this concern to the individual's life merit attention. In exploring the notion of collecting it is important to stress that both consumers and sellers 'collect' in very similar ways and are active practitioners in this process.

2.6 The Alternative in Vintage

Vintage as a concept has been discussed in the fashion theory literature but is lacking exploration within the consumer research domain as such this research study will employ a wide range of literature to explore the practice in the marketplace. As highlighted in the introductory chapter, the term vintage is multifaceted and problematic due to a lack of consensus regarding the time periods necessary to constitute 'second-hand', 'retro', 'vintage' and 'antique'. Semantically, 'secondhand' or vintage implies that it is a used good, and that there is some sort of journey taking place between one consumer to another. DeLong et al. (2005:23) argues that vintage clothing "usually involves the recognition of a special type or model, and knowing and appreciating such specifics as the year or period when produced or worn." A more broad definition from Palmer and Clark (2005:175) suggests, "It is used to cover a huge spectrum of clothes that are not newly designed." Tungate (2008) provides a more focused view highlighting the complex nature of the term, "any one particular item may change through time and usage by the fashion media so that second-hand becomes known as retro then in turn as vintage." The growth in the popularity of vintage clothing has been attributed to multiple sources: the revival of fashion trends from the 1940s -1970s (Dyer et al., 2006), an influence of celebrity

(Palmer and Clark, 2005; Odulate, 2008), a growing distrust of global brands (Tungate, 2008) and concern over ethical practices of large fashion chains (Jackson and Shaw, 2008; Brace-Govan and Binay, 2010). This move from niche subculture to more mainstream fashion choice has been capitalised on by the high street, luxury retailers, and online platforms and can be evidenced in the number of fashion blogs, websites and books on selecting and wearing vintage clothing (Palmer and Clark, 2005; Tungate, 2008). In Gregson and Crewe's (2002) study, they discuss the revival of 1970s clothing styles results in vintage consumers viewing second-hand clothing as "*aesthetic objects to be valued, understood, and worn in terms of, and as an appreciation of, their design, construction and authenticity*." In modern consumption landscapes, vintage is heralded as a highly commoditised fashion alternative to wearing new designs (Palmer and Clark, 2005: 197). As Tungate's (2008) work explores the notion of 'vintage' is an attitude rather than a style of dress:

"It's a rejection of 'exclusive' yet global brands, an affirmation that cheap and unusual is better than expensive and everywhere- and a message to marketers that the fashion consumer of the future will be harder to snare" (Tungate, 2008: 245).

Vintage is regularly featured in the pages of leading fashion magazines and is held up as a sign of individuality and authority. Silverman's (1986) work identifies a flourishing market for vintage goods in the middle class and youth population. This is further developed by Crewe and Gregson (1998) who describe that second-hand is purchased by this group of consumers for fun and sociality and also for display purposes. Dubin and Berman (2000) further this and comment on the notion of innovation in regards to vintage fashion, suggesting the exclusivity of wearing vintage *"is about having and wearing what nobody else can; it's about looking* *wonderful while not looking like everyone else*" (2000:13). This suggests that there are hierarchies within vintage at work. Further the adoption of vintage by the mass market could be seen to dilute the appeal of vintage for some consumers who are concerned with aesthetics and individuality (McColl et al., 2013). Vintage in this way can be seen as highlighting the complex knowledge required in these judgements and the associated value.

The consumption mode of vintage is often confused with second-hand consumption which is due to the overlap which exists in practice: vintage pieces might be secondhand and second-hand pieces might be vintage, but not all vintage pieces are used and not all second-hand pieces are old. The relationship of individuals to fashion is explored in the following chapters as socially mediated and this is an important consideration when considering vintage fashion. Clarke and Miller (2002) assert that there has been a democratising of individual relationships to fashion and a much greater ability to create niches consisting of consumers, rather than created by the industry. In looking at how traditional fashion authority is subverted, DeLong et al.'s (2005:24) study focuses on the development of vintage as a subculture and the complex process these aesthetics demand. Their work on vintage proposes that being *'hooked on vintage'* is not a haphazard process but a rather that it is a complex process that involves the consumer possessing the relevant "aesthetics, taste, clever dressing, historical curiosity, and an ability to discriminate the authentic product, and revalue it in a new setting" (DeLong et al., 2005:24). This coupled with the pleasure gained from "others in the know" recognising that the items are authentically vintage, which raises notions of cultural capital around vintage (Bourdieu 1984, 1985). A strong appeal of wearing vintage clothing, DeLong et al.

(2005) find is the perception that wearing vintage allows the wearer to create something original by reconfiguring something old. The knowledge that is used in the process of creating something unique acts as a way to assert an individuality through *"revaluing and reuse"* which in turn *"redefines fashion in the process"* (DeLong et al., 2005:40). They express that wearing vintage is primarily about being involved in a change of status and a revaluing of clothing beyond the original time period or setting.

Vintage and second-hand has been shown to involve an experience that is more emotional than functional for the consumer. For instance, Roux and Guiot (2008) conclude that there is a direct influence of nostalgia proneness on recreational motives, which encompasses treasure hunting and social contacts. Consequently, it could be proposed that vintage consumption may be more driven by hunting for a treasure or the unique piece rather than hunting for a bargain. As well as alternate exchange channels the literature of alternative marketplaces depicts characteristics of this marketplace providing consumers with unpredictable goods, a platform to search for bargains, visual stimulation, social interactions and feelings of affiliation (Belk, Sherry and Wallendorf, 1988; McRobbie, 1988; Gregson and Crewe, 1997; Sherry, 1990a). The move away from consumption as a utilitarian exchange to a social experience highlights the complexity of consumption spaces and practices (Crewe and Gregson, 1998). The work of Crewe and Gregson (1998) on car-boot sales comes from a geography perspective but is relevant in unpacking the turn to alternative consumption spaces as they explore unexpected spaces that display the sociality of consumption and are held it in contrast to conventional retail environments. They depict car boot sales as "dirty, cluttered and unpredictable but

as such are also exciting, challenging and fun" (Crewe and Gregson, 1998: 42). They propose that crucially this pleasure is not just experienced on an individual level; it is intrinsically embedded in the social character of the space for both vendors and buyers. They assert that these exchanges provide the consumer with a greater degree of agency than more traditional contexts of exchange. They further propose that second-hand exchange is the production of a spatialised practice of exchange: "this is a space for quasi-fun, quasi-leisure, quasi-work activities; an arena in which conventional boundaries are blurred and/or transgressed" (Crewe and Gregson, 1998; 50). Moreover they focus on the attempts to impress meaning in second-hand exchange in which practices are situational specific. Space and the practices that it affords is an important consideration. For example the car boot sale is deemed the site of "locally embedded exchange," in that it is a site in which the possibility and potential for object reuse is tied to the individual's knowledge of the specific neighbourhood and their situated knowledge (Gregson and Crewe, 2002:138). In considering the marketplace, the spatialities and the trading in certain spaces and retail locations can be seen to add to the construct of the alternative. In their later work, Crewe et al. (2003) find that respondents position themselves through their discourse on retail spaces, with a constant distinction drawn between their choice of location as individualistic and on the cusp, compared to the blandness and confining nature of what they consider the centre. This notion of having the correct space to embody the creative scene is also tied to the idea of being part of and knowingness to enliven these alternative spaces (Watson, 2009; Gregson and Crewe, 2002).

In exploring the construct of vintage in the marketplace as alternative it is important to consider to what it is alternative. Vintage is constantly constructed as the 'other' -

as an alternative to the mainstream, of being different and doing things differently. The vintage marketplace is conceptualised as being in stark contrast to the imagined homogenised culture of mainstream fashion and retail (McRobbie, 1988). As described by Nobbs' et al. (2011) work on vintage and visual merchandising, the popularity of vintage has grown within the last decade and it is no longer considered the domain of the niche market, it could be argued to have crossed into the mainstream. This notion of a move away from the mainstream and differentiating vintage was explored in detail by Crewe et al. (2003) as they argue *'retro retailing'* is:

"Slippery, elusive and mutable, retro is a facet of the second-hand market which, whilst it has certain relatively transparent, defined boundaries and core constituents, is simultaneously about opacity, and necessarily so" (Crewe et al., 2003:62).

This excerpt highlights that the ambiguity over the term is linked to a constantly shifting search for value in the marketplace; the elusiveness of the terms defies definitive boundaries. Semantics have potentially moved on since the Crewe et al.'s (2003) study and what they describe as 'retro' could be superseded with what is discussed here as vintage. Importantly with the concepts of 'retro' and in turn 'vintage' boundaries can be drawn but they are not of a fixed entity, there is permeability to the constructs as they are continually in flux because of market perceptions. The use of 'retro' as a term is most prevalent in the literature from the late nineties and early noughties, especially around the nostalgia literature (Lowenthal, 1985; Brown et al., 2003; Brown and Sherry, 2003; Borgerson and Schroeder, 2003; Blanchette, 2014) however there appears to have been a semantic shift to the use of vintage over retro. Jameson's (1984) early eighties work highlights

retro and sees it as '*the cannibalistic plundering of the past*, ' he posits that it is random and without meaning or content (Jameson, 1984: 65). However this idea of the randomness of retro can be argued to stand in stark contrast to the process and preciseness with which some subcultures have adopted retro styles. In comparison to Jameson's work, Jenss' (2005) research on sixties clothing, highlights the complexity of the concept of retro but he asserts that in English no definite definition of retro exists. He proposes that it is an individualistic term: some used it to describe the mix of old and new in one outfit, others to describe original second-hand dress only and many used it to apply to copies of old designs. Jenss (2005) classifies it as a visual or materialised recourse to objects and materials of the past which stimulates memories and associations with other times or decades. This ambiguity over the terms and their theoretical underpinnings again highlights the lack in cohesive understanding of the concepts.

This ambiguity in the vintage marketplace, of its '*between-ness*, ' is further explored by Crewe et al. (2003). Their study demonstrates that vintage commodities are often situated within the mainstream, such as mass-produced fashions during their first cycle of consumption and at the time are not considered rare or unique. Crewe et al. (2003) tie in the concept of retro industries to the creative industries. By their very nature they argue they share certain similarities in that the retail landscape and marketplaces as they are frequently positioned in market stalls and small retail outlets, in which second-hand goods are sold alongside their own designs. However, the creative industries are predicated on the creation of new and innovating designs, retro in contrast can be seen as the reconfiguration and re-entering of previous designs into the marketplace. In Crewe et al.'s (2003) study, the boundaries of the

creative (vintage) practitioners are held up as constituting 'the alternative as imagined'. Their respondents speak of being embedded within networks of likeminded individuals, with an oppositional discourse established between the unknowing mass of consumer society and the knowing, elite collective. In describing this membership to an alternative scene it can only be articulated in this stark comparison. In describing vintage as an alternative subculture to the mainstream held in stark contrast, there is a notion of stability and fixity of constructs. In positioning the constructs as oppositional the notions of purity and authenticity become questionable. It is important to acknowledge the instability of this boundary as vintage has been popularised as a concept and therefore it could be argued that there has been a blurring and even hybridity of the two.

2.7 Fashion Context

The examination of fashion consumption has to be situated within its historical, social and cultural contexts. Historically second-hand clothing has attracted a stigma as a mark of poverty (Veblen, 1899; Horne and Broadridge, 1995; Tarrant, 1983). The exploration of second-hand clothing in the marketplace is not a new construct but it was one that has garnered little specific academic attention. Clothing and its interpretation could be seen as culturally, socially and historically constructed and bound. Progressing from the basic need of clothing to cover the body, in relation to the material world there is an acknowledgement of the polysemous nature of clothing: with clothing as a possession, social marker and commodity. Dress can be seen as a basic fact of social life and all cultures dress the body in some manner from clothing, tattoos, cosmetics and body painting (Entwistle, 2000a). Within

and inner worlds as well as defines a social status (Campbell, 1996; Evans, 1999; Clarke and Miller, 2002; Crane and Bovone, 2006; Scaraboto and Fischer, 2013; Blanchette, 2014). Fashion can be viewed as "the practice of the people" (Clarke and Miller, 2002). In this respect it can be regarded as a collection of social practices that emerge in spite of rather than because of the fashion industry. In Simmel's (1957) seminal work on the theory of fashion, he proposes that fashion occurs largely from a tension between creativity and conformity. However, many fashions appear to be regressive and nostalgic towards past cultural traditions. In accordance with Levi-Strauss' (1966) theory it could be argued that material culture objects allow consumers to connect with history and for the object to have a place in everyday life. Objects can therefore be seen to function as communicators and link to the past, through these material objects, history could be maintained in the present. In consumer research, consumers have been shown to form close connections with objects of material culture, through which both the consumer and the object sustain each other (Zwick and Dholakia, 2006; Fernandez and Lastovicka, 2011). They could be seen to form webs of significance (Geertz, 1973) that embrace a wide variety of material artifacts and social histories. This position compliments McCracken's (1986: 71) position that "cultural meaning is constantly in transit" and objects of material culture could be seen to be the embodied essence of past events. This study is particularly interested in the material world as it relates to the concept of 'vintage' as a means of understanding and enactment.

Within the domain of consumer research, Thompson and Haytko (1997) contest that novelty, rapid changes, a proliferation of styles and the mass consumption of goods

represents Western fashion. In their exploration of fashion discourses they cite Faurschou (1987:82):

"Fashion is the logic of planned obsolescence – not just the necessity for market survival, but the cycle of desire itself, the endless process through which the body decoded and recoded, in order to define and inhabit the newest territorial spaces of capital's expansion" (Faurschou 1987: 82, cited by Thompson and Haytko, 1997).

In considering Thompson and Haytko's (1997) account of the discourses surrounding fashion consumerism, each could be seen to champion value in a specific appearance style whilst de-valuing other styles. They conclude that consumers have considerable potential to appropriate fashion discourses, generate personalised fashion identities and resist the dominant fashion narratives. Similarly in readdressing similar questions, Murray (2002) found that consumers are able to construct desired identities through fashion successfully and to employ fashion discourses to mediate identity tensions that arose. More recent research from Scaraboto and Fischer (2013) highlights the dual (potentially incompatible) logics at work in the fashion industry as the logics of culture and the logics of commerce. These dual logics they argue both constrain and enable the industry in their co-existence. In their exploration of the *'Frustrated Fatshionistas'* they portray an agentic view of consumers as "*strategic, purposive actors in organisational fields*" (Scaraboto and Fischer, 2013: 1254). This perspective of consumer agency is useful in considering how alternative markets materialise.

2.7.1 The Fashion Object

Before exploring the literature further it is important to define and position the concept of the fashion object that is subsequently developed throughout the thesis. Fashion as shown above is conceptualised in relation to clothing, but fashion could be seen to reach far beyond this into lifestyle choices, interests, hobbies, views and values. Davis (1992) argues that fashion is not static or a definite article: it is something that transforms, mutates, converges and even regresses over time (Davis 1992:28). In the consideration of fashion operating within the social landscape, it becomes evident that it could be used as a mechanism for distinction and division between social collectives. Rafferty's (2009) work on the class-based emotions in fashion consumption suggests that this distinction through fashion could happen in highly nuanced ways. Rafferty (2009) found that fashions are constantly redefined and renegotiated and embraced or dismissed by collective groups of consumers because they become associated with particular groups and they are then subsequently either rejected or approved by individuals. In regards of contemporary society's vast array of self-fashioning choices Rafferty (2011) argues that the practices of consumerist choices continue to be embodied by their cultural background. Building on Bourdieu's work, Rafferty's (2011) work suggests that fashion functions as a psychosocial existence through which competitive social relations are fuelled. Fashion consumption can therefore be conceptualised as a culturally specific practice. Rafferty's (2011) paper highlights the role of emotional experiences in social life that structure consumption patterns. She explores the relationship between social class and consumption, suggesting that consumption practices are interconnected with emotional tendencies. In conceptualising fashion as

a force that functions within the psychic and emotional context of social relations, she argues that a space of social relations through which class discrimination actively unfolds is created. Rafferty (2009) stresses that fashion should not be considered as a purely commercial force that controls consumers; she posits that it is a complex social phenomenon. Rafferty (2011) asserts that the most commonly agreed academic definition of fashion is as "a specific aesthetic for dressing the body" (Rafferty, 2011: 242). This definition offers a definite form rather than considering the psychosocial aspects of fashion. Bourdieu's (1984, 1985) work has similarly conceptualised fashion as 'an embodiment of objectified capitals', stressing that fashion is a set of class-based principles of existence, conditioned by taste and competitive relations. Referring back to Bourdieu's (1984) definition of fashion as the result of competitive relations, fashion could be conceptualised as a determining influence on the consumption choice the individual consumer makes. Entwistle (2000a: 2) adds to this argument positing that fashion is a practice born out of competitive social relations and as such it could be said to be 'a competitive psychsocial mechanism' that calls for styles to be revised and reworked frequently. In framing fashion in this manner and acknowledging the mobility of aesthetic ideas, fashion could be considered as a variety of dress styles at any one point in time that are culturally deemed to constitute being 'fashionable'.

Bourdieu's (1984) work and in particular his work on practice, is a useful framework to unpack the variety of fashion consumption approaches in contemporary society. His work suggests that taste; style and aesthetic sensibilities are *'not an individual concern but are the result of 'cultural capital' from socialisation.* 'Taste therefore becomes the expression of cultural capital and is embodied in practice, such as in

consumption choices. Bourdieu's (1984) work focuses on 'fields' and is concerned with how position within social hierarchy allows individuals to gain specific habitual dispositions and capitals. In turn these are then important in the attainment of social status and the subsequent rewards that this affords in consumer society. Sandicki and Ger (2010:32) draw on Bourdieu's concept of habitus and posit "consumption practices and fashion in particular can have an important role in the construction of a new parallel taste structure. Fashion can be both a key medium and a marker of a new habitus." This perspective allows consideration of the assertion that the individual actor's taste is constructed in relation to their larger social standing and resulting issues such as class. Entwistle (2000a) suggests that dress is an embodied practice and one that is embedded within the social world. This emphasis on the social nature of dress highlights that individuals are active agents in their engagement. This reinforces the earlier work of Mary Douglas (1973, 1984) who proposes that the social highly restricts the medium of expression of the body and that the body thus becomes a symbol of its cultural location. This positioning of the body within the social was suggestive of Bourdieu's (1984, 1989, 1993) accounts of embodiment that links between the social and the individual. For Bourdieu (1984) bodies are structured by social position and these structures are only reproduced through the embodied actions of individuals. According to his 'habitus' individuals are predisposed to a particular way of dressing and for social collectives this is a way through which individuals could orientate themselves towards the social world. The body could therefore be seen as a social entity with fashion as the outcome of both social factors and individual actions. Clarke and Miller (2002) go further than Bourdieu's suggestion and posit that the social setting of an individual choice has to

be acknowledged in the juncture of daily aesthetic decisions by consumers. Bourdieu (1990) defines cultural capital as the specific set of cultural tastes, skills and knowledge that is cultivated and used in the social field. This becomes intrinsic in so much as individuals make consumption choices that do not appear culturally and/or class bound. Bourdieu proposes that through the exercise of cultural capital, respect of others through consumption of objects become linked to the ability to do so. Bourdieu (1984) proposes that these class structures dominate individual's tastes and lifestyles. Warde (2007:1) built on this and argues that the concept of 'good taste' could be used as a 'weapon' in competitive social relations. In this respect individuals could position themselves through consumption into an advantageous position as cultural battles occur.

In considering a consumer research approach and building on Bourdieu's argument, Holt (1998:4) has further argued that all actions and interactions are inevitably classifying practices, deeming them *"micro-political acts of status claiming.*" Based on Holt's argument, individuals must therefore negotiate through the social struggle in order to achieve a position. Bourdieu (1984; 1990) also highlights the varying sources of capital required to obtain cultural capability: social capital through networks and outlets, economic capital having the financial proficiency and symbolic capital in that through the possession and display of items to which value has socially been ascribed. Bourdieu's depiction of consumer culture has been criticised for lack of an emotional capital that could be deemed necessary for social advancement (Illouz, 2007, 2009). Illouz's (2007) work argues that individuals use consumption as a way to connect to others, therefore the social relationships that consumption activities enable act as *'moral goods in which the content of self-hood and well-being*

are at stake ' (Illouz, 2007: 67-8). Her argument suggests that by revisiting the Bourdiuesian model a focus on the emotional habitus could be developed to find new insights into the relationships between identity, class, social relations and gender. Building on the central tenet of Bourdieu's (1984, 1985) work of fashion as social practice, an emphasis on the larger social context for fashion choice has been explored thoroughly in the literature with issues such as identity, gender and sexuality being at the foreground (McRobbie, 1998; Entwistle, 2000a; Rafferty, 2011) as for the purposes of this thesis an exploration of the communal, social element of the practices is crucial.

2.7.2 Fashion Subcultures

The historical development of fashion could be argued to be the result of an exclusive, elite activity that became popularised and pluralised over time, however in modern society fashion could be stimulated by sub-cultural or lifestyle propensities. As discussed earlier in this chapter, there is now a long history of youth sub-cultural groups that have garnered academic attention (Hebdige, 1979; Hall and Jefferson, 1993; Gelder and Thornton, 1997; McRobbie, 1988). Lury argues that punk was often viewed as *'the last authentic youth culture'* (Lury, 1996:198). She suggests that the increase in commercialisation of popular culture has resulted in the demise of authentic youth subcultures. Lury (1996) believes that it is impossible to maintain authenticity in the subculture because of commercialisation; she suggests that the growth of the media has subsumed youth culture. The literature of subcultures is an important consideration in the discussion of vintage, as the subcultures are shown as traditionally based on an aesthetic way of seeing the world around them and how this in turn influences their resulting consumption choices. Abercrombie and Warde's

(1988) study distinguishes these subcultures by three general characteristics: firstly that they are cultures based on leisure rather than of work, secondly that they are focused on the collective rather than an individual and lastly they are concerned with style and in particular a youth specific style. The collective is seen to have unity in dress, rituals, attitudes and language (similar to tribes Cova and Cova 2002; Cova et al., 2007). Again in Jefferson's (1976:86) early exploration of the Teddy boy's, the subculture and style highlighted that the Teddy Boys dress *"represented a symbolic way of expressing and negotiating with their symbolic reality; of giving cultural meaning to their social plight."* These accounts identify the role of the market and consumer culture in the formation and maintenance of the subculture.

The idea of 'bricolage' is important when considering concepts of style and subcultures. Taken from the work of Levi-Strauss, it was used to describe the process through which objects acquired new meaning through recontextualisation and redefinition. McCracken's (1998) work compliments this as he asserts: "combination and recombination take place until a concept and an aesthetic emerge that help give substance to a group's wish to differentiate itself from the mainstream" (1998:136). In this way vintage clothing could be seen as cultural commoditisation as an item's status is culturally shaped through a process of being withdrawn and introduced in a new setting (DeLong et al., 2005), through this redefinition of the object, 'the imaginary potential of the object could be unlocked' (DeLong et al., 2005).

In examining fashion subcultures, one contemporary manifestation relevant to a discussion of vintage, is that of the hipster. In contemporary culture, discussions of 'hipster style' have become frequent (The Guardian, 2010: NY Times, 2013: Times, 2013) and synonymous with style, nostalgic ideals and a search for the alternative.

Frank (1997) describes 'hipster' as a term that is employed to reference an avantgarde, 20-something who has worked hard to cultivate an alternative style that in some respects is considered the revival of bohemian culture. This rhetoric has continued in contemporary manifestations in a privileging of localised consumerism, a focus on independent consumption practices and rejection of mainstream aesthetics. Schiermer's (2013: 168) study asserts: "in contrast to a traditional subculture, hipster culture is characterized by a lack of generational distinction and by a genuine veneration of certain cultural expressions and objects of the previous generations." He argues that the two salient traits of the hipster are "irony and redemptive conservatism" (Schiermer, 2013: 168). Bringing the hipster domain to consumer research's attention, Arsel and Thompson (2011) adds to the discussion in their Journal of Consumer Research paper proposing that as a 'fashionable *counterculture*', the hipster myth is developed through marketing intelligence of cultural producers and widespread co-optation of media imagery. This myth positions hipsters as consumers who are part of a participatory lifestyle group which is "largely denuded of any connotations of social protest or deviance" (Arsel and Thompson, 2011: 795). They argue that the hipster myth devalues their tastes and interests that such individual's will have worked hard to develop the status in their collective through their consumption practices and necessary knowledge. The marketplace myth they argue ultimately devalues their tastes and aesthetic practices and as such they have to protect their tastes through a process of "demythologisation". This focus on a collective and their consumption practices that

are rooted in a desire for the alternative, style focused and a search to separate from

the mainstream is especially pertinent in a consideration of the vintage collective and how value is framed.

2.8 Consumption and Materiality

In considering Arnould and Thompson's (2005) discussion, the social environment through which consumer actions take place and the objects that they utilise in the material world, is critical. This thesis is interested in an exploration of marketplace practices and as such an understanding of linking the material goods to the consumer as means of sense making is developed and was examined in the extant literature. Miller (1988b) proposes that consumer culture is the study of material culture in which consumption is regarded as the primary mechanism for interaction in society. Further material objects can be argued to shape the structure of daily consumption (Shove and Araujo, 2010: 13). In consumption studies, objects are often described as mere aids in the exchange practices of consumption, in applying a material culture perspective, it is argued that the narrative of the object cannot be separated from the object itself. Whilst this thesis is not a sociological material studies research undertaking, it does reflect on the importance of the material in the vintage world in performing in the marketplace, therefore material culture theory is unpacked further to explore the experience of consumers in this lived experience. The following section considers the broader material culture field and how objects are situated and understood in this domain.

One of the most important ways that people relate and interact with each other socially could be said to be through material goods (Lury, 1996). In recent decades the study of material cultures has grown in prominence grounded in the premise that

social and cultural relations are characterised by material objects and so material culture studies focus on objects in use (Dant, 2004, 2005). Consumer culture could be viewed as a particular form of material culture, as it frames consumption away from the common definition of consumption⁴ as 'using up.' This move to a more comprehensive understanding follows Dant's (2005) sociological perspective in which he argues for a more in-depth focus on the 'material stuff of life'. This approach suggests a consideration of the social and cultural meanings of the object, rather than just considering the meaning as inextricably linked to the object (Dant, 2005; Miller, 1998b). The conceptualisation of objects is an important consideration for material studies. Latour (1991) argues that objects should go further than highlighting social structures or being treated as symbolic entities, he suggests that the material world should be considered in line with social practices and that objects should be understood within their materiality: "we are never faced with objects or social relations, we are faced with chains which are associations of human and nonhuman" (Latour, 1991; 110). Douglas and Isherwood's (1979) much cited study discusses objects as vessels for thinking and for rational beings to make sense of the modern world. They suggest that through making visible the cultural categories that objects function as cultural markers they are the "visible bit of the iceberg which is the whole social process" (Douglas and Isherwood, 1979:74). In addition Reckwitz (2002) proposes that contemporary cultural theorists have to consider both the material and the symbolic in order to aid the sense making of the familiar and modern and should explore the meaning in the everyday lived experience. For

⁴ The Oxford Dictionary defines 'consume' as "to eat, drink ingest, or to completely destroy, or to use up a resource." <u>http://oxforddictionaries.com/definition/consume?q=consume</u> {Accessed on May 28th 2012]

Reckwitz (2002) objects are conceptualised as 'hybrids': they are used for and have effect in their materiality. Therefore objects are also symbolic in that they are defined, interpreted and handled culturally; this suggests that objects have cultural and material specificity and are not easily replaceable. Lury (1996:1) compliments this in her work on consumer culture describing the appropriation of an object as a moment of both consumption and production, of doing and undoing and of destruction and construction. Through these dualities the cyclical nature of consumption is conveyed. Lury (1996) asserts that through a material culture lens, consumption could be seen as 'conversion'. She believes that in shaping Western consumer culture the notions of the circulation of commodities through which they are produced and exchanged in the market for capitalist gains is a crucial concern, as are the changes in the relationship between production, consumption and value. Possessions, for Lury, in turn function as key definers of interests, lifestyle and attachments. With regards to these attachments, Dittmar's (1992: 205) work suggests "material possessions also serve as expressions of group membership and as means of locating others in the social-material environment." Considering how objects affect groups in relation to the social and material environment is especially relevant in considering the vintage marketplace.

Miller's (1998a) work has focused on shopping and objectification as he explains that shopping, as a practice is a manifestation of key social relations, of family, class, ethnicity and gender. He asserts that objects are *"social relations made durable"* (Miller, 1998a: 19) this suggests that shopping is practiced as a medium for facilitating relations of love and care. He argues that consumption is one tool for giving objects meaning and transforming them from *"anonymous and alienated*

conditions of their production. "For Miller (1998a) the site of this shopping is one in which a devotional practice is actioned and commodities are imbued with significance. Miller's work depicts a shopping space through which objects are considered the material culture of love. This creates notions of how space is constructed, how shoppers construct space and how they make sense of such spaces.

2.8.1 The Potential of Objects

There is a common assumption in marketing literature that material objects follow a linear trajectory of consumption (production, valuation, consumption and decline) When looking specifically at the vintage market, it is constructed of goods that have been discarded from the traditional market. This expulsion could potentially lead to a re-examination and re-evaluation of objects. Douglas and Isherwood's (1979) anthropological study proposes that consumption happens in all societies and that it is *'beyond commerce*, ' in that it is always both a cultural and economic occurrence. They argue that it has as much to do with value, meanings and communication, as economic exchange. In considering this perspective objects are not only used to do things, but have intrinsic meaning of their own and function as a marker of social relations. Douglas and Isherwood (1979: 66) assert that objects are the *'visible parts of culture'* and that it is through acquiring, using and exchanging these objects that individuals are afforded social lives. It is suggested that objects are a way to make visible the categories and groupings that classify people in society.

The classical anthropological themes of culture and economy are explored further by Appadurai (1986) with a slightly differing focus than Douglas and Isherwood as he focuses on the *'objects in motion.'* Appadurai (1986) asserts *'objects have lives,'* in

that objects are treated as living beings with social lives through which their meaning changes, they acquire and loose value and their identity shifts. Appadurai (1986) proposes that objects move in and out of the commodity trajectory throughout the course of their lives. With this focus on the motion and the potential of objects, he argues that the trajectory of objects is constrained by *'scale and temporality'*. Appadurai's approach is important to consider, as it does not focus exclusively on a particular moment in the object's life, but appreciates for example its production, marketing journey and reputation. His attention also on the social dimensions of objects as they move along the trajectory. From a cultural value paradigm further insights can be gained from Appadurai's (1986) work as he suggests fluidity to value from following objects in motion. He expresses the importance of the classification and valuation structures linked to wider cultural organisation:

"...define the world of things, lumping some things together, discriminating between others, attaching meanings and values to these groupings and providing a basis for rules and practices governing the circulation of these objects." (Appadurai, 1986:14)

This deepens and extends our understanding of how value is produced and reproduced within social structures. Extending the Marxist idea of commodity value being created as use value is produced and shifted through exchange by combining it with Simmel's concept of the *'exchange of sacrifice being the source of economic value,* ' for Appadurai views exchange as the main source of value creation of *"different regimes of value in space and time"* (1986:4). From a material culture studies perspective, the understanding that objects have *'social lives'* (Appadurai,

1986) or 'biographies' (Kopytoff, 1986) suggests that at some stage in this trajectory, that objects are defined by their relationship to monetary value and then at other stages they become 'de-commodified' (Miller, 1998b), with spatial and temporal considerations (Appadurai, 1986; Kopytoff, 1986) affecting the commodity states as they move through contexts and owners. In considering when goods are withdrawn from this commodity stage and reconfigured, it can be argued that they preserve their tacit exchange values (Kopytoff, 1986). When considering material culture, Appadurai's work forces the researcher to consider each of these moments rather than favouring one over the other or to join them all together. His approach allows the sequence of these relationships and pathways to be explored and gives weight to the positioning of objects within individual's lives - they have the power to influence and change actions. Building on Appadurai's (1986) proposition, this *commodity potential* becomes especially pertinent in the discussion of the vintage market. With regards to vintage objects there is a focus on journeys and a passage of time and space for the object. For objects that have been 'disposed of' from the marketplace, to re-enter, their potential has to be perceived by a consumer and/or seller (Parsons, 2008, 2009). Taking up Appadurai's (1986) argument, the object must wait to be re-discovered and moved on in their journey, the object is reliant on an opportune moment of another consumer finding it and being brought back into the consumption sphere by having new knowledge applied to it. This framing of the marketplace and staging of the object, shifts the literature focus towards a more performative perspective. As Araujo et al. (2008, 8) describe markets could be conceptualised as "ever-changing performances, rather than as stabilised entities, shaped by multiple and distributed calculative agencies." They suggest that market

practices are the necessary performances and practices needed to contribute in the market and that markets are in a constant evolution regarding the practices employed to sustain them and their resulting forms. This thesis focuses on the objects within the practices, through adopting Appadurai's (1986) approach of following things in motions. For Appadurai (1986: 31) consumption is *"eminently social, relational and active rather than private, atomic or passive."* Appadurai (1986) follows the objects rather than the people to reconsider the dynamics of the exchange process. He argues that objects are important because *"their meanings are inscribed in their forms, their uses, their trajectories"* (1986:5). In discussing the movement of objects, Appadurai posits that objects have no absolute value, that rather their value is created through the social process of exchange. This echoes Kopytoff's (1986) suggestion that objects move in and out of *'commodity stages'* in their trajectory. Both authors question what constitutes an object and how permanence and value are enacted in the marketplace?

2.9 Conclusions

2.9.1 Positioning this research and Key Literature Gaps

This chapter has shown the development of Consumer Culture Theory (CCT) as a theoretical domain within the consumption landscape that focuses on the sociocultural, experiential, symbolic and ideological aspects of consumption rather than the production, consumption and exchange of the traditional marketing approach (Arnould and Thompson, 2005, 2007). Positioning this study within this CCT tradition, this thesis approaches an understanding of consumption by unfolding collective practices of the agentic consumer to gain an appreciation of the collective framing of value.

The construct of the 'consumer' has been depicted in this chapter as active and dynamic and this agentic nature of the consumer will be key in exploring the vintage context as this thesis is coming from the CCT approach in which markets are seen as social and cultural constructs (Arnould and Thompson, 2005, 2007). With a sociomaterial focus, this chapter has illuminated the move away from consumption as a utilitarian exchange, to a social experience that in turn highlights the complexity of consumption spaces and practices (Crewe and Gregson, 1998). Building on Holt's (1995) typology 'consuming as play' this chapter explores how consumption and objects are utilised by consumers to interact with other consumers in the marketplace (Arnould and Price, 1993). The literature suggests that fashion functions as a psychosocial existence through which competitive social relations are fuelled (Rafferty, 2009, 2011; Bourdieu, 1984, 1985; Illouz, 2007). Further this chapter has also explored the scholarly attention that has been devoted to consumption communities as a construct in their various formations: subcultures, brand communities and tribes. The marketplace will be explored further in this study with a focus on the collective effort needed in constructing the market space and exploring the social links of 'vintage'.

This chapter has shown that vintage as a concept crosses many theoretical understandings; vintage was understood as an emergent scene in an alternative consumption landscape. The lack of cohesion of vintage as a concept was evident from the literature as discussions of the creative and retro highlighted the complexity of vintage (Gregson and Crewe, 2002; Jenss, 2004, 2005; Crewe et al., 2003). In
foregrounding objects and their inherent materialities (Dant, 2004; 2005: Miller, 1998b, 2005; Appadurai, 1986) and following objects in use, this thesis is particularly interested in the material world as it relates to the concept of 'vintage' as a means of understanding and enactment. Further through situating consumption within the materiality of everyday life (Slater and Miller, 2007:22) and understanding consumption as part of social practice, has necessitated a turn to practice which will be explored in the next chapter (Reckwitz, 2002; Schatzki, 2002; Warde, 2005).

Chapter Three: Practices and Value

3.0 Introduction

This chapter explores the theory of practice approach to the study of consumption and value. It considers the concept of practice (Schatzki 1996; Schatzki et al., 2001; Reckwitz 2002) and how it has been used in consumption studies (Shove and Pantzar 2005; Warde, 2005). Further this section explores the theoretical basis for the use of the practice lens, how it is employed in the literature and explores the merit of conceptualising vintage consumption as a form of practice. The use of the practice theory lens allows the researcher to explore how practices and consumption activities are a continuous, dynamic and interrelated process (Magaudda, 2011; Shove 2003a, 2003b; Shove and Pantzar, 2005).

This chapter aims to provide a holistic understanding of practice to underpin the thesis. As such it explores practice discussions in depth, firstly from an overall understanding before turning to look at how practice has been used by consumption studies and the possible implications for CCT. It then explores different perspectives of practice focusing on the material and then the spatial elements. Next it turns to other domains, Market Studies and Organisation Studies, to aid this practice understanding. It looks then at the understanding of collective practices before the final sections focuses on understandings of value and practices of value creation. In employing a practice theory perspective this section seeks to unpack the connections between practices, the consumption of objects and the value surrounding both.

3.1. Defining Practice Theory

Within the context of the debate in social theory on structure and agency, in recent years there has been a shift to social theories of practice to further explore this problematic. Practice theory has its background in the work of philosophers such as Wittgenstein (1984 [1953]), social theorists such as early Bourdieu (1972, 1984, 1985) and early Giddens (1979, 1984), cultural theorists such as late Foucault (1969, 1984) and Butler (1990), and theorists of science and technology, such as Latour (1991, 1992) and Pickering (1992, 1995), their insights have been integrated into a philosophical ontology of practices developed and mapped together by Schatzki (1996, 2002) and then by Reckwitz (2002). As noted above, practice theories are a set of philosophical and culturally grounded accounts that explore the conditions surrounding the practical carrying out of social life (Halkier et al., 2011). Both Schatzki et al. (1996, 2001) and Reckwitz (2002) have aided the development of a contemporary systemised theory of practice.

In his ontology of practices Schatzki (1996) highlights two different notions of practice: practice as coordinated entity and practice as performance. Schatzki (1996:90) explains that practice as performance refers to the carrying out or undertaking things, the performing of the doings and sayings that "actualizes and sustains practices in the sense of nexuses." For him practices consist of "embodied, materially mediated arrays, and shared meanings" (Schatzki, 2001: 3). Schatzki's (1996:12) theories of practice "present pluralistic and flexible pictures of the constitution of social life that generally oppose hypostatized unities, root order in local contexts and/or successfully accommodate complexities, differences and particularities." Schatzki proposes that both social order and individuality result

from practices (1996:13). Reckwitz (2002) also synthesises elements from Bourdieu and Giddens with a slightly different emphasis than Schatzki in his understanding of practices, he emphasises the importance of routines within practices. Reckwitz's (2002) much quoted definition draws on Bourdieu and Latour as he explains:

"A practice...is a routinised type of behaviour which consists of several elements, interconnected to one another: forms of bodily activities, forms of mental activities, things and their use, background knowledge in the form of understanding, knowhow, states of emotion and motivational knowledge." (Reckwitz, 2002: 249-250)

Drawing out from Reckwitz's definition, there is a focus on the idea of interconnectness and a centrality of 'things and their use'. In arguing that a practice ''necessarily depends on the existence and specific interconnectedness of these elements, and which cannot be reduced to any one of these single elements'' (Reckwitz, 2002: 250), the notion of shared practical understanding is also highlighted by Reckwitz. Reckwitz (2002) proposes that practice theory focuses on the practical construction of life in a world where both humans and nonhumans have an ontological status of their own: in so much as they exist as separate, individual elements. The practice is therefore composed of both the consumer and their context, which is in turn embedded with meanings in practical terms (Holt, 1995). Reckwitz's (2002: 259) concludes: ''practice theory should develop more philosophical perseverance and at the same time not give up its embeddedness in 'empirical social and cultural analysis'. Then in future the loose network of praxeological thinking might 'yield some interesting surprises,'' he thus highlights the possibilities for the future development of the field around practice theory.

For both Reckwitz (2002) and Schatzki (1996) the existence of a practice is dependent upon the specific interconnectedness of several elements - mental activities, bodily activities, background knowledge in the form of understanding, things and their use, competence, emotional states and motivation. A practice can be said to form the "block" whose existence depends on the existence and interconnectedness of these elements, and which cannot be reduced to any of these elements. The practice therefore becomes the entity from which consciousness emerges, as a form of integration of different resource elements. Therefore, a practice-based approach is not focused on who is doing what, what these subjects think about, or what they are like. Rather, the focus is on the process – what is done, how doing is constituted, how resources are used and how the doing has developed over time. Shove and Pantzar (2005) propose that a practice is skills, objects and images "in use". As proposed by Shove and Pantzar (2005), three elements constitute the practices: *firstly an object* (what tools or resources are required in the practice); secondly an image (what images are involved); and thirdly a competence (what competences does the practice require). Practices can be argued not to exist in a static state, but are reproduced and reconfigured in the acts of everyday life. In practice theory, the social is not found in the mental acts or discourse or interaction, but in the reproduction and enactment of practice (Reckwitz, 2002). Practice consists of both doings and sayings therefore importantly for research, the analysis must be concerned with both practical activity and its representations (Warde, 2005:134). Bourdieu (1984, 1985) emphasises the role of practice and embodiment within the social world. Embodied preferences refer to the implicit knowledge or skills in a practice and have been described by Holt (1998: 3) as "a set of decontextualised

understandings...that are readily recontextualised across new settings. "These embodied practices can be seen to be imbued into individual's thinking, feeling and subsequent action in their daily life and consumption choices. In cultural studies the idea of embodiment of practices has been prevalent from the 1990s onwards. In Thrift's (1996, 1997, 1999) work he is concerned with practices through which the 'subjects' become decentred, but embodied and expressive and involved with others and objects in a world continually in process. He focuses on 'the body-subject, not the body, engaged in joint body-practices of becoming' (Thrift, 1997: 142). The emphasis is on practices that cannot adequately be spoken of and on forms of experience and movement that are not only or never cognitive. Thrift is concerned with the ways in which subjects know the world without knowing it, the "*inarticulate understanding' or 'practical intelligibility' of an 'unformulated practical grasp of the world*" (Taylor, cited in Thrift, 1996: 10).

While there is a common focus throughout the practice literature, with a centrality on how social action is carried out, practice theory is not a coherent theory (Halkier and Jensen, 2011), it is fragmentary in nature with a lack of homogeneity in approach, as different theoretical readings exist which will be detailed further in the following sections. Across the literature all use the term *'practices'* in disparate ways, but similarly emphasise routines, shared habits, techniques and competence in their work (Shove and Pantzar, 2005: 44).

3.2 Developing Practice Theory

From the mid-2000s through the work of Warde (2005) and Shove (2003a, 2003b, 2005, 2009, 2010) there is a perceptible shift in the view that what is involved in

consumption. They propose that the relationship between materials and practices was deserving of more theoretical attention. Warde's (2005) article in *Journal of Consumer Culture* is regarded as the first problematising piece presenting an examination of the potential of practice theoretical perspectives in the analysis of consumption. The 1990s had seen a focus of the symbolic meanings of consumption with identity formation and the configuration of exchange relationships in consumer research. This shift in the view of consumption beyond the marketplace from Warde considered the social organisation of consumption and broadened out the view of consumption sociology. Warde (2005) argues that most practices require and entail consumption of some sort; he asserts that the term 'consumption' fuses together two contrasting ideas, of purchase and of using up. He proposes that consumption cannot be restricted to or defined by market exchange. Warde (2005: 137) explains consumption as:

"A process whereby agents engage in appropriation and appreciation, whether for utilitarian, expressive or contemplative purposes of goods, services, performances, information or ambience, whether purchased or not, over which the agent has some degree of discretion."

In this view, consumption is not in itself a practice, but could rather be conceptualised as a moment in almost every practice (Warde, 2005) and this lens has been used to situate consumption within the materiality of daily life (Slater and Miller, 2007:22). The conventions and the standards of the practice can be said to steer the behaviour (Warde, 2005:137). Warde (2005) further explains that wants are fulfilled only in practice, their satisfaction attributed to effective practical performances. Warde's (2005) seminal paper on practice theory constructs an overview and discussion for the use of practice theory in consumption research. He proposes using the theoretical approach of Reckwitz (2002) and Schatzki (1996) into an empirical investigation of consumption. Moving on from Schatzki's (1996) concept of practice, Warde bases his theory on the idea of practice as a nexus coordinated by understandings, procedures and engagements (Warde, 2005:134). Consumption can be therefore argued to occur as goods are appropriated; this involves the consumer undertaking certain practices including knowledge to carry out these practices and the tools to engage as a practitioner (Warde, 2005). Warde suggests: "practices, which are logically and ontologically prior to action, steer consumption" (Warde, 2005: 135). In understanding consumption as something that occurs whilst in the course of doing something else, a focus: 'less to individual choices and more to the collective development of modes of appropriate conduct in everyday life' is necessary (Slater and Miller, 2007:22). Warde (2005) acknowledges that objects are only one type of thing that can be consumed and that these are all part of practices. In decentring consumption, it allows Warde to examine its processes through a different set of frames. Within this view of consumption, people are viewed as the starting point for making sense of the social world. Warde conceptualises consumption as pervasive yet intermittent, for him it is a momentary accomplishment within many practices (Warde, 2005: 150). In order to exist, practices must be continually enacted and reproduced as distinct packages of organised activities. This has important methodological considerations, which will be explored further in the Methodology Chapter.

There is a multiplicity of practices available to people and the external and internal rewards for involvement in these practices differs greatly. Warde (2005) asserts, "Judgements of performance are made internally with respect to the goals and aspirations of the practice itself, and proficiency and commitment deliver satisfaction and self-esteem" (Warde 2005: 148). With regards to external rewards of the practice and of understanding they focus on the gaining of access to social networks and respect within the collective (Warde 2005). Warde's (2005) article can be regarded as the first 'programmatic' piece offering an exploration of the potential of practice theoretical approaches for the analysis of consumption. This was as a response to recognition of the shortcomings of existing foci and theoretical approaches to consumption in the social and cultural analysis (Halkier et al., 2011). Warde (2005) argued throughout the 1990s for the questioning of consumption to move beyond the marketplace and to consider the social organisation and move beyond examining the purchase stage of consumption. The broadening out of the consumption discussion includes a focus on the mundane and routine consumption habits, as Gronow and Warde (2001) elaborate in their text Ordinary Consumption, in which the mundane and everyday were crucial. This shift was again highlighted by Randles and Warde (2006: 226):

"Ordinary consumption is best understood in terms of concepts like habit, routine, constraint, and so on and can be summed up as a recognition of the conventional nature of consumption."

Within the practice theory lens, the material environment – objects, tools, devices and apparatus - and the implicit and explicit practical knowledge stored in them are central in the process of creating interaction, continuity and reality (Halkier et al., 2011). Further adding to the role of practice theory with regards to consumption, Shove (2004, 2005) develops a focus on practices and their routine reproduction through her research. Her version of practice theory highlights the performative nature of social life and examines it analytically (Shove and Pantzar, 2005: 44). She questions change and continuity in practices and their associated objects and resources. Shove and Pantzar (2005) base their approach with consideration of Reckwitz and Schatzki's definitions of practice. Therefore they define practices as involving the active integration of materials, meanings and forms of competence (Shove and Pantzar, 2005: 45). They focus on the failure in the consumption literature to convey fully what is involved in the way things are acquired, appropriated and subsequently utilised. Their work is also greatly influenced by the arguments of Actor Network Theory (ANT) and science and technology studies (STS) - both place an importance on the centrality of the relationship between humans and objects. Shove and Pantzar (2005) move the literature forward by conceptualising consumers' not as simple users but as creative practitioners and view appropriation of things as just one dimension of practice. Shove and Pantzar (2005) with their study on Nordic walking add to the discussion of symbolically focused theories of consumption, to re-evaluating consumers and the material dimension in which they operate. Shove and Pantzar (2005) are interested in how new practices emerge within and through consumer - producer interaction arguing: "many products are quite directly implicated in the conduct and reproduction of daily life" (Shove and Pantzar, 2005: 44). They demonstrate the importance of the relationship between material objects, competence and associated images and place an emphasis on the 'non-human actors' who share the focus along with consumers. For Shove, the

practice exists in the unit of analysis and 'the consumer becomes de-centered in the analysis of reality' which constitutes different elements of reality: material, competence, and images (Shove, 2004; Shove and Pantzar, 2005). This has implications for how both the object and the consumer will be viewed in this study.

3.3 Consumer Research Moves to Practice

From a practice perspective, consumption is argued not to be restricted or defined by market exchange, but can be thought of as an integral part of everyday life (Warde, 2005). Consuming can be considered as a mode of action that is undertaken by people and in which they make use of consumption objects in a multitude of ways (Holt, 1995). In considering practice theory and its use in consumer research, it is worth highlighting the basic purpose of consumer research, as Holt (1995) explains is:

"...to comprehensively describe the variety of ways in which people consume, to understand how these differences vary across groups and situations, and to explain the unacknowledged conditions that structure how different groups consume" (Holt, 1995:1).

Rather than looking at the processes and cultural significance of objects, practice theory allows a focus on the skill, processes, practicalities of their use and how these objects are subsequently consumed. With a focus on the mundane, practice theory explores the interconnected dynamics and how performativity is organised through multiple shared practices (Halkier and Jensen, 2011). Practices are therefore the contexts where actions are carried out (Schatzki, 2001). The concept of practices refers to ways of doing things, which are embedded in the context of both inward/subjective and outward/objective interlinked elements. The outward elements are embodied in the sense that the subject is constituted within the practice (Schatzki, 2001). This means that practice is not synonymous with action or behaviour, but as a total it explains the systematic whole that generates action (Dourish, 2001).

In looking at the application of practice theory in recent studies, Magaudda's (2011) paper is an example of the application of practice theory to the field of cultural consumption. He highlights the role of materiality in relation to practices and questions whether the music consumption practices have indeed been 'dematerialized'. Magaudda (2011) proposes the 'circuit of practice' as a heuristic device to map routines of transformation linking objects, meanings and doings to emphasise the ways in which practices are created, stabilised and transformed. Within the field of digital music consumption the dramatic changes in this practice are explored in how it affects meanings and ways of doing. His 'circuit of practice' illuminates how individuals change practices when transforming or adopting new patterns of activity. In his study he highlights that specific subcultures shape the way that new listening practices emerge and also new technologies. Similarly in Truninger's (2011) study on cooking practices she highlights the importance of specifying the scope and scale of practices as she argues that there are different types of agency of practitioners and trajectories of practices. With this example she highlights the dialogue between economic, social and cultural market forces with domestic conventions. By combining the practice theory turn with conventions theory (Boltanski and Thévenot, 2006 [1991]; Thévenot, 2006) she emphasises issues such as the sequencing of tasks, the tensions that emerge between instituted and personal practices and also the links between images, objects and competences. Both

of these examples manifest the subtle relationship between practices as individual performances and how practices are embedded in cultural structure. Both demonstrate how culture and social structure may form a site of resistance or challenge as practices are 'reproduced through imitation' but subsequently may have to be interpreted and adjusted accordingly.

3.3.1 Implications for Consumption Studies

Within marketing and consumer behaviour, practice theory has been explored in relation to the theorisation of markets and market practices (Araujo et al., 2008), value creation (Schau et al., 2009) and resource theory (Arnould, 2008), these examples illustrate the linking of consumer research with practice theoretical input. With regards to consumption, practice theory focuses on the sociology of consumption, on the collective ordering of consumption, the social construction of 'need,' how practices become 'normal' and how they change (Bourdieu, 1984; Shove and Warde, 2002; Shove, 2003a, 2003b; Hand et al., 2005). This approach to the study of consumption could possibly be viewed as an attempt to fill what Latour (1992) deems the 'missing masses' of social theory.

Consumption theory has moved on, previous consumption studies have focused on the symbolic and identity creation powers of items (McCracken, 1988a; Featherstone, 1990), as many theorists stress the symbolic as a signal of status and identity, in contrast with practice theory it is implied that material is used directly in the reproduction of everyday life (Shove and Pantzar, 2005). Practice theory focuses on the relationship between products, competence and practice. It also stresses the importance of routine, and the collective and conventional nature of consumption (Warde, 2005). Consumption when viewed from a practice theory perspective is not only what the customer thinks, feels, and decides to do, but also is the act of consumption that the consumer takes part in. The act of consumption is not decided by the customer or by context, but more specifically as part of the whole system of the subject and context in interaction, which consists of both human (other people involved) and non-human (the tools required) elements. The concept that objects are acquired and used in the undertaking of social practices has important implications for the study of consumption. The challenge for the researcher is to explain 'how practices shift' and with what subsequent consequences for forms of consumption.

Everyday consumption practices are often characterised by habits and routines as by intentional reflection and choice (Halkier cited in Gronow and Warde, 2001:26). Therefore the concept of consumption can be integrated into the framework of everyday analysis. Warde (1997:19) understands consumption as "who obtains what goods or services, under what conditions are they delivered and to what use are they put." Consumption experiences can thus be thought of as bodily practices whereby goods and services are obtained, used and disposed of, such practices are experienced in interaction with relevant others and across different spaces in the contingent of everyday sociality (Halkier, cited in Gronow and Warde, 2001:31). A practice can be said to relate to the unconscious dimension of consumer decision-making. Consumer practices suggest a focus on the "…more or less routinized actions, which are orchestrated by tools, know-how, images, physical space, and a subject who is carrying out the practice" (Korkman, 2006: 27). Consumption could be argued to rarely occur purely for its own sake. In this way practices could also be argued to create wants (Warde, 2005), Warde (2005) uses the 'hot rod aficionado' as

an illustration of this - modified vehicles, specific magazines, memorabilia, are the result of practice in a particular activity rather than a result of the individual's taste. It is therefore, Warde (2005) argues, the involvement in the practice rather than a personal decision about conduct that accounts for the nature and process of consumption. This also highlights that items are delicately entangled and are often crucial elements for the practice to take place and act as a channel for its performance. This idea of integration and the intertwined nature of practice are again highlighted by Shove (2004) and Shove and Pantzar (2005), who emphasise the elements of material, image and competence. Innovation they propose is based on the active integration of all the elements of the practice. The authors explore the practice of Nordic walking, digital photography and the game of floor ball to explore and understand the relationship between practice elements and the development of specific practices. As highlighted by their findings, the instruments used, such as the sticks in Nordic walking, are lacking in individual value, but when integrated by the consumer into practice they become invaluable.

Consumer research literature often focuses on the objects at the moment of acquisition by the consumer, rather than how the consumer uses the objects. CCT in particular has focused on the meanings and symbolic properties of commodities and their use for example in identity construction; this can be argued to only present an incomplete view of consumption. By employing practice theory, how the action is carried out in the process becomes of crucial importance (Hand and Shove, 2007:81), as Holt (1995) comments that practice is the embodied skills that people bring to bear in their everyday activities. This focus on how things are used has been examined within consumption practices such as showering (Hand et al., 2005), eating

(Cheng et al., 2007; Halkier and Jensen, 2011), DIY (Watson and Shove, 2008) and wooden boating (Jalas, 2009). In their *Journal of Marketing Management* paper, Murphy and Patterson (2011) employ a practice theory perspective in their exploration of motorcycling edgework. They use the practice lens to tease out connections between the practice, the consumption of objects and the meanings surrounding both. They again highlight that CCT has focused only on a partial view of consumption, without considering the actual 'doing' of consumption. This focus on objects, doings and meanings by Murphy and Patterson (2011) takes inspiration from Magaudda (2011) and *"opens up the horizons of investigation,"* to illuminate that motorcycling cannot take place without the object – the motorcycle or the knowhow required to use it correctly. This again illustrates that using practice theory to explore consumption allows more abstract questions about the relationship between practices, technologies and infrastructures to be posed. Further it allows probing into what the consequences are for these relationships and the 'fixity' and 'fluidity' of patterns of consumption and everyday consumer routines (Hand et al., 2005).

3.3.2 The Consumer as the Agent of Practice

Previous discussions place the consumer as a 'dupe', with no real choice in the symbolic selection that they make (as shown in Table 1 below). However in practice theory, the consumer is an active agent carrying out social action. Warde (2005) significantly argues that the idea of 'the consumer' disappears and the focus is situated on the organisation of the practice and the moments of consumptions enjoined (Warde, 2005). Differing understandings, levels of competence and sphere of involvement generate behavioural differences in relation to practice. According to Reckwitz (2002: 249-250) the consumer becomes not only a carrier of patterns of

bodily behaviour, but also of certain routine ways of desiring, understanding and participating. Reckwitz (2002) notes, *"The social world is first and foremost populated by diverse social practices which are carried by agents."*

Following on from the discussion in Chapter Two of the active consumer, from a practice perspective it can be suggested that material objects obtain value through and as a result of their enabling of a practice. Shove and Pantzar's (2005) work on Nordic walking further highlights that as a form of consumer-producer interaction, practice and value are created through the competence of practitioners. Using practice theory to study consumption allows the researcher to conceptualise the emergent and co-constitutive relationship between object and the reproduction of shared discourses and the skills of consumption and differentiation.

Table 1 – Role of Consumer

Consumer Profile	Consumer Competence	References
Dupe	Obedience and acceptance of the status quo.	Slater 1997
Hero	Constant consciousness, clear thought process and ability to maximise utility.	Slater 1997
Postmodern Identity Seeker	Ability to introspect.	Featherstone 1991; Firat and Dholakia 1998
Craft consumer	Rediscovery and appropriation of manual skills and capacities.	Campbell 2005
Prosumer	An active consumer who through immaterial labour adds affective and cultural value to market offerings.	Tofler 1980; Kotler 1986; Cova and Dalli 2009
Reflexive Consumers	Capacity to dialogue through internalisation and sharing of values.	Beckett and Nayak 2008
Practitioners	Possession of the requisite skills and knowledge to conduct practices.	Warde 2005; Watson and Shove 2008; Roepke 2009.

3.4 Practice Perspectives

Whilst this thesis aims to address practice from a CCT orientation, it has been necessary to turn to other perspectives to fully explore the concept. This will be discussed further in this section around practice understandings of the material, practice as space, practice as markets and practice as organisation.

3.4.1 Practice and the Material

Practice theory allows researchers to understand ways of consuming as continuous relational accomplishments at intersections of multiple practices in daily life (Halkier and Jensen, 2011). Practices are not held in place by meaning alone, Hand et al. (2005) posit the need to focus on the relation between the material, conventional and the temporal, as configured through practice. With regards to a focus on the material, the practice of dressing is an everyday norm, in that it is a practice undertaken by the majority of human beings on a daily basis, it is a taken for granted feature of life. Trentmann (2009: 297) highlights that using theories of practice offers a different take on the role of objects in consumption:

"The life of objects, in other words, is not prior to or independent of social practices but co-dependent. This also means that value is not based in a product or its meanings but in how it is put to use."

This is idea of 'putting to use' is crucial in this study which considers the notion of vintage value and how it is framed in practice. This is again highlighted in the work of Rouse (2006) who explains that practices range from ephemeral doings to long-standing patterns of activity. Rouse (2006) proposes: *"attention to practices often requires extensive examination of relevant equipment and material culture, but can also assign constitutive roles to vocabulary and other linguistic forms or performances"* (Rouse, 2006:490). In understanding practice as 'performance' the active nature is crucial in value creation. Practices can also be framed as 'active' in that a practice of doing has a recursive and co-constitutive relationship or alternatively practice as 'entity' in which the practice is durable, embodied, materially mediated and has shared meaning. Through this focus on practices

researchers are able to explore ways of consuming and how these are related to changes and the webs of social reproductions (Halkier and Jensen, 2011). The activities of social life are continuously carried out and this performativity is organised through multiple, collective shared practices. This is echoed by Røpke (2009) who argues, "Primarily, people are practitioners who indirectly, through the performance of various practices, draw on resources." Warde's work further adds to this argument with "a performance presupposes a practice" (2005:134). This focus on the performative nature of practice is crucial and at the forefront of a discussion of the active nature of practices, it is developed further by Truninger (2011) who argues that the concept centres on the idea of an 'on-going practice'. She illustrates this concept in relation to the practice of cooking, "cooking cannot be understood as the end result, but instead as the on-going reproduction of 'doing cooking', where competences are developed through performance" (2011:38). Practice according to Truninger (2011) consists of three elements: *stuff* (nature and objects), *image* (symbols and meaning) and skill (competence, know-how and technique). This again demonstrates the importance of the idea of skill and knowledge when discussing practice theory. Theories of practice go one-step further than Goffman's (1959) notions of performance that emphasise the embodiment and processual nature of social interactions. Through the work of Shove and Pantzar (2005) consumption activities are portrayed as the result of individual performances that overlap and intertwine with the socio-material context through which the practices take place, this is able to occur as a result of the arrangement of meanings, activities and objects. Their research brings the concept of social practices as performance to the foreground of discussion. They simplify the complexity of the area of social practices

and depict three interrelated streams: firstly of meanings and representations, secondly consisting of objects, technologies and material culture and thirdly depicted by embodied competences, activities and doings (Shove and Pantzar, 2005; Magaudda, 2011:20). Their argument therefore proposes that practices are characterised by the outcome of the performative linkages between these three interrelated streams. In these links materiality performs an integral function in the creation, change and stabilisation across the range of practices (Magaudda, 2011:20).

3.4.2 Practice and Space

Theories of practice articulate the importance of space, as Schatzki (1996: 89) comments practices are '*spatially dispersed*.' This compliments the work of Giddens (1979:3) who articulates that '*time-space relations are inherent in the constitution of all social interaction*' and also of De Certeau (1984b) who understands 'space' as "*the intersection of moving bodies*." De Certeau's concern is the ways in which, in everyday, unconscious practice, people alter and adapt, and appropriate cultural products and make them their own. De Certeau (1984b) draws attention to the ways in which people create their own meaning for spaces, individually and collectively, through the specific ways in which they move through those spaces and put them to use. This concern is built upon by Schau et al. (2009: 31) who maintain that practices comprise a "*temporally unfolding and spatially dispersed nexus of behaviours*". This focus on the spatial occurrence of practices is central in this thesis, but it appears underrepresented in the literature. Referring back to Magaudda's (2011) *'circuit of practice'* in his much-used model, the relationship of context or space to the meanings, doings and objects is omitted. This concept of space is an important

consideration when attempting to understand how consumption occurs through practices within specific spaces and not one that is addressed in the literature.

Building on the work of Warde (2005) and Appadurai (1986), the link between spaces, objects and practices is of central importance in this thesis. Unlike with the 'Diderot effect' (McCracken, 1988a) in which a change in one object leads to multiple transformations with the whole set, Hui (2012) argues in her study of mobile practice networks that previously discarded things are reclaimed easily when mobility's change. Using the context of patchwork quilting she explores how spaces are made up of networked mobility's, asserting that consumption is a moment in the course of a practice but also a moment in the course of the object's movement, in which their social life consists of use and disuse (2012: 203). Hui's (2012) position echoes Lury's (1996: 77) suggestion to 'think in terms of object-people practices'. Hui (2012) builds on this and proposes mobile practice networks are formed from objects and people together, as they both shape their shared movement. She argues that moments of consumption are multi-sited and that through careers of practice and looking at object's lives, moments of travel can be said to facilitate moments of consumption. Hui (2012) asserts that mobile practice networks are enacted as temporary accomplishments or as moments within the circulation of objects. Consumption therefore can be suggested to occur during performances in relation to objects of practice that are used thus forming 'symbiotic chains'. When consumption is understood as part of social practices the movement of the objects and how this movement shapes the moments of consumption, the perception of space becomes critical.

3.4.3 Practice as Markets

Within marketing literature, the value in the practice turn is being explored out with the CCT paradigm; the field of Market Studies is also utilising practice theory to explore the disconnection between everyday practices and the shaping of markets. Drawing on the perspectives of Finch and Geiger (2010: 238) the relationships between producers and consumers are performed in market spaces and these can be argued to be indirect as a market's objects, the intricacies and interactions of this relationship, bind them. With regards to consumption studies, Shove and Araujo (2010: 14) state that in marketing literature the term 'consumption' is often utilised in relation to the attainment of goods as a result of market exchange. Using methodological approaches linked to Actor Network Theory and Technology Studies and coming from a more economic sociological perspective, in Market Studies practices are used to highlight the routine character of the marketplace. It suggests that the constitutive role of objects and infrastructures in shaping and transforming these practiced routines is omitted from the previous market studies. Shove and Araujo (2010:18) conceptualise that a practice approach includes *"the social field of* interwoven practices anchored in habit, routine, shared understandings and embodied skills". They use the practice approach to enable them to reframe consumption along with the relationship between users and material objects.

There has been much discussion in marketing literature of how practitioners mobilise knowledge, but far less focus on how this knowledge is apportioned into their practices such as value creation. Callon (1998a) has been an advocate in furthering conceptualising markets as practices - considering what practices and expertise are necessary for a market to exist, and how the market is performed by the cognitions,

technologies and actions by market actors. Callon (1998a) proposes that markets are composed of a range of practices that are the result of different forms of expertise and material objects. These objects can be seen as materialised insights as they are linked directly to consumer's involvement with practices. In considering value creation practices the space of the marketplace must also be acknowledged. In the marketplace, Shove and Araujo (2010:20) suggests that value is only created through practitioner skill and that objects only receive value when "in the hands of someone with a requisite level of competence." Building on the central tenet that "markets are *practical outcomes*" Araujo et al. (2010: 5) propose that a creative practitioner must have adequate understanding of the practice, the boundaries of the marketplace and how value is constituted in the constantly shifting market. As Araujo et al. (2008: 8) describes markets can be conceptualised as "ever-changing performances, rather than as stabilised entities, shaped by multiple and distributed calculative agencies." This is an important consideration when considering a marketplace such as the vintage market that is constructed on the basis of items being re-seen and reevaluated.

From a Market Studies lens (Araujo et al., 2008; Geiger et al., 2012; Finch and Geiger, 2010) practices of value creation are both a social and market activity. Slater (2002a) proposes that for an economy to exist at all, objects must become stabilised through *'processes of materialisation'*. Therefore the object must become calculable and be given meaning by marketplace actors at a given moment in time. He argues that all commodities are calculated and materialised in practice within the market and economic action (Slater, 2002a, 2014). Further Trentmann notes: *"practices have a dynamic force of their own, creating sensations, competencies and plans of doing*

more or doing things differently. They are entangled in a creative interplay with materiality" (2009: 297). In this way the life of an object is not independent of social practice, but is argued to be co-dependent and therefore value is not found in the object but in how it is utilised. Kjellberg and Helgesson (2006) add to this and identify three specific and interconnected market practices that configure the exchange market: exchange practices (activities that are involved in consummating individual economic exchanges of goods); normalising practices (activities that result in norms and rules guiding the actions of market actors involved in exchange); and representational practices (that portray markets, the way they work, and produce shared images of the market). This compliments Callon (1998a) who proposes that markets are composed of a range of practices that are the result of different forms of expertise and material objects. Turning to the Market Studies domain answers calls for reform in marketing to connect both classic marketing thought and developments within the social sciences field (Araujo et al., 2008: 5). A focus on markets rather than marketing helps to define 'market practices' as bundles of practices, inclusive of material arrangements that allows for the performance of markets (Araujo et al., 2008: 8). Moving forward with this performative idiom (Pickering, 1995) Market Studies adds to a holistic understanding of practice as the discussion moves beyond the theory-practice dichotomy and suggests that markets should be studied as sites of multiple and often conflicting arrays of practices (Araujo et al., 2008: 6).

3.4.4 Practice from an Organisational Lens

Another domain that has been practice orientated is Organisation Studies, which has explored a theoretical view of practice as a transactional social process involving experience and action as mutually informing aspects of human conduct (Simpson, 2009). This is a useful perspective as it highlights the ideas of 'temporality' and 'transactionality,' which shifts the view of practice towards a dynamic, emergent and socially agentic (Simpson, 2009) understanding. Practice has been explored in the Organisational Studies literature and has shown that the dynamics of human practices are just as important as the outcomes produced (Simpson, 2009; Nicolini et al., 2003; Whittington, 2004, 2006; Johnson et al., 2003). This also has important methodological considerations for gaining understanding into how practice is constituted through experience and action and through looking at 'practices in use'. Practice is conceptualised as a dynamic, temporal process that can be seen to converge and diverge (Simpson, 2009), however it goes further than this and practice involves human conduct and the exercise of agency. In this way identity can be seen to be a co-constituting process along with practice. Further in aiming to understand collective practices, turning to the Organisation canon illuminates the formation of communities of practice is significant, as through this formation consumers are able to transform the market through creating new market ideologies and forms of exchange. These consumer communities can be seen to perform both a literal and social form of 'bricolage' in their organisational practices. As in order to gain acceptance into the community the individual has to perform ritual and symbolic activities, which will then give them access to shared resources (Cova and Cova, 2002:17). The concept of communities of practice is similar in some respects to Cova and Cova's (2002) consumer tribe perspective which notes that people gather together in tribes and that this community is more influential on their behaviour than the market or other institutions. However with communities of practice, Wenger and Snyder (2000:139) comment that communities of practice are "groups of people"

informally bound together by shared expertise and passion for joint enterprise."

Although Wenger and Snyder are applying their definition in an organisational context, it is useful to consider the language when applied to consumer collectives, *'bound together,' 'shared expertise'* and *'passion'* are all applicable. This thesis turns to the Organisational Studies domain to enable a holistic comprehension of practice with the aim of enriching the CCT understanding of practice.

3.5 Practices of the Collective

As explored in Chapter Two, this thesis is grounded in practices and value creation of the collective and as such an exploration of consumer groupings in the literature has been necessary. The literature showed that the study of communities has become a staple topic within the realm of CCT studies (Cova et al., 2007; Muniz and O'Guinn, 2001; McAlexander et al., 2002; Schau et al., 2000; Kozinets, 2002; Muniz and Schau, 2005; Cova and Pace, 2006) and as shown above in the Organisation Studies domain with the communities of practice (Wenger and Snyder, 2000) framing. In exploring the practices of the collective, from the literature there has been a shift away from a singular brand discourse to collective groupings functioning as organic meeting places that are focused around practices rather than a brand. Overall, as seen in the previous chapter brand communities have been characterised as consumer collectives that share norms of behaviour related to the brand discourse and the institutionalised practices that arise from them. Commonly, the brand discourse has been seen as the overarching driver of consumer interaction (Cova and Pace, 2002; Kozinets, 2002), and value has commonly been conceptualised as arising from all practices that pertain and emerge from the social interactions within the community (Schau et al., 2009). In exploring these consumer groupings from a

practice theory and value perspective, Schau, Muniz and Arnould (2009) explore the process of collective value creation within brand communities using social practice theory. They illustrate a common set of value creating practices that operate within nine brand communities. They suggest that practices have an "anatomy" that is comprised firstly of general procedural understandings and rules or what they term "explicit, discursive knowledge", secondly of skills, abilities or "tacit, embedded knowledge" and lastly of emotional commitments expressed through actions and representations. Schau et al. (2009) identify twelve practices common across their nine brand communities in which consumers are able to co-create value beyond what the firm has offered. Their collectives exhibit community like qualities that include status of members, identity construction and meaning making. Within these collectives they found that practices could be said to empower participants with cultural capital. Interestingly within the communities they "work together and drive one another" (Schau et al., 2009:35). Competition within the community Schau et al. (2009) found to enable consumers to distinguish themselves and to create a social hierarchy. These practices can be seen to refer back to Holt's (1998: 22) 'micropolitics of consumption.' However their findings challenge Muniz and O'Guinn's (2001) work on brand communities as of 'limited liability'. They suggest that social networking practices within the collectives evolve and exceed brand boundaries. It is interesting in this respect as to how knowledge is transferred within the collective, from 'insiders to initiates' and that value is found in the collective enactment of practices. The Schau et al. (2009) paper is interested in practices in the building and nurturing of brand communities because of their collaborative value creating potential. Based on Vargo and Lusch (2004), Schau et al. (2009: 31) contend that value resides in actions, interactions and projects. This compliments work on open source innovation (Etgar, 2008; Von Hippel, 2005; Koerner, 2006) and Lusch and Vargo's (2006) co-creation work, which attests that organisations will eventually collaborate with consumers to create their entire marketing offering. Schau et al. (2009) visually represent their interaction of practices with a gear metaphor and refer to practices as being 'bundled' (2009: 32) and they do not attempt to formally theorize the nature of this relationship. Arsel and Bean (2013: 913) build upon this in their Journal of Consumer Research (JCR) paper and argue "the teleoaffactive structure of brand specific values, norms, and mythologies that orchestrate the integrative practice of brand community." They highlight that a practice-based approach allows for the connections that exist between marketplace cultures that are formed around broad interests are held together as complex entities. Arsel and Bean (2013) depict how 'taste regimes' regulate practice through continuous engagement in their JCR paper. They argue that taste is performed as a practice that affects the material. They demonstrate how aesthetics are linked to practical knowledge and become materialised through consumption choices. Using Magaudda's (2011) 'Circuit of Practice' Arsel and Bean (2013) argue that a taste regime is maintained through the interaction between objects, doings and meanings and further that practical knowledge is a mediated phenomenon that is acquired and habitualised through engagement with mass mediated regimes (Arsel and Bean, 2013: 913). They move the discussion of practice theory to consumer research forward by suggesting that aesthetics are grounded in practical knowledge and become materialised through everyday consumption practices. This section has shown that collectives are grounded in practical knowledge for both participation and value creation, this

section has highlighted that CCT is starting to turn to a discussion of practice to understand the '*doings*' of the collective. The next section turns to the value associated with this practice perspective.

3.6 Value in Practice

All of the discussions from the literature above highlighted that practice and value creation are intrinsically linked. The concept of value has received scholarly attention from sociology to philosophy, economics, anthropology and marketing. This thesis focuses on how value is framed in practice in consumer research and as such considers domains such as cultural anthropology (Appadurai, 1986; Arvidsson, 2011; Baudrillard, 2000; Belk, 1996; Cherrier, 2009; Graeber 2001; Holbrook, 1999a; Kopytoff, 1986; Simmel, 1978; Thompson, 1979) and CCT (Holbrook, 1999a; Cova and Cova, 2002; Holt, 2004; Gummeson, 2008; Lusch et al., 2010; Vargo and Lusch, 2004; Karababa and Kjeldgaard, 2014). Considering value from a practice lens, Hartmann et al. (2011) asserts that a 'conceptual blind spots' exists when understanding value creation practices in consumer behaviour. A theoretical gap has emerged as to practice consumption and their role in value creation (Schau et al., 2009). Hartmann et al. (2011) adds to the practice consumption and value creation discussion by moving beyond equating practices with performance, they develop two families of practice: firstly as 'socially intransitive practices' through which performance of a practice involves a non-human object and secondly 'socially transitive' practices through which the performance of a practice involves another human subject. This social transitivity of practices elevates understandings of practice consumption. Practice theory gives an empirical base for understanding the use of resources in the situations of value creation or in establishing 'use value': Holt

(1995) illuminates how spectators use various artefacts in their practices of supporting their teams, Shove and Pantzar (2005) study discusses the use of sticks and other objects in Nordic walking, Korkman (2006) describes twenty different dominant practices on board cruise ferries that due to the distribution of space configure the enactment and reproduction of everyday practices such as eating, playing and sleeping. In all these depictions the focus has not been in exchange value, but in the understanding of what is 'valuable' as a part of everyday life.

This attention on practice theory and value has been advanced by Arnould (2014:1) as he suggests practice theory offers a solution to the perennial structure-agency problems in social theory, as it takes structures and agents both to be contingent outcomes of practice. He argues that a practice approach allows for the suggestion that human creative agency is at the foundation of value creation. In reconsidering "individuals as carriers of practice" (Reckwitz, 2002:252) and the idea of practices as "carriers of value" (Schau et al., 2009), use value can be seen in the performance of practices. In accordance with Warde (2005) practice performances have productive and consumptive moments: in productive moments a resource is offered (experience, object) and in a consumptive moment (information, material, experiential) that are received, it is through both of these moments that value is found. These moments are ongoing in social fields and "all the carriers of practice in a field are potential performers whose knowledge of 'what is going on here' is updated through active and passive participation in practice performance" (Arnould, 2014:2). This highlights that value is realised through active performance, participation and practice. In exploring this further the next section details how value has been conceptualised in the marketing domain to date.

3.6.1 Marketing and Value

Value is seen as one of the most '*controversial issues*' in the marketing literature (Sánchez-Fernández and Iniesta-Bonillo, 2007) as value is seen as complex and multi-dimensional and can be perceived to have different meanings depending on time, situation or person (Holbrook, 2006). Differing notions are used frequently in the consumer research literature without definitive grounding of 'value': aesthetic value, use value, social value, economic value, emotional value, brand value and community value. Distinctions between price and value are commonplace in everyday life - price is viewed as an economic construct, whilst value is often seen as a social process: practices of value are a personal process but it is grounded in social practice.

From Adam Smith's (1776 [1937]) assertion that value is conceptualised in terms of effort, money, trade and usefulness and Marx's (1974 [1865]) understanding of value with his labour theory of value in his critique on capitalism, to Baudrillard's (2006, 1972) development of value in terms of the material consumer objects, value has been a key topic of scholarly focus. Understandings of economic value stem from classical and political economic theory, with the Marxian approach conceptualising value as the necessary labour that is required to produce an item, this labour theory of value stresses that it is the labour time involved in the production that leads to the value creation. Value is therefore seen as the difference between the exchange value and the sum of value necessary for production. In this view, use value is only achieved through consumption or in use of the item. From the traditional Kotler (1967) marketing viewpoint customer value is considered the basis for marketing activities. Further the foundation of all marketing and consumption activities can be

seen with Levy's (1959) sign value along with Bagozzi's (1975) exchange value and Holbrook's (1999a) use value. In the marketing domain with branding discussions 'sign value' takes on prominence as a way to conceptualise markets as social and cultural constructions with a system of meanings created. In marketing the dominant approach has been to define marketing behaviour as a system of value exchanges amongst a variety of parties (Bagozzi, 1975). For the firm, pricing is viewed therefore as an example of market behaviour that defines the economic value of the product. Within the context of market exchange, price is achieved at a certain point in time and could be considered in line with relation to production costs and/or competitor's pricing strategies. However the understanding of consumption from a CCT perspective in which it is rooted in the social, individual and psychological makes this economic understanding more complex.

3.6.2 CCT and Value

Turning to CCT research, 'meaning' is used interchangeably with 'value' as a term, further at individual and collective levels, for the substantive cultural elements and categories consumers draw from and reproduce in formulating identities and relating to other people and the natural environment (Penaloza and Mish, 2011: 10). Value is also understood as a symbolic meaning (Shankar et al., 2009), as a value added concept (Woodruff and Flint, 2006), as a linking value (Cova and Cova, 2002), and as value-in-use (Vargo and Lusch, 2004). These different notions of value are grounded in the different fundamental assumptions of distinctive theoretical approaches in the social sciences – economics, sociology, anthropology and psychology. In considering value creation practices, much consumer research has focused on the consumer-object meaning making process (Belk, 1988; Gabriel and

Lang, 1995; Lury, 1996) however there has been little research on the impact of other actors or practices. In marketing theory, domains of value are not separate and abstracted but are interrelated. CCT research has long grappled with these complex notions of value and since its emergence the notion that semiotic value is created between marketer and consumer has been evident (Levy, 1959; McCracken, 1986). When considering value from an emic level, it can be associated with socio-cultural contexts and market co-creation. However, this socio-cultural approach to value has been critiqued for being reductionist in nature (Karababa and Kjeldgaard, 2014), with cultural categories and principles (McCracken, 1986) the construct of value as simply 'meanings' (Thompson and Troester, 2002; Venkatesh and Meamber, 2006) and the economic lens of value as exchange and 'use value' (Bouchet, 2007; Yip, 2012). Research has problematised the assumption that value creation practice only occurs in the context of exchange (McCracken, 1986; Holbrook and Hirschman, 1982; Penaloza, 2001; Vargo and Lusch, 2004).

In an early attempt to bridge semiotic value and exchange value, Holbrook and Hirschman's (1982) seminal article articulates the importance of experiential dimensions of consumption – fun, fantasy and emotions. These hedonic aspects of consumption are also shown by Babin et al. (1994) to enhance identity value. This highlights that experiential value in the market context contains both semiotic value and economic value. Holbrook (1999a) further conceptualises consumer value as the interactive, experiential and subjective relation of goods and he identifies the dimensions of aesthetics, efficiency and spirituality as important. Aesthetic value is considered to emerge as consumers experience beauty and pleasure through style and design (Holbrook, 1999a). Linking or relational value is derived from an object or

practice when it allows consumers to form a connection with other individuals to form a community (Cova, 1997; Arvidsson, 2011). Garage sales are one site that have been illuminated as enhancing linking value, as they create a strong sense of "we" between buyer and seller (Herrmann, 1996; Lastovicka and Fernandez, 2005). Consumer value literature highlights the context specific nature of value (Holbrook, 1999a). In this contextual focus and development around universal social values and materialism, cross-cultural CCT research depicts that social values alter in different contexts (Ger and Belk, 1996; Venkatesh, 1995; Wallendorf and Arnould, 1988). In the study of micro-cultures, the value systems formed are shown to be intertwined with consumer's experiences and cultural meanings (Thompson and Troester, 2002).

In a particular focus of CCT orientated research, value has been viewed from a semiotic perspective in which it is seen as sign value or meaning (Baudrillard, 1993; Graeber, 2001; McCracken, 1986; Levy, 1959). In this respect, meaning is created through consumption and is continually reconfigured by multiple actors. One influential research stream focuses on how companies and consumers interact to co-create value in terms of co-production (Bendapudi and Leone, 2003; Etgar, 2008; Grönroos, 2006; Wikström, 1996; Woodruff and Flint, 2006) and consumer involvement (Andersson et al., 2008; Kalaignanam and Varadarajan, 2006). In many circumstances, consumers are also viewed as value co-creators, using their skills and knowledge to produce or to create the objects of their own consumption as *'prosumers'* (Bagozzi and Warshaw, 1990; Toffler, 1980; Xie et al., 2008), or working consumers (Arvidsson, 2006; Cova and Dalli, 2009; Gabriel and Lang, 2006; Zwick et al., 2008). Consumers may also co-create value not only by participating in the market or company, but also by out performing companies or

marketers through defiant or oppositional consumption practices, such as consumer empowerment (Denegri-Knott et al., 2006; Kozinets and Handelman, 2004) and consumer resistance (Dalli and Corciolani, 2008; Holt, 2002; Kozinets, 2002; Pongsakornrungsilp et al., 2008), or 'cultural hijacking' (Kozinets and Handelman, 2004; Izberk-Bilgin, 2010). Due to the dynamism and multi-dimensionality of value (Lawrence and Phillips, 2002; Sánchez-Fernández and Iniesta-Bonillo, 2007; Vargo et al., 2008), value co-creation processes often depend on how consumers interpret market offerings, marketing communication, quality, performance and value. When considering the consumer perspective, the potential 'use value' for the consumer has to be acknowledged (Yip, 2012). In applying a CCT approach, through which consumption is viewed as social, psychological and economic, the understanding of value from the consumer perspective is seen as more complex in nature. Denegri-Knott and Molesworth (2009) in their discussion of eBay with objects as stock and latent value sources, assert that the relationships of consumers with possessions as stock, may produce 'promiscuous' consumer behaviours - they define this as *"casual, opportunistic and short-lived relationships with goods for the sake of* variety or as means to sample goods before committing to long-term ownership" (2009:308). A consumer may perceive a personal possession as valuable but it is only through exchange that the economic value is fully realised (Denegri-Knott and Molesworth, 2009). Their study suggests that value, in particular economic value, is axiomatic to practices of disposal. They found that the spatial architecture of eBay intensifies the professionalisation of practices. This focus on disposition in their study highlights that objects are disposed through re-commoditisation through e-Bay, in which the objects become stock that provide monetary value to the seller. They
show that the object's life is extended and the seller's capital is enhanced through perception of being seen as a bargain hunter or skilful seller.

This focus on co-creation practices of value has been a widely discussed area within CCT (Holt, 2004; Gummeson, 2008; Lusch et al., 2010), value is thought of as meaning and how producers and consumers co-create this meaning in the marketplace explored. From a CCT perspective, Arnould, Price and Mashe (2006) stress that consumers may derive value from value propositions in a more creative fashion that the firm had originally intended. This highlights that the way in which value creation was proposed by firms does not always translate into the one materialised in consumer practice. In the literature much attention has been paid to how consumers engage in the co-creation of value through individual co-creation experiences and interaction with brands, companies and other consumers (Prahalad and Ramaswarmy, 2004). Muniz and Schau (2005) examine the case of the Apple Newton brand community, in which the stigma of an 'abandoned' brand led consumers to co-create the brand meaning, thus extending the brand's life. For Apple Newton, consumer faith in the brand emerges through the co-creation of myths and religiosity in a largely online community, and this faith kept the brand alive for a select group of consumers. Further in 2009, Schau et al. identify the creation of brand value within brand communities through a network of actors. In considering fields of consumption and practices, Arsel and Thompson's (2011) study establishes how market-facing actors' values shift within their field dependent identity in the hipster scene. They show how semiotics of the hipster style shifts into exchange values and mass marketisation. Developing this focus on multiple actors, Karababa and Kjeldgaard (2014:5) argue that the concept of value can be seen to necessarily

develop further from a linear value chain perspective, to a view of value as cocreation through the interactions of multiple actors.

3.6.3 Disposal Practices and Value

Exploring product value discussions back to the consumer research landscape the focus has been extended to illuminate practices of disposal and re-configuration. Cappellini's (2009) study of consuming the leftovers of family meals highlights a reversal of many taken for granted assumptions regarding value. Her study explores how practices of disposal, revaluation and reuse allow for identity in individual, family and community contexts to be materialised. Further Penaloza and Mish (2011) discuss that a product's value is closely linked to its perceived disposability and Parasuraman and Grewal (2000) suggest "redemption value" as the enduring benefit at the end of the object's life. Türe (2014:3) develops this in her assertion that value regimes allow objects to move across cultural boundaries, among parties with non-similar interests or standards of valuation. As with Appadurai's (1986) theory as explored in Chapter Two, each exchange is open to the individual's value perceptions, consumers can therefore remove and re-enter objects away from their prescribed paths, through consumer creativity (Thompson, 1979). Türe (2014:3) contends that in light of this perspective, the value of objects is temporally and culturally constructed through paths and diversions and the relating social relations, power contests and value regimes. Her study highlights the complex relationship between acquisition, usage and disposition and reflects that an object's value is:

"dynamic, shaped by consumers' commitments to different and usually conflicting value regimes (altruism vs. frugality) and constructed beyond the dyadic relation between objects and consumers through the inclusion of imagined or actual value partners and their value estimates "(Türe, 2014:16).

This view places emphasis on value construction as transferable and based on consumer competence, consumption practices and the surrounding social network. In exploring the relation between value and disposition, her study stresses the importance of movement for the creation of value. This reiterates the findings of Gregson et al. (2007) as they found consumers are influenced not only by their own competences but also by their social positioning.

In Gregson and Crewe's (2002, 2007) work, when goods were seen to be disinvested through sale, valuation becomes situational and contextualised through information on what the goods could be potentially worth. Gregson and Crewe (2002, 2007) found that if an object was perceived as too common, too used or too fashionable, the subsequent economic value would be affected. They propose that labour was involved in moving the object forward and this would suggest that complex practices are necessary to move it forward and release the value (Parsons, 2008; Denegri-Knott and Parsons, 2014; Epp and Price, 2010; Cappellini, 2009; Cappellini and Parsons, 2012). Gregson and Crewe (2002) illustrate value rather than being intrinsic to the object, is shaped by consumer use, practice and interaction. They therefore argue that the symbolic meaning is affected by value sources such as object styling and location of sale. Adding to this discussion, Parsons' (2008) work on antiques and value construction posits that the movement, flow and circulation of objects are overlooked in the discussion of alternative markets: objects are 'moved on' by the action of the marketplace and the circularity of the production and consumption process highlighted (Parsons, 2009). This has important implications for potential value

consideration of the object, if the marketplace is viewed, as constantly in transition, with objects moving along the consumption trajectory, then value cannot be deemed to be fixed. Using *Thompsons' Rubbish Theory*, Parsons' (2008) argues that value is emergent through consumers' ways of seeing and placing objects. However, Parsons proposes that Thompsons' theory does not fully consider the practices of value creation. In the literature, value is emergent and consumers utilise and "absorb objects into their lives" (Parsons, 2008:393) as they require. She suggests that rather than being intrinsic to an object, value is an emergent process as a result of how the object is seen and placed. This again highlights the importance of materiality in a discussion of practices. Further in Denegri-Knott and Parsons' (2014:93) study on 'Disordering things' they explain: "turning to the marketplace, we think that secondhand sites and spaces of exchange might also be instructive in thinking through the reversal and transgression of dominant norms or orders found in commercial market spaces." They suggest that these spaces represent locations whereby ambiguity and disorder reside and the linear nature of capitalist logic is challenged (Denegri-Knott and Parsons, 2014:94; Slater, 2014). This thesis would problematise this assertion of organising and ordering further in relation to practices of value.

A focus on practices such as transformation or reuse during disposition can (re) associate objects with associated regimes (Türe, 2014; Gregson and Crewe, 2002; Parsons, 2008; Cappellini and Parsons, 2012) by reasserting their aesthetic value. For as Entwistle (2009) reminds, aesthetic value must be seen as something that is not necessarily a quality intrinsic to the object per se, rather it becomes processual and an attribution of value that is gained over its journey. As a result, the process of qualification has to be continuous and commodities need to be examined

continuously and should remain open to interpretation and transformation as they move from one actor to another in the marketplace (Entwistle, 2009: 68). Here aesthetic value is *"created, accrued and attributed"* along the network in which the object moves (Entwistle, 2009: 58). In order to perceive value, frameworks of meaning and understanding must be present to situate the object and pass a valuation judgment. A mobilisation of knowledge around the objects can be seen to extend beyond their current state and use. This understanding of the object's inherent aesthetic value displays a layer of work but further labour is required to manifest the economic value in the marketplace.

3.6.4 Contemporary Considerations of Value within CCT

Considerations of value within the CCT dialogue have continued to elicit scholarly discussions and Arnould (2014:3) suggests moving away from a typology of value (Holbrook, 1999a; Karababa and Kjeldgaard, 2014; Türe, 2014) to think in terms of 'regimes of value creation' and the relationship between them. He questions that in a *'praxeology of value'* existing models of value creation may be built upon to ask what kinds of value are produced within the regimes and what classes of value are produced in each and how do they vary (Arnould, 2014:3). Penaloza and Mish (2011) further propose that within the marketing system, different stakeholders produce value via discourse and practice employing interactions, meanings and artefacts. Karababa and Kjeldgaard (2014) move this forward by announcing a *'cultural paradigm'*. Based on Graeber's (2001) tripartite systems are of value (economic, sociological and semiotic); they highlight the *"interrelatedness of value,"* arguing:

"...marketing is involved in the evaluation of the socio-cultural differences and the articulation of the economic worth of these differences...marketing can be understood as a practice of configuration of commodified value system potentials rather than a meaning transfer institution." (Karababa and Kjeldgaard, 2014: 6)

This statement echoes Graeber (2005:45) who depicts in advanced consumer cultures the key value is the self. Arnould (2014:3) concurs and suggests: "the implicit project of consumption is restoring the subject divided by capitalism to unity." From a cultural anthropological view, macro theories of value that focus on exchange and the circulation of objects aid our understanding, as they go further than the moment of exchange to describe the circular nature of value (Graeber, 2001). In applying a socio-cultural approach the interrelatedness of the social; economic and semiotic (Graeber, 2001) strands of value construction are highlighted. Based on Graeber (2001) and Simmel (2004) value creation can be seen to be neither objective nor subjective but rather is grounded in the contingent effect of interaction. Arnould (2014:2) proposes value resides in the actions and interactions, which resources make possible or support. The Karababa/Kjeldgaard framework is further added to by Venkatesh and Penaloza (2014) who suggests that it could be further developed to include social and environmental issues, they add 'socio-cultural value' and 'environmental value' to the market value system. This addition is especially relevant in the global and cross-cultural context. These recent discussions depict that value is still very much a CCT research concern and as Arnould (2014) suggests, a practice focus may illuminate further aspects to the complex concept of value.

3.7 Limitations of Practice Theory

Before concluding this chapter it is important to consider the use of practice theory in this CCT thesis. As discussed in this chapter within practice theory a variety of conceptualisations of practice theory exist and with this limitations and ambiguities are present. As this chapter has shown, practice theory does attempt to allow for a coherent critique of action however within the academic world, it is still a battle to provide a persuasive alternative conceptualisation that does not focus on individual choice and decisions, but privileges habit and routine. The theoretical mapping by Reckwitz (2002), detailed at the start of the chapter, drew out the theoretical narratives of practice theory against the broader domains of 'cultural theories'. Conversely in his ontology of practices, Schatzki (1996) 'side tracked' objects as an outcome of practices and this is problematic for those interested in the social and cultural manifestations of consumption, in which objects are crucial. Reckwitz (2002:250) argues that a practice "necessarily depends on the existence and specific interconnectedness of these elements, and which cannot be reduced to any one of these single elements," he makes it clear that objects and their subsequent use are central to the reproduction of practices in everyday life. Using the context of food consumption amongst Pakistani Danes, Halkier and Jensen (2011) propose that two important implications of the use of practice theory is firstly to see all types of qualitative data as enactments of social action and secondly to enable the *'construction of analytical generalisations that are not based on methodological individualism.* They contend that there are operational difficulties in the analytical translation between applied practice theoretical concepts and operative methodological procedures are therefore necessary. Halkier and Jensen (2011)

explore the methodological implications for using practice theory in consumption studies; they argue that using practice theory enables a conceptualisation of consumption as processes of practical and social accomplishment and an empirical focus on '*ways of consuming rather than a focus on individual choice*'. Practice theory has also been critiqued for limited application to the macro level.

3.8 Conclusion

3.8.1 Positioning This Research and Key Literature Gaps

Against the background of the continuing debate in social theory and philosophy regarding the structure-agency problematic, scholars have turned to theories of social practices. As this chapter has shown a practice approach presents a way to move beyond the dualisms of other traditions of social/material, actor/structure, body/mind, and theory/action. Practice theories are an assembly of philosophical and cultural accounts that focus on the practical conditions of carrying out social life (Halkier et al., 2011:3). A practice lens from Ortner (1984), Schatzki (2001, 2002) and Reckwitz (2002) produces a radical shift in understanding of the social. In viewing the social world as actively constructed rather than passively absorbed the practices of the individual become of paramount importance.

From the practice theory perspective, consumption occurs within and because of practices (Warde, 2005). Objects are argued to be consumed by skilled practitioners to engage in certain practices, this in turn requires the appropriate consumption of relevant goods and services and that value is emergent from consumer interactions and practices and rather than being intrinsic to objects it is shaped by consumer use and perception (Parsons, 2008; Gregson and Crewe, 2002; Thompson, 1979; Türe,

2014; Denegri-Knott and Parsons, 2014:94; Slater, 2014). Hand and Shove (2007) propose that the practice itself, the doing and the knowing how to do, holds the dimensions of materiality, conventionality and temporality together. The literature illustrates the importance of exploring both the range of practices, dominant practices of the consumer and the value created. It is further crucial to question how people are positioned within those practices and how they engage and how value is manifested. The issue of what level of commitment to differing practices is also important, as is how consumers develop an understanding of the practice, what is required and their role within the practice. Within consumer research the common assumption has been to consider that objects follow a linear trajectory of valuation, consumption and decline, however as the literature (Türe, 2014; Parsons, 2008; Gregson and Crewe, 2002; Thompson, 1979; Cappellini, 2009; Cappellini and Parsons, 2012; Denegri-Knott and Parsons, 2014:94; Slater, 2014) has shown objects move through spaces, different contexts and biographies and this disrupts this linear conceptualisation as not accommodating the circularity in the movement of objects and consumer practices. However an exploration of the practices that an object must encounter to become valued have been under explored in the literature (Parsons, 2008:390). In adopting a cultural perspective and building on Karababa and Kjeldgaard's assertion (2014: 6) that the value of an object is dynamic, subjective and context-dependent, that it has to be "constantly co-created within a network of actors," value will be further explored in this thesis from a socio-cultural perspective in which value will be considered "a bundle of multiple values created by practices" of a collective of networked actors (Bourdieu and Wacquant, 1992).

3.8.2 Applying Practice Theory

Within this thesis the application of practice theory will contribute to CCT studies as it focuses research on the performative processes of social life, which involve consumption activities, whilst at the same time not lessening the emphasis on cultural conditioning. From the literature contained in this chapter, practices are formed as the resources of consumers interlink with different contextual elements (Reckwitz, 2002) and practices are conceptualised as 'doings' rather than cognitions or emotions. As this chapter has detailed, within the theoretical turn to practice theory multiple perspectives of practice have emerged. Schatzki's (1996) ontology of practices highlights two different notions of practice: practice as coordinated entity and practice as performance. In conceptualising practice as entity, scholars build on the notion of 'a temporally unfolding and spatially dispersed nexus of doings and savings' (Schatzki, 1996:11) whilst practice as performance is seen as processes of doing through which practice is sustained and reproduced (Shove and Pantzar, 2005). Within contemporary debates over the practice turn in consumer research (Schau et al., 2009; Murphy and Patterson, 2011; Arsel and Bean, 2013; Magaudda, 2011) the literature has presented a plethora of ways in which the two notions around practice have started to be employed in discussions of consumption. This turn to practice marks an ontological departure in consumer research and underlines that consumption choices are more directly connected to engagement in particular practices than they are to individual taste (Warde, 2005; Murphy and Patterson, 2011). The particular contribution of practice theories to consumer research allows for a focus on the performative processes of social life, which by necessity involve consumption activities, while not diminishing the importance of either the cultural

conditioning of consumption, or the consumption of practitioners (Halkier et al., 2011). As such, this thesis builds on Schau et al.'s (2009) understanding of practices as:

"Practices need to be known in order to be repeated and must be repeated to become part of the value creation repertoire. By providing opportunities to demonstrate competencies, practices allow members to accrue cultural capital through adroit performance which creates value for the consumer" (Schau et al., 2009: 27).

In this thesis using the practice lens and understanding culture as a field of practices, the analytical focus will shift the focus from individual experience, to culture and the vintage collective as a site of embodied social action (Bourdieu, 1984; Shove and Pantzar, 2005; Entwistle, 2000a). This perspective is thus orientated on the practices through which culture gains meaning and value in the marketplace (Arnould, 2014). The practice view proposes that value creation happens when actors undertake practical things in their life. New variations of practices can occur as consumers start to do things in new subjective ways. Practices have been seen to tie value creation to a specific socio-cultural and spatio-temporal context. A turn to practices in this thesis allows for a focus beyond the consumption meanings and to the use value of material objects and the everyday doings of people. Murphy and Patterson (2011) argue that a practice theory approach provides the opportunity to examine consumption phenomena in a more holistic and concrete way. In using practice theory within the CCT context, it allows the researcher to focus on the 'doing' of social practices rather than the moments of individual decision-making within the vintage collective. Consumers will be conceptualised as competent practitioners who negotiate and perform varied practices in their daily life (Shove and Pantzar, 2005) and practitioner

competence will be regarded in line with Watson and Shove (2008: 71) as the skills implied in the use, integration and desiring of items required for the effective accomplishment and performance of everyday life.

Chapter Four: Methodology

4.0 Introduction

This chapter discusses the research methodology utilised in this study. It considers the philosophy, reasoning and approach to this thesis. It acknowledges how the data has been collected and highlights the rationale for the decisions made. It outlines the research aims and objectives that emerge from the literature review of Chapter Two and Three. The research philosophy from which this study emanates is discussed and justified and the qualitative approach within the interpretive framework is explored. As outlined in the previous chapter, a turn to the Practice Theory perspective has influenced the research philosophy and strategy, as such this research utilises Interpretivism, in which the notion of the social self as being dynamic and fluid is central. The chapter then establishes the research as sitting within the ethnographic tradition of methods (Geertz 1973; Levi-Strauss, 1967; Agar, 1996; Spradley, 1979) which has an established tradition within the social sciences (Madden, 2010) and within Consumer Culture Theory (CCT) (Arnould and Price, 1993; Schouten and McAlexander, 1995; Kozinets, 2002; Arnould and Wallendorf, 1994). The research design is detailed and discussed with consideration given to the role of the researcher and ethical issues encountered. The data collection process is described in detail as this research specifically uses semi-structured depth interviews, observation with field notes and visual analysis to collect detailed ethnographic data surrounding around the practices of the marketplace. Methods for analysing and evaluating the data are outlined and explored with consideration as to how they are applied in the context of this study. The chapter finally concludes with a consideration of the

limitations. Within each of these sections a review of the relevant literature is provided and justification for the approach taken within the thesis presented.

4.1 Research Aim

This thesis aims to gain an understanding of how the practices, spaces and socialites intersect in the consumption of vintage. Using the context of the 'pop-up' vintage market this exploration requires a non-positivistic approach that aims for an understanding and interpretation of these practices in the marketplace.

4.2 Research Strategy

The literature review provides the basis of the research strategy with a focus on practices, consumption theory and specifically the domain of CCT. The qualitative research tradition is explored in detail and as Maxwell (2005) underlines, sequential models for qualitative studies are not appropriate, as the research process is seen as iterative in nature and the research design is argued to be a "*reflexive process operating through every stage of the project*" (Hammersley and Atkinson, 1995: 25). Maxwell (2005) adds to this and argues that there are three main activities in the research study: collecting and analysing data, developing and modifying theory, elaborating and refocusing research objectives. This exploratory research study adopts a multi method focus, as Denzin and Lincoln (1999: 5) assert, qualitative research is inherently multi-method in focus, as this allows for an in-depth understanding of the subject of the study. This plurality of methods is an important aspect of qualitative research and Flick (2002: 229) adds to this arguing that multiple methods allow for "*rigour, breadth, complexity, richness and depth*" to the inquiry. This study uses a multi-method ethnographic approach to capture how the practices

of vintage consumption are undertaken in a marketplace and the materials, values and competences of the actors. The researcher assumes an interpretivist position, being interested in the meanings, practices and active interpretations of vintage.

4.3 Research Philosophy

Positivism and interpretivism have differing underlying philosophical assumptions, as the table below highlights. Positivists believe that a single, unchanging objective reality exists and that it is possible to make specific observations of the world. In contrast, interpretivists believe that multiple realities exist and that reality is a mental construct that is constructed by individual and social perceptions. Interpretivists therefore propose that society is a changing entity that is constructed by people themselves (Brewer, 2000). With regards to human behaviour interpretivists argue that people actively create their own environment, whilst positivists believe that behaviour is caused by influences outside the individual. In line with the research aims, this research study employs the philosophical traditions of interpretivism. Adapted from Hudson and Ozanne's (1988: 509) work, the table below highlights the differences in approach between Interpretivists and Positivists:

Assumptions	Interpretive	Positivist
Ontology		
Nature of Reality	Socially constructed, contextual, multiple	Objective, single, tangible
Nature of Social Beings	Proactive, voluntaristic	Reactive, deterministic
Axiology		
Overriding Goal	An understanding based on ' <i>Verstehen</i> '	Prediction- explanation through subsumption, generalizability
Epistemology		
Knowledge Generated	Idiographic, context dependent	Nomothetic, context independent
View of Causality	Multiple, simultaneous shaping	A real cause exists
Research Relationship	Interactive, cooperative, no privileging of observation point	Dualism, separation, privileges point of observation

Table 2 – 'Summary of Research Approaches', source: Hudson and Ozanne (1988: 509)

In considering the axiological assumptions, positivists believe in explanation through a generalisable set of universal laws that allows them to be able to predict behaviour. For interpretivists the fundamental belief is of gaining an understanding of behaviour or of *'Verstehen'*. Originating from Weber (1947) the concept of *'Verstehen'* refers to the understandings of subjective meanings that are attached to behaviour. There is an emphasis placed on empathic identification from the researcher to the subject (Patton, 2002). This main tenet of *'Verstehen'* fits well with the aim of this research study of gaining an understanding of the practices of the vintage collective.

4.3.1 Ontological Position

A researcher's ontological position provides the starting point from which to construct a conceptual framework, which in turn allows them to see and make sense of the social world around them, but it also defines the basic assumptions that they make about the nature of both reality and knowledge (Guba and Lincoln, 1998). This emphasis on the importance of the researcher's perception of reality is accentuated in Spiggle's (1994: 491) seminal article on methodology as she argues that: "*debate centres on the question of how we can know and represent what we know about reality*." Within the social sciences, ontology could be separated into two positions: positivism is rooted within the realist ontology and holds the belief that a reality exists that can be objectively discussed, in contrast interpretivism stems from a postmodern sensibility through which there is a belief in multiple realities and that reality is a social construct that is dependent upon the perceptions of individual actors (Guba, 1990; Denzin and Lincoln, 1998). The philosophical assumptions of interpretivism indicate that the interpretive paradigm is the appropriate mode for the current research aims and research objectives. The following section explores this with a discussion of the ontological position and CCT landscape.

4.3.2 Interpretivism within Consumer Research

As highlighted, this study sits within the interpretivist tradition that is closely associated with qualitative inquiry. Although qualitative methods are used in positivist studies, they are considered within that tradition to have less rigour (Denzin and Lincoln, 1998). Originating from the idea that the world and reality are not objective and exterior, rather that they are socially constructed and attributed meaning by individuals (Husserl, 1946), builds into Geertz's (1994) notion that the researcher should appreciate the individual constructions and meanings that people ascribe to their experiences, rather than focusing on patterns, occurrences and fact gathering. Within the wider tradition of qualitative inquiry, it can be seen to encompass a broad range of methods and activities (Denzin and Lincoln, 1998) and cuts across disciplines and research contexts. Creswell (2007) argues that there are

five different commonly used approaches to qualitative enquiry: ethnography,

phenomenology, narrative study, case study and grounded theory.

Qualitative Inquiry	Characteristics	
Ethnography	Used for investigating cultures by collecting and describing	
	data that was intended to help in the development of a theory.	
Phenomenology	Understanding the essence of experience, studies several	
	individuals with shared experiences to explain the	
	phenomenon.	
Narrative Study	Tells the story of individual experiences – exploring their	
	lives or events within to explore their individual stories.	
Case Study	Focuses on developing and providing a depth description and	
	analysis of a case or multiple cases. Typically studying an	
	event, activity or program that comprises of more than one	
	individual's perspective.	
Grounded Theory	An inductive form of research, which is based or 'grounded'	
	in the observations or data from which it was developed. It	
	uses a variety of data sources- interviews, observations,	
	surveys, reviews of records. Quantitative data can also be	
	used.	

Table 3 - Types of qualitative inquiry, source: adapted from Creswell (2007)

Interpretivists argue that there is no rigid relationship between paradigm and

methodology and that there should not be a privileged position over data collection

or analysis techniques. As Denzin and Lincoln suggest:

"Qualitative research, as a set of interpretive activities, privileges no single

methodological practice over another" (Denzin and Lincoln, 2004:6).

In practice, those with an interpretivist mind frame typically engage in qualitative inquiry. For interpretivists the aim of research is to understand not predict behaviour. This idea of understanding is crucial and is viewed as a process rather than an end product (Hudson and Ozanne, 1988). Qualitative methods have been considered a holistic form of inquiry. The interpretivist's research approach is one of constantly

evolving research design (Hudson and Ozanne, 1988). Oualitative methods highlight a depth approach to understanding, through which a rich or "thick" description of the area is involved (Geertz, 1994). This aim of thick description has according to Denzin (1989:33) four key elements: firstly it gives context of an action, secondly it states the intentions and meanings that organises the action, thirdly it traces the development of the act and lastly it presents the action as a text that could then be interpreted. The role of the interpretivist researcher then in accordance with Denzin's (1989:3) outline is to bring this world of lived experience and make it accessible to the reader. Fine and Deegan (1996: 6) note: "qualitative researchers are explorers making 'first contact' with alien civilisations, courting experience through observing diverse times and places, later reanalysing and rewriting, hoping for heightened awareness to ignite insight." This alludes to the multifarious nature of consumer behaviour as social, complex, often irrational and unpredictable (Venkatesh, 1992) and as such qualitative methods have become popular for studying consumption from a macro perspective (Pettigrew, 2000). Consumers cannot be considered in the same manner as inanimate objects as Szmigin and Foxall (2000: 188) comment they need to be viewed as "in a process of continuous emergence." In reaction this creates an active role for the researcher, according to Denzin and Lincoln (2004) the interpretive researcher as 'bricoleur' understands that research is an interactive process but that the product is a complex, guilt-like bricolage or a set of fluid interconnected images. The interpretive ways of seeking knowledge have become more accepted in the study of consumers (Hudson and Ozanne, 1988; Belk, Sherry and Wallendorf, 1988; Holbrook, 1986). Within consumer research, interpretive consumer research emerges in the 1980s, with Belk et al.'s, 1989 'Odyssey' paper

marking a shift in the accepted perspective of the marketing discipline. Following the interpretive paradigm consumer researchers have employed a wide variety of methodological approaches from participant observation (Schouten and McAlexander, 1995; Belk et al., 1998), phenomenology (Thompson et al., 1990; Thompson and Haytko, 1997), in-depth interviewing (Hirschmann, 1994; McAlexander et al., 1992; Belk et al., 2003), ethnography (Penaloza, 2001; Arnould and Wallendorf, 1994), grounded theory (Goulding, 1999, 2000) hermeneutics (Arnould and Fischer, 1994), semiotics (Mick 1986, 1988), symbolic interactionism (Soloman, 1983), projective techniques (Rook, 1988) and psychoanalytic techniques (Holbrook, 1987). Hogg and Maclaran (2008: 132) assert that interpretive consumer researchers are faced with the issue of convincing readers that their knowledge is worth paying attention to when it develops from a field-dependent situation, when it is offered as an interpretation rather than 'absolute knowledge' and when it is presents reality portrayed through description and conceptually-mediated analysis of social experiences. They suggest that based on Golden-Biddle and Locke (1993) that interpretivist consumer researchers need to ground their accounts with 'authenticity, plausibility and criticality' (Hogg and Maclaran, 2008: 133). This has important implications for the research that will be further discussed in relation to the analysis of the data.

 Table 4 - Overview of analytic framework for interpretive consumer research. Source: adapted from Hogg and Maclaran (2008: 133)

Authenticity	Convincing the readers that:	
	The interpretation is drawn from the data.	
	The researcher has 'been present in the field and grasped how	
	members understood their world.'	
Plausibility	Accounting for as much of the information as possible, so that there	
	is some degree of well argued 'fit' between the information (or data),	
	and the explanation offered to account or the interpretation offered of	
	the data.	
Criticality	Incorporates reflexivity:	
	'Carving out room to reflect, provoking the recognition and	
	examination of differences, and enabling readers to imagine new	
	possibilities.'	

Hogg and Maclaran (2008) suggest that within the wider qualitative research domain it is still difficult to persuade an audience that the resulting knowledge is deserving of merit. When considering qualitative data, the researcher has therefore to ensure that they are convincing of the study's worth. Golden-Biddle and Locke (1993) suggest that this is achieved through the writing and focusing on authenticity, plausibility and credibility. Denzin and Lincoln (2004) add to this and assert that trustworthiness is considered the indicator of merit in a qualitative study; this is based on the study's credibility, dependability, confirmability and transferability. These issues are recognised and further considered in the analysis of the data in this study.

4.3.3 Consumer Culture Theory and Ontological Positioning

Consumer research along with many other disciplines has a long tradition of positivist research and a quantitative preference that is associated with this approach. However there is an increasing focus upon post-positivist, interpretive scholarship particularly coming from Europe (Belk, 2006; Askegaard, 1991). Within the consumer research tradition, Hudson and Ozanne (1988) argue that there are multiple ways through which to seek knowledge about consumers and they note that each has its own merit. Both qualitative and quantitative research methods have roles to play in allowing for theoretical contributions to be made to consumer research (Ger, 2005). Consumer research is still dominated by positivistic research however interpretivism has become a "well-trodden path" (Hogg and Maclaran, 2008:1). As discussed in the literature chapters, this could be attributed to the philosophical conceptualisation of consumers that has shifted over the past century - with consumers moving from "passive" to "active" (Firat and Venkatesh, 1995:240). This move towards conceptualising consumers as individuals that participate "in multiple interacting cultures" (Belk, 1995b: 62) lends itself to gaining an understanding of them, rather than a prediction of behaviour. From a historical perspective, Belk (1991, 1995b) identifies that the methodology underpinning consumer research has evolved in four distinct stages. He outlines that the first stage was the 1920/1930s and that consumer research was focused predominantly on understanding and predicting consumers' needs within a supply and demand context, in which the consumer was viewed as a rational being and cognitive agent. Secondly he asserts that in the 1950s consumer research was centred on the emotional relationship between consumer and goods. In the third stage in the 1960s, he suggests that consumer research was grounded in positivistic scientific methodologies through which they generate models to gain an understanding of consumer purchasing behaviour. In the fourth stage in the 1980s, major changes took place, which he refers to as the "new consumer behaviour research" (Belk, 1995b; Sherry, 1991).

This new movement employs non-positivist research methods and Belk (1995b: 64) suggests that this "opened up a Pandora's box of 'new' substantive questions to be investigated." This naturalistic inquiry was spearheaded by the Consumer Behaviour Odyssey (Belk et al., 1991) and from this work they provide new understandings of the holistic consumer experience. The belief that consumers are "socially connected human beings participating in multiple interacting cultures" (Belk, 1995b: 62) emanates from this study. Although positivism remains the dominant approach in consumer research for gaining knowledge, the Odyssey led a turn to a diversification of research approaches and opened up a new era for consumer research. Moving forward within consumer research, in their seminal article Firat and Venkatesh (1995:243) argue that consumption is a complex human behaviour and could not be separated from "culture, language, aesthetics, narratives, symbolic modes and *literary expressions and meanings.*" Therefore this postmodern perspective calls for an approach that is "eclectic in thought and practice" (Firat and Venkatesh, 1995:243). CCT responds to this call, as it is not considered to be grounded in one specific research tradition or to privilege only qualitative methods. In arguing for methodological breadth, CCT research aims to reveal the "dynamic relationships between consumer actions, the marketplace, and cultural meanings" (Arnould and Thompson, 2005: 868). Algesheimer and Gurau (2008) suggest that within CCT research methodology is focused and motivated by the level at which culture is being conceptualised and explored. In considering ontology, CCT argues against methodological hegemony. Fischer and Sherry (2009:1) suggest that as a discipline it has emerged to allow for "interpretive perspectives on consumer phenomena." The growing link between CCT and qualitative inquiry is explained by Sherry (1991) as it focuses on the experiential and socio-cultural dimensions of consumption that are not easily accessible through experiments, surveys or data modelling. This focus on understanding participant's point of view - the emic perspective - allows for researchers to highlight broader cultural meanings - the etic perspective. These methodological preferences could be argued to be driven by the *"aims that drive CCT rather than from a passion for qualitative data or vivid description"* (Arnould and Thompson, 2005). Hogg and Maclaran (2008) situate CCT as a new disciplinary brand for those working within the interpretivist tradition and one that is concerned with the 'lived' experience that cannot be removed from its socio-cultural context.

Within CCT the methodological privileging of naturalistic research from the first wave of consumer research, has come to be questioned from the late 2000s. Moisander et al. (2009a) outline the epistemic community of CCT as having implicit norms, received ideals and practices. They take a self-reflective, genealogical perspective (Foucault, 1970) and explore the evolution of what they term CCC – consumer culture community. They challenge the predominance of 'the individual as the unit of analysis' in CCT work:

"...it is important to be attentive to the specific representational systems – the socioculturally normalised and institutionalised ways of thinking, talking and representing knowledge about the consumption agent, object or activity that are available and offered by the researcher to those being studied as resources for making sense of the topic in the specific context of the study" (Moisander et al., 2009a: 20).

In a subsequent article Moisander et al. (2009b) further challenges the limitations of the individually and experientially based perspective on consumer culture, as a result of the phenomenological interview being the predominant mode of data collection within CCT. They argue that this perspective has a tendency to downplay the critical aspects of consumer's lived experiences but this does not appear in the phenomenological interview. Similarly critiquing CCT methodological positioning, Askegaard and Linnet (2011) argue for an epistemological positioning of CCT beyond this 'lived experience of consumers' in their Marketing Theory article. They argue that the contextualisation of the consumer experience needs to be expanded to include the systemic and structuring influences of both the market and of systems not always discursively expressed by respondents. They argue for a consideration of 'the *context of context.* 'Methodologically they suggest a bridging of the 'analytical terrain,' they pull on Douglas's (1986) theory of institutional agency, to give their argument broader epistemological grounding. Building on Douglas's (1986) argument of 'the prevailing thought style,' Askegaard and Linnet (2011) propose that the consumer culture researcher must include an account of the institutional thought style in the particular field that extends beyond the life-world of the consumer. Responding to the critiques of Moisander et al. (2009a and 2009b) and Askegaard and Linnet (2011), Thompson et al. (2013) discuss how consumer researchers who have employed the humanistic/experientialist discourse have reconfigured its original epistemological premises. They suggest that humanistic/experientialist CCT represents a "coherent paradigmatic alternative to information processing theories" (Thompson et al., 2013: 7). CCT has studied consumption for its own intrinsic value and its role in everyday life. They suggest that the foundational axiological premise of CCT is not of managerial relevance but aligns with the philosophical and intellectual stance of sociology and anthropology. With the first waves of CCT

research, humanistic and experientialist CCT researchers embraced naturalistic enquiry (Belk et al., 1988) and existential-phenomenology (Thompson et al., 1989) that are orientated towards documenting the emic perspective of consumers. Both of these recognise the researcher's role as a sense-making instrument. An epistemic shift they propose has resulted from internal tensions and the norms of naturalistic research displaced by more open ended, hermeneutic and post-structuralist epistemic principles. Thompson et al., (2013:9) asserts that CCT researchers support their interpretive claims through rhetorical means by:

"...systematically explicating layers of cultural meaning, elucidating sociohistorically grounded connections between emic articulations and cultural and ideological frames, and demonstrating novel theoretical insights through comparisons to more orthodox or established theoretical frameworks." (Thompson et al., 2013:9)

They stress that CCT researchers and the contemporary CCT heteroglossia is axiomatic and that personal understandings take inspiration from complex cultural systems and are discussed within specific socio-cultural fields and unfolded through specific social structures, power relations and marketplace structures (Thompson et al., 2013:11). Thompson et al.'s (2013:13-14) discussion illustrates the development of the CCT tradition: that the first wave of CCT research primarily draws upon humanistic/experientialist discourses and has since developed to afford a broad range of "dialectical and dialogical theorisations that focus upon recursive relations between structure and agency." Moving forward Thompson et al. (2013:18) call for CCT to begin to "disentangle the material and discursive webs that both sustain and

potentially destabilize these global networks of political, cultural and socioeconomic distinctions and hierarchies."

4.4 Epistemology

The previous sections highlight that the ontology of the research study is coming from the interpretivist tradition. In the following section the related epistemology is discussed in further detail. Epistemology considers how knowledge and meaning are understood. Hudson and Ozanne (1988) separate this into three elements: how knowledge is generated, the research tradition approach to causality and finally the relationship between the participants and the researcher. These are considered throughout the discussion of the paradigm that the research emanates from.

4.4.1 Ethnography

Originating in anthropology, ethnography is originally used to study small indigenous groups however it became an accepted method in the exploration of consumer behaviour. Embedded in cultural theory, ethnography could be thought of as a form of naturalistic inquiry with a specific interest in culture (Pettigrew, 2000). Ethnography seeks to establish the context and significance of experience for particular groups but also attempts to convey the comparative and interpreted cultural significance of this experience (Pettigrew, 2000). Stemming from the view that the world and reality are not objective and exterior, but are rather socially constructed and given meaning by people (Husserl, 1946), the researcher focuses on appreciating the different constructions and meanings that people place on their experiences (Geertz, 1994). Ethnography is based on aiming to understand people's actions and experiences of the world and how they reflect back on these (Brewer, 2000). Ethnography is suitable for gaining 'intimate familiarity' with the social world and

'capturing the voices of people who inhabit it' (Brewer, 2000).

Table 5 - Characteristics of Ethnographic Studies. Source: adapted from Hammersley and Atkinson (1983)

The culture that drives the behaviour of people in a small group was studied in its everyday settings and contexts.

The researcher participates in the group for a long period of time, so that he or she has to live two simultaneous lives - one as an insider in the group and the other as an outside observer of the group.

Observations and relatively informal interviews were the major sources of information, although other sources could be used.

The data was gathered in a relatively unstructured way so that data collection was 'played by ear' and flexibly covers as much of the everyday life of people as possible.

The masses of data that were rich descriptions of the group were analysed to extract the meanings and functions of the group's behaviour and the report takes the form of verbal descriptions and explanations.

As demonstrated in the above table, as a form of inquiry, ethnography involves

observing and analysing behaviour in naturally occurring conditions. For a researcher

'ethnography' means 'the writing of culture' (Atkinson, 1992). Ethnographers aim to

look beyond what people say to understand the shared system of meanings called

'culture' (Goulding, 2005:298). The ethnographic research approach is seen as

labour intensive and is characterised by fieldwork. As Geertz's (1973) influential

work notes:

"From one point of view, that of the textbook, doing ethnography is establishing

rapport, selecting information, transcribing texts, taking genealogies, mapping

fields, keeping a diary and so on. But it is not these things, techniques, and received

procedures that define the enterprise. What defines it is the kind of intellectual effort

it is" (Geertz, 1973:6 cited in Hudson and Ozanne, 1988).

As Geertz (1973) highlights the labour involved in ethnography is seen as immense – both physically and mentally. The necessary immersion in the field according to Emmerson et al. (1995) provides the fieldworker with access to the fluidity of lives and also improves their sensitivity to interactions and processes. Goulding (2005) asserts that fieldwork is the hallmark of ethnography. Within the ethnographic process, the researcher attempts to capture the interactions with the phenomena verbatim - the emic perspective - however the researcher does not retain this perspective when writing up the results - the etic perspective (Hudson and Ozanne, 1988). This is again illuminated by Madden (2010) as a key characteristic of ethnography, to explore the relationship between an 'emic' and 'etic' understanding of human behaviour (Madden, 2010: 19). An 'emic' perspective considers the insiders' view point or participants' stance and an 'etic' perspective reflects the outsider or researcher's viewpoint. The importance of ethnographic reflexivity requires that the researcher does not lose sight of their own etic perspective in the field as reflexivity is crucial in the explaining the particular human phenomena against a wider cultural backdrop (Madden, 2010: 19-20). This importance of the wider cultural framing is depicted by Arnould and Wallendorf (1994: 485) who assert that ethnography was: "not just a form of data collection; it aims to clarify the ways culture (or micro-culture) simultaneously constructs and is formulated by people's behaviours and experiences." Arnould and Wallendorf (1994: 485) highlight further what they consider the four distinctions that mark ethnographer's research practice: firstly the data collection and recording of individual action within natural settings is systematic, secondly that it involves extended experiential participation by the researcher in the cultural context with an evolving sampling plan,

thirdly that it produces interpretations of behaviours of the persons studied that the intended audience find credible and lastly that it incorporates multiple sources of data to generate varying perspectives on the context of interest. Similarly Willis and Trondman (2000) reject the idea that societies can be understood without reference to mediating ideas and culture, but they also consider it limiting for a researcher to analyse culture as if it were free-floating, they argue for *"the ethnographic recording of lived experience within the social"* (Willis and Trondman, 2000: 10), suggesting that through such experience a researcher can obtain knowledge which has a wider reference. Within the narrower consumer research domain, ethnography is deeply rooted in gaining a cultural understanding of the context of study.

Table 6 - The role of ethnographic research within consumer research. Source: adapted from Arnould (1998)

Ethnography should aim to explain the ways that culture constructs and was constructed by the behaviours and experiences of its members.

Ethnography involves prolonged participation within a specific culture or subculture.

Ethnography in consumer research tends to be particularistic rather than generalizable, espousing pluralistic accounts of consumption.

The potential for ethnography lies in applying multiple data collection methods to a single phenomenon. These may range from surveys to observational data, videotapes, photographs and recordings of speech in action.

Ethnography requires tactics for representing research findings. These representations should aim to unravel the layered meanings that marketing activities hold for the customer (or in this research study the marketplace practitioners).

Building on Arnould's (1998) summary, for the ethnographer place is also an important construct as it represents a dual context: one that is inhabited by the researcher and one that is studied by the researcher (Pink, 2008). This understanding of place is central to the exploration of practices in this thesis. The ethnographer is faced with the task of understanding place that is something that could be said to be in constant flux of being made and remade. Pink (2008) describes that the researcher is *'emplaced'* but that they are seeking to explore the emplacement of others and the practices through which the places that they construct are continually reconstituted. Pink (2008) asserts that this is a complex process through which the researcher is firstly dealing with how research participants make place, secondly reflecting on how place is a collaboration with respondents in the research through practice, thirdly how in representing research place is reconstituted and lastly how the audience of the research create place and add to its existing narratives.

In the field of consumer research, there is also an extensive body of ethnographic literature that the researcher could use as a firm base for inspiration: from Hill's (1991) study of homeless women and the meaning of possessions, Arnould and Price's (1993) 'River Magic' study, McGrath et al. (1993) farmer's market, Otnes et al.'s (1995) shopping with consumers and Schouten and McAlexander's (1995) biker culture exploration. These pioneering studies create an ethnographic tradition within the CCT field and have been added to by technological advancements including netnographic studies (Kozinets, 1998, 2002, 2006, 2010; Hamilton and Hewer, 2009) and videographic accounts by Belk and Kozinets (2005, 2007), de Valck et al. (2009) and Rokka (2010). Within the context of this research study, the researcher is looking for rounded, holistic explanations achieved by working with people in their natural settings (Goulding, 2005: 299). Therefore as the focal point of the research is refined, the approach taken changes accordingly. Belk et al., (1988) describes that the researcher moves along a continuum from passive to participant observation generating insights from each vantage point. This focus on the importance of fieldwork is again highlighted by Arnould and Wallendorf (1994) who propose:

"long term immersion in the context increases the likelihood of spontaneously encountering important moments in the ordinary events of consumers' daily lives" (Arnould and Wallendorf, 1994: 486). Within the turn to practice, this fieldwork approach allows for an active understanding of the practices that practitioners engage in and how space is constituted and negotiated through practice. For this study an ethnographic approach is a way to gain an understanding of the practices of the collective, to view practitioners in their marketplace context and view them holistically. Viewing the vintage collective members in the marketplace is vital in order to observe and explore the active role that each member has and also to gain an understanding of the interactions and practices between practitioners. Long-term immersion in the field of study has been sought to enable the researcher to move along Belk's (1988) proposed continuum.

4.4.2 Criticisms of Ethnography

Some general criticisms of the ethnographic approach are that ethnography provides only one interpretation of the phenomenon of interest because of the partial and local nature of the analysis (Pettigrew, 2000). The notion of objectivity suggests that a researcher is able to obtain knowledge of an external world as it exists independently of the research process (Hammersley and Atkinson, 1995; Silverman, 1985). If this is accepted, then the aim of research is to maximize accurate observation and reduce distortion and bias to a minimum. A somewhat different but related argument is that objectivity is not attainable because people's perceptions and interpretations are inevitably selective and are shaped by the understandings they bring to any situation; it is not possible to perceive the world as a separate step prior to attaching meaning to it. As Denzin expresses: *"Humans are always already tangled up in a second*- *hand world of meanings*" (1997: 246). Issues of power and objectivity in the ethnographic relationship are prevalent and the role of the researcher as observer and interpreter of a community carefully considered:

"...And even where the researcher and researched are social equals, power is still involved because it is the researcher who makes the decisions about what is to be studied, how, for what purposes etc." (Hammersley, 1990: 13)

This consideration of the power balance in the field is further developed by Emmerson et al. (1995) as he conveys that even though immersion in the field is sought for close proximity to the subject, the researcher should remain at some distance from the world that they study. This tension of *"going native"* (Geertz, 1973) or ethnographic marginality is ever present in ethnographic studies. Also the ethnographer's authority has been questioned with regards to issues of validity, generalisability and reliability (Denzin and Lincoln, 2005). The researcher acknowledges these issues for the thesis however as the researcher is striving for an understanding of the collective not a definitive understanding, ethnography as a method is appropriate as it could access what consumers really do, rather than what they say they do (Elliott and Jankel-Elliott, 2003:222)

As detailed above in relation to CCT, there has been a shift in the discourse of naturalistic, humanistic discourses to a more multi-layered approach in CCT. As Thompson et al. (2013) express how the original epistemological positioning of CCT has limitations and CCT has developed to encompass a matrix of rhetorical, theoretical and methodological practices. They warn of an *'anachronistic influence'* of humanistic/experientialist dualities as a result of CCT's embeddedness in the

marketing discipline, they argue for a CCT that focuses on *"recursive relations between structure and agency playing out in varied consumption or marketplace contexts*" (Thompson et al., 2013:14). Thompson et al. (2013:2) further question what they term the *"conflicting polydiscursive current and future state of CCT."*

They argue that the readings of the CCT heteroglossia stem from the epistemological and ontological ambiguities that have resulted from CCT's complex institutional history. Recently Askegaard and Linnet (2011) argue that it is imperative that consumption is regarded as a practice, which has important implications for CCT research. They appeal for consumer researchers to pay attention to the contexts that condition these practices of consumption. Their suggestion of a contextual awareness, argues that the subject has to be placed within the wider social and cultural context within the research study. In considering the epistemological considerations of considering consumption as practice, Askegaard and Linnet (2011) note that consumers act to maximise the material, symbolic and social capital within various types of fields and that they may not feel constrained by rules or obligations, this they argue is the point of habitus; that the social actor holds an embodied, prereflective competence which are not necessarily the object of conscious reflection. Methodologically this raises issues with research that repeats the claims of consumers of individual uniqueness or self-realisation without adequate consideration of the social and cultural constraints in which the consumer is operating. Askegaard and Linnet (2011) propose that the strength of the concepts of practice and habitus lies in the underlining epistemology. They argue for giving due attention to the contexts that shape practices of consumption.

4.5 The Practice Theory Lens

Referring back to the literature presented in Chapter Three, this thesis adopts a practice lens to explore the practices, socialities and spaces of consumption. In considering practice as the conduct of transactional life, when viewed through the lens of practice theory, necessitates the temporally unfolding, symbolically mediated interweaving of experience and action, methodologically this has implications as to how to best capture these experiences and actions. Simpson (2009:1338) highlights that through this definition of practice, it is an emergent process that sustains routines while allowing for the possibility of creative action. This depicts practice as an anticipatory act, in that practices are the enactment of the future that materialise as actors anticipate the possible outcomes of their social actions. As detailed in Chapter Three, Schatzki (2001) in his critical review of the practice turn in social theory classified the field into three categories. Firstly he positions the theories that originate from philosophical and sociological concerns with the establishment and maintenance of stable social structures. This privileging of the social draws upon influences from Nietzsche and Heidegger and views the capacity for human agentic action as sceptical. Schatzki's (2001) second categorisation is of the convergent dynamic. He adopts a micro perspective focusing on the embodiment of meaning and the tacit understandings and knowledge that both shape and are shaped by practices. In his third categorisation, he stresses the dynamics of emergence and transformation to incorporate creative action. Simpson (2009:1329) builds on this and positions the practice turn with a theoretical view of practice as a "transactional social process involving experience and action as mutually informing aspects of human conduct." For Simpson (2009:1332) practice is a dynamic, temporal process but that also

involves human conduct and the exercise of agency – an embodied social agency that transcends traditional boundaries between the individual and the social. Simpson (2009) highlights the issue of dualisms that become evident in practice theory, in the separation of theory vs. practice, individual vs. social, rational action vs. normatively orientated action. The problem of dualisms has been illuminated in the work of Bourdieu's notion of 'habitus' and Gidden's 'theory of structuration', both of which critique the dualistic separation of the individual and the social. Simpson (2009) argues that whilst both of these use practice to draw together these binary opposites, neither she suggests engages fully with the social actor. This has important methodological implications as Simpson (2009:1339) positions practice as neither privileging agency nor structure but as a dynamic process of social meaning-making: "that is not limited to either the real-time present or the historical past and neither is it constrained to a unitary sequence of events in spatialized time." Simpson (2009:1342) demonstrates that the fundamental unit of ontology in this approach is the 'transaction' – whether this be intra, inter or extra-personal. This transactional view attests that the focus of enquiry is upon the agentic and temporal dynamics of meaning making rather than on particular meanings or meaning-makers. Simpson (2009) develops a perspective of practice as a transactional social process that involves action and experience, the practice lens is seen as dynamic, emergent and socially agentic. Also writing within the Organisation Studies domain Elkjaer and Simpson (2011:71) explain that through the lens of practice the transactional nature of social agency and the reflexivity in social practices is highlighted. They suggest that experience is both active and passive and that habits are tied towards specific actions. This links well with the ethnographic methods of this study and grounding in
actual experiences and consumer practices. This focus on the practices of everyday life is ontologically, epistemologically, and methodologically central to the thesis. This shift to practice and the subsequent methodological considerations has become more pertinent in consumer research as highlighted in the preceding literature chapters.

As presented in the literature review, consumer research literature often focuses on the objects at the moment of acquisition by the consumer, rather than how the consumer uses the objects. CCT in particular has focused on the meanings and symbolic properties of commodities and their use for example in identity construction. By employing practice theory, a focus on how the action is carried out in the process, becomes of crucial importance (Hand and Shove, 2007:81). This practice lens has included recent discussions that focus on the more mundane aspects of consumption, proposing that there is more merit in the discussion of the normally inconspicuous routines and systems (Gronow and Warde, 2001). There is a broad range of studies in CCT that address institutional, historical, ideological and sociocultural shaping of consumption and the broader market and social systems that situate consumers' practices and identities (Arnould, 1989; Arsel and Bean, 2013; Arsel and Thompson, 2011; Askegaard et al., 2005; Bardhi et al., 2012; Crockett and Wallendorf, 2004; Firat and Venkatesh, 1995; Giesler, 2006, 2008, 2012; Holt, 1995, 1997, 1998, 2002; Humphreys, 2010b; Kjeldgaard and Askegaard, 2006; Kozinets, 2008; Murray, 2002; Penaloza, 1994, 2001; Scaraboto and Fischer, 2013; Thompson, 2004; Thompson and Arsel, 2004). Continuing with the practice orientation, Moisander et al. (2009a: 331) suggests that CCT researchers 'understand culture as a *field of practices*'. The research focus they then argue is on the '*practices of*

objectification' through which culture and other "things" gain specific meanings. They argue that attention has to be paid to the cultural complexity of social action and the marketplace activity. This has methodological implications, as there is a focus on the minutiae of consumption practices.

In considering practices as "the active amalgamation of meanings, skills and objects", the relationships involved can be seen to be in a constant evolution (Shove and Araujo, 2010: 23). In viewing the social world as actively constructed rather than passively absorbed, the practices of the individual become of paramount importance and methodologically this calls for attempting to capture this active movement. Lury (1996: 77) conceptualises following this movement in her suggestion to 'think in terms of object-people practices'. Further Nicolini (2012) has written extensively on practices and suggests that in studying practice it is necessary to encapsulate movement in the methodology or "a zooming in on the accomplishments of practice, and of zooming out of their relationships in time and space." Nicolini (2012) creates a 'palette for zooming in' it is concerned with gaining a richer and more nuanced understanding of the world (Nicolini, 2012). From a methodological standpoint, the study of practice could be said to always start in the middle of action - what are practitioners using to assemble collaboratively and accomplish these scenes of action? Attention therefore needs to be given to what people are doing and saying (Nicolini, 2012). Nicolini (2012) asserts when studying practice, two practices are always looked at in unison: the researcher's epistemic practice and the focus of their interest.

This has important methodological considerations for gaining understanding into how practice is constituted through experience and action and through looking at

'practices in use'. This thesis proposes that the 'deep hanging out' (Wolcott, 1999) of an ethnographic approach is a way to capture practices in action. Participation in the community through an ethnographic methodology involves a reflexive engagement with 'knowing in practice' at all stages of the ethnographic process (Wenger, 1998:141). By utilising ethnography the researcher is seeking to develop a 'thick understanding' of practices in use and using a practice lens allows the researcher to gain an understanding into the multiplicity of practices that are knotted together in the marketplace. In attempting to capture the social and cultural experiences of respondents in the vintage scene, using practice theory coupled with ethnography allows for a more focused appreciation of the practices of consumers. Elliot and Jankel-Elliott (2003) propose: "ethnography reaches the parts other research cannot *reach.*" Based on the research aim of exploring marketplace practices and gaining an understanding into habitual behaviour, group dynamics and practitioner-object interactions, the coupling of ethnographic observation and depth interviews is seen "as a way to grasp and render" practices that are perceived in the marketplace (Geertz, 1973: 10). In her domestic kitchen practices study, Martens (2012) highlights the complexity of capturing practices and the shift between topics and practices. Responding to these concerns she asserts that her use of interviews and language in the study: "Arguably, our discussions were not about dish washing as a practical activity, but about a set of interpersonal concerns that are implicated in dish washing (thus pointing to what makes practices interpersonally social)" (Martens, 2012: 24). She uses the interview format to tease out how language conveys activity and the practical performance.

4.6 Research Objectives

The objectives of the research seek to meet the research aim and to provide clarity and structure to the data collection and analysis of the study. As outlined this study adopts an interpretivist, qualitative approach seeking to explore practices within the vintage marketplace. This thesis aims to gain an understanding of how the practices, spaces and socialites intersect in vintage. Coming from the CCT domain and employing the practice theory lens has had implications for the research methodology and research design as explored in detail above. Following the underpinnings of the interpretivist tradition the research objectives that emerged are:

- 1. What practices are used in the marketspace to materialise vintage?
- 2. How do practices, objects and knowledge intersect in the marketspace?
- 3. Does vintage create a shared consumption space?
- 4. How is value ascribed and framed in vintage?

The research objectives have guided the sampling, the data collection, the research relationship with the participants, the data analysis and the writing up of the data.

4.7 Data Collection

There are several data collection options that could be employed in this research project. These stem from the researcher's position around ontology, epistemology and paradigm position. Belk, Wallendorf and Sherry's (1989) interpretive study is an early exemplar of a multi-method ethnographic study: they use a combination of observation, field notes, interviews, photographs and video. This ethnographic research study employs multiple data collection methods to gain a holistic understanding into the vintage collective's practices. Ethnographic research by its very nature is multi-method and involves different data collection approaches for triangulation. A multi-method ethnographic study is undertaken as a way of gaining an understanding into practices undertaken in the marketplace. It is felt that what the participants say and do needs to be interpreted alongside the material and spatial surroundings, as they play an active role in the shaping of practices. Over the two years of the research study the researcher is embedded in the collective to gain an understanding, as such there is a daily interaction (being active in the wider vintage collective) but table 7 below gives an overview of the specific data collection methods of the fieldwork and the volume of data that is generated:

Method	Data collected
Observation days at the 3 different markets	70 visits
Depth interviews	25 interviews - 650 Pages of transcripts
In situ interviews	30 interviews - 120 Pages of transcripts
Field notes	380 pages of field notes
Researcher Diary	3 hand written notebooks - 250 pages of reflections
Photographs	260 photographs (iPhone and DSLR)
Visuals and artefacts collected (leaflets, posters, business cards, branding, press cut outs)	Approximately 45 visual materials

 Table 7 – Overview of the ethnography and data collection.

Hurdley and Dicks (2011: 278) propose that insufficient attention has been paid to the extent to which meanings are made in situ and utilise a full spectrum of sensory phenomena in which the actors engage, as such in situ occurrences (ad hoc interviews, visual collection, photographs, articles) in the vintage collective are continually recorded by the researcher over the two years to add to the understanding of the field. The data collection process for this thesis is described in further detail in the following sections as this research specifically uses semi-structured depth interviews, observation, a researcher diary and visual analysis to collect detailed data in context around the issues of value and the vintage marketplace.

4.7.1 The Research Context

4.7.2 The Wider Context - Glasgow

Glasgow is the largest city in Scotland and the third largest in the UK. It is known for its retail presence and was considered the UK's second largest and most economically important after central London (Harrison, 2012). Away from the more mainstream retail outlets, Glasgow has a long history of markets and trading in the city. The Barras market was once the largest open-air market in Europe,⁵ originating in the 1920s and 30s, the marketplace in the East End is seen as Glasgow institution although it has now fallen into somewhat a state of disrepair. Another significant market in the city was Paddy's Market⁶, which operated for more than 200 years and was originally where the immigrant Irish went to buy and trade second-hand clothes. It has developed into a market selling a wide variety of second-hand goods but was closed in 2009 following allegations of crime and counterfeit goods.

4.7.3 Glasgow's Vintage Scene

As depicted above Glasgow has long roots of market culture within the city however the research area of this study focuses particularly on vintage rather than second hand

⁵ The Barras Market has a long tradition in the East End of the city of Glasgow, with hawkers selling from handcarts (Barras) in the early part of the 20th Century. This became more formalised in the 1920's and 30's with the building of the McIver's sheds, and the Barrowlands Ballroom (The Barras Market).

⁶ The street market, Paddy's market, originated after the influx of poor Irish immigrants to Glasgow in the 19th Century (Stewart, 2009). As part of a city redevelopment scheme it was closed in 2009.

and in this way it is important to note the differences between the contexts. The second hand markets illustrated above as a Glasgow staple were based on economic necessity and as such carried with them stigmas and class connotations. The context of this research study focuses on what could be considered the gentrification of these market formats that started to arrive in more affluent areas of the city in their 'pop-up' format from 2009 onwards.

Within the research study, the context of the study focused on the vintage scene in Glasgow and in particular on the vintage pop-up markets across the city – with a particular focus on the student filled west end of the city. Observation was carried out over a two-year period at the markets. The researcher attended vintage styled events across the city, regularly visiting vintage stores and interacting with the collective on social media sites.

4.7.4 Sites of Enquiry

The first market entitled '*Granny Would be Proud*' runs twice monthly in the upstairs of a converted old cinema, which now functions as a restaurant in the westend of Glasgow. The wider venue of the west-end of the city acts as a cultural hub and has a high student population. The market has been running for four years, selling vintage clothes, homeware, accessories, upcycled vintage and independent craft. The sellers differ at every market, with the exception of a couple of longstanding stallholders. Established in 2010, it is one of the first of this format of 'pop-up' markets in the city. The market is free for consumers to enter and stalls cost £25 for sellers to hire for the afternoon. Doors open to the public at 12pm and the market runs until 5-5.30pm. The market is run by a young couple that promote the market predominantly through the use of social media. They also promote the market by flyering local bars, coffee shops and stores around the Byres Road area of the city.

The second market is called '*The Little Birds Market*' which operates from Sloans Glasgow, which is a restaurant and bar with a historic past that is situated in the city centre. The market runs every two weeks and takes over the spaces of the first and second floor grand ballrooms on a Sunday afternoon from 12-5pm. The market is free for consumers to enter and focuses on vintage, craft and independent design. Established in 2011 this market is run by a woman in her early thirties who left her job as a PR director to establish the markets. She uses social media, paper leafleting around shops and bars and e-newsletters to their own database (consumer details were collected at the markets).

The third market was the 'Judy's Affordable Vintage Fair'. Set up in 2006, this market tours the UK visiting over 30 cities, mostly university towns and also large festivals (including Vintage at Goodwood, Vintage at Southbank). Ran by Judy and her small team, they bring together sellers from across the UK to bring the events together. They also run a monthly market in Spitalfields – which states it is the largest in Europe. In Glasgow, the market visits two to three times a year and is situated in the Glasgow University Student Union. The event is actively promoted through their presence on Facebook and Twitter. They also establish an individual event for each city using Facebook. The markets run from 12-5pm and they normally target times of the year that will be busy with students (September/October and March). The events are heavily branded and the organisers also have a website presence and mailing list that they also used to market the events.

4.8 Theoretical Sampling

In sampling as with all other elements of the research design, this judgement is based on the preceding decisions of ontology and epistemology. Maxwell (1998: 234) suggests that there are four major considerations in deciding upon the methods for qualitative research: sampling, the research relationships that develop with participants, data collection and data analysis. Each of these is considered in turn.

Sampling in ethnographic research is often a combination of convenience sampling and snowball sampling (Bryman and Bell, 2007: 458.) Ethnography makes use of small samples: seeking out people who have knowledge in an area, judgemental samples or "opportunistic samples" (Elliott and Jankel-Elliott, 2003). Watters and Biernacki (1989) highlight that all qualitative sampling contains some degree of convenience, as without willingness and availability of participants to speak or be observed, and then the research would soon stall. Based on the research aim and objectives, the three pop-up markets sample sites were selected as they were the longest running and most prominent in the city. Further in this study the sampling unit are individuals who regularly participate in the vintage scene within Glasgow. The researcher gained access to the markets through attending regularly and building relationships with key characters in the marketplace. The participants were identified through a snowballing technique (Morgan, 2008) after spending time observing the marketplace and developing relationships. In line with the field of enquiry the sample criteria was based on the individual being an active practitioner in the Glasgow vintage scene – as market seller or consumer. Time and context were both considered with respondents sought from different times of the day at the market and also contextually from local stores and businesses that supported the marketplace. Further

respondents recommended other participants who were present at the market that day who may be useful in the study. Some twenty-five respondents were included for indepth interviews (further details displayed below in table 8).

Respondent	Age	Role in Marketplace
Adam	28	Vintage fair organiser, he studied Business at
		University.
Alice	42	Vintage/second-hand storeowner in the west end of the
		city, married to Bob.
Andrea	28	Lawyer with an interest in crafts and vintage homeware.
Annie	62	She sells vintage jewellery part time at the markets; she also has an Etsy store and attends markets regularly.
Ben	28	Works in graphic design. Enjoys shopping for vintage to mix with his high street and designer pieces.
Carrie	29	Works in marketing for a legal firm. She likes to attend the markets regularly.
Charlie	25	Vintage business owner, she sells at fairs and online through Asos Marketplace.
Cherry	26	Vintage dressmaker with her own store, she also sells at markets and is active in the vintage club scene.
Claire	32	She studied textile design at college; she used to sell at markets.
Derek	28	Works in retail, he studied illustration at college. He visits vintage stores in city center often and goes to the markets at the weekend with his wife.
Elena	35	Vintage fair organiser.
Gillian	27	She owns her own fashion shop with a vintage section, she also runs an online Asos Marketplace store and wholesale vintage store, she has recently stared to attend markets to sell and buy stock.
Jackie	42	Vintage storeowner in the west end of the city.
Jessica	31	Works in digital marketing for a city centre agency. She enjoys visiting vintage stores on a Saturday and regularly attends markets as part of Sunday routine with her husband.
Judy	35	Started the vintage fair in Leeds in 2005, after attending art school. Run the markets.
Louise	35	She left her job in PR to set up vintage fair and vintage china/homeware business.
Lindsay	29	Sells vintage jewellery at the markets and she also works as a social worker.
Luke	30	Works in website development, visits vintage markets at the weekends to browse.

Table 8 – Overview of in-depth interview respondents

Melanie	27	Works as a teacher. She is a regular eBay buyer; she visits west end shops and markets most weeks.
Michael	43	Vintage seller at the markets. He is an optician who has set up vintage spectacles business.
Michelle	28	Studied fashion at university. She visits the markets and car-boot sales frequently.
Molly	30	She works as a PA in an advertising agency and visits the markets with friends on a Sunday.
Sarah	26	Vintage fair organiser, she studied business at University.
Sheila	60	Vintage storeowner runs EBay and Asos Marketplace store.
Tina	30	Vintage wedding business owner sells at fairs.

Further, purposeful sampling (Morgan, 2008) was employed for this study as it involves samples being selected purposefully to allow for understanding of a subject in-depth (Patton, 2002). This led to the selection of participants that were used to provide in-depth information that was relevant to the study in accordance with the research aim and objectives. Within qualitative research it could be argued that all sampling is purposeful as the sample is selected in accordance with the needs of the study (Coyne, 1997). Opportunistic sampling was also used in the field of study, to compliment the research aim and identify informal research participants for in situ interviews within the context of the vintage marketplace. This allows for new leads to be followed up and allows for flexibility within the field. All research participants are treated in line with Ruth and Otnes' (2006: 562) "etiquette of qualitative research." Ruth and Otnes (2006) stress how integral etiquette practices are in the research process and interacting with informants. As the research is dependent upon successful social interactions, they reframe qualitative research as "opportunities for interpersonal communication and relationship building with informants" (Ruth and Otnes, 2006:561). They separate out etiquette interactions into: firstly "the 'before' phase: making and managing contacts with informants," secondly "the 'during'

phase: respect and deference during focal interaction," and lastly "the 'after' phase: enhancing the chances for prolonged relationships" (Ruth and Otnes, 2006:562-63). With both practical and theoretical implications, etiquette considerations and respecting informants and their experiences is at the forefront of this research. The importance of this relationship is stressed again by Hammersley and Atkinson (1995) who discuss that particularly with ethnographic enquiry, it unavoidably requires the researcher and the research participants to influence each other, and as a result it is crucial for the researcher to be explicit about their relationship within the research study. O'Reilly (2009:175) again highlights this and spoke of 'a gradual building up of trust'. This is important as within the ethnographic process there is an importance placed on building rapport with participants in the ethnographic process as these human relationships are seen as crucial in ethnographic practice.

4.8.2 Observation and Field Notes

In consumption studies observation has focused on naturally occurring behaviours in a variety of settings (Baudrillard, 1968; McCracken, 1989; Boyd and Levy, 1963; Soloman and Assael, 1987; Wallendorf and Arnould, 1991; Belk et al., 1991c) and provides researchers with a perspective in action. Madden (2010) suggests that ethnographers are a form of recording device and as such must always be 'on'. With regards to the consumer research context, Penaloza and Cayla (2006: 280) argue that the consumer behaviour domain has been dominated by studies that focused on *'what consumers said, over what they did.* 'They argue for field-notes and photographs as a mode of social representation. In developing a practice perspective and ethnographic methodology for this research study, a development of the researcher as participant observer becomes key as the researcher gains both visceral as well as contemplative appreciation for the social action being focused upon.

Observation is at the forefront of the methods employed in this research study. Agar (1996:163) describes that this is the essence of ethnographic methods, *"you are directly involved in community life, observing and talking with people as you learn from their view of reality*" (Agar, cited in Elliot et al., 2003:216). Due to its focus on the socio-cultural patterning of behaviour, ethnography according to Arnould and Wallendorf (1994: 486) gives primacy to observing behaviour and speech events that are naturally occurring. Goffman (1959) suggested that observers become insiders over time and that they are granted access to the 'backstage' areas. Ethnographers therefore can be seen to observe everyday events, settings, uses of objects and interactions over time (Jorgensen, 1989). Ethnographers provide a written account (field notes) of what they have seen, heard and experienced in the field.

Table 9 – Recording with field notes. Source: adapted from Spradley (1980: 78)

Space	the physical place or places.	
Actor	the people involved.	
Activity	a set of related acts people did.	
Object	the physical things present.	
Act	single actions that people did.	
Event	a set of related activities that people carried out.	
Time	the sequencing that takes place over time.	
Goal	the things people were trying to accomplish.	
Feeling	the emotions felt and expressed.	

As shown in table 9 above, writing field notes should not be a passive act as it involves the active process of interpretation and sense making of events and behaviours. Field notes Emmerson et al. (1995) argues are *"inscriptions of social life* and social discourse." He further suggests that through ethnography the researcher is able to experience first-hand how meanings emerge through discussion and "collective action" and how these understandings shift over time (Emmerson et al., 1995). Penaloza and Cayla (2006: 279) suggest that writing field-notes is a standard activity in ethnographic research - the researcher enters a social setting, participates in the routines of that setting and writes down what he/she learns and sees. Penaloza and Cayla (2006: 287) concur on the importance of field notes as a data set and posit that field notes allow the researcher to work through and document their initial thinking and how this evolves over time and could be related to the extant theory, which in turn is useful for illuminating points of difference and building theory. Within this research study, observation informed all of the researcher's actions in the research context. The researcher approached the observation and fieldnotes as a way to gauge an understanding of the practices emerging in the vintage market before any interviews took place. Following Spradley's (1980) advice, the researcher used the observation and accompanying field notes as a way to capture the marketplace practices and behaviours in the specific space of the pop-up markets. The researcher noted how each of the spaces was used and negotiated by both sellers and consumers, what people did in the space (how they moved, interacted, existed) and what objects were involved in this action, further the timings and sequences of the actions and behaviours were noted in the field notes and the feelings (facial expressions, gestures, body language) that were observed were recorded. These observations allowed some initial sense making to occur and Spradley's (1980) elements of observation guided the practical recording of the field notes throughout the study.

The researcher gained access to the first market, Granny Would be Proud through personal interactions with the market organisers. They allowed the researcher access to the marketplace to observe behaviours and practices. From a practical perspective the researcher positioned themselves behind one of the main participant's stalls – this stall acted as a hub for the market as this was also where the organisers would gather throughout the day to discuss how the market was progressing and they set this up as a make shift office at each market (to store leaflets and their personal belongings). This positioning also allowed the researcher access to the 'backstage' area of the stall – how it was set up and staged but also how consumers approached and interacted with the stall and participants. Throughout the afternoon of the market, the researcher would alter their position around the market to gain different perspectives of the day, but they would always return to the main area as a touchstone. As the time spent in the market progressed, participants would come and speak to the researcher or suggest people that would be interested in participating.

At the second market, Little Birds the researcher gained access through an introduction at the Granny Would be Proud market to the organiser, Louise. Having interviewed Louise, the researcher had asked whether it would be possible to carry out observation at her city centre market. The Little Birds market operated over two floors and in different rooms in 'Sloans'– with this is mind the researcher moved position more frequently to try and gain an understanding of the market practices. Having spent time at the market and the wider vintage scene, the researcher began to see the same people at the markets, one stallholder invited the researcher to sit at their stall and this again acted as a focal point for the researcher to operate from.

The final market Judy's Vintage Market was less frequent and the researcher gained access through email contact with the organisers and then after interviewing two members of the Judy's staff. The researcher observed the market from different vantage points – they positioned themselves at the door to watch the arrival, initial negotiations of the space and then departures from the venue. They also observed the wider market place scene from within the venue and the interactions, actions and practices carried out.

At all the markets the researcher took handwritten field notes that were as detailed as possible in situ – these were then after the market ended typed up and the researcher diary was also used at this point to reflect on the day's observations and interactions. This was done on the evening of the market day so as not to lose insights and observations that had occurred throughout the day.

4.8.3 Interviewing

Within qualitative research interviewing is one of the most prevalent sources of data. Interviewing provides the researcher with the opportunity to enter the other person's perspective (Patton, 2002). Johnson (2002: 106) proposes that in-depth interviewing allows for deep information and understanding. In addition Burgess (1982) comments that depth interviewing is flexible, but controlled. Building on Spradley's (1979) *The Ethnographic Interview*, which highlights the conversational structure of the 'speech event', Madden (2010: 67) suggests the ethnographic interview remains one of the most important ways of '*knowing others*'. Spradley's (1979) ethnographic interview explains that it should give the researcher insight into how the participant sees the world in analytical, typological, and relational ways, and should give a

glimpse into their world-view or '*Weltangschauung*.' A key focus of the ethnographic interview is on the use of non-directive questions. Using a semistructured approach allows the researcher to adapt the questions to fit the respondent and allows them to probe issues that emerge further. Hammersley and Atkinson (1983: 113) suggest that these act as "*triggers that stimulate the interviewee into talking about a particularly broad area*." In creating a more flexible interview approach it allows for the appreciation, as Denzin (1970: 125) explains that different people have a "*unique way of defining the world*." By allowing freedom for a more conversational approach to the interviews it provides respondents with the freedom to discuss points that they feel are important, this could have added value for the data. The interview data supplements the observational data and provides practitioner perspectives on the actions and practices of 'doing vintage' in the marketplace. The individual actors are assumed to have different cultural interpretations (Miles and Huberman, 1994) within a '*shared social milieu'* (Perren and Ram, 2004: 90).

With the research study the main aim is to explore the internal reality of the vintage collective practitioner (Carson et al., 2001). The style of interview adopted for this study was of the depth interview (please refer to Appendix i for the sample depth interview with one of the key respondents Michael). As detailed above within the research study twenty five ethnographic interviews (McCracken, 1988b) were used to gain an emic perspective of vintage behaviours and practices through respondent's stories and descriptions of their own and others behaviour. The interview was planned so that it was, as natural and conversation like as possible although it is open to flexibility. The researcher ensured that time is taken at the start of the interview to ensure that respondents feel comfortable in the interview situation. Within the body

of the interview the questions were grouped around the research aim. Consideration was given to Krueger's (1998: 31) assertion that questions should be presented in an open manner, so that "the answer is not implied, and the type and manner of response not suggested." The interview finishes with the researcher reflecting with the respondent on whether they are happy that they had felt their views are expressed adequately in the interview. The interviews were arranged at times that suited the respondent and take place in locations that are convenient for them – their homes or behind their market stalls. Due to the personal nature of the research all interviews were conducted in comfortable contexts, either in situ at the market or in practitioners' homes and stalls. This ensures that interviews were carried out in a familiar and comfortable environment. This also creates an observational opportunity of the material conditions of the respondent's homes. Discussions cover topics such as sourcing goods, the nature of vintage consumption, how value is ascribed and the collective practices around the vintage markets. Interviews lasted approximately one and a half hours and with the respondent's permission all were audio-recorded and later transcribed. All interviews were transcribed and annotated with initial impressions and observations noted to crystallise the main themes that emerge (Bryman and Bell, 2007). Interpretation and analysis involved multiple iterations of coding, with the researcher repeatedly returning to the multiple forms of data to refine thematic codes. In aiming for an understanding from the subjective point of view of the respondents, audio recording allowed for the natural language of the participants to be captured. Confidentiality was maintained throughout the research study. The audio-recordings were stored securely with password locked files. Respondent's names, addresses and other details are not disclosed and in accordance

with best practice pseudonyms are used in the thesis. The importance of transcription was considered as it allows the researcher to be extremely familiar with the data. In line with the tradition within consumer research brief extracts from the transcripts are utilised in the study to highlight themes and concepts.

The researcher also carried out thirty in situ, informal interviews in the marketplace. These are an attempt to get as "experience near as possible" (Geertz, 1973). These interviews took place in the marketplace and were performed without a list of questions but rather talking points that were informed from the observations carried out. Elliot and Jankel-Elliott (2003) argue that much of the richest data that could be captured with ethnography occurs from this informal talk between researcher and respondent. These interviews act as a way to capture natural insights and 'thick description' (Geertz, 1973) from practitioners whilst in the act of 'doing vintage'. Conversation was natural and free flowing with the respondents talking points focused around what items they purchased and why, their experience of the markets and unpacking their practices around valuing vintage. These interviews also function as a method to gain insight into practices and thought processes that could be explored further in the in-depth interview setting. These interviews were recorded and transcribed in a similar fashion as to the in-depth interviews or when recording was not possible, the researcher took notes immediately after the interview of points raised.

4.8.4 Photographic Data

The visual elements of ethnography (video, photography and other forms of visual media) are being used increasingly in ethnography (Pink, 2007: 1). The importance

of visual ethnography is growing rapidly however it should be considered in line with the larger ethnographic process. Madden (2010: 106) explains that photographs could be used as a reference point or to add validity to descriptions but also to move the reader through the scene in a way that the textual representation sometimes fails to do. This view draws attention to the need to contextualise photographic data as part of the 'ethnographic toolkit' (Madden, 2010: 109). Within the CCT tradition photographic data has been used to develop interpretations of behaviour including the temporal flow of events, culturally significant occurrences and human-object interactions (Arnould and Wallendorf, 1994). Penaloza and Cayla (2006: 279) comment that ethnographic studies of visual images and material objects aid the researcher to "draw meanings from the experiential and kinaesthetic aspects of consumption: people moving their bodies through consumption spaces." Belk (2013: 94) comments on the importance of the visual for the researcher: "we are most apt to immediately apprehend our environment visually". Penaloza and Cayla (2006: 284) add to this statement by arguing that as researchers, they study consumption venues and what consumers do in them. Using visual data allows the researcher to position consumers and their consumption behaviour within their social practices and spaces and allows the ethnographer to consider how they mediate these spaces. Arnould and Wallendorf (1994) establish that visual data could be useful in developing the interpretations of consumer behaviour including: the temporal flow of events, significant cultural occurrences and human-object relationships.

In this research study the photographic data is used in tandem with the observational field-notes as they both capture different aspects of the consumption phenomena (Penaloza and Cayla, 2006: 281). Photographs were taken in the marketplace using

both the researcher's DSLR camera and also their iPhone – permission from participants was sought in line with ethical approval from the University. In the markets, the iPhone camera allows less intrusion when taking images and is a way to capture in situ occurrences in the marketplace. Photographs are used to capture notions of space, staging, performance and to detail interactions. They are used to accompany the field notes and interview data. All images were saved digitally in a password-protected file and were coded along with the field notes. Using both field notes and visuals allows for the researcher to follow consumer representations at different moments in the '*circuit of culture*' (du Gay et al., 1997). With the researcher performing as 'bricoleur' and taking inspiration from Schroeder's (2002) '*image economy*', the photographs allow a focus on the modes of visual enactment of vintage.

4.8.5 Researcher Diary and Reflexivity

The concept of reflexivity of the researcher has emerged from sociology and ethnography. In considering the role of the researcher, positivists adopt the role of observer and do not influence the research subject; in contrast interpretivists interact with people and aim to gain an intimate familiarity with the subject (Brewer, 2000). Within this study, as an interpretive researcher, the researcher actively constructs interpretations of their experiences in the field. The context within which the practices or behaviours have taken place, are explained by the researcher to provide the reader with the associated meanings (Geertz, 1973). Baranov (2012: 164) proposes researchers adopt a position of *"reflective advocacy"*. In keeping with the idea that research is an interpretive activity, it is important for the researcher to reflect on how the research is being carried out and also to allow an insight into how the process of doing research could shape its outcomes (Hardy et al., 2001). Madden (2010: 136) asserts that all good ethnographic projects start with an embodied experience in the field and it is the subsequent recording of these experiences through field notes and other forms of data that are the beginning of all good ethnographic data sets. As noted by Lincoln and Guba (1985: 236) "the instrument of choice in naturalistic inquiry is the human". Ozanne and Murray (1991) further suggest that the researcher should become an active agent in the process of constructing modes of living and being. As such, in a qualitative research study researchers bring in their inherent biases, which must be acknowledged and identified (Miles and Huberman, 1994). This is a recurrent theme throughout the CCT literature as Penaloza (2001:376) suggests that as an interpretive researcher, researchers should be aware of their role as "*a primary data collector and analyser*." Coffey (1999) further asserts on the researcher role "fieldwork is necessarily an embodied activity" (1999: 59). He describes that the researcher must attempt to gain another's habitus and train their bodies to fit into the field (Coffey, 1999:65). In this respect, it is important for the researcher to keep in mind the embodied history (Bourdieu, 1990) that they bring to the field and be reflexive of this in the research process. Madden (2010) builds on this suggestion as he articulates that good ethnographers use their whole body as an organic recording device. In this way a consideration by the researcher of their positioning of their self is necessary for the reader to understand how their personal position influences all stages of the research process. Gould (2006, 186) asserts "we are all introspectors", in considering the introspective consciousness he further comments: "we reflexively examine our own inner spaces to varying degrees and with varying sensitivities; that is, we engage in

the practice of watching ourselves" (Gould, 2006: 186). Denzin and Lincoln (1999: 6) state that the interpretive bricoleur appreciates that research is an interactive process that is shaped by their own personal history, gender, biography, social class, race, as well as those in the subject setting. Throughout the research study the researcher kept a hand written researcher diary - this allowed the researcher to explore their feelings and impressions whilst carrying out the study. It was a place of sensemaking and recording of feelings and ideas throughout the research process (from initial concerns and nerves at being a 'researcher' presence in the field, to negotiating boundaries with respondents) it was used throughout marketplace research activity such as with the observations and interviews but also as a space to explore feelings and reflections about the research process more generally over the two years (on average it was annotated once a week on a Sunday evening, every week over the two years of the study – with differing lengths of entries but on average, at least a page long). The diary was especially used in the obsevations, as a space away from the field notes to explore personal reflections, ideas, and tensions in carrying out the field research. It explored issues such as the balance between being a researcher and a part of the collective and the personal experiences of the fieldwork journey such as making friends, the serendipity moments and the physicality of the process (such as nerves, tiredness and feelings of anxiety over the project). It was also a useful tool, to allow the researcher to express their personal feelings (stressed, distant, excited, nervous) and acted as a way on reflection to bring the field notes more alive in the analysis stage (the field notes were referenced against the diary to gain a move holsistic perspective). Overall a researcher diary was used to consider the reflexive aspects of the lived experience of meaning in motion in the

marketplace. Emmerson et al. (1995) highlights that reactivity is important in the research process and that the ethnographer must be aware and perceptive of how they are treated and seen by others. A researcher diary allows the researcher to be aware and explicit about:

"...their own epistemological position, thoughts and feelings about how the research is progressing, exploring how we experienced the research situation as a social encounter and what influence that had on the interpretations produced" (Nadin and Cassell, 2006: 215).

Discussion of the 'ethnographic self' (Coffey, 1999) highlights the importance of discussing the researcher in the ethnographic process. Landen (2011) stresses the importance of 'the researcher self' in relation to ethnographic research as a point of identification. She argues that the researcher is often viewed as the main instrument in ethnographic research and that as a result an analysis of what the researcher brings to the research is necessary. Geertz (1988:144) has long questioned the "very possibility of unconditioned description" and Landen (2011) builds upon this and places the researcher within the field of study and calls for the constant querying of the status of the researcher 'self'. Wallendorf and Brucks (1993) refer to this as "reflexivity within research" and they highlighted the importance of acknowledging the researcher in the writing and analysing process of consumer research.

Elliot and Jankel-Elliott (2003: 219) suggest that the researcher's diary should be linked to the field notes taken to explore this 'researcher self' but that it could serve as a useful record of the cognitive and emotional experience of fieldwork. It is with this in mind that the researcher used a diary. In this study the diary is utilised as a

tool for the researcher to capture their own feelings and experiences in the field (becoming immersed in the collective brought with it personal experiences of becoming close with respondents, of having trouble accessing desired sites or managing respondents – overall the human elements of observing and interacting with research respondents). In this way it is intentionally designed for an audience but as a safe tool to explore their experience as an ethnographer and researcher in practice.

4.9 Overview of Data Analysis

With regards to data analysis, the process is referred to by Klag and Langley (2013:150) as, *"the conceptual leap in qualitative research"* and that this analytical process involves both 'seeing' and 'articulating'. Both of these are intertwined in reality as 'seeing' refers to making sense of the social world in a new way and 'articulating' suggests how this new understanding is represented. With the wider process of analysis, Madden (2010) describes moving from interpretation to story in the writing 'up' of ethnography. Clifford (1986) asserts: *"the making of ethnography is artisanal, tied to the worldly work of writing."* This highlights that in translating the reality of others, researchers need to be aware that in their writing they are writing within and against specific traditions, disciplines and audiences and will employ expressive conventions.

In practical terms the ethnographic data analysis process can be divided into two stages (Jones and Watt, 2010: 159): stage one involves ordering, collating and managing data in a way that makes analysis possible and stage two involves the actual data analysis. For many ethnographers any form of system for data analysis goes against the interpretivist values of ethnography (Hammersley and Atkinson, 1995). However such an argument ignores the reality of the research process in which the ethnographer is continually editing and managing both consciously and unconsciously what they are recording in the field. Hammersley and Atkinson (1995: 203) provide a strong critique on the limitations of using software in data analysis:

"...we must recognise that they only provide adjuncts to the sociological or anthropological imagination. They certainly do not provide 'automatic' solutions to problems of representation and analysis. Understanding and interpretation are the outcome of interactions between ethnographer and the data, which are themselves constructs. There are no mechanic substitutes for those complex processes of reading and interpretation."

This statement highlights that the key skills required for the analysis of ethnographic data are reading, interpretation and reflection (Geertz, 1984) and for these tasks the computer package may aid the storing and organisation of the data but not the process itself. For this reason the data analysis was approached manually with highlighters, post it notes and file cards. In practical terms, once the data is organised, analysed and interpreted, the researcher begins to shift from an analytical role to an interpretive role. Organising the ethnographic field notes is the starting point for the first step in the analysis process of the ethnographic data. The field notes are indexed and coded to identify themes in the notes that relate back to the outlined research aim and objectives. These identified themes are saved electronically in password-protected files. Madden (2010: 141) describes coding field notes is about indexing what happened in the field and what this might mean for the questions that the researcher may want to ask. Miles and Huberman (1994: 12)

highlight the active nature of the analysis process: "qualitative data analysis is a continuous, iterative enterprise." Ethnographic analysis they argue is descriptive and should be tasked with "uncovering" and "explicating" (Miles and Huberman, 1994: 12). From the transcripts from the interviews first order codes were used that were relevant to the research study aims and objectives. These codes were assigned a particular colour and these were used to gauge an initial understanding of patterns and relationships between themes. These were then coded again with second and third order themes, in this way it was a holistic process that evolved over the time of the analytic and interpretive phase of the project. The visual materials collected were indexed alongside the field notes into themes. Images that were relevant to more than one theme were saved in more than one thematic folder. The visual data was treated similarly to the field notes in that they were broken down into thematic pieces that were used to display the frequency that certain themes appeared in the data and the relationships between themes. Madden (2010: 140) highlights the data has already been 'partially cooked' by the choices that the ethnographer makes during the primary inscription process and this framing becomes more evident in the themes that emerge from the field notes. The importance of critical reflexivity is again highlighted during this process of analysis.

4.10 Methodological Limitations

The limitations of the research approach taken will now be considered along with the researcher's response to the identified challenges. A practical limitation that is identified and is reflective of the nature of the qualitative tradition is the time commitment required in both data collection and the interpretation stage. Due to this time commitment, only three sample pop-up markets are used and as small samples

of participants are employed, issues of reliability and generalizability could be common. Transferability is highlighted by Guba and Lincoln (1998), as to how insights from the data could be applicable in other contexts or populations.

Ethical considerations are crucial throughout the study and the close relationship between participants and researcher are recognised. Wallendorf and Belk (1989) suggest that problems of integrity could arise due to problems between the researcher and the informant. To limit this, the researcher places an emphasis on creating a good rapport and building trust with respondents. The subjectivity of the researcher is also important and continually acknowledged in the data collection and the subsequent analysis and 'write up'.

4.11 Conclusion

This chapter has considered the researcher's methodological approach to the research study. This chapter has situated the research within the interpretive ontology that compliments the CCT focus of the experiential and socio-cultural dimensions of consumption. This chapter has defined the research aims as focusing on the exploration of practices, spaces and socialites of vintage consumption. This study adopts an interpretive approach and searches for understanding. It is based on gaining an understanding into the practices of the vintage collective rather than generating universal laws of prediction. The qualitative tradition has been explored in line with the aims of the study and considered that this was in keeping with the theoretical tradition. The chapter has also outlined the research design of the study. The research framework has been discussed in conjunction with the research aim and the analysis of the ethnographic data highlighted. Finally the assumptions, limitations and practices that this study has adopted have been explored.

Chapter 5: Findings – The Practice, The Space and The Socialities

5.1 Introduction

"The market is this strange place. It emerges and fades. It started me questioning what people gained from this kind of space rather than a normal shop experience? What is it about the pop-up markets that are different and exciting? Which in turn makes me question what it is about the market I enjoy, which I suppose is returning to the initial curiosity I had in the field. It is different, it is cool and quirky. Firstly I like that people talk to you. It feels more personal. People know where to buy interesting things, unusual vintage finds and generally they share stuff with you. You get to have interesting chats about design and art and fashion. It seems more alive somehow. I like that when you walk down the street and you see someone from the market they say hello or at least smile. It feels like you get to know the city you live in, in a different way." (Researcher diary, May 18th 2013)

This personal reflection begins the findings chapter as it starts to tease out the researchers own bond with the marketplace and the collective. As a reflective starting point the researcher diary was approached as a sense making tool to parallel the field notes, but this narrative highlights the undercurrent of the market being emergent and a different way to consume and questions the turn to this marketplace. Further the notion of accessing a different form of community in the city than had been accessed before, one that was grounded in the 'unusual,' in art and fashion, became apparent. And finally the social element, that it is personal and bonds are built with sellers. As an ethnographic study over two years, the time spent in the markets involved

questioning and problematising the practices that the researcher and the collective took for granted in their market interactions of vintage as both consumers and sellers. The findings chapter starts the journey in discussing the practices of vintage and of the market, before moving onto the spaces of consumption and practice, before ending the chapter with an exploration of the socialities of vintage.

5.2 The Practice

5.2.1 The Practice: The Role of the Practitioner

The researcher first met Elena at her vintage market when it had just started in the city. Elena and her husband Adam were a young couple living in Glasgow and they have organised one of the cities longest running vintage markets, Granny Would be *Proud.* Elena originally ran the markets - she was working in hospitality and noticed that the hotel she worked in was usually empty on a Sunday afternoon. She explains that she has always been interested in vintage or second-hand. Originally from France, Elena has moved around the UK and explains that she likes Glasgow because it is 'creative' and there is always something new happening. She describes that she has never been into mainstream fashion but that she has always preferred searching something more unusual. She talks of going to club nights and making an effort with her clothes and styling - victory rolls and sixties eye makeup - but she says now that she has had a baby, she is more laidback about it. She spends a lot of time on Facebook building up the community and starting discussions. She does not have a personal Facebook profile and therefore does not distinguish between her personal life and market persona as 'Granny Elena.' Adam and Elena have been together for three years - they are recently married and have a young daughter. Adam has an interest in marketing and had worked in market research as a student. He started to work more in the organisation of the markets over the two years of the research study. He took over the design aspect of the flyers and e-promotions. He also works in different part time jobs to bring in another income. Born in the North of England, Adam spoke of 'thrifty behaviour' being a family trait and had always been taught to go to jumble sales and charity shops, rather than shopping on the high street. They highlight their interest in vintage:

Elena: "For us, it is a style and lifestyle more specifically. With our home and our clothes, I always prefer to buy vintage or second-hand, I go to car boot sales most weeks and it is an outing for everyone. There is always something happening and it feels more like an achievement than a store purchase. Vintage for me - I love it. I used to be head-to-toe and I think people expect that at the market, but now for me it is about the finding and bringing alive a special piece."

For Elena, it is the practice of finding that she enjoys whereas for Adam vintage is a way for him to construct a way of dressing that highlights his knowledge and individuality and resistance to the broader culture at work.

Adam: "...it frustrates me when you see everyone trying to do vintage. We try to be quite picky with our sellers, but the emphasis is on it being different. I have always shopped that way and wanted to get more unusual pieces that are like a little bit of history and are much more me than Topman or something. My style isn't wacky or anything but I like the quality and look that I can put together..."

For Adam the market becomes a limitless place of possibilities of performance. For both, the vintage markets become a space to practice understandings and appreciating potential value. This raises the initial question as to how value is attached to objects that are constantly moving 'in' and 'out' of the marketplace? As Parsons (2008:392) argues "*the transition from thing of little or no value (rubbish), to thing (durable) can result from a relatively minor shift in the way we see something.*" As described by Elena, this aesthetic appreciation moves objects forward in the marketplace, imbuing them with new value. This builds on Gregson and Crewe (2002: 146) who stress:

"The commodification of goods bought for consumption need not be terminal – the cultural and economic longevity of a commodity's life can be infinite, value can lie dormant, meaning can be hidden from view, awaiting rediscovery."

In the markets vintage, by its very character is seen to be active and in practice a fluid construct that is open to interpretation by practitioners (Osterlund and Carlile, 2005; Cova and Dalli, 2009; Kotler, 1986; Zwick et al. 2008; Humphreys and Grayson, 2008; Ritzer, 2010; Shove and Pantzar, 2005). This leads to a questioning of how those inside the marketplace know what will be valuable? As an entity, vintage is seen to require the skills of practitioners to materialise it into action in the marketplace:

"...for some people it is just junk, something that they would throw away or would walk past or dismiss, for me it is like these special little moments when I find something vintage that I maybe haven't seen before, or I've been looking for or that I love the pattern on, it is like the dress might be displayed nicely and it catches my eye or it might be that I just see the pattern of the sleeve sticking out of a box, just something about it peaks my interest and it is so much more exciting than just going into a store and handing over money. It feels a bit like giving a dress another chance at a life and I love that about vintage." (Elena)

As Elena highlights the nature of vintage is fluid and without the necessary understanding as a marketplace practitioner, the potential value of the object is not seen and may easily be missed if it is not stabilised. It becomes an act of taste (Hennion, 2004; Arsel and Bean, 2013) and embodied capital (Bourdieu, 1984). In this way practitioners are seen to make temporal sense of objects and adopt carefully played roles in the market in *"re-enchanting objects"* with new lives (Parsons, 2006: 189). To address the research objective fully of how new forms of value are created through practice, firstly the role of the practitioner must be addressed. In the discussion of the marketplace both consumers and sellers are seen as practitioners in action. In this study they will be referred to as 'practitioners' as both roles were seen to be adopted and enacted at certain points in time. Vintage is viewed as a form of practitioner interaction and through this practice value is created from the *"rediscovery"* and competence of practitioners such as Elena and Adam. This compliments Sherry's (1990a) original work on flea markets whereby searching and examining were key activities.

The marketplace becomes a place for these active discovery practices and also for roles to be interchanged as necessary. The cyclical nature of the marketplace and the fluidity of boundaries are clearly highlighted in the data: consumers' spoke of selling and sellers spoke of consuming. Sellers also depict consuming for pleasure without the intention of selling items on. This constant intersecting and interchanging of the role of the consumer in the marketplace is best represented when Sarah spoke of this fluidity so that she can move the object forward:

"I will find something vintage like a seventies leather jacket and I might love it for a while, like I will wear it to death for a couple of months. Then I get sick of the sight of it, so I will sell it again, I will probably take a stall at one of the markets...some people say it is like recycling or something, I like that I can just keep stuff moving, it keeps it all fresh." (Sarah)

This adoption of the role of seller when convenient for the consumer illustrates the ephemeral boundaries of the marketplace and the unsentimental way in which objects are considered. The roles and the object's value are not fixed or constrained and interest in the vintage scene and knowledge of the marketplace is seen as legitimacy as a practitioner. As Sarah outlines, sellers translate objects from one context to the next and success in the marketplace is reliant on the development and display of knowledge of the market, consumer desire and discernment for the potential of objects. The activity of vintage is reliant on this enacted knowledge to access what Campbell (1987) calls the *"unknown object of desire."* In the vintage scene, practitioners function as actors in the meaning-making process and in the creation of value. By playing this active dynamic role, practitioners take on the role of *'working consumers'* (Cova and Dalli, 2009). In this working consumer framework, individuals become key in the practice of value creation, as Gillian highlights:

"I will find something at a car boot or a charity shop and I will know that if I style right and dress it up on the stall, it will become vintage and wanted. Or I will find something and see a bit of magic in it, I might keep it for a while, even a few years until I think the market is ready for it, it is a judgement call... Sometimes you have to use the eye that you develop to help others see the potential." (Gillian) Gillian's comments highlight the 'magic' around vintage, that it is created, staged and manifested by her practice, judgement and skill. She expresses "it will become *vintage*" and thus shows the active socially agentic nature of the practice of vintage. She speaks of a transformation and enactment for an 'imagined' consumer, which through the knowledge that she possesses as a practitioner in the marketplace. changes an object from second-hand to the elevated status of vintage. She was able "to read economic value" into the object to facilitate this transformation (Denegri-Knott and Molesworth, 2009: 308). In this active role she employs her cultural capital (Bourdieu, 1984) to add value to the vintage object. This notion was further expressed by Gillian, that any object has potential value and meaning with age, creates the role of investor for the practitioner in that they are able to buy with a view to what will be collectable in the future. Storing clothes and objects 'to age' calls to mind practices of wine collecting and valorises contemporary objects as potential sources of value and exclusivity, once they have aged. This dynamic role also illustrates the underlying mobility of the objects and the transience of the marketplace. Jessica spoke of the practice of mobilising objects when she wants to access the capital in them:

"If I need extra cash or something then the things I no longer wear become something to earn money out of. It is not something I do often, maybe once or twice a year...there are some things I would never sell, but overall I like to think that they will find a better home." (Jessica)

Jessica is in her early thirties and recently married. She works in creative design for a website company. She has been shopping vintage and second-hand since she moved to the city to attend University. She goes along to all of the Glasgow markets
regularly to pick up homeware pieces (furniture and glasses particularly) and to look for fashion treasures. She normally attends the markets by herself and explains that she prefers to shop alone and have a chance to have a conversation with the stallholders. She spoke of her 'Sunday routine' – she goes for coffee with her book in the morning and then visits the markets not long after they have opened at 12 o'clock, she arranges to meet friends for coffee at two o'clock and depending on what she has seen she sometimes pops back into the market on her way home. She spoke of attending the markets in familiar terms as if it habit and routine for her. She explains that she has sold at the markets when she has cleared out a few times but prefers to be a consumer rather than a seller. She spoke of her mum being passionate about antiques when she was growing up and she said that this has influenced her style. She is open to all styles but has particularly focused on the 1940s look in the past, as she believes that this suits her shape. Jessica's assertion above highlights that there are some pieces that she treasures but that she is able to see other objects in a different frame, as potential stock (Denegri-Knott and Molesworth, 2009). This change in how objects are viewed (worn, adored item - changing to a piece to be sold and money to be reinvested) speaks of a shift in the viewing of objects (Slater, 2014). Treated without reverence or sentimentality, wardrobes are seen as stockpiles that are waiting to be capitalised on. In this way, objects are materialised into a key piece for a fleeting moment in time before the practitioner moves on (with their tastes, attachments, enjoyment of the object) and sets in motion the object again for its next life cycle. This transient life of objects is best highlighted by, vintage jewellery seller, Annie:

"The main thing as a seller that I consider is 'where to store my vintage items', it is much less about the capital tied up in them, that isn't so much of an issue for me, it is more the question of where do I practically store things? If I had unlimited space I would be able to buy more and also to hang on to things much longer, quite often I will sell something because I think 'oh that is just taking up space.' The lack of storage almost forces me to part with things that I would maybe have hung on to longer. But I know most sellers have the same issue unless they have retail store and storage rooms. (Annie)

Annie articulates that the space 'forces' her to mobilise objects at a moment in time as she is constrained by what she can store, without this constraint she would mobilise fewer objects potentially. In the marketplace, this judgment of the seller as to which objects to mobilise and when, is adopted when deemed necessary: whether it be economic to mobilise capital tied up in the objects, a lack space issue for storage as mentioned by Annie or to aid the purchasing of another object. In this way the marketplace can be seen to shift to meet the needs of the practitioner and the marketplace becomes a space to nurture proactive practitioners. Conceptualising both sellers and consumers as practitioners further builds on the earlier discussion in Chapter Two of the changing role of consumers and the evolution of the consumerproducer relationship: of Campbell's 'craft consumer', the shift to 'prosumers' (Kotler, 1986; Zwick et al. 2008; Humphreys and Grayson, 2008; Ritzer, 2010) and Cova and Dalli's (2009) "working consumers". In turning to a practice perspective (Arsel and Bean, 2013; Warde, 2005; Shove and Pantzar, 2005) practitioners are shown to fully participate in new forms of value creation as new forms of value are created through their competence.

5.2.2 The Practice: Conceptualising Vintage

By its very nature vintage is a finite resource and this has implications for the object's perceived value from the outset. Although possibly once mass-produced, the objects have outlasted their counterparts and re-entered the market. Described by practitioners in the marketplace as 'vintage', 'second-hand', 'pre-loved', 'pre-owned', there is a plethora of ways in which stallholders describe the selling of objects that have been removed from the consumption sphere and reintroduced in the vintage marketplace. This conscious use of certain terms over others suggests associations about how best to position their objects in the marketplace to create value.

One tension that became prominent in the data early on in the study is the notion of what constitutes 'vintage', as it exists without clear definition and intangible by nature, from the observations a practical acknowledgement of the characteristics of vintage is prevalent in the market. Unlike with antiques, no accepted definition is at work in the marketplace. The shared collective acknowledgement of the perimeters and accepted understandings of vintage elevates goods from simply second-hand to cherished vintage objects. The wisdom of the collective in denoting vintage was found to be an important theme that ran through the research. This tension is further problematised in a conversation as Elena and Adam discuss what they consider the differences between vintage and second-hand to be in the marketplace:

Researcher: Can you tell me about your understanding of 'vintage' and 'secondhand? Elena: "I think second-hand nowadays – what we call second-hand is the new clothes that you can buy from the shops – like Marks and Spencer – you know the High Street shops – and then you know they are handed into charity shops. Vintage, I think is something that was made in the 1920s, 40s, 50s, and you know is original from that era. Whereas what is second-hand is basically just made in our day and then just passed on – do you know what I mean? I think that's second-hand."

Adam: "What I think is it depends who you are speaking to – last year, we were sat down and talking about it – and a lot of the things that we consider vintage are becoming – or perhaps where we are now with it – they're gradually becoming antique - so the things that we are looking at now from our grandparents' time – some of them we might call vintage - a lot of them are now becoming antique. They are coming out of that 30 or 40 years contemporary-ness and coming up to 100 or 150 years old - so they are becoming more antique so the price – the value of them is going up – but as we are coming out of this – I notice this with other kids from the 90s - the dress is very 80s but if you think what was around in the 80s it was very disposable. Especially in the markets it's becoming, the price is going up of things because the stuff that people really want from the 40s, 50s and 60s and 70s is becoming more rare – so as it goes up it's becoming antique as it becomes available – and you've got a lot of stuff from the 80s that's a lot less - you know there's a lot less of it... "

In this conversation, Elena and Adam begin to highlight the slipperiness of vintage as term. Here they identify their tacit knowledge (Entwistle, 2009) as practitioners of the marketplace and unable to fully agree on definitions of vintage and second-hand, they highlight the fluidity and episodic nature of the marketplace. Without precise definition, the concept continually lacks an anchor to stabilise objects in the market in the same manner as with antiques. In this way the label 'vintage' is found to be multifaceted and problematic, carrying with it a feeling of ambiguity; expressed oftentimes as a form of uncertainty and hesitancy (Goffman, 1986a: 302). These forms of ambiguity are often made explicit through the lack of consensus in the collective regarding the time periods necessary to constitute 'second-hand', 'retro', 'vintage' and 'antique' (Slater, 2014). The collective acknowledgment of 'vintage' is continually questioned and made explicit:

"I'm not sure how I feel about tie-dye, I mean I know it is a 70s thing but I am not sure that it counts as vintage. I mean I think most people in the market think that it is cute but I don't think they see the point in it really or that it is truly vintage." (Adam)

These tensions and dilemmas in the marketplace of what is suitably 'vintage' emerge, as Adam articulates an anxiety regarding the collective judgement and their appropriateness of being deemed vintage. This highlights that the practice is questioned and produces an anxiety over the use of the definition of vintage. In this sense, vintage in the market is fluid and open to interpretation by the collective:

"Vintage means like something from yesteryear. Like something that's got a bit of history. It doesn't mean defined by time as such because in terms of like fashion nothing is new – everything is evolving – so I would say that something that is older than today's look would be described as vintage." (Melanie)

Unlike Adam who prizes the age of an object as a qualifier of vintage, Melanie highlights the fluidity of vintage. Again separated from antique with its expensive price tags, vintage appears to be defined by its affordability and openness. With links to fashion and the trends that follow this, vintage is treated as less precious, more accessible and is seen as more ephemeral in its existence. Without the aged crowd as seen in antiques markets or the bargain hunters of the jumble sale, vintage exists in a liminal moment – both created and critiqued for its link to fashion. When asked what makes something 'vintage', all participants found it hard to qualify or even quantify. For Cherry: *"it is about desirability, if the piece is current or unusual then it becomes more than just second-hand."* However, becoming more than second-hand was not easy for an object; systems of value had to be at work for an object to be deemed 'vintage'; or as Carrie explains: *"it must have something special about it, it is more than just old, it is the quality, cut, the design"*. Whilst for Ben quality became a byword for style: *"it has to be about the quality but also whether it is wearable and can be mixed up to look modern again."* This collective negotiation of vintage depicts practices of value creation and Othering within the consumption group, as the collective share practices and values including the discourses employed to make sense of the vintage world.

As explored in the literature of Chapter Three, practice theory contributes to CCT studies as it focuses research on the performative processes of social life, which involve consumption activities, whilst at the same time not lessening the emphasis on cultural conditioning. Practices are formed as the resources of consumers interlink with different contextual elements (Reckwitz, 2002) and practices are conceptualised as *'doings'* rather than cognitions or emotions. One of the most experienced practitioners of the 'doing' of vintage is Alice. She owns a vintage shop in Glasgow's west end, just off the busy main street of Byres Road - down a small cobbled lane, Ruthven Lane, where there are several other independent shops. She

has been trading in second-hand goods for over thirty years. Her shop in Ruthven Lane has been there for thirteen years. She knows the owners of the other stores that are also 'down the lane'. Her shop focuses on more modern pieces that are two or three seasons old but they are all second-hand, either sourced by Alice herself from charity shops, jumble sales, car boot sales or she sells clothes on behalf of people and takes a fifty percent cut. Both she and her husband, Bob, work in the shop. She works in the shop two days a week and at other times she is sourcing stock. All other staff in the shop are friends or family members, including her mother. Alice is in her early forties and she is married with two teenage children. She explains that she has always been interested in 'stuff' and spent her youth collecting and selling at markets and jumble sales around Glasgow (including many years selling at Paddy's Market). She started selling at a weekly market in Glasgow and then moved into a more permanent market fixture before moving to a retail presence. She explains that she used to focus exclusively on vintage pieces from the 1940s, 50s and 60s but has found it harder and harder to source stock and eventually she noticed a gap in the market for selling more recent fashion pieces second-hand. She explains that she constantly struggles with questioning over whether what she does is suitably 'vintage'. She visits all of the markets frequently as she is keen to keep up with trends and the researcher meets her often at Granny Would be Proud where she was frequently buying.

Figure 2 - Alice's store: photograph by researcher May 12th 2012



"People come into my shop and go 'this isn't vintage' and I'm like 'I'm not saying I'm vintage, I'm not vintage.' I changed years and years ago, I stopped upselling properly old things like 1950s prom dresses because it got harder and harder to find. I would now say that I do second-hand and a bit of vintage but people get confused about the labels and what I do. I suppose as well people want to be able to say that an item is 'vintage' because that is what is cool." (Alice)

The vintage context illuminates that object lives consist of moments of use and disuse and are transformed by practitioners' understandings of vintage such as Alice. Understanding this mediation of knowledge around the definition of vintage is captured with reference to Bourdieu's (1984, 1985) notion of cultural capital. This tension surrounding the interpretation of 'vintage' is present for all and is further articulated by another seller Michael:

"My definition of vintage is something that is actually of the era that it is from, aged in a certain way. Retro is a copy of a vintage look, when I am selling I am very careful to define the two and make sure people understand the difference and what they are buying. I can't tell you the number of times in the market that I have had people in saying, 'I bought this pair of vintage glasses on eBay, what do you think of them, can you put lenses in them,' and I look at them and they are some £2.99 frame from China or a copy which Topshop would sell for £12 or something and you don't know how to tell them that it is not a vintage frame, people are protective of their purchases, so you can't really contradict people but at the end of the day, I find that annoying when you sell the real thing, it is stopping trade and devaluing what I do."(Michael)

Practitioners mention the discussion over the semantics frequently as they are deliberate in how they describe their business and whether it is 'vintage' or 'second-hand.' Michael highlights seller frustrations around misunderstandings of 'vintage' in the market. When discussing the construct of value further in the vintage marketplace, Michael suggests that vintage is fundamentally elusive:

"I've stopped so much valuing them (glasses) on the age of them, it is not like antique value, it is not that, it is intrinsic, the quality, rarity and the look of them...it is the how unique or individual they are...how irreplaceable they are." (Michael)

Michael's excerpt demonstrates that vintage far from being a fixed construct is open to active negotiation based on skill and knowledge practices. In this way value can be seen as an embodiment of capital (Bourdieu, 1984). In considering the vintage context this notion suggests that stallholders act as 'gate keepers' regarding which objects will merit another consumption cycle. This practice of finding and bringing to market is significant in that value must be seen in the object that others have discarded and further practitioners act as 'taste-shapers:' bringing a range of objects back to market. The seller functions as mediator and must also know that this value that they see in the object will translate to the market and the object will be 'saleable'. In this way through practice, 'gate keepers' are created in the marketplace: a judgement is made on the potential value of the object and a decision made on whether it would be desired in the marketplace.

As Sheila further explains:

"It is strange that people say that I have a vintage shop, I know I do, but I never called it vintage then, vintage was a word that wasn't used maybe apart from you would say 'a vintage car' or a 'vintage Chanel dress', do you understand? But vintage wasn't, just wasn't in my vocabulary, the people that I knew, vintage the word wasn't used." (Sheila)

Sheila's discussion of vintage conveys the tension in the construct of 'vintage'. She believes that vintage is a new construct that had previously only been used to articulate the rarity of an object; in this way the semantics can be seen to have shifted in the marketplace from second-hand to vintage.

Alice, Michael and Sheila are all seasoned practitioners (both as sellers and as consumers) and their comments articulate that vintage has a fluidity attached to it as a construct and that it is ambiguous by nature, in this way the label of vintage acts like an added value or functions as a brand name in the marketplace. Practitioners adopt the role of active culture creators through their use of brand 'vintage'. This aesthetic merit and knowledge of the marketplace elevates goods from merely second-hand to the added value of 'vintage' and through this discourse, objects are labelled or badged with value. For sellers, vintage acts as a label that allows marketplace categorisation and realisation of value to occur in the marketplace for economic gain or could be argued to create a commercially mediated marketplace

myth (Arsel and Thompson, 2011). It also orientates the objects and roots them within a marketplace discourse that is accessible to consumers. Eckhardt and Bengtsson (2010) describe brands as important agents in marketplace interactions, which further develop Holt's (2002) argument of brands providing heuristic shortcuts for consumers in contemporary culture. Jessica explains the essence of vintage:

"... To me it becomes like a label in itself, someone decided that item was more than second-hand, it was deemed 'vintage'. That then gives it with more value and you pay more for it." (Jessica)

This active practice of labelling an object as 'vintage' is a dual process in which the consumer expectation of what was classified, as 'vintage' has to marry with the seller expectation of 'vintage' and in this way the market is in a constant process of negotiation. This use of brand vintage highlights that as a construct it is fluid and created in the marketplace to enable second-hand objects to be re-invigorated for market re-entry. This use of marketing branding practice is almost contradictory to the marketplace myth (Thompson, 2004; Arsel and Thompson, 2011) propelled by participants of wanting to consume differently from the mainstream. As Jessica comments:

"Vintage has kind of created like a reverse label, where not having a label is more attractive than having labels." (Jessica)

Jessica's insights suggest that vintage acts like an overarching brand name - the objects have been removed from the marketplace first cycle and their original markers and labels are no longer signifiers in this next consumption cycle: these are subsumed by the 'badging' of vintage (Duffy and Hewer, 2013). This creates a

marketplace in which practitioners are attempting to distance and distinguish themselves through their consumption practices, but operate in a marketplace in practice that replicates branding discourses of the mainstream culture. It is through this ambiguity between the commercial mainstream and the discourse of alternative consumption that vintage is culturally produced and replicated. This active role required of the vintage practice will be unpacked in the next section before exploring the competence needed as a practitioner to participate.

5.2.3 The Practice: The Knowledgeable Practitioner

"...It is almost like the process of walking into a shop and buying a vintage inspired design in Topshop or Urban Outfitters defeats the point. The people at the markets want to have that story of a 'find' and of being different in how they came across the object." (Researcher diary, Little Birds Market, March 4th 2012)

The act of vintage and 'being different' is tied to the physical activities such as finding, examining, evaluating, haggling, socialising and interacting observed in the marketplaces. This talent for spotting potential treasures is heralded as a venerated skill in the collective. Vintage dressing is seen to encapsulate mixing vintage items with new, accessorising from a variety of styles and periods, using *knowledge in practice* to enact these style preferences. The markets provide a space for the exchange of advice and encouragement, but also of aesthetic and tacit knowledge, along with the trading of hints and tips and the prediction and enforcement the vintage aesthetic. In the vintage marketplace there is a cache and cultural capital (Bourdieu, 1984, 1985; Rafferty, 2009, 2011) at work in being able to uncover hidden 'treasures' that differ from the mass production approach of the high street.

Practitioners pride themselves on knowing the markets, but this leads to a questioning of what sort of knowledge they have and how it is materialised in practice?

From the observations of the sellers in the marketplace, the undertaking of their business involves 'hanging out' in the market and a deep absorption within the scene. Much of the knowledge distributed (Entwistle and Slater, 2014; Araujo et al., 2010; Slater, 2002a) in the collective is therefore dependent upon 'being there', within the face-to-face interactions of the marketplace. As Crewe and Gregson assert regarding car boot sales, *"Vendors are also buyers, and buyers in turn are also browsers, sightseers, momentary tourists"* (Crewe and Gregson, 1998: 50). Such knowledge needed to navigate the site is in turn integrated as second nature into their practices and performances; as Jackie comments:

"It is an ability. I don't know whether it is because you look at textures, at the stitching, it just sticks out like a beacon. You are always looking for quality. Secondly I will always look at the label and then I will always look at the decade, for me something has to scream decade: it will have to make a statement for instance to say, this is the 80s, the 70s or else it will have to be quirky or else it will have to be totally current." (Jackie)

Here Jackie, a vintage storeowner in her forties demonstrates her affiliation with vintage clothing and establishes the practiced skills essential to the art of selling within this marketplace. She runs a vintage store in the west end of the city on the busy Byres Road. She is petite and professional to meet in person. She runs numerous businesses across the city. She has been involved in the textile industry for

more than twenty years and prior to opening the store she had been selling vintage as wholesale. She explains that her mother had been into fashion in the sixties and that she has always had an eye for details and quality. For Jackie the appeal of vintage resides in the quality and cut of the fabric. She explains that as she is so petite in frame, vintage silhouettes work well on her. She spoke of her particular draw to 1960s two-piece suits and Grace Kelly inspired fitted dresses. She attends the markets to pick up pieces for herself and also to 'see what people are doing out there.' Jackie's store is updated daily with new items and they operate on selling pieces at affordable prices and selling in quantity.

For Jackie calculation (Slater, 2002a) may be at work in vintage; but equally passion and knowledge are two essential factors for this process to succeed. As she articulates, knowledge and skill have to be developed in the marketplace, although happenstance and serendipity sometimes play important roles: in this way vintage remains a labour to be practiced and accrued over a lifetime. Building on Callon's (1998a) proposal that markets are composed of a range of practices that are the result of different forms of expertise and material objects, although on the surface these practices appear like 'gut instinct.' As Schau et al. (2009:27) suggest: "practices have a common anatomy and varied physiology evincing discursive knowledge, explicit procedures for doing; know-how and tacit elements, taken for granted knowledge of worthy projects; and affective commitments to brand-centred practices." Throughout the research practitioners spoke of learning and spending time developing these understandings of the value, knowledge and use of objects. One noted way through which they explain this learning about goods, is from their upbringing, of family practices of searching for items and of older objects being prized:

"My Mum likes going into antique stores, second-hand stores – and I think that's been instilled in me since childhood. In that way it is more like a value that I have been passed on about seeing the merit in older things - that not everything has to be new and shiny. She would see a beauty in something even if it was torn or too big and knew that she would be able to make it work. Now I suppose that is just my style." (Michelle)

"My mum has quite a lot of sort of old fashioned, not necessarily old-fashioned style but an old fashioned approach. I would say that was especially true with furniture as she is all about the quality of older pieces. I spent a lot of time when I was younger with her going to look at pieces in old dusty antiques places and she would inspect them and feel the grain of the wood and smell it – it was all really tactile. Which when I was a kid was really dull and awkward at times [laughs] but I now do the same, especially with old leather chairs." (Jen)

"I can tell you how I got into it, if that helps you, so when I was a wee girl what I liked was stuff. From a very young child, from a very early age, I am talking about the ages of 3, 4 and 5, going to my Gran and Granddad's, and my Gran liked stuff. My favourite thing was to empty her kitchen drawers, they would just be packed with stuff, and I would organise them and categorise them and just play with the stuff." (Alice)

From these excerpts respondents demonstrate an embodied feeling for vintage, that increasingly it was a part of their make-up and an aesthetic that they had practiced from childhood. This shows that practices develop over time and the marketplace offers an opportunity to foster and cultivate practice (Schau et al., 2009). The embodied nature of this knowledge can more fully be captured through a turn to Bourdieu's (1990) habitus. Habitus refers to: "principles of the generation and structuring of practices and representations...objectively adapted to their goals without presupposing a conscious aiming at ends or an express mastery of the operations necessary to attain them" (Bourdieu, 1990:53). Within the vintage context, the habitus structures ways of doing, being and knowing, this is useful to consider when exploring how taste and aesthetic practices come to be acquired. In this way vintage is seen as a practice cultivated over many years and one that is not only of the doing of dressing and behaving, but of valuing objects and seeing the quality and craftsmanship in them: it was an appreciation of the practices that form these goods and make them last the test of time in the consumption sphere. Lave and Wenger's (1991) work conceptualises practices as 'apprenticeships' as over time consumer engagement with the practice develops and integrates further. Marketplace sense making has to combine both rational calculation and codified knowledge with a temporal and learned understanding of the practice of vintage. This is seen also in Osterlund and Carlile's (2005: 97) study, as they note members "do not merely learn about practices, they become practitioners." Dedicated vintage practitioners such as Jackie favour getting their hands dirty in the finding process even though the time spent sifting and 'vintaging' (term used to dentote the entire labour of the vintage practice in action) does not necessarily result in immediate gratification. For vintage practitioners the buying process differs from more mainstream retail practices in that items are purchased without specific occasions or events in mind and buying becomes about a provision for the future. Seasoned practitioners spoke of 'buving things when you see it and figuring out what to do with it later' (Annie) and 'buying something and knowing that I will use it at some point, when it is right' (Carrie) or this ability to unlock this potential value without an immediate need or want was a common theme in the data. When an object is uncovered by a potential buyer in the marketplace it is seen to be in a more natural state than if it had come directly from a store, this in turn serves to reinforce the feeling of achievement upon uncovering an object. Jackie spoke of being able to focus in on the best goods because of her experience and that the rest was left to practitioners without such a trained eye for quality. This reinforces Parsons (2010: 294) view that dealers translate objects from one context to the next as "market intermediaries," as they have developed a "finely tuned aesthetic sensibility to discern the good from the bad, the beautiful from the ugly and the authentic from the inauthentic." Practitioners are situated within the marketplace network, which allows for learning and knowledge exchange for both consumers and sellers. The market context illuminates that knowledge and the social practices of market exchange are entrenched within vintage. Market actors learn through trial and error but this specialised knowledge and object discernment serves as competitive advantage within the marketplace. Practitioners must mobilise their knowledge and skill to thrive in the dynamic marketplace.

This specialised knowledge is especially pertinent in the pop-up format of the marketplace as it lends a transient and sometimes hectic atmosphere, which serves to add to the heightened sense that the objects are 'one of a kind' and only being sold on this particular day. In this way, the objects have liminal moments and present

opportunities for their potential to be discovered in the marketplace before they will be boxed up again and moved on by the seller:

"It feels like you always are under pressure in case you don't see the item again. Most of the sellers are only there for that one market and then you might not get another chance. I suppose that is always my underlying feeling at the market, if I don't buy it, I know I will spend ages looking for that exact same thing again and never find it." (Ben)

Here Ben demonstrates the essence of the vintage market: the goods are hard to find and are only for sale in a small window of opportunity. Ben spoke of feeling pressured around the consumption decision and reflects on the opportunity presented at the market. The ambiguity around the terminology also means that prices could rise and fall, leaving opportunities for a bargain to be struck. The transient nature of the marketplace in this way serves to add value to the objects, as they are often seen as 'finds' and would thus be hard to track down again after the market had closed for the day. Further the vintage market can be framed as different from the traditional marketplace, or as Mary Douglas explains: "...a ritual provides a frame. The marked off time or place alerts a special kind of expectancy, just as the oft-repeated "Once upon a time" creates a mood receptive to fantastic tales" (quoted by Goffman, 1986a: 252). In considering the marketplace, the actors but also the stage and performance itself are critical in this construction of vintage value. As active practitioners, their skill and competence sustains and enacts the marketplace with an air of expectancy and a feeling of 'something' may happen. Stallholders are thus crucial intermediaries in the vintage exchange; they are crucial in moving the object from one lifecycle to another and in the process re-creating meaning and value.

This focus on how an object becomes valued in the marketplace necessitates a consideration of the movement in value and the subtle transformation in how an object is consumed. For Shove and Araujo (2010), value is created through practitioner skill, such that objects only received value when "*in the hands of someone with a requisite level of competence*" (2010: 20). This is an important consideration when considering a marketplace context, such as vintage, that is constructed on the basis of items being re-seen and re-evaluated. Appadurai (1986) posits that objects have no absolute value, that rather their value is created through the social process of exchange. This echoes Kopytoff's (1986) suggestion that objects move in and out of 'commodity stages' in their trajectory. Following this argument, in the vintage marketplace the object is reliant on another consumer finding it and it being brought back into the consumption sphere by having new knowledge applied to it:

"I will see something and know that I can breathe new life into it and cherish it, others might not see that or appreciate the beauty of these items." (Elena)

For Elena vintage therefore becomes a practice of understanding and appreciating potential value. This illuminates that for vintage value to be accrued objects must be transformed in an active process of interpretation and understanding by marketplace actors. Her suggestion that the notion of vintage is fluid and "one person's junk, is another's treasure" is further evidenced in expressing the magic of the moment of finding and that on occasion the object has been brought to life for her though presentation and that through her own practice of finding and uncovering, but the marketplace facilitates this uncovering process for her:

"It seems like the presentation of clothes makes them come alive almost. They will be styled and presented on the hangers or just displayed so you can see them. It feels a bit like some of the work is done for you but it definitely makes it easier to visualise how you can use the clothes and integrate them into your everyday in a less scary way" (Researcher diary, Judy's Vintage Fair, Sept. 2011)

Figure 3 - Judy's Vintage Fair 'Oh Happy Days' stall, Sept. 2011



Elena's insight, the visuals and the researcher's notes start to illuminate the notion that the objects have to be made evident to the viewer for them to enact the potential value they possess. Practitioners adopt the role, as 're-enchanter' and 'storyteller' bringing these long forgotten objects that exist in the periphery of consumer lives, to the foreground and repurposing them for the contemporary world. Vintage objects have a 'history' to them, which can potentially be seen to add or detract from their value. The practitioner to increase the value, for example through styling, elaborates these 'stories' or histories of previous consumption cycles or conversely sellers spoke of attempting to distance the object from its history through practices of mending and re-appropriating of the object: "I will get something and it will have a rip in it or something and I will know if I spend some time with it and fix it up, it will be able to be sold on and desired."(Lindsay)

This is representative of a complex intimate relationship with the object, as Parsons (2006: 189) posits in regards to antiques "*dealers are involved in creating new lives for objects, as they die in one context, they are revived in another, thus becoming re-enchanted.*" This *'re-enchantment'* for the marketplace audience is crucial in the vintage scene. As Melanie spoke passionately of 'the story' from the object:

"I like the idea of someone loving this dress before me, or loving this bracelet before me – and something happening or I like the interaction, the story that happened before the object became mine." (Melanie)

For Melanie, part of the consumption experience of vintage is 'the imagining' of previous others. Melanie is in her late twenties; she studied fashion business at University but now works in an office. She describes browsing charity shops and jumble sales with her mother when she was younger and how as an adult she embraces this way of consuming. She describes that at University 'anything went' and vintage was a way to stand out and show that you had interesting style. She explains that for her she liked the 'back-story' that goes with the item and that it might have a history that you could imagine beforehand. She is not loyal to any one particular period and mixes vintage with high street frequently. She visits the markets regularly and often would travel to attend the market. She knows most of the long-time sellers by name and greets Elena warmly. For Melanie vintage is a method to illuminate her fashion skills and knowledge:

"For me vintage is about putting a look together, vintage is unique and you pull on past eras to add to your modern day style... I love it, that excitement when I find an unusual piece and can tell a story about it." (Melanie)

Continuing:

"...It is about how people make it work for me, new stuff, mixed in with much older pieces – for me someone like Alexa Chung or Fearne Cotton get it right...a vintage floral dress, with a Mulberry bag and Topshop boots, simple." (Melanie)

Here Melanie describes the imaginative process of bricolage, she highlights the epitome of vintage for her is that it should be used along with current design and trends to showcase fashion knowledge. This describes a careful mixing and that within the tribe it acts as this playful arena (Kozinets, 2002) to pull on this creativity and knowledge. In this way, vintage objects can be viewed as *'trophies with complex cultural and economic histories'* (Palmer and Clark, 2005) and as Melanie alludes to can be seen to be *'signifiers today of unique fashion'* (Palmer and Clark, 2005:174). The objects do not exist on their own; they are carefully negotiated and reappropriated through these tribal aesthetic codes and practitioner learning (will be discussed in more detail in Socialities sections).

This section has shown that vintage becomes an everyday object that is (re)imbued with knowledge and through the marketspace it is made visible. Vintage is therefore not an inherent characteristic of the object, but is a result of emotional and social relationship, in which consumption is an outcome of this engagement and knowledge in practice. In this way, value is embedded within the objects and in the practices of appreciation and competence. Vintage goods have characteristics that have merited their inclusion in a second or third cycle of exchange. The meaning attributed and value constructed around the object are created and manipulated by marketplace practitioners. In this way vintage is culturally and socially shaped through practices as the object's status is created through a process of being withdrawn and introduced in a new setting (DeLong et al., 2005). In turning to the vintage context, the tacit knowledge of practitioners is illuminated and suggests that practitioners are 'embodied locations of knowledge' (Bourdieu, 1984, 1985, 1990; Entwistle, 2009). In seeing value as created and nurtured as a result of this knowledge, ties in with the work of Slater (2002a) who proposes that for an economy to exist at all, objects must become stabilised through 'processes of materialisation'. With vintage, the object must become calculable and be given meaning by marketplace actors at a given moment in time. Slater (2002a) argues that all commodities are calculated and materialised within the market and economic action: "The very notion of a market depends on an anthropology of goods – the boundaries of markets follows the categorisation of things" (Slater, 2002a: 97). This also illuminates the cultural coding of objects and how this is stabilised by the market (Slater, 2002a) and then evaluated as vintage by the marketplace. In this way, the marketplace provides the stage for this materialisation to occur. There is a complexity to where vintage sits in the consumption, production trajectory. This in turn has important implications for value consideration, if the marketplace is viewed, as constantly in transition, with objects moving along the consumption trajectory, then value cannot be deemed to be fixed either, rather value is constantly in flux and being renegotiated according to changing circumstance.

5.2.4 The Practice: The Materiality

This section provides an overview of materiality and links back the literature around the construct of materiality as an interdisciplinary approach to exploring the links between the material world, social organisation and culture. In considering the research objective as to how such material conditions come to be associated with the subjective construct of vintage, Miller's (2010) material culture perspective is considered. As outlined in the literature chapters, for Miller (2010) wherein to understand people, he argues it is necessary to understand things and the dialectics of our relationships with them. This approach stresses the dialectics of objectification (Miller, 1988b) as value and meaning constantly shift between people and objects in their everyday use. In applying this material culture perspective, it is argued in this section that the narrative of the object cannot be separated from the object itself. In conceptualising practice as an interaction between ideas, ways of doing and the material (Shove and Pantzar, 2005), this focus on the material is necessary to fully understand the practices of the vintage collective. In the context of the vintage marketplace, objects although possibly once mass-produced, have outlasted their counterparts and re-entered the marketplace: given a second or third opportunity to shine and garner value. As shown in the previous section, vintage is in practice worn, not simply collected, venerated and admired. Some items are treasured and preserved but it is without the reverence in often cases that would be bestowed on antiques – the objects are seen as wearable and active:

"I have a 1940s cashmere baby pink twinset and I wear it all the time. Mostly I will throw it on with jeans or a cute skirt; I'm not fussy about how I wear it. I know I should probably keep it special or whatever but I think it was made to be worn and it will be lost sitting in a wardrobe, forgotten about...it feels like I give it another go at life when I wear it." (Melanie)

As Melanie's comment illustrates there is no sanctity shown towards tradition or original purpose of the object, with the practitioner selection guided by fit, originality, wear-ability and desirability. The transformation of old clothes into 'vintage' and the acquisition of value and elevated status and creates hierarchies of finds. It is seen that objects fully participate in the marketplace along with the actors and consumers and how they are presented generates credibility and above all value. There is a constant reconfiguration of the meanings and affection towards the object; however some objects stand out for practitioners as a 'special find' or holding significant personal attachment:

"I've got all my old vintage stuff that I would never sell and I don't see any new stuff anymore but if I did see a lovely fifties dress that was £50 I wouldn't buy it now, rationally it is pointless cause I would just be buying it to covet it, to have it. I mean I really covet them, it is special, I really love old rose print things, and there is quite an inquisitive nature to it. I want to see things and have things and I can't let things go. With vintage pieces, special pieces, I do have an emotional attachment - possibly even there isn't any emotion there but I think I do. I'm scared that if I sold something and let it go that I wouldn't get it again, does that make sense? Because I know how difficult that find is, because I am in the business I know how difficult it is to actually get things now." (Alice)

Alice reveals a tension, that in her selling role she knows that she should not pay too much for a purchase but then as a consumer she attempts to explain the relationship and attachment that she has to the objects and the complex nature of her feelings. In applying this tension to the marketplace, the practitioner has to elaborate a material strategy to allow them to integrate the object into their practice. A change in the attachment and affection for the object is necessary and the material object is embedded in this complex shift of actions, behaviours and forms of doings in the marketplace. All respondents spoke openly of an object that was a memorable find for them and the feelings around this uncovering are detailed in personal stories of loving, adoring and caring. In contrast to Melanie's carefree discussion of wearing her twinset, Michael highlights the responsibility he feels to certain items and the relationship he has with these objects:

"I have a couple of things tucked away, pristine 1940s shirts, I haven't even brought them out recently cause it is almost like I know they are so good, that I don't even want to wear them, every time I think about wearing them I just look at them. I just love them. I just love the actual items." (Michael)

He goes further and expands on his feelings of finding something he considers special:

"There are just one or two pieces that I just don't know - it is almost like a sexual feeling towards them. It is a strange thing; it is larger than life considering it is just an item." (Michael)

In this account there is recognition of the strong feelings about the objects and Michael considers the objects to be almost intimate: with the love and sexual feelings that he bestows upon them. Although he is self-aware of this practice occurring, for him the objects become more than practical objects to own, he imbues these finds with desire and intimacy. The idea that he has to save them, to preserve them from the everyday depicts the fragility of the objects but also the reverence with which he treats them, they become almost sacred in his affection (Belk, 1989). Vintage in this way was part of the everyday for Michael but he felt a duty of care and responsibility to the objects he has kept for so long. This can be seen in the meaning and subsequent value attached to individual possessions (Belk et al., 1988), the provision of this value and therefore the concept of vintage is argued to be a socio-material practice (Miller, 1988b, 2010; Campbell, 1996).

In the markets a fundamental marker of vintage is seen as the wear that the object presented with. An object possessing a certain amount of wear to indicate its vintage nature is seen as a positive thing but there is a carefully considered line of objects being 'too worn,' as Adam explains regarding a leather bag:

"It was battered and bruised and so well lived in, I think leather always looks better when it has been handled for a good ten years plus. I would always take vintage well-used leather over a pristine new leather piece any day. It looks too like you just left a store with it." (Adam)

The previous life of the bag is evident for Adam from the creases and markings on the leather but he only sees this as a positive indicator of 'vintage', as the bag has lost it's just purchased, immaculate condition. For Adam, being able to see the wear and history of the material is what holds the value of vintage over new goods. He prizes the material conditions and the suitable wear of the object and he labels this wear as 'vintage.' In the market, discussions around the object centre on mending and care of objects to overcome this wear that they had sustained if it was considered 'too much.' Achieving this balance between finding an object and the condition of it is a constant presence in the markets. At Judy's Vintage Fair, sequined 1920s dresses are heralded as unworn and worth more money because the beading and stitching is intact, in the Granny Would be Proud market, as described by Adam, leather jackets are seen as scuffed and showing signs of wear and rather than diminishing value this increases their desirability. Finding objects with suitable wear is attached to cultural capital (Bourdieu, 1984, 1985) of the practitioner in the marketplace:

"Conversations frequently started about items of clothing: where it was from, where they had purchased it from, who it was designed by, how much had been paid for it. Audible gasps of excitement; chattering and admiration were heard over exchanges as stories of finds were told. These added to the atmosphere of frenzy in the marketplace." (Researcher notes, January 22nd 2012)

Part of the practice of vintaging is in being able to retell these narratives of uncovering an object, recounting its wear and giving it relevance again. For all respondents the condition of the object is a significant consideration, Jessica mentions that her favourite bag is *'in near perfect condition'*, Michael spoke of repairing and cleaning his glasses, all seem to accept that the care and time taken over objects is part of the vintage practice and adds to their value. The process of removing labels and brand identifiers from the material objects was also seen to be at work in the marketplace, ridding them of previous lifecycle attributes and positioning objects to invite intrigue and imagination. Attributes and aspects of objects are hidden through the disorder of the retail marketplace environment, which adds to the ambiguity of the objects (Bardhi, 2003; Slater, 2014). The vintage marketplace provides a temporal distance from the past owners of the goods as a negotiation by experienced practitioners of this past and the imagined Other is sanitized in the staging of objects. Jessica explains how as a consumer she negotiates the imagined past of the object:

"...I mean I know at the end of the day it is second-hand but vintage is that bit more special. It is something that has lasted and someone has discovered and spent time and energy revivifying. Sometimes I like to think about who might have owned it before and where they wore it to and how they styled it, but I mean I do like the things I buy to be clean and quite pristine. But I mean finding something in near perfect condition is always the vintage dream." (Jessica)

Jessica spoke of vintage as a way of consuming, for her the imagined past is a way to connect to the object. Within the consumer research landscape the role of imagination in the consumption experience is prominent, especially when related to play and fun (Holbrook and Hirschman, 1982). In Illouz's (2009: 399) work imagination is based on *"sensations, feelings and emotions to make present that which is absent."* This imagining of an absent Other by Jessica is part of the consumption experience of vintage. But like many consumers she still expresses a desire for the vintage to be as 'new' as possible. Vintage in this way stylises the old and attempts to make it new again by creating consumer desire and intrigue. Michael suggests that in reality vintage was a stylisation of second-hand, which could then be reflected in the prices charged:

"Everyone except the lower classes has hijacked the vintage or technically the second-hand trend a bit. Charity shops are now all posh and merchandised, second-hand places are now fairs instead of jumble sales and bring and buy sales, the prices have elevated. If you can make people want something, then you can charge more for it." (Michael)

Michael reflects that vintage objects have long existed but that have been reorientated for a new audience. He spoke of the skill of vintage as "trying to read trends and judging what will sell." As with Denegri-Knott and Molesworth's (2009:313) exploration of eBay practices, a key focus is revealed in the 'work' that is involved in trying to 'maximise the economic value of goods.' Campbell (2005) notes that concepts such as 'possession,' 'personalisation,' 'customisation' and 'appropriation' overlooks the transformative processes and practices of consumption. Vintage here could be seen as being rooted within the objects as notions of 'value' appear highly bound to this intricate relationship with the object. Value as a result appears transient and enacted by the object through context and by practitioner competence. Entwistle (2009) notes that similarly with aesthetic value, it must be seen as something that is not necessarily a quality intrinsic to the object per se, rather it becomes processual and an attribution of value that is gained over its journey - aesthetic value is "created, accrued and attributed" along the network in which the object moves (Entwistle, 2009: 58). As a result, the process of qualification has to be continuous and commodities need to be examined continuously and should remain open to interpretation and transformation as they move from one actor to another in the marketplace (Entwistle, 2009: 68). In referring back to Miller (2010) in which this findings section starts out, he illustrates how objects, far from being either permanently alienating overpowering ideological objects or simply existing as the ephemeral background to life, actively make people as much as people make objects. This section has shown that objects are inextricably

linked with the both the practices of vintaging and the practitioners who actively shape the narratives of vintage.

5.2.5 Conclusion of 'Practice'

This first findings section has shown that the marketplace of vintage was constructed through networks of objects, practitioners and social practices. Sellers translate objects from one context to the next and success in the marketplace was reliant on the development and display of knowledge of the market, consumer desire and discernment for the potential of objects. This section has shown that ways of marketplace knowing and doing varied out with purely cognitive or economic domains, instead what counts as knowledge is questioned and knowledge as learning and performance emerges emphasising the practice of knowing. In line with Wenger's (1998) understanding of practice, within the practice of vintage, it is based on the assemblage of different kinds of objects, understandings and processes. Various knowledges (knowing as practice and performance) are intricately woven together across objects, spaces and tools to make sense of the market. In order to calculate and attribute value within the marketplace, practitioners must have an understanding of the market. Knowledge in vintage is based on sense making and engaging non-cognitive abilities, it is formed through encounters with the objects and marketplace experiences, in this way it is embodied (Bourdieu, 1984; Entwistle, 2009). Knowledge of vintage could only be acquired within the markets and as such the body emerges as an important location for knowing (Entwistle, 2009). Practitioners become internal to this transient world as such they are best placed to negotiate and predict trends and marketplace shifts. As with Denegri-Knott and Molesworth's (2009:313) study, a key focus was revealed in the 'work' that was

199

involved in trying to 'maximise the economic value of goods.' The fluctuations and instabilities of the pop-up market demanded particular practices to be adopted by participants for coping with this aesthetic labour involved. The findings reveal that the practitioners themselves transform the marketplace and the material environment. Practitioners act as 'gatekeepers' in the market, actively calculating (Entwistle and Slater, 2014; Slater, 2002a) taste in the process. The marketplace setting and aesthetic labour (Entwistle, 2009) recreates the experience of searching for finds and the physicality of the vintage process allows a heightened sense of accomplishment for the practitioner in the searching process. These marketplace behaviours were encoded and embedded in the norms of the collective and relied on individuals being educated or socialized into the vintage tribe (Bourdieu, 1984, 1985; Rafferty, 2009, 2011).

In considering Lury's (1997) consideration of '*object-people-practices*, ' within the frame of this research study, the marketplace illuminates how value was enacted around the vintage label or brand (Eckhardt and Bengtsson, 2010). In order to endure, practices must be repeatedly enacted and reproduced as specific bundles of organised activities. Moments of practice were shown to be multi-sited and the objects themselves integral to the networks of which they are a part (Hui, 2012). In looking at consumption within practices, as Appadurai (1986: 31) suggests is *"social, relational and active"* and this approach allows for a more thorough understanding of the meta practice of consumption.

5.3.1 Introduction

Referring back to the literature in Chapter Two, Arnould and Thompson (2005:869) highlight the value of "study[ing] in consumption contexts" and the first findings section highlights that vintage happens in a physical space in the markets through which the material environment is actively transformed by practitioners. Returning to the literature, the consumer research discussions of shopping malls and retail stores as 'nonplaces' (Auge, 1995) highlights that they are lacking in social interaction (Sherry, 1998), this led to the development of commercial settings as 'experience driven' and a 'places for co-creation' (Maclaran and Brown, 2005; Sandicki and Holt, 1998; Kozinets et al., 2002; Sherry, 1998). Maclaran and Brown's (2005) study of the 'utopian marketplace' illuminates the commercialised spaces of the festival mall in Dublin and an understanding centred on the creation of 'playspace' and 'artscape' (Maclaran and Brown, 2005). Moving this discussion forward this findings section explores the socio-cultural environment of the markets and the role that the space itself plays in the materialisation and consumption of vintage. This led to an initial questioning of how the transient and ephemeral nature of the vintage market marked them as different spaces to practice and consume?

5.3.2 The Space: The Market Setting

The notion of the staging of the marketplace has developed in the consumption literature around the language of Sherry's "brandscape" (1998), Ger and Belk's (1996b) "consumptionscape" and Arsel and Thompson's (2004) "hegemonic brandscape". Described as social environments grounded in entertainment and fantasy, these accounts explore the relationship between consumers and the environment but omit the active role of objects in these marketplaces. In using the practice lens, the active role of objects in the marketplace is key and in further applying a material culture focus to the vintage marketplace, this section highlights the spaces of materiality (Arsel and Bean, 2013; Borgerson, 2005; Miller, 2005; Denegri-Knott and Parsons, 2014). This focus allows the researcher to explore the particular aesthetics of the vintage scene: but also the modes of presentation, illumination and action made material. The concept of time and its appreciation is tied to vintage from the outset. By the very definition of vintage, the second-hand object is conceptualised as 'vintage' only through a certain amount of time passing. This led the researcher to question how was this re-appropriation of the past through objects related to marketplace practices?

As a created market the notion of bringing vintage to life and offering an alternative space from the high street or online retail punctuates discussions of it being "*a different environment to shop in*" (*Tina*). The shared social space of the marketplace creates the backdrop to the exchange and as has been shown in the previous sections, it frames consumer encounters with others. The location of the vintage markets within the urban domain of the city also serves to reinforce the vintage stock as authentic. In the west end of the city, Granny Would be Proud is positioned just off Byres Road, the hub of student life in the city and Sloans in the city centre is located in one of the oldest pubs in the city, accessed through cobbled lanes, whilst Judy's market uses the Glasgow University Union that was founded in 1885. As a transient market the feeling for vintage has to be created and staged every two weeks in both the Granny Would be Proud and Little Birds Market and every quarter for Judy's.

The architecture of the buildings offers an opportunity for observing the unfolding of the markets, through time and space and the practices that materialise the market.



Figure 4 – The Balcony setting of Granny Would be Proud market

In considering conceptualisations of the marketplace as an arena of exchange and of place as having a defined meaning, Tuan's (1977) understanding of the difference between place and space becomes relevant in a sensemaking of the vintage markets. Tuan (1977) highlights that place refers to security and a space that has meaning attached, whilst space denotes freedom or where no value has as yet been added. In applying this to the vintage pop-ups, the understanding of a marketspace conveys the ambiguity and transient nature of the space – it is a space that is constantly reworked and reconfigured in practice. Taking over disused spaces, the upstairs of a restaurant (Granny Would be Proud), hall of a Student Union (Judy's) and spare rooms in a bar (Little Birds), the construction of a retail space and creation of a suitable atmosphere from wooden tables and mismatched chairs is imperative to the feel for the markets. This is illustrated strongly in the arrival and set up of the stalls:

"Vendors painstakingly bring their stalls to life: prized objects are unpacked from tissue stuffed boxes and containers carried carefully from cars outside. It is amazing to watch, they all have a basic space but dress it with hat boxes, china cups, table cloths, even standard lamps and then their wares... before you know it they have created a little vintage haven that reflects their personality and the goods for sale" (Field notes, Little Birds Market, February 26th 2012.)

In these alternative locations, the space is transformed through practitioner action and styling. This cycle of creation and destruction, this active process is seen from the observations as being critical in creating acceptance in the market. To start with there is an air of urgency as vendors arrive. The most pressing demands at this time are the practices of setting up the table, unloading stock and securing the transport of this stock to the venue. Having moved such objects, vendors busy themselves in dealing with the urgencies of time and their calculations over spatial display:

"Vendors arrive to be shown to a rectangular wooden table, a lamp and a chair. From these objects they have to create their own area that showcases their wares and establishes themselves as different from their close neighbours. The unpacking and setting up process seems to be crucial in the cultivation of a vintage look. The more seasoned vendors arrive early and with a careful system unpack daintily embroidered table cloths, dressing the wooden table as they go, cake stands, hat boxes, weathered suitcases, all revealed to store treasures and present them to the awaiting crowd." (Researcher fieldnote, Granny Would be Proud, February 12th 2012).

The construction and practicing of the 'look', both for oneself and for one's stall, thus adds a degree of credibility (*as aesthetically belonging with the collective*) to the seller in the marketspace. Once the market commences an air of expectation and anticipation can be felt and the market day is marked by moments of action:
"The day ebbs and flows as consumers enter and move through the space.

Throughout the day this rhythm changes as the audience alters: inner members of the collective arrive early and eagerly anticipate rummaging for "treasures", with the Sunday strollers of mid-afternoon creating a more relaxed and less exclusive scene, with newspapers tucked under arms and hushed tones of 'don't touch' to children. The crescendo of the day falls mid-afternoon as the sound of consumers' chat hums in the air, the beats of the 1960s pop music sets the tempo, with long conversations between stall holders and consumers ensuing the past is continually invoked as a reference point for understanding and interaction, to hurried exchanges in which objects are inspected and paid for. As the end of the day approaches the pace slows, stallholders chat to one another about their 'take' for the afternoon and attentions turn to carefully packing and organising, and the act of deconstructing the stall" (Researcher fieldnote, Granny Would be Proud 15th April 2012).

In the marketspace there is an abundance of niches and corners with stalls and this creates the feel of something spontaneous and without strict retail control measures. Stalls often have a disjumbled appearance with suitcases and boxes brimming with items around the floor of their stall area. This marketspace setting recreates the experience of searching for finds and adds physicality to the vintage process and allows a heightened sense of accomplishment for the practitioner in the finding process. In this way the marketspace appears as a version of the traditional jumble sale, however the merchandising and staging allows for the process of discovery to take place in a simulated environment. This compliments Bardhi's (2003: 375) work on thrift shopping whereby she asserts that *"disorder is created by placement of products of different categories next to each other."* Or as Denegri-Knott and

Parsons (2014:94) propose: "The second-hand objects sold at car boot sales are somewhat distant and disembedded from these systems of meaning, their very mode of display often involves disarray, and cues can only really be sought from the individual store holders as to their potential uses and past lives." This scene setting of the vintage marketspace structures the finding practices and consumer experiences. This manipulation of the marketspace context displays that the placement of objects and the staging of stock serves to allow the practitioner to discover what has previously gone unnoticed.



Figure 5 - Judy's Vintage Fair: 'Practice of browsing', 16th September 2012

Figure 6 - Judy's Vintage Fair - 'Practice of Organising' in the marketspace, 18th May 2013



Figure 7 - Granny Would be Proud: 'Practice of display', October 7th 2012



The marketspace is created from this careful management of spatial confines (organising and displaying) to facilitate practices of browsing and finding. The lighting is often quite dispersed - to illuminate goods and displays and to create a warm atmosphere the stallholders use lamps - but this highlighting does not extend far beyond the table where consumers gather. The use of materials such as lace doilies or floral print tablecloths supports an atmosphere of a homely and welcoming space (McCracken, 1989; Linnet, 2011). This setting up one's table to best display stock became an art of practicing: an exercise in staging and performance through objects. Gregson and Crewe (1997: 87) reveal that alternative market spaces can be one of play and performance: *"the conventions of retailing are suspended, and where participants come to engage in and produce theatre, performance and spectacle and laughter.* " The vintage markets come to represent sites for sociality and performance, as practitioners suspend formalised retail behaviours and instead engage in and produce theatre and create a whimsical space. Vendors place objects to best display their items, sometimes constructing a front and back stage through the layering of the table to best attract the attention of mobile customers, and thereby encourage them to linger:

"You start to learn what makes your stall attractive, sometimes it is through trial and error or being inspired by seeing how someone has put their stall together. We try really hard to make it easy to see the stock but also to get across some of our personality, so it is never going to be too 'done' it will always be a bit like a dressing up box..." (Charlie)

Charlie's account places an importance of conveying a personality to the consumer and of this portrayal being a learned experience. This shows that practices develop, evolve and deepen over time. Importantly she emphasises that it should not be interpreted as being 'too done', this was a common thought throughout the interviews, the practice of the creation of the vintage aesthetic should appear naturalistic and almost effortless in its construction regardless of the observed labour involved. The production of 'the look' of the vintage marketspace is seen to be integral to practitioners and suggests concepts of authenticity, acceptance and stylisation. The 'right look' is seen to add to their reputation and either attract or detract consumers from engaging with them. This creative process creates a feeling for the market and for the sellers. The well-managed atmosphere of Granny Would be Proud, Little Birds and Judy's is a mechanism for stallholders to show their own personality within the vintage context but also to act as a framing of their aesthetic vision. This striving for 'the look' also ties in with the consideration of the socialites of consumption, as the achievement of this is also crucial for the staging of the stall, but also significantly for both the stallholder and consumer. The practice of 'the look' is seen as key in gaining approval in the marketspace:

"The sellers arrive and are greeted with a brown table and a lamp, on returning an hour later, they bring the space to life with ornate, embroidered table cloths, rose adorned cake stand holders overflowing with costume jewellery, tweed suits and silk scarves hanging over the walls, suitcases brimming with goods at the foot of the table..." (Researcher fieldnote, Granny Would be Proud, March 4th 2012)

The communicative staging of 'the look' influences the practices and processes of all market actors. It shapes the objects chosen for sale, the aesthetic presentation of the stall and consumer desire (Belk et al., 2003; Campbell, 2005) achieved:









The above images show Annie's photo diary of setting up her stall in the Granny Would be Proud market, it was a practiced act and one that she spoke of streamlining and adjusting with experience. In this manner the performance of vintage is not just a singular act, but also a repetition of assumed norms within the space. The importance of 'the look' is mentioned frequently by respondents, as are the material conditions, objects and sensory factors that differentiate the marketspace. A look of what will be termed in this study 'eclectic homeliness' is evident, as if cupboards and wardrobes have been opened and stock makes its way effortlessly to the market (McCracken, 1989; Linnet, 2011). This feeling that the stall could have been formed from overflowing cupboards creates a feel of playful abandonment for practitioners or disorder to navigate (Denegri-Knott and Parsons, 2014) – as if left to riffle through someone's forgotten treasures. The marketspace is manifested in a practice of blurring the boundaries between the commercial space and the familiarity of the domestic domain. This recreation of the familiar through the performance of staging and negotiating 'the look' adds to a sense of belonging for practitioners in the marketspace. This extends Debenedetti et al.'s (2014: 918) recent work that

illustrates that 'homeyness' can reside outside domestic settings and can be experienced in commercial settings. As Debenedetti et al. (2014:911) asserts:

"Spatial knowledge is often also connected to a desire to mentally, and sometimes physically, appropriate the place because for customers, my place also means the place where I belong."

The vintage marketspace creates a porous boundary between practitioners (or sellers and consumers); as such the consumption experience is not just focused on the object or service given but on this 'practice of belonging' (Brownlie and Hewer, 2011). Or as Crewe and Gregson highlight with their car boot sale sites: "This is a space for quasi-fun, quasi-leisure, quasi-work activities; an arena in which conventional boundaries are blurred and/or transgressed" (Crewe and Gregson, 1998: 50). This thesis extends this argument further as through the creation of 'eclectic homeliness,' the marketspace exists in an ambiguous space, between the conventional retail market and the familiar domestic sphere for temporal escape from the everyday.

In practice, the seemingly haphazard ordering and categorisation of objects is carefully produced and practiced in action within the confines of the marketspace. In Judy's market, the strategic merchandising of the clothing is both ordered and at once seemingly jumbled. With rows of rails with neatly organised denim shorts, to the box of t-shirts, scarves, jumpers and ties that overflow at the bottom of stalls: Figure 9 - Little Birds Market 'Practices of disjumbling'



Figure 10 - Granny Would be Proud 'Practices of disjumbling'



Figure 11 - Judy's Vintage Market: 'Practices of disjumbling' at work in stall displays



For the collective, within the allocated space the stall table acts as a main focal point for the vintage practice: the interaction, accepted behaviours and vendor reactions operate around this space, as Jessica explains:

"I expect someone to acknowledge that I have entered their space, it almost feels like quite a private space, whether that's with a nod or a smile or starting to make chat, but it almost feels like you need their permission to touch things and through saying hello it feels like you get the go-ahead." (Jessica)

This excerpt portrays that the created space around the table is a private area, almost as a homely sphere that has established practices that must be acknowledged before progression can be made. A dialogue of understanding is enacted around the stall space. It emerges from the observations that a relationship and common ground is established that assesses levels of commonality and aesthetic understanding between practitioners - both stallholder and consumer. This dialogue provides a means for stallholders to describe their objects, offer a discussion of the object's history and previous lifecycle and ultimately navigate a use value and potential value in the mind of the consumer. For the stallholder it is a way to show their care for the object, the rarity and exclusivity of the object. These market routines, practices and conventions produce a creative environment in which values of independence and Othering from conventional market logic is put forward:

"We are quite laid back about it, people want to tailor the space to their own needs and to highlight their own stuff that they are selling, it isn't really something we dictate. I would say we are pretty flexible and are really here to oversee and help if anything comes up." (Elena)

The material conditions and aesthetics along with the presentation of the objects, establishes an atmosphere of something organic and without strict boundaries that is contrasted with market norms. Elena's quote encapsulates the idea that the physical environment of the market is unsystematic and is a product of its temporal nature and the market site location reinforces the knowledge in practice needed by the practitioner to negotiate the disordered and informal retail setting. This section has explored an understanding of the pop-up market format as a space of *'eclectic homeliness*, *'* that exists in a liminal moment between commercial market logic and the familiar domestic sphere. Practices of organising and display create this temporal distance from the banal and every day and transports practitioners to a liminal space of play and performance.

5.3.3 The Space: The Marketspace Performance

Building on this notion of performance, the theatricality of the marketspace is central to the social connections observed. The construct of performance is unpacked in this section in regards to the consumption exchange in the marketspace. Through the active construction of the market and the notion of *"knowing audiences"* (Gregson

and Crewe, 2002) suggests that the marketspace is enacted and highlights the idea of a performativity for a marketspace audience. One of the most prominent characters of the Granny Would be Proud market is Michael. He is an optician who trades at the markets under the name 'Eyedresser'. He has several traditional optician practices throughout central Scotland. Now in his mid-forties he has always collected vintage frames and started actively selling them three years ago at the Granny Would be Proud market. He explains that he heard about the market by chance though a part time member of staff who works in his practice and he went along to see what it was all about (he frequently questions the 'serendipity' aspect of this encounter). He now sells his frames at vintage pop-up markets around Glasgow. Since setting up the vintage side of his business he has shut one of his practices so that he has more time to focus on this aspect of the business. He has been selling at the market since it started and is now involved with the market organisers in discussions of the market's future direction. He is one of the only sellers who is offered a table to sell as frequently as he wants (all other sellers are told that they will not be able to take a table at two consecutive fairs to ensure that the stock is different on every occasion). In talking to other stallholders and consumers his name is mentioned frequently as 'The Eyedresser' and it becomes evident that he has established himself as a prominent figure in the market:



In playing the role of educator, Michael creates an air of authority around the objects as he guides practitioners with their purchases and their own stock:

"...with vintage it is almost like a black art, you know they do like to ask my advice and leave it up to me a lot of the time... it is quite complex and I need to get to know them a bit. I mean what they associate with, what history they like." (Michael)

In this role Michael guides the novice consumer through the sharing of taste and experience (Fournier and Lee, 2009). Michael draws on the understanding of value as fluid in the market, this transient enactment by practitioner competence is revealed when he spoke of the action of *"going back and selecting the best of what was available at the time."* (Michael). This highlights the idea again of the *'the thrill of the find'* (Bardhi, 2003) but also of the expertise needed in re-introducing an object to the market. Practitioners can be seen to mobilise their *'field dependent cultural capital'* (Arsel and Thompson, 2011) in the marketspace depending on the situation to realise value. In considering such practices, knowledge and the ability to spot

potential in an object appears pivotal within the collective. Michael's role of educator is reminiscent of Goffman's (1959:19) suggestion that everyone is consciously playing a role and the marketspace provides not only a 'showcase' but creates an opportunity to present the consumer (and seller) with a space for play and experimentation (Holt, 1995).

This role of play in the marketspace was prominent in the conversations, similarly Jackie comments:

"...It is like an Aladdin's cave, you get to try new things with vintage or new sides of your personality, you can take it to extremes or stay in your comfort zone, but it is exciting and I just get a buzz from it all." (Jackie)

Jackie explains that her mother had been into fashion in the sixties and that she has always had an eye for details and quality. As Jackie expresses with an almost childlike reverence of 'the dressing up box,' the objects of the stall are used strategically to produce forms of invitation to explore and stage a personality beyond the confines of what may be understood as mainstream selling. In the vintage market play and the unveiling of difference is bound by the financial rewards to be gained by the seller for creating an immersive space. In the vintage market, this communicative staging and performance of the 'look' of the table and one's' self is rehearsed and refined by the seller in the course of their career in the marketspace through the practice of organising their stall and 'prepping' for the event. Without a suitable audience, its very existence can be questioned. Magaudda (2011: 19) argues that consumption activities are: "... The result of individual performances that are imbricated and intertwined in a complex socio-material context where meanings, objects and embodied activities are arranged in specific configurations of 'practices.'"

In the marketspace of vintage, for this performance to occur, this study finds that the interconnection of bodily activity, mental competency, understanding, and motivation has to interlink within the marketspace context, with the objects and their use, for the practice to occur.

5.3.4 The Space: Performing Politics

In the context of the pop-up market, vintage is positioned as part of a creative scene and this is evidenced in the variety of ways that participants talk about the practice of dressing and obtaining the 'correct look' for both themselves and for their stall. Market performance is key for success; the market therefore becomes a stage for a performance of value through *'practices of belonging'* (Brownlie and Hewer, 2011); or as Molly echoes:

"... The market is an opportunity to dress up and show off some of the pieces I have found and how I put them together, I always feel so good if someone compliments me on a find and asks about where I got it. It feels like I belong and that someone appreciates my skills."(Molly)

For Molly, the market becomes an opportunity for positioning oneself as a skilled and competent practitioner. By achieving 'the look' and through calculations over objects to bring and discuss, the event becomes a showcase for oneself. This transformation is achieved through the management and organisation of objects for producing a feeling of acceptance and belonging. Here Molly ascribes to the notion that without collective acknowledgement that her pieces are suitably 'vintage' as an aesthetic appreciation, they would be valued less. This builds on the work of Thompson and Haytko (1997: 26) who propose that fashion meanings can be used to gain a sense of standing out or they can be used to negotiate a sense of affiliation with others and to offer a sense of belonging. Acceptance by the collective means that these sharing practices benefit individuals and also create consumption opportunities: stallholders keep stock to the side for consumers that they know or mention other items that they have that may interest returning customers. Members of the collective share tales of new finds with stallholders, and sellers describe new items that they have in stock to sell at upcoming markets. There is a constant discussion at the stalls around topics such as how involved they were in the vintage tribe, where did they normally shop, who attends other markets, what they thought of the other markets and what would be sold at the next market. This building of a relationship between stallholders and consumers and the acceptance into the vintage collective is marked also by the time spent over the interactions around the exchange. As a marketspace there is an embodied approach to vintage unlike with a brand community (Muniz and O'Guinn, 2001) it is a personalised experience of bringing forth individual values to the collective more fitting with consumer tribes (Cova et al., 2007). This will be explored further in the next findings section on sociality.

The approach to vintage as a knowledge form, allows participants to wear their capital (Bourdieu, 1984, 1985; Entwistle, 2009) in the marketspace. In this way the market emerges as a space for connective interplay and *'practices of belonging'* (Brownlie and Hewer, 2011): acts of performance and participation are constitutive of the collective. The previous sections highlighted that the learning of 'doing

vintage' is a social endeavour and is bound by the time commitment given to this learning. In this way vintage could be considered a form of *'learned nostalgia'* (Goulding, 2003). This section reveals that vintage must also be understood as a form of embodied nostalgia, a preference for a time that has long since passed but that is brought back to the modern consumption sphere through the revival of an object:

"It isn't really about enacting the past; it is more ironic than that. Like you have the choice to buy new, but it is more interesting and the aesthetic is more layered than that...most of the people that I know that buy vintage, are coming from an educated background, have jobs and could afford to buy new products but they see vintage as a way to show their intellect and that they aren't mainstream. You are taking a bit of a stand against fast fashion and showing a preference for quality and long forgotten about skills. It is kind of like showing off that knowledge and style that you have accumulated." (Jessica)

As Jessica articulates, rather than a reliving of the past, vintage objects and the vintage aesthetic are employed to gain social status and differentiate oneself from the mainstream. It can be seen as a form of consumer politics in action (Chatzidakis et al., 2004) against the fast fashion habits of contemporary society but moreover vintage is a way for individuals to display their knowledge and possible social standing through their consumption practices. In this way it is a conscious practice to align themselves with a particular group or set of beliefs as a consumer. The turn to the pop-up marketspaces allows for a transient space to be created by practitioners that embodies their ideals and social norms. Vintage is this way is utilised as a social power in the marketspace, further it is a self-conscious aesthetic and brings to mind Arsel and Thompson's (2011) *'Taste Regimes and Market Mediated Practice'* in

which they discuss that a hipster is ideologically and culturally shaped and has appropriated the indie consumption field. The hipster discourse is inextricably linked with the vintage aesthetic, as Schiermer (2013:12) asserts "...*hipsters exert a form of cultural hygiene, either ironically burning the objects of the recent past which deserve it or redeeming authentic cultural expressions from oblivion.*" The word 'hipster' entered the narrative of the vintage marketspace later in fieldwork and prior to this has not been a term that was used in discussions frequently. The respondents in the vintage space did not use the term hipster often and this was not how they identified themselves, rather this was seen as a media caricature or extreme. Instead respondents spoke of finding value in the collective and the creativity of the pop-up space, vintage for them is a practice that is rooted in seeking the alternative, of doing something interesting, by practitioners who could spot opportunity and potential. Within this context, value is found in the knowledge needed to participate and in the cultivation of a vintage style accrued over time.

As Jessica continues:

"I suppose when you think about it seems silly, but all my creative friends are a bit hip and we joke about it but it is still the main style even though they are self-aware. It is more like you can't buy that style in Topshop or just off the hanger - it is that effortless ability to throw pieces together and mix old and new and still look cool, of knowing what vintages pieces will work for you and what will look too 'try hard'. I do think about looking like a little old granny, which in my 'rain mate'⁷ I might, but it shows my personality and I always have a little thought somewhere about where

⁷ A 'rainmate' is a plastic bonnet or hood that is traditionally worn by females on the head to protect their hair - it is normally worn by older ladies to protect their perms.

the piece would have come from before me, how many tea parties or bridge games it saw."

Vintage in this way allows for an unlocking of an imagined past in every wear of an item. Jenss (2004: 396) discusses that by wearing vintage *"the originality and the uniqueness of the garment literally rubs off on the wearer."* As Michael similarly asserts:

"The very essence of it, of vintage is wearing something that your granny would have, might have been proud of you to wear..."

The marketspaces of Granny Would be Proud, Little Birds and Judy's facilitates a community based on these shared ideals and aesthetics and creates a consumption space in which the community's preference for the past have been cultivated through practice and associated forms of understanding. As Gregson and Crewe (2002) conclude on their work on second-hand, consumption is accompanied by an unsentimental nostalgia in which second-hand re-appropriates and reinvents consumer objects, rather than imbuing them with a longing for the past. This mirrors notions in McCracken's (1988) early work of the possibility of objects, of the *"combination and recombination take place until a concept and an aesthetic emerge that help give substance to a group's wish to differentiate itself from the mainstream"* (1988: 136). As a knowledge form vintage is embodied: vintage is worn and performed on the body (Entwistle, 2009). In this way vintage allows the consumer to play with the stylistic norms and even gender boundaries through fashion. The marketspace provides the backdrop to these performances of what will be termed *'cultivated assemblage'* (Dant, 2004: 2005). Similar to Dant's (2004: 74)

'driver-car', vintage is not a thing or a person, it is a spatially assembled social being that adopts properties of both and conversely cannot exist without both. The actions of the practitioner are key to the very existence of vintage. The marketspace functions as a safe space to trial new aesthetics, pull on strands of current interests and enact past eras of interest in practice. Vintage in this way can be viewed as less costume wearing and more material assembling of objects of interest. In this manner, vintage can be seen as the ultimate form of bricolage or a reinterpretation of fashion that reconfigures used clothes, mixed with new. Vintage is practiced as a sensibility: a valuing of the old and discarded, such applications of knowledge and cultural investment act to differentiate the collective. As Deighton (1992: 370) argues the idea of performance is central to creation as "many consumer transactions involve not possessions but performances," and as Kozinets et al. (2004:671) further suggests "the performance turns the consumer into a producer." Vintage here should not be understood as the end result but as a continual and ongoing process of improvisation in which competences are developed through play and performance in the market setting.

5.3.5 The Space: Inherent Instability of Vintage

The pop-up markets are transitory by their very nature and their lifespan is grounded by the movement of time, this temporality is one of the central markers of the market format (there is a literal popping-up, before they are dismantled and disappear again). This led to a consideration of the notions of tempo and rhythm which brought forward this sense of being-in-the world of vintage, where practices must be understood as unfolding within particular parcels of temporal invigoration (Duffy and Hewer, 2013). Such an assertion on the time-like qualities of everyday practices brought the researcher closer to the notion of the vintage scene as an ensemble of practices orchestrated and oriented around notions of time:

"The market almost acts like this stand-alone little enclave of your normal shopping routines and rituals I suppose, but they are heightened by the market surroundings. It is hard to explain the dullness and boredom of some periods of the day, building to the frenzy that happens when people all crowd round your stall." (Annie)

As Annie conceptualises, to theorise time as significant to the constitution of practices forces a consideration of the value of particular consumption spaces. Using the practice lens and by *'following the actors'* (Latour, 1987) looking at how vintage is organised within the marketspace, vintage can be seen to be defined in terms of being an unstable and constantly changing form:

"I will look at something and I have to decide if it looks vintage, or is it just old or is it not old enough? That is something that I run over in my head every time that I pick something up. I mean I will inspect the item for wear and cracks, smell it, feel the fabrics, but at the same time I will be running over in my head if I think it is on trend or does it have something about it that makes it a stand out piece." (Jackie)

For Jackie, she explains this practice of categorising (Slater, 2002a, 2014) an item as vintage: it is both a physical practice of examining the object and categorisation of things and the tactile process of feeling out the object but also applying contemporary marketspace knowledge to it and whether it will be a desired object – whether *'it is just old or not old enough'*. In this way market logic is applied to contextualise the object. Jackie continues:

"...It also depends on the time of day, I rarely take chances on an item but if it is near the end of the day and I can get a good price, I will quite often take it. I trust my judgement and there will be something about it that stood out to me."

The time of day in the marketspace is frequently mentioned in the research as being a factor in the strategies employed:

"...You can tell the people who know how to play the game, they maybe come along and see what you have at the start of the day and they will ask you about prices and inspect the items, then they might reappear later and hope that as the day has gone on your prices have gone down. I mean sometimes I am just happy to shift items and would rather sell them and cover the cost of my table, than having to pack them all up again." (Lindsay)

As Lindsay and Jackie's comment depicts the passing of time on market day and that price is fluid and tied to the rhythm of the day. Expressing practical concerns such as covering her overheads is considered along with physical considerations such as having to repack unsold stock and transport it home again. In this way knowledge of the marketspace and the flow of the day, is seen to add to the strategy when looking for objects. Vintage is seen as temporal and ephemeral, it exists in pockets of time, to both be seized and capitalised on by the practitioner, or for the potential opportunity of discovery to be missed. The marketpsace acts as an arena for the practice of vintaging to occur.

The field of the marketspaces allows for the opportunity to observe the spatially dispersed practitioners circulating ideas, trends and images. In gaining a further understanding into the space and materialisation of vintage, turning to the literature

of Warde (2005) and Appadurai (1986), the link between spaces, objects and practices becomes critical in exploring the movement of objects. As shown in the first findings section, to be a competent practitioner requires a thoroughly reflexive engagement with the objects: their current use and potential usage even if they have fallen out of use. Ben comments:

"...Some stalls do such a good job of pulling it all together, it makes the objects come alive from this overlooked thing and you start to imagine how it would work on your mantelpiece or in your wardrobe."

Appadurai's (1986) 'commodity journey' contains context as crucial in highlighting the value of goods, however this focus on the social framing of commoditisation is lacking in empirical studies (Gregson et al., 2007; Gregson and Crewe, 2002; Parsons, 2007; Denegri-Knott and Molesworth, 2009). This lack of focus on context and commoditisation/disinvestment is especially pertinent in the discussion of vintage goods through which different ways of framing and negotiating value are present. The findings have shown that the process of qualification of the material object is continual and in some cases circular, as objects are constantly examined in the market, by both seller and consumer, constantly open to interpretation and transformation as they move from one space to the next: *peripheral existence, to stock, to marketplace, to new owner and possibly re-entering of the marketplace.* Practices are thus contextually embedded, value or meaning is not "*created*" by the consumer but formed in practical arrangements, spaces in which actions are carried out and resource elements are integrated skills and images related to practices (Schatzki, 2001).

In further considering the practical arrangement of the marketspace, De Certeau (1995) considers space as *"an intersection of moving bodies"* and of it being a practiced place. Carrie alludes to this 'intersection' and the power of the moving bodies:

"I like all the hustle and bustle I suppose, you navigate it, by the middle of the afternoon it is busy and you are aware of the music and people talking loudly. You are trying to look at the stalls and have a 'game plan' of where to go... I always bump into people I know and you are trying to chat but people squeeze past you and things catch your eye and it's like you get shifted along, you can't really stop for long." (Carrie)

Carrie highlights the physical and mental navigation of the marketspace, there is almost a flow of the market that takes over and practitioners are unable to stop, as the collective moves them forward in the space. Hereby within the vintage context, an understanding of everyday practices is enriched through understanding consumers as moving entities in time and space, through which questions notions of fixed structures are questioned, and a more nuanced appreciation of agency begins to emerge, perhaps best thought of in terms of the ability to *"negotiate movements within those structures"* (McDonald, 2004: 200). As this understanding illuminates it necessitates a turn to movement to make explicit the routines and conventions, which are revealed by taking on board this aspect of practices. Through this constructed space in the market, an understanding of vintage practices as a *"structured movement system"* (Kaeppler, 1985:116) begins to emerge; what Williams would prefer to visualise as a *'structure of feeling'*. This dynamic 'feel' in the marketspace, the relation between vendor and consumer, the subtle dance around the table that was

performed, was seen as key in the acceptance into the collective, the building of relationships and ultimately an exchange. Referring back to Michael's earlier discussion of guiding the consumer through the exchange, it could be considered in these fluid and dynamic terms, of learning the steps to engage with the collective practices of vintage. This focus on movement and fluidity further calls attention to Nash's (2000: 654) assertion: "dance is taught, scripted, performed and watched". Building on the observations of people approaching, navigating and vacating the stall space, the consumer acts as willing participant to be led through the process by a knowledgeable tutor, seeking acceptance to continue onto the next step of being able to interact with both the objects and the possible stage of exchange. Rules of etiquette in the market are brought to the foreground in the stall space and are reminiscent of the distinctions drawn by Hanna et al. (1979) between competence (internalised rules for dancing) and performance (what someone does on the basis of knowing such rules) and creator (choreographer) and imitator (Hanna et al., 1979: 315). As such when consumption is understood as part of social practices the movement of the objects and how this movement shapes the moments of consumption, the perception of space becomes critical. In viewing this vintage marketspace as actively constructed rather than passively absorbed the practices and performances of the individual become of paramount importance.

5.3.6 Conclusion of 'Space'

In this findings section practices have been sought to be revealed as an invaluable perspective for capturing the sense that actions unfold in space and time; and through such a practice turn, a sensitivity to the vintage marketspace, as Lash and Lury suggests, as a "*space of intensities*" (2007: 15) could be captured. Referring back to

Warde's (2005) assertion that consumption is a moment within social practices, these findings show that consumption occurs during performances that are related to objects and are spatially bound. Practices are always carried out in context, as Gregson et al. (2007: 198) comment "practices are always spatialized and spatializing." The afternoon of the market was revealed as a continual shift between time, space and actors, with a changing rhythm and tempo that marks the passage of the day. This time-space dynamic nods to Giddens (1979: 3) in that the dichotomy is inscribed in all social interactions. In the suggestion that vintage practices bring forth movement, reference can also be drawn on what Bourdieu in *Outline to a Theory of Practice* (2010, orig. 1977: 7) refers to as the 'dialectics of strategies:' whereby the urgencies of time must be included as a central constituent of everyday practices. As this discussion has explored, vintage knowledge is not only temporally organised it is spatially located. Such a sense of time takes a number of forms but makes explicit the ways that practices are structured temporally.

Returning to the suggestion that markets should be studied as sites of multiple and often conflicting arrays of practices (Araujo et al., 2008: 6) or *'in the making,'* this study comes to understand the markets not primarily as sites of commerce but as sites of social connection and interaction that occur in these alternative spaces. This builds on Watson (2009) who highlights that markets play an important social function of social inclusion and care of others. She depicts that as a shared social space, markets provide informal care of others and act as sites of social association and inclusion. This thesis adds to this understanding as the findings reveal that the vintage collective makes use of under-utilised spaces (empty rooms and closed off areas in commercial spaces), away from the high street, to enact their own market

values and aesthetic preferences. The marketspace site is therefore under constant motion, reorganisation and rearrangement (Tuan, 1977). The collective create value in these urban spaces that were otherwise empty and unused (Watson, 2009; Goss 1993).

In understanding consumers as moving entities in time and space, consuming speaks of the importance of considerations of space (Hewer and Hamilton, 2010) and how these spaces were unsettled and transformed in practice. The importance of the site of consumption was underpinned by the notion that they are living spaces and as such consumers actively construct lived experiences through their use of consumer goods and services. As discussed in the methodology chapter, the researcher observed participants in the markets but also observed the context in which these practices occurred (Spradley, 1980). Using visual data allowed the researcher to position consumers and their consumption behaviour within their social setting which was especially important in the market context. The wider sites of consumption was still a relatively under researched topic of study in the literature – exceptions include Goths (Goulding et al., 2005), Burning Man Festival (Kozinets, 2002) and white water rafting (Arnould and Price, 1993). In addressing this second research objective and employing the practice perspective, this focus on space and context of consumption from this study adds to an understanding of how value was performed in practice through an exploration of the vintage marketspace. With a focus on the ephemeral space of the vintage pop-up market, that invoked the concept of 'the alternative,' these marginal sites of consumption are seen as active places that are an amalgamation of market objects, actors and influences that encode particular practices (Slater, 2002a; Geiger et al., 2012). The market became a space to enact

differentiation and knowledge and '*practices of belonging*' (Brownlie and Hewer, 2011). This thesis further highlights that Othering of the market takes place for it to embrace its positioning and value as '*alternative*.' In '*invoking the alternative*,' vintage sites became spaces of differentiated exchange and were juxtaposed as Other throughout the research to '*spaces of the new*.' It was a space of aesthetic enjoyment of objects of the past, not simply a nostalgia or temporal longing, but a collective appreciation of the possibilities and sensibilities of forgotten aesthetics (Arsel and Thompson, 2011; Schiermer, 2013).

In further articulating the importance of this turn to space, Schatzki (1996: 89) highlights that practices are 'spatially dispersed' and Giddens (1979:3) illuminates the 'time-space relations inherent in the constitution of all social interaction.' The construction of the market 'look' within the space is shown to add credibility to both seller and consumer in the collective hierarchy (unpacked further in the following section on Socialitities) and has implications for the construction of value as in this way value is created anew within the marketspace. Further the development of an interaction and bond functions as a key differentiator in separating the vintage marketspace from more traditional conceptions of the marketplace. This discussion has shown that the space itself plays a key role in the exchange: it is a space of familiarity, support and learning (Goulding et al., 2013; Linnet, 2011; McCracken, 1989; Watson, 2009). With different levels of commitment and feelings towards vintage, the market creates a space for experimentation, performance of *cultivated* assemblage' and critique. The pop-up format allows for a space in which the practitioners are made visible to each other and the performance sustains and enhances these collective ties, as will be further explored in the next section on

sociality. As shown with Goulding and Saren's (2007:27) study on Goths, communities of consumption are "sites of contestation where orthodoxy is challenged and identities are constructed and performed." This also links to the work of Schau et al. (2009) who depicts that rather than focusing explicitly on individual participants, a focus on the collective and the participatory actions is needed in studies of the collective. Following the construct of practices as implicit and interrelated ways of understanding, saying and doing things, practices are argued to be spatially and temporally bound behaviours and performances in action. In focusing on this performative enactment of vintage, this findings section has shown that it is a social entity grounded in the temporalities and spatialities of commodity journeys, with vintage and value creation as the outcome of this embodiment and consumer action (Gregson and Crewe, 2002; Parsons, 2006, 2010). The spectacle of the markets is central to the notions of collectivity and displays of style whether through practices of bricolage or 'eclectic homeliness'. In this way the practitioners are not mere consumers of the vintage market, but actively construct it through their marketspace practices. By focusing on the collective understanding and practices, the marketspace discourse between practitioners highlights that value is made a new in these collective interactions. This collective understanding will be explored in detail in the following section.

5.4 The Socialities

5.4.1 Introduction

The previous findings sections have shown that the marketspace of vintage is actively constructed through networks of objects, practitioners and social practices.

In addressing the research objective of how value is ascribed and framed in the marketspace by the collective, it is necessary to revisit the literature on various consumer collectives and how they are considered to be sites of value creation (Cova, 1997; Giesler, 2006; Kozinets, 1999; Muniz and O'Guinn, 2001; Pongsakornrungsilp and Schroeder, 2011). In considering the marketspace context of the research and the collective nature of the exchange, Schau et al.'s (2009) discussion is important and illuminates that through their discussion of brand communities, practices have clear typology of value creating effects. As discussed in the first findings section on practices, the development of practices in the marketspace, similarly to Schau et al.'s (2009) study leads to the exchange of collectively valued goods. Moving this discussion forward and seeking to explore the interaction of the practitioner and the community, the strategies employed to negotiate collective membership are further explored in this section in order to address the first research objective as to whether vintage creates a shared consumption space.

5.4.2 Socialities: The Tribal Nature of the Collective

Coming from an ethnographic perspective and in considering vintage as knowledge in practice in the collective, the objects, actors, tools and locations are considered in exploring the *'discourses of community'* and *'collective creativity'* along with the online social space of the collective. As explored in the previous sections, practice in this way is seen to be multifaceted involving the assemblage of different objects, processes, understanding and assumptions. As Wenger (1998: 47) articulates: "A concept of practice includes both the explicit and the tacit. It includes what is said and what is left unsaid; what is represented and what is assumed. It includes the language, the tools, the documents, the images, the symbols, well-defined roles, specified criteria, codified procedures, regulations, and contacts that various practices make explicit...it also includes all the implicit relations, tacit conventions, subtle cues, untold rules of thumb, recognisable intuitions, specific perceptions, welltuned sensitivities, embodied understandings, underlying assumptions, shared worldviews. Most of these may never be articulated, yet they are unmistakable signs of membership of communities of practice."

As shown in the literature in Chapter Two, the theoretical construct of community has a rich history within the field of marketing and consumer research. Community is defined as 'communitas' (Arnould and Price, 1993; Goulding et al., 2004), 'gemeinschaft' (Belk and Costa, 1998), brand communities (Muniz and O'Guinn, 2001; Muniz and Schau, 2005), subcultures (Goulding et al., 2004b; Schouten and McAlexander, 1995) brand and marketplace communities (McAlexander et al., 2002) tribal formations (Cova et al., 2007) and micro-cultures (Thompson and Troester, 2002). The community feeling is reiterated throughout the research as an important aspect of the current success of the vintage market model in the city. It becomes necessary to question how the collective is formed and maintained in the marketspace.

Within the tribal framing the collective does not need to be located around the vintage labelling (unlike with a brand community) necessarily but around the interactions and engagement. In discussing the market culture of vintage (Arnould and Thompson, 2005) within the collective each practice encourages deeper

commitment to the shared understanding of vintage as shown in the first findings section. The development of relationships and shared understandings suggests that the act of vintage is social by its very nature. Elena asserts:

"It is like a family almost, we see the same people every few weeks and we all chat on Facebook in between the markets. It is like finding an extended group of friends that all value the same things as you." (Elena)

As Elena remarks there is an almost familial relationship that takes place in the collective around the core members: they have shared interests, values and speak frequently – both in person and online (developed further later in the following section). From the observations, the building of these relationships and these sharing practices form a large portion of the interaction between consumer and stallholder and could grant status to both through this dialogue. Knowledge of the social actors creates a hierarchy of social interaction and adds to the familiarity that Elena describes.

Vintage consumption from the observations is both an individualistic and collective act that enlivens the marketspace. The tribal perspective of consumption as outlined in Chapter Two, proposes that people gather together in tribes and that these social groupings are more effective and influential on behaviour than marketing institutions or cultural authorities (Cova and Cova, 2002). Tribes are considered unstable, small-scale, affective and not fixed by any of the established parameters of modern society (Cova and Cova, 2002). The previous section on 'The Space' has shown that the tribe exists as ephemeral performances rendered through the commercial stage of the pop-up marketspaces. Cova and Cova (2002) suggest these are spaces that 'anchor'

the tribe as it allows them to celebrate their own existence through these tribal performances of the collective. Within the vintage scene the act of being together in a communal space and having shared aesthetics is seen as key in adding value to the vintage practice. In considering a tribe as a group who have a common interest and create a social sphere with their own: rituals, language, hierarchy and values, through which they enact their passion for an object or brand (Kozinets, 2007), in the vintage context this collective action could be seen to form a hybrid (Latour, 1991; Entwistle and Slater, 2014) between market and community:

"Yes the market is important, it provides this space where you can go along and know that you will meet others that value the same things as you. It that weird thing of whether the market exists because of the people or the people exist because of the market? I sometimes think about if I would have connected with so many people without that opportunity...it is like our secret club." (Adam)

Adam's comments illustrates that the market provides a location for the enactment of this market performance around vintage and the shared values of the collective or *'secret club'*. It starts to tease out the question of whether the practitioners make the market or whether the market makes the practitioners? Engagement in the collective is seen to be distinguished through practice. Within tribes the social links established between consumers - or consumer-to-consumer linking value – are more important than the object being consumed (Cova et al., 2007). There is an important focus on re-appropriation in the acts of being, gathering and experiencing together as a group. The vintage collective is not a homogenous group of community participants but is diverse and could not be defined on basis of gender or age, what binds them together are their consumption practices. Cova (2003: 78) suggests *"belonging to a tribe is*

not the result of individual characteristics, but the result of a common experience of reality." This focus on 'sharing' highlights that ways of doing things, actions and language shape the collective. The vintage tribe expresses pride in differentiating themselves from what was considered the 'mundane mainstream' (Ben). Through their consumption practices they mark themselves as different in their consumption tastes. In this respect they consider themselves outside the mainstream and Othering could be seen to be at work. This is in line with the work on tribes whereby:

"Tribes need to feel unique as averse to mainstream; they need a story, myth or narrative around which they can unite" (Goulding et al., 2013: 822).

In considering tribal practices, the search for legitimacy and authority by practitioners is prevalent within the community. Jessica speaks of the exclusivity of the vintage scene and the etiquette that should be employed: that it should be *"work", "a little bit scary"* and *"not overly accessible to outsiders."* This language of exclusivity and drawing boundaries between insiders and outsiders in the collective conveys a hierarchy at work.

5.4.3 Socialities: Collective Hierarchy

When considering the collective practices, a competitive spirit underlines many of the market interactions. Members of the collective could be seen to compete on knowledge and experience in the marketspace in the searching, finding and uncovering process. This air of competition is present in all three of the markets and enables participants to distinguish themselves strategically through their 'finds' to gain advantage and in turn create a social hierarchy. In the Granny Would be Proud market, Claire spoke of Lindsay having *"amazing pieces, I don't know where she* gets them but it is always really unusual, unique finds that she sells." One way in which the social hierarchy of the collective is created is through the stock that sellers have to sell and the reputation that they have built up as to the quality of their stock. Although all the practitioners on the whole are friendly and welcoming to each other, this hierarchy is marked out by the knowledge that sellers have regarding the best places to find stock and starts to differentiate members:

"I get asked everyday [about stock] and I never tell, it is what sets me apart and then people would all flock, you'd have to kill me before I'd tell you." (Jackie)

This social hierarchy is further displayed in the way that participants spoke of the length of time that they have visited and consumed at markets. As core members also set themselves apart within the collectives. Both Alice and Michael assert their place in the collective's hierarchy by establishing a history of their market trading, Alice spoke of the long closed Paddy's market in Glasgow where she had first had a stall in the 1980s and Michael spoke of travelling to Edinburgh during his student years to look for vintage men's clothing. Michael's status within the collective is a source of pride for him:

"Put it this way, I am now someone who has kicked around in the west-end most of his life, the first time was when I was a student at Glasgow University when I was in my early twenties or late teens, and then again now. In the middle period, I went away from Glasgow and didn't have much to do with it. Back then in my 'uni' days, I was just anonymous, just invisible, no one gave a damn what I was doing, who I was or whatever, now all these years later, I come back and I've only been back for two years and the only two times I have went up and down Byres Road shopping over the summer there, I've literally been stopped two or three times by people that know me in the space of ten minutes. It is quite nice you know, it is all positive stuff you know it is all good stuff, I get really good vibes behind the whole thing so that's nice."(Michael)

Michael's discussion highlights how capital is embedded in the interactions with the collective and the satisfaction that he garners from being a known part of the vintage scene, it also has given him fulfilment with his practices as a seller. Although his expression that it is a 'community thing' is central in what he considers the essence of the vintage moment, Michael expresses his surprise at being a known presence within the scene and how this impacts his daily life. In considering his tribal capital, Michael asserts his history as a long time vintage connoisseur and that his practices have shifted alongside the market formation.

Within this fluid hierarchy of the tribe, in the social field acceptance and solidarity is a theme throughout the marketspaces. At Little Birds the wider selling community treats the original sellers (that had been there from the start of the market two years ago) with respect and reverence. If questions are raised about the market, they defer to their expertise. Louise the market organiser comments:

"There is a core group of people that have been selling and coming to the markets from the start and obviously I encourage lots of new people to come along but we have like a little gang that have seen the market change and grow and we can't help but be a little bit of an inner circle...it is not like we have rules as such but more ways of doing things." (Louise)
Louise's extract begins to unpack the social embeddedness of the marketspace and the accepted 'doing' of vintage. In her early thirties, Louise left her full time job as a PR Director to set up her own vintage market in the city centre. She has been selling vintage homeware at a variety of markets in the central belt of Scotland and decided to set up her own market. She explains that she felt the markets lacked a good marketing and PR strategy and that is something that she felt that she could offer that was different. She takes great pride in the professionalism of the markets and that all sellers are vetted before they could sell and that a good mixture of different goods offered. She chose to open the market in the city centre in the historic bar and restaurant 'Sloans' - the market takes over their Grand Ballrooms and Victorian Style tearooms twice a month. Louise explains that she thinks the location is important as it offers an alternative to the busy high street that is just minutes away on the main Glasgow shopping street, Buchanan Street. Louise uses Facebook, Twitter and e-newsletters to promote the market and she also uses her PR contacts to gain exposure for the markets. Louise describes buying and selling in the markets in personal terms rather than grounding it in economic exchange. She expresses her love of the objects she sells, but she privileges discussions of the people she meets at the markets in our conversations:

"I usually come to the markets with my mum or niece and in thinking about it, the market is a proper family affair and not just your actual family but the people you meet who sell and you see every few weeks and you learn from, to the customers that tell you all about their lives and big stories about stuff. I suppose I have always been a people person." (Louise)

For her the market is underpinned by this personal and emotional discourse with knowledge of the market grounded in practice and experience. Within the collective this notion as Louise asserts of differing types of membership is evident throughout discussions. There are understood differences between the practitioners and a hierarchy emerges that is reliant on their level of sociality in the collective. The levels of membership are dependent on time spent in the marketspace, connections within the collective and perceived expertise of vintage. This connects to Parsons' (2006) work that suggests that the degree of social embeddedness of dealers within the antique context and their motivation for trading linked to the morality of the market. Acceptance by the vintage collective means that sharing practices benefits the individual and also creates consumption opportunities, further ways of understanding and doings of vintage are shared and appropriated as knowledge within the marketspace. In accordance with Thompson and Haytko's (1997) work, vintage allows practitioners to use fashion to self-define, to construct a personal discourse of their history and to negotiate the social. As explored in the sections on practices, practices could be seen to bestow participants with the cultural capital necessary to engage in the marketspace and individual participation is seen as being differentiated through performance (Bourdieu, 1984; Holt, 1995).

This division between practitioners of firstly long-term enthusiasts and secondly marketspace visitors was also constantly marked out by the interviewees. The vintage practitioners were typically constituted as having *'instinct'* or an appreciation and long-term interest in the aesthetic style as unpacked earlier in the chapter around the discussions of practice. In contrast visitors and spectators in the market were constructed as simply being interested in the trend, or of not caring enough and of an

indifference towards the purity of the marketspace. Practitioners spoke of 'the hobbyists' who had started selling at the markets and they were referred to as ignorant in their knowledge and in their lack of engagement with the objects and wider collective scene. Relationships and social history separate out spheres of influence in the collective - from accepted member to peripheral practitioner or 'hobbyist' or as Judy mentions people who are 'playing at it':

"We get lots of applications to sell at our markets, but we tend to focus on the people that know what they are doing. We find that it changes the tone of the market if we have sellers that are really just playing at it or aren't very experienced. We pride ourselves and the customers expect it, that our markets look professional and are well set up...there are only so many 'cupcake' type stalls we need to see." (Judy)

In line with the work on tribes (Cova and Cova, 2002; Cova et al., 2007; Goulding et al., 2013) the core members of the vintage tribe challenge the practices and norms of those that do not confirm to their own standards. These standards are negotiated by market interaction, socialisation and learning. With this marketspace hierarchy at work, passionate statements about the ideals of vintage being corrupted abounded in the layers of the hierarchy. The markets are constructed, as complex and learned spaces that centred on discourses of taste, style and aesthetics and that could not be easily navigated by *'the hobbyists'*. Annie explains that she is relatively new to the tribe as she describes learning the rites and codes of practice in the market:

"I came to vintage quite late, I have always been style conscious but once I retired I was looking for something more creative to do. I travel to France a good few times a year and had always picked up jewellery and 'bits and bobs', I started to see from coming along to the markets, that maybe I could do something and resell some of my pieces and buy more on my trips. It has been a steep learning curve, of what will sell and what makes your stall stand out, even how best to pack and set up. People have been welcoming, though I am a slightly different age bracket than some and it can be hard to stay up to date with the online stuff, Etsy and Facebook, but I am not doing it to make my living or am reliant on it, but because I enjoy it." (Annie)

Annie depicts that her initial stages of membership with the tribe have involved a socialisation process that is grounded in extensive learning about market practices. As a practitioner learning to fully align (Goulding et al., 2013) with the tribe is crucial and the newcomers do not always possess the subcultural capital (Thornton, 1995) to enable them to adhere to the tribal codes. The more experienced members administer the rites and codes of the market and oversee the learning of new community members. Annie demonstrates that full participation in the tribe is reliant on the acquisition of skills and the norms of the group. This relates to Warde (2005:148) who posits "people come to an understanding of what is required by the practice and their role within it." Vintage in this way can be seen as a complex social practice that is learned in context, which further establishes that practitioner competence is not fixed; it evolves and shifts in the market. As shown in the previous section around the spatial considerations, the marketspace creates an environment for practitioners to interact with others of a similar taste and mind-set and to accrue this 'learning to be tribal' (Goulding et al., 2013). Practices evolve and extend over time and members' legitimacy increases.

The market is a communal setting, which is enlivened by the social bonds of the collective and allows tribal practitioners to develop and demonstrate their

commitment to the collective. When asked about the role of the markets Tina explains:

Researcher: "Can you tell me about what you think about the markets?

Tina: "What I think about is selling that object, discussing with the customer about the item, building a relationship with them and thinking about what new lease of life they will give it... I love those interactions...I have made so many friends in this line of work and I always thought that I was alone in how I valued vintage stuff over new things..."

Tina reveals in her language the 'doing' of vintage in the collective – *selling*, *discussing*, *building and thinking* – through this she conveys the sociality of the encounters and the overall feeling of not being alone. This social role of the market and the physical proximity of the marketspace enforce the relational nature of the exchange and in turn enliven a stronger feeling of community. Similarly for Crewe and Gregson (1998: 42) car boot sale consumption is a communal activity in which *"sociality, tactile interaction and tribal solidarity"* are all part of the distinction of the space from a more traditional shopping context. They argue that the informality of these ephemeral spaces provides a space for sociable consumption in which the *'character and texture'* of the meeting are crucial. Their study suggests that exchanges within marginal sites of consumption are embedded within a socio-cultural milieu. They assert that these exchanges provide the consumer with a greater degree of agency than more traditional contexts of exchange. This section has depicted within the tribe belonging has to be practiced into being and negotiated over time. The hierarchy is developed around these dynamic social practices, which the

marketspace provides a space for the performance of these socialities and collective gathering.

5.4.4 Socialities: Collective Creativity

Within the collective framing, the marketspace acts a hub for creative thinking and development of ideas around the collective understanding of *'vintage'* further it serves as a location for this creative sense making practice. As perceptions of the vintage practice change and develop, so did the collective's resulting marketspace practices. In the marketspace practitioners were seen to act as innovators and were entrepreneurial in their action: over the study the researcher observed the development of services and products to meet changing market demand. The collective was anchored in the marketspace understanding of vintage and as opportunities arose, practitioners developed to meet the changing parameters of the market. Cherry explains that she had been asked along to the markets because a potential opportunity had opened up which provided a marketspace opportunity:

"I was talking to the market organisers about how it was progressing and ideas for the future and we started chatting about how people buy vintage dresses even if they don't fit and that they always mean to get them altered but never do, as a seamstress they suggested that I should market my services at the market as a way to make that dress fit...it is strange because I wouldn't necessarily have thought about it, but like everything it has just kind of developed and my own business has flourished because of it." (Cherry)

From a discussion with the organisers, potential was seen in the marketspace and as a result Cherry developed a sewing service as part of her own market practice and due

to the success of this venture, had opened her own vintage custom dress shop in the cobbled streets near the Granny Would be Proud market. This creative discourse of the market was constant which highlighted the dynamic nature of the marketspace. Further marketspace practices create and offer sites of new opportunities for consumption. The collective acts as a space for nurturing of ideas and potential, with the caveat that the practitioner had to be accepted within the collective. Similarly to Goulding and Saren's (2007) *'Tribal entrepreneur'* through whom cultures of consumption develop to meet the needs of the created markets, the vintage marketspace could be seen to nurture proactive entrepreneurs as well as passive consumers. Building on being a place of *'play'* (Holt, 1995) the markets offer tribal members the opportunity to customize market offerings (Goulding and Saren, 2007) rather than a dependence on available consumption resources. Further as trends develop within the collective the market responds and alters:

"People think they know what vintage is and what the vintage markets are, we are continually trying to expand these ideas and be innovative with our thinking. We have focused on different areas, such as homeware as that has become more fashionable and I am continually working with my team to develop new areas for us to expand into. It feels like for the markets to stay alive you have to stay ahead of the trends." (Judy)

Judy explains the vision of her markets for them to stay relevant with the collective, she employs traditional market logics of strategic planning, forecasting and differentiation to enable the markets to grow and develop. Although in many ways a grassroots approach and more traditional, she depicts that as the market organiser she is clearly orientated towards ensuring that the market will continue to develop and remain current for the tribe. As with other trend orientated, creative industries, over the two years of the study the researcher witnessed original innovators being pushed out of the collective and moving onto new sites to explore interests. As such Sarah had organised a market in the city centre for two years however she had become disillusioned by the mass marketisation of vintage:

"Our markets were a success and maybe that is part of the problem, when I started seeing vintage in Topshop and in every magazine, it started to lose its appeal, it was no longer this special thing that my friends and I were interested in, it became everywhere. The markets became less about this select group of people with shared interests and more people trying to find the latest thing from Topshop at a cut price. Now I am more focused on independent designers and under the radar events. Once it gets too mass, you know it is time to move on." (Sarah)

Sarah's phrasing of a 'select group' of people illuminates the collective hierarchy at work and that by its own success and increased visibility, vintage became too ordinary for Sarah and the appeal started to diminish. Her interest has continued to develop around a further seeking out of 'the alternative' and 'under the radar' and she continues to position her consumption practices as different from 'the mass.' For Sarah the appeal of the alternative is in the ability to differentiate and develop her interest in line with a group of similarly minded consumers. Sarah's comments further illustrate that as a consumption field, vintage does not exist in isolation. It is embedded within the wider socio-cultural structures and does not exist as an autonomous field. In this manner, participants often questioned such change and enacted a feeling of protection around vintage as a form of stylization; as Sarah continues to explain: "Vintage to us used to mean something exclusive and special - an aged one off that no one would have the same. But I think that in the last few years the term 'vintage' has been used and abused. When you have shops like New Look using the word, it kind of ruins it all a bit. It has become a sort of cliché. I find that really off putting all round for something that started out so special." (Sarah)

She highlights the paradox of the market, that it needs a collective effort to be sustained in practice, but that once this interest reaches such a large group, the original premise of the interest can be lost or damaged and sections of the collective move on. Sarah's comments reveal her attempt to disengage from the tribe as for her the appeal of vintage is in its uniqueness and the capital that she garnered from being involved in something outside mainstream. This depicts that unlike with a brand community and Muniz and O'Guinn's (2001) *"moral responsibility,"* Sarah's attitude to the collective is based on her own needs and she is not concerned about exiting and moving on once her needs are no longer met. In this way the collective is shown to be a social hub but is fluid in construct and is tribal in nature, members disperse and reform as something else once the wave of shared emotions end (Cova et al., 2007). When viewing vintage through this tribal (Cova et al., 2007) understanding, it is seen as active and continuous and participation is experiential, tied to individual goals and furnishes participants with the knowledge and doings needed to practice.

Sarah decided to move on from the vintage collective in her search for the alternative however she was not alone in conveying dismay and frustration at how vintage has been co-opted by high street retailers. Further many respondents voiced a mistrust of

249

the apparent appropriation of the term and a sense of anger felt that the essence (and purity) of what vintage constituted was being lost and endangered:

"...If you have a space that you've nurtured and created and then if all these people don't really grasp what this space was created for and try and adopt it – you get kind of protective. With the high street finding vintage, it just makes me more puritanical about it...it defeats the whole point of being vintage – you now have this vintage dress code instead of just wearing it cause you like it." (Cherry)

Cherry illuminates a '*finding process*' of the high street in discovering the market value of vintage. She further highlights the resulting 'codes' at work in the market and a less organic feel to the scene. As highlighted by Cherry's discussion, this resulting tension and feeling of protection around the concept and created space was powerful. Cherry felt a sense of duty that the concept of vintage was being undermined and that a lack of appreciation for the aesthetic labour (Entwistle, 2009) involved in the practice was evident. This also illustrates that the marketplace myth (Arsel and Thompson, 2011) of vintage, in that it is propagated around vintage standing in contrast to the mainstream and as a counter culture consumption field (Bourdieu, 1984) associated with creativity and being different for the collective.

5.4.5 Socialities: Sociality of Exchange

The marketspace is seen as a dynamic and active field, it serves as a meeting place, a shopping venue and a temporary home for the vintage collective. In the research context vintage is shown to be a social practice that constructs a communal marketspace setting, which is enlivened by the social bonds of the collective. Over the past twenty years, consumer researchers have explored a variety of consumption-

orientated collectives ranging from extreme sports groups (Celsi et al., 1993), surfers (Canniford and Shankar, 2013), temporary anti-brand festivals (Kozinets, 2002), inline skating tribes (Cova and Cova, 2002), brand cults (Belk and Tumbat, 2005), brand communities (Muniz and O'Guinn, 2001) and consumption subcultures (Schouten and McAlexander, 1995). From this critical understanding a further question emerges as to how the tribe shapes practices of vintage? This collective social space of the market creates the backdrop to the exchange and frames practitioner encounters allowing them to interact with others of a similar taste and mind-set. The market context and the variety of materialities present further support this sense of *'belonging'*. Vintage consumption from the observations is both an individualistic and collective act that enlivens the market space:

"These otherwise empty and unused spaces are brought to life by a group of people on a Sunday morning. Without traditional advertising and marketing techniques the markets operate on an 'in the know' basis almost. Knowledge of vintage and the local creative scene seem to be passed on face to face or followed up online." (Researcher field note, March 11th 2012)

In the Granny Would be Proud Market, the collective creates a core group of frequent practitioners and the marketspace functions as a social encounter every second Sunday. The development of practices in the marketspace, similarly to Schau et al.'s (2009) brand communities, leads to the exchange of collectively valued goods and knowledge. As well as providing friendship opportunities, the collective is based on economic exchange and loyalties are displayed to sellers and their objects. The researcher observes Adam, one of the Granny Would be Proud fair organiser in the weeks before Christmas. He explains that he is hoping to buy all of his Christmas gifts at the markets as it would be a good thing for all his friends who are stallholders, as he would always rather give them the business and also that his partner would know that he had bought all of her gifts from the market. In his discussion of this practice, the vintage objects became secondary to the links to the people he has purchased them from. The sellers all take great pride in showing him items that they thought Elena would like and allow him to pay up over a few weeks some of the larger items. In this example, the thought that he was purchasing within the collective and that the Christmas gifts would have more significance because of their association to the market is a strong presence. This loyalty and favouring of buying from friends in the market, is traditional and reminiscent of a retail model whereby the personal, social element to the buying process is as important as the object itself. Knowledge of the social actors, knowing their names, interests and backgrounds creates a familiarity in the market and blurs boundaries between the commercial marketplace and the family orientated private space. This could be seen as a 'nostalgic discourse of community' with characteristics of traditional marketplaces evident. In this way, it is argued that the marketplace provides a space that differs from the high street model, wherein the exchange is impersonal and purely about the end product, in the vintage marketspace the social exchange and collective interaction is prized.

Further the building of friendships is important to both stallholders and consumers. In all of the markets there is a tangible feeling from consumers that stallholders are the 'gatekeepers' to market knowledge and potential finds. Consumers spoke of a feeling that there is always more stock to be seen than is present on the stall and that the only way to access this potentially more interesting and more valuable stock is through the stallholder as a gatekeeper:

"You always think that if you can chat to them maybe they will be able to tell you about where to get what you are looking for or how rare an item it is. Mostly I get so much information during those quick chats; I usually try and pay it forward to my friends or people when I hear that they are looking for something in particular." (Lindsay)

By building relationships in the collective one of the most important exchanges is of knowledge of stock as Lindsay highlights. In this way the collective can be seen as strategic in manner, all practitioners are highly motivated to ensure success in the marketspace (what would be considered success varied from finding out about new supplier or clearance sale that is coming up, to being directed to a source of a hard to find object). The collective is further tied to being part of an economic strategy of practitioners:

"We tend to find out lots of things from other people like what markets to sell at, what to avoid, where there might be good stock coming from, it is like a little network here. You only start to find out about how big the scene is and everything that is going on once you are in it really."(Tina)

"...You see what other people manage to sell, how much they manage to get for it, it is we are like learning as we go along and people are normally pretty good at sharing things that don't work for them or markets to avoid, maybe they don't always share the things that do work for them– I suppose we are all trying to sell at the end of the day. But it has to be a close knit kind of thing, you have to leave your stall sometimes and you ask people beside you to watch it for you, you don't always have the right change and need to borrow some from a neighbour, it is this community thing that happens because we all have to trust each other a little bit in the market." (Louise)

Louise describes that the market functions on these community practices, that even practical aspects of selling in the market setting are reliant on sellers being able to trust their neighbours to look out for their best interests in the market or as Tina depicts with 'networks' at play. Louise highlights that through discussing with other sellers; she is able to find out about best practice in the market and is able to learn from their past experiences. Further these narratives and interactions facilitate and extend consumption behaviours. The value in this continual exchange of information on a face-to-face basis is characteristic of the marketspace. Exchange in the marketspace is based on these discourses of knowledge, style and collective acceptance. The vintage objects become tangible resources that are used to construct style, identity and signal group membership. In the exchange process, time is spent on establishing shared values and interests: stallholders asked the consumer questions about their style and influences and where they normally find vintage objects. From the observations it emerges that a relationship and common ground is established in the exchange encounter that assesses levels of commonality and aesthetic understanding between both stallholder and consumer. This dialogue provides a means for stallholders to describe their objects, offer a discussion of the object's history and previous lifecycle and ultimately navigate a use value and potential value in the mind of the consumer. For the stallholder it is a way to show their care for the object, the rarity and exclusivity of it. This development of an

interaction and bond is a key differentiator in separating the vintage marketspace from more traditional conceptions of the marketplace. This is also in line with Wenger's (1987) work on communities of practice, which suggests that an emergent sense of group membership and identity develops from practices that facilitate the interchange of collectively valorised resources. A number of respondents link these elements of marketspace interactions and values back to a broader sense of social belonging, for example Ben discusses:

"It is nice to think that you pop along on a Sunday afternoon, you do some shopping and pick up some new interesting pieces but you feel like you are in a place where people get you and you belong...you feel they understand how you like to shop and what you think is important [pause] maybe that is quite 'hipsterish' but it seems like one of the main reasons why I shop here."(Ben)

Ben prizes the feeling of belonging to the collective and that influences his rationale in choosing to shop at the markets. He places less emphasis on the particular objects and more on the interactions and feeling of people understanding him. He is conscious that this may play into connotations of *'hipster'* behaviour but it is an element of the consumption practices that is important for him. He actively seeks out others that he considers as sharing similar aesthetic values and considerations. In exploring this relationship further with Tina, who owns a bespoke vintage dress service, she speaks of building up almost familial relationships with the consumers she works with:

"It is my business but I love working with people for months on creating something that will mean so much to them but it always shapes me as well...it is important that you have the same vision and share a vintage aesthetic...long afterwards I still like to have a drink with them as we have been so much part of each other's lives."(Tina)

Tina conveys esteem around these relationships with consumers she highlights the time and involvement with each other, the idea that they both invest in each other around the exchange - in both a personal and a monetary fashion. She also illuminates the idea of the importance of the shared 'vintage aesthetic', that this similarity in how they view the world, their style preferences and appreciation of the vintage scene is key in the development of this relationship. The importance placed on these practices of a shared aesthetic and the social demonstration of these practices around the exchange, portrays a social investment in the object from both stallholder and consumer in which the value is constructed. Stallholders describe their consumers as friends and these friendships are longstanding, with some spanning years. They are relationships that although built around an exchange are spoken about in personal and affectionate terms. There appears to be friendships based on a shared understanding of aesthetics and style similarities but also of gaining advantage and knowledge in the marketspace. Practitioners spoke of the "camaraderie" and "community" of vintage and this conveys forms of sociality that are emergent also to the practices of consumption. The vintage marketspace can be understood as a 'performance of community value', which is a collective and social undertaking, and in which negotiations over objects, use and ultimately price, are intimately woven into the social. By this account, while appearing individualistic vintage consumption is also understood as a communal form of consumption whereby the reclamation of social relations and the enlivenment of social spaces are

made possible through the marketspace. This community feel was summarised by Michael:

"There is no doubt about it, there is a community thing going on, I think this is what will keep the thing going, no matter what happens to the vintage, that undercurrent of a vintage trend at the moment." (Michael)

And further added to by Louise:

"Vintage is a lifestyle and not just how you choose to shop: it is about a way of seeing the world but also I suppose how you see your friendships. Yes, I sell at markets as my job, but it is something that I love, when I am up at 4 am to go to a house clearance or car boot sale on a rainy Glasgow November morning, what I think about is selling that object, discussing with the customer about the item, building a relationship with them and thinking about what new lease of life they will give it... I love those interactions...I have made so many friends in this line of work and I always thought that I was alone in how I valued vintage stuff over new things..."(Louise)

The marketspace provides a context for this collective gathering and for the exchange of shared ideals. Louise describes how when selling her vintage homeware pieces she likes to "get a feel for the customer," and to "try and understand their vintage style and preferences" so as to make recommendations and know what they would be comfortable with. She feels that this exchange is crucial in the marketspace and that people value this interaction. This building of a relationship between stallholders and consumers and the acceptance into the vintage community is marked also by the time spent over the interactions around the exchange. This relationship

and interaction of the stallholder and consumer are intrinsic as to how the objects are viewed in their new lifecycle, as illustrated through researcher field notes:

"At the stalls it seems like there is a window of opportunity to talk around the object and really bring it to life. The exchange seems quite personable in that way and the stallholder's personality has to come through with their chat and establish a shared vision of vintage. Whether that is through a tale of how they came to possess the object or how they had cleaned it up, or even how the object could be used, it seems like this dialogue that takes place over the object either establishes a purpose for the object or not." (Researcher field notes, March 25th 2012)

As this excerpt depicts, consumption is viewed as a way to connect to others, in conceptualising consumption in this way it allows for an understanding of the complex emotional and social relations that unfold in the marketspace. The above extract demonstrates that the object is only one part of the exchange, the practice of building of relationships and the view that vintage is a lens for viewing consumption and through this a community has developed. In previous consumer research, friendship has largely been conceptualised as a form of member reference group. These groupings have been argued to have the power to exert a normative influence over their consumption choices (Tarnanidis et al., 2010). In the literature Turner and Rojek (2001) suggest that friendship is a direct consequence of sharing ritual activities. This parallels the discussion of Muniz and O'Guinn (2001) who assert that consumption activities can be placed centrally in the creation of friendship and ideas of belonging. The vintage context illuminates that friendship in the vintage marketspace is key to marketplace performance of the tribe and as a practice of belonging.

5.4.6 Socialities: Virtual Community Spaces

As a forum for social practices and knowledge exchange, this emergent collective relies on social media to connect, market and maintain their offline social practices. As a transient market, social media is used to give a concrete presence to the markets. Instead of websites and personal sites, social media is used to connect practitioners and promote the marketplaces. This move online was evident in the exchanges in the marketplace around the object and stall table, which were often followed up with encouragement to *'befriend'* each other on Facebook. Social media also serves as a way to market the events of the marketplace to an interested audience. As transient marketspaces these social media practices facilitate the marketspace, this allows knowledge of the marketspace to reach the collective. Molly comments on her use of social media:

"Of a weekend I normally just look on Facebook to see what markets are on that day or I will have seen friends or the organisers posting about an event during the week and I will then remember to stop by." (Molly)

In this way, social media reaffirms the physical connection for Molly and also allows for the community to move online. It takes these notions of social exchanges and the desire to be part of 'the club' to a more visible, tangible state. In considering community engagement practices, the practices of social networking are seen to be focus on the creation of enhancing and building alliances and networks. This move onto a further online connection represents a lasting, tangible social connection that results from this initial exchange. In this way the stallholders again act like gatekeepers to the vintage tribe and community extending invites to members of the collective that they assert merit this connection. Michael highlights the idea that this online platform allows for an interaction and acceptance:

"I think social networking has a lot to do with it... say you had a particular thing for vintage clothes, you would maybe meet other people who liked vintage clothes at the vintage clothes shop but the chances of hitting it off with them or going for a pint with them were pretty slim... because now people can express their interests up front on Facebook and places like that, there is more interaction with people, with groups that are like-minded..." (Michael)

For Michael social media becomes a vehicle to maintain and build emotional links within the collective. His extract also conveys that social media acts as a coordinator of the marketspace; it connects tribes (Cova et al., 2007; Canniford, 2011b; Goulding et al., 2013) that may have been unable to connect without this medium. This builds on the assertion, that the social web allows for the gathering of people with similar interests to converse and interact, therefore social networking sites can be regarded as a platform for social interactions and depict the collective's tribal affiliations (Maffesoli, 1996; Cova, 1997; Hamilton and Hewer, 2010).

The Facebook sites of all three of the markets act as tribal spaces (Maffesoli, 1996) which functions as a cultural resource that allows for an establishment of group unity which discussions of *'us'* and *'them'* can be used to differentiate the marketspaces from the masses. Discursive knowledge (Schau et al., 2009) is shared online and becomes a practice that is expected of the tribal members. The markets use social media to facilitate marketing practice and importantly to build relevance within the collective. As a tribal context, social media platforms allow for a tangible presence of

the member's emotional investment in the collective. In general Twitter is used as marketing platform to advertise and raise awareness of events, market dates and upcoming sellers, whereas Facebook acts as the main platform for emotionally engaging within the collective. Participation levels vary and the amount of personal information varies extensively across the collective. All three of the sample markets use Facebook to set up events pages for the market dates and showcase the sellers that will be present. They use this as the main vehicle to advertise the markets, with leafleting and e-newsletters used to supplement this. In the case of Granny Would be Proud, they use Facebook on a personal level to engage with the collective. On their market page they upload pictures of the markets, alongside pictures of their toddler daughter and do not distinguish between work (vintage markets) and private lives. This gave the feel of being more than work orientated but personal in nature and inextricably links the private and work sphere.



Figure 13 - Facebook screenshot of 'Granny Would be Proud' market page

Facebook provides an emotional cultural context to overcome geographical barriers to bring together like-minded practitioners and provides a space to share their passions. These active consumer practices are further grounded in Holt's (1995) 'consumption as play' and build on the suggestion by Kozinets (2008:351) that virtual communities "bring back some of the intimate feelings we enjoyed in childhood when we were deeply immersed in play: intimacy with ourselves and our capabilities alongside closeness with friends to share our passion."

As highlighted by Cova and White (2010) the most important resource in on and offline consumer gatherings, is the crowd. The online outlets also transform the fluidity of the social hierarchy of the vintage collective. The participatory nature of social media enables a collective authority to be tangible and positions of knowledge created within their tribal roles. This online organising and gathering of the tribe builds on connections of the marketspace and sparks new conversations and bonds to be built. This interactivity between practitioners is evidenced in a space that is used for advice and information, from discussions on 'the best vintage friendly dry cleaners' to 'restoration of blanket boxes' and 'sellers of the next marketplace,' as Jessica highlights:

"I don't always comment on stuff on the Granny [Facebook] page, a lot of it isn't for me, but I like to have look at the sellers that are coming up or I will look for a name of someone that I met at the market." (Jessica)

The dialogue observed in the social media landscape demonstrates within the vintage collective roles of directing, legitimising and accepting within this online tribal space, as Hamilton and Hewer (2010: 271) depict as space for *"interaction, connectivity and creativity."* This consumer community construct of *'virtual tribalism'* (deValck, 2007: 271) shows a togetherness that follows on from the marketspace interactions, it is personal in nature and the markets use social media to

create a relevance and everyday location in the collective's lives. The community spirit of personal engagement and interaction underpins the feel of the social media encounters. Although each of the markets uses the landscape of social media slightly differently, all use it as a way to navigate a more permanent feature in the vintage collective. Social media allows a platform away from the markets for socially agentic consumers to utilise a space for community building, self-promotion and value creation. This study of online interactions highlights how these spaces of virtual community gathering shape vintage consumption (Hamilton and Hewer, 2010; Denegri-Knott and Molesworth, 2010).

5.4.7 Conclusion of 'Socialities'

This final findings section has sought to explore the value of the tribal lens to the understanding of the sociality of vintage. The findings and preceding discussion focused upon 'the hierarchy at work', 'discourses of community', 'collective creativity' and 'the virtual collective space'. Within the tribal lens, vintage is framed as different from the mainstream, of doing, valuing and consuming differently. The pop-up markets become a site where tribal understandings and collective allegiances are negotiated and sustained through practices such as 'discourses of community' and 'collective creativity'. The collective and the sociality are negotiated through distinct forms of materiality and their interaction with these prized objects. In this way the exchanges are seen as testing emotional practices, such as commitment to vintage as a lifestyle. The material culture of vintage is produced through these interactions and creative practices and as a result vintage can be framed as a social object. The pop-up vintage markets play a social purpose for the collective and they offer visible forms of social connection in less formalised sites of consumption. The marketspace can be

understood as a '*performance of community value*', which is a collective and social undertaking, and in which negotiations over objects, use and ultimately price, are intimately woven into the social.

The literature from Chapter Two highlights studies of marketplace communities and how communal consumption activities form collective identifications grounded in shared beliefs, complex value systems, rituals and systems of meaning whereby consumers develop emotional links with one another (Schouten and McAlexander, 1995; Muniz and O'Guinn, 2001; Cova et al., 2007). Through this ethnographic thesis, this research builds on this and further identifies that communities of consumption often demonstrate their joint emotional attachment through a united opposition to the dominant cultural norms and mainstream consumer practices (Muniz and Schau, 2005; Arnould and Thompson, 2005; Dalli and Corciolani, 2008; Goulding and Saren, 2009). Further the vintage marketspace was highlighted as embedded in a socio-cultural system of collectively shared cultural knowledge. aesthetic tastes, social networks, and systems of social distinction and hierarchies (Fonarow, 2006). As Wenger (1998) asserts with his communities of practice discussion, even casual communities lean towards some form of internal leadership. In the vintage tribe, the hierarchy was based on experience and time as a practitioner, indeed all of the discussions with participants pulled on discourses surrounding a family tradition of prizing antiques and being educated in their youth of alternative retail outlets and aesthetic understandings. Community hierarchies (Kates, 2002; Thornton, 1995) were therefore constituted and maintained through this learned knowledge of the tribe's history and understanding of collective aesthetics. 'Newcomers' or 'hobbyists' in the marketspace were depicted as struggling to

negotiate boundaries or as those who had not yet learned to fully align with tribal codes, and therefore had not acquired the subcultural capital (Thornton, 1995; Bourdieu, 1984) to be fully accepted members of the collective or the 'secret club' that Adam explains. Through processes of interaction in the collective new knowledge was found when members interact, discuss and exchange knowledge with community members. Insider language and terminology around the doing of vintage was combined with an understanding of the discursive, tacit and performed knowledge needed to succeed in vintage. Value was formed through the lens of tribal learning (Goulding et al., 2013) and marketspace engagement. These findings add to an understanding of tribal learning, as Goulding et al. (2013) contend that learning to engage in the tribe is practiced and accrued over time, as competencies are learned from more experienced members. Within the vintage tribe respondents cultivated an embodied feeling for vintage that had been practiced from childhood. Vintage was seen as a practice realised over years of the doing of dressing and behaving, but also of valuing objects and appreciating the practices of this consumption landscape. This study's findings contribute to an understanding that the value in vintage was embedded within wider socio-cultural structures (Türe, 2014; Arnould, 2014).

The tribal nature of the vintage collective shows also that individuals participate and find value in the collective for firstly personal goals – information gathering, social interaction and the experience of the marketspace, secondly on a collective level – to develop a sense of community and identity and to support this grouping in their practices. This also links to the work of Pongsakornrungsilp and Schroeder (2011) who find that each member in their co-consuming group has two goals in their participation - on both an individual and group level. This further compliments Schau

et al.'s (2009) study in suggesting that the beneficiary role is a basic function of social practices in co-consuming groups. By assuming the role of beneficiary, consumers were able to engage with the community and facilitate value creation. Schau et al. (2009:25) study argues that through providing opportunities for members to demonstrate competencies, practices facilitate members to build cultural capital through *'adroit performances'*, which creates value for the consumer. This thesis develops this and proposes that practices actualise opportunities for new consumption experiences for tribal members.

The notion of 'communitas' was more ephemeral in the vintage context due to the transient nature of the marketspace, value was placed on the potential to inspire passion and form social links through deconstructing and assembling marketspace resources (Cova et al., 2007; Dant, 2004). The connected collective shared practices and values including the discourses employed to make sense of the vintage world. The application of the tribal lens shows how consumer practice can be understood as a situated social practice and 'cultivated assemblage' in action (Dant, 2004). As shown in this findings section, the move online to social media served to further add to their creation of a tangible social sphere to share their narratives, experiences and values around vintage. Further this reiterates the findings of Gregson et al. (2007) as they found consumers were influenced not only by their own competences but also by their social practice. This thesis contributes to an understanding of value as the vintage context illuminates that value was seen as an enactment of relations, practices and socialities constructed in a specific context that was spatially and temporally bound. From a consumer perspective, value was seen to emerge from what people do, through the distinctions between actors in social pursuit and

266

exchange (Arnould, 2014:2). The findings illustrate that value was mobilised by practitioner agency and skill and led to a collective valorisation of 'vintage' (Wenger, 1987).

This section concludes the findings chapter. This chapter has shown that vintage whilst constructed as an alternative to the mainstream, of being different through its specialness was seen as becoming all too ordinary. Within the research this sense of tension was made explicit with practitioners often categorising the vintage marketspace for its 'between-ness' (Crewe et al., 2003). In this manner, participants often questioned such change and enacted a feeling of protection around vintage. Further this thesis argues that the vintage collective has been shown to have the cooperative ability to threaten the mainstream norms and establish their own through their consumption choices, spaces (on and offline) and practices. Tribes therefore could be considered not simply as groupings of consumers but as a collective of 'prosumers' (Zwick et al. 2008; Humphreys and Grayson, 2008; Ritzer, 2010) or as in this study, socially agentic *practitioners* in the marketspace. In considering consumer agency in this way, members act as collective actors in the marketspace and the market develops and shifts as a result of their practice. This also shows that as part of the wider creative scene, as a consumer driven market, vintage is in a constant process of flux and regeneration. The following chapter synthesises these findings in relation to the research objectives and emergent themes and concludes the thesis.

Chapter 6: Conclusion

6.0 Introduction

The overall aim of this study was to explore how the practices, spaces and socialites intersect in the consumption of vintage, with a specific focus on the pop-up market format. Therefore this final chapter offers a summation of the findings, offering the core theoretical contributions this thesis makes to consumer research and to the CCT (Arnould and Thompson, 2005, 2007) domain. Further the overall contributions are explored noting the limitations of the study before detailing future directions for the research. The chapter concludes with the researcher's reflection on the study.

Each of the contributions will be addressed around the research themes of practice, space, sociality and value, these emergent themes are explored in line with the research objectives that guided the study:

- 1. What practices are used in the marketspace to materialise vintage?
- 2. How do practices, objects and knowledge intersect in the marketspace?
- 3. Does vintage create a shared consumption space?
- 4. How is value ascribed and framed in vintage?

6.1 Theoretical Contribution

The following section will detail further understanding into the core contributions of the thesis around with the emergent research themes of practice, space and socialities before providing insight into the other dominant contribution from the research theme of value.

6.1.1 Contribution: Practitioners as embodied locations of knowledge

The first contribution to practice refers to the conceptualisation of the socially agentic nature of the consumer in the market. In approaching an understanding of the practice of vintage, the construct of the consumer was explored early on in the literature review. As a result this study moves the understanding of consumers forward, by conceptualising them as socially agentic practitioners in the marketspace. The construct of the consumer as shown in Chapter Two was extended by the findings of this study, the consumer research focus of 'prosumers' (Bagozzi and Warshaw, 1990; Toffler, 1980; Xie et al., 2008) and 'working consumers' (Cova and Dalli, 2009) was combined with understandings of aesthetic labour (Entwistle, 2009; Denegri-Knott and Molesworth, 2009) needed to practice in the marketspace. The resulting contribution was that the framework of the 'practitioner' was extended to encapsulate the agentic nature of the consumer combined with the aesthetic labour needed to fully participate in the market. The construct of the practitioner was developed further as the findings reconsidered the intersections of the socio-market relations and the inter-relationships between consumer and seller, both as practitioners in action. Consumers were conceptualised as practitioners who negotiate and perform varied practices in their daily life (Shove and Pantzar, 2005). Practitioner agency in this thesis was regarded in line with Watson and Shove (2008: 71) as the skills implied in the use, integration and desiring of objects required for the effective accomplishment and performance of everyday life. This study demonstrates that the vintage practitioner was seen as both location and locus of market knowledge (Bourdieu, 1984, 1985, 1990; Entwistle, 2009; Entwistle and

Slater, 2014). This contribution adds to consumer research as the findings established that embodied knowledge was understood as performative and enacted. The findings showed that various knowledges (knowing as practice and performance) were inextricably woven together across objects, spaces and materials to make sense of the vintage market. Knowledge in vintage was based on sense making and engaging noncognitive abilities, it was formed through encounters with the objects and marketspace experiences and in this way it was embodied (Bourdieu, 1984; Entwistle, 2009). From the findings, practitioners attempted to verbalise their doing of vintage in conversations using embodied metaphors such as 'gut feelings' or 'the eye. 'In the vintage tribe, the findings demonstrated that hierarchy was based on experience and time as a practitioner, indeed all of the discussions with practitioners pulled on discourses surrounding a family tradition and being educated in their youth of alternative retail outlets and aesthetic understandings (Bourdieu, 1984). As the knowledge of vintage could only be acquired within the markets, as such the body emerges as an important location for knowing (Entwistle, 2009) and learning (Goulding et al., 2013). 'Newcomers' or 'hobbyists' in the marketspace were depicted as struggling to negotiate boundaries or as those who had not yet learned to fully align with tribal codes, and therefore had not acquired the subcultural capital (Thornton, 1995; Bourdieu, 1984) to be fully accepted members of the collective or the 'secret club' that Adam explains in the findings. This thesis reveals that what makes vintage was the practical activity of vintaging: through which the objects were actively organised and collectively framed through tribal learning (Goulding et al., 2013) and understanding. Through this the identity of the collective was created and maintained around the distinctive way of seeing the world or as Arsel and Thompson

(2011) highlight 'a marketplace myth'. The findings demonstrated the consumer collective was an active site of value creation, with value as a social construct that was tribal in nature. To conclude this contribution the evocativeness of vintage as practice, object and collective was perhaps best expressed as this study has sought to reveal as the marketspace performance of vintage as marked out by an ensemble of practices, which speaks of a form of longing for alternative worlds (Duffy and Hewer, 2013).

6.1.2 Contribution: Marketspace as practiced into being

This study builds on the conceptualisation of consumption as a consequence of practice (Warde, 2005) as portrayed in Chapter Three and explored in the findings, with the organised activities of consumers becoming the starting point for understanding the social world. This thesis employed practice theory within the CCT tradition, which allowed the researcher to focus on the 'doing' of social practices of vintage rather than the moments of individual decision-making within the collective. In returning to the literature review as a starting point for the contribution, Nicolini makes explicit: "Practices only exist to the extent that they are enacted and reenacted. Focusing on practices is thus taking the social and material doing (of something: [where] doing is never objectless" (Nicolini, 2012: 221). Extending Nicolini's (2012) argument, this study contributes to an understanding of practice in consumer research, as from the findings vintage was seen to be both emergent from and sustained by an extended apparatus of interconnected practices, performances and arrangements. Drawing on the theoretical underpinnings of practice theory to highlight the necessary knowledge of practitioners and the role the markets play in facilitating consumer and seller learning, the findings demonstrate that to be a

competent practitioner requires a thoroughly reflexive engagement with the vintage objects: their current use and potential usage (even if they have fallen out of use). The thesis contributes the emergent construct of '*cultivated assemblage*' (Dant, 2004, 2005) referring to different kinds of objects; understandings and processes at work in the marketspace. This was the result of the findings that revealed that vintage was not an item or an individual; it was a spatially assembled social being that adopts properties of both and conversely cannot exist without both (Dant, 2004: 74). The contribution that the '*marketspace is practiced into being*' was a result of this assemblage, therefore this thesis understands that practices were contextually embedded and value was not "*created*" by the consumer but formed in practical arrangements and contexts in which actions were carried out and resource elements were integrated with skills and images related to practices (Schatzki, 2001).

6.1.3 Contribution: Practice is spatially bound

The literature review revealed that a consideration of the spatial dimensions of practice were lacking from previous CCT studies (Arsel and Bean, 2013; Magaudda, 2011) which focused on the 'circuit of practice' or a visualisation of 'objects, doing and meanings' (Shove and Pantzar, 2005). This study moves this forward by proposing that the practice cannot be considered out with the context. The findings of this thesis demonstrate that 'space' needs to be more fully addressed in the discussions of practice within the CCT dialogue. The focus on the spatial occurrence of practices was central in this research study, but it was underrepresented in the CCT literature. De Certeau (1984) originally drew attention to the ways in which individuals utilise and move through spaces, Schau et al. (2009: 31) moved this discussion forward in the CCT landscape and argued that behaviours are 'spatially

dispersed', Denegri-Knott and Parsons (2014) accentuated the importance of the framing of the space of second-hand exchange, however work such as Arsel and Bean (2013) omitted the importance of space in a discussion of practices. This thesis contributes that a holistic CCT understanding of practice cannot be fully achieved without this acknowledgement of space.



Figure 14 - Practice bound by space. Adapted from Magaudda's (2011) 'circuit of practice'

This study demonstrated that these marginal sites of consumption, the pop-up markets, were active places that were an amalgamation of market objects, actors and influences that encode particular practices within the space (Slater, 2002a; Geiger et al., 2012). This discussion of space was key from the findings in conceptualising the

markets as bounded locations and also to understanding the circulation of *'knowledge in practice'* within. This thesis contributes to consumer research in understanding market practice as spatially bound.

6.1.4 Contribution: Practicing the familiar - 'eclectic homeliness'

Extending works on space, this thesis argues that in marketplace contexts by *invoking the alternative, i* vintage sites became spaces of differentiated exchange and were juxtaposed as Other throughout the research to 'spaces of the new.' As Denegri-Knott and Parsons (2014: 94) assert regarding car boot sale sites: "the location of these spaces outside the commercial marketplace both metaphorically and literally (as they are usually located on the margins of towns and cities) offers interesting flexibilities in practices of meaning making and ordering." In this thesis the marketspace was manifested in a practice of blurring the boundaries between the commercial space and the familiarity of the domestic domain. This recreation of the familiar through the performance of staging and negotiating 'the look' demonstrated a sense of belonging for practitioners in the marketspace and a space for 'consumption as play' (Holt, 1995). This study embraced the Glasgow scene as a particular instance of vintage in practice, for as Michael 'The Eyedresser' asserts in the findings, "vintage has become a safe space for people." This thesis extends this argument further as through the emergent theme of 'eclectic homeliness,' the marketspace exists in an ambiguous space, between the conventional retail market and the familiar domestic sphere, as a created temporal escape from the everyday. This notion of 'homeliness at work' extends Debenedetti et al.'s (2014: 918) recent work that illustrates that 'homeyness' can reside outside domestic settings and can be experienced in commercial settings. The vintage marketspace literally 'pops-up' and

creates a permeable boundary between practitioners (sellers and consumers) as such the consumption experience was not just focused on the object or service given, but on this 'practice of belonging' (Brownlie and Hewer, 2011) or as this thesis found, the latent familiarity of the space. More than just a marketspace aesthetic, the findings showed that 'eclectic homeliness' became a roadmap for practitioners, a way to negotiate the finding process of vintage and navigate the accepted codes of tribal behaviour (Cova et al., 2007). The marketspace setting recreates the experience of searching for finds and the physicality of the vintage process allows a heightened sense of accomplishment for the practitioner in the searching process. The findings demonstrated that the physical environment of the marketspace becomes a product of its temporal nature and the market site location reinforces the knowledge in practice needed by the practitioner to negotiate the disordered and informal retail setting. This turn to the familiar in commercial spaces or third spaces was highlighted initially in Linnet's (2011) work on 'hygge,' as he spoke of 'coziness' encountered and sought by Danish consumers, this thesis builds on this and argues that through the space of the 'pop-up' marketspaces, a platform of familiarity is practiced into being. Practices of organising and display create this temporal distance from the banal every day and transports practitioners to a liminal space of play (Holt, 1995) and performance (Ustuner and Thompson, 2012). For practitioners, this space exists in an ambiguous moment between commercial market logic and the familiar domestic sphere.

As explored in this thesis, alternative markets such as the 'pop-up' have become commonplace throughout the UK. This thesis suggests that consumer culture has a role to play in allowing access to contexts whereby this practice of '*eclectic homeliness*' can take place and as such this study has implications for both policy

275

and practice. As shown from the findings, as a concept, vintage has continued to grow in popularity and this has implications for all levels of the fashion chain – from charity industries with vintage sections and strategic merchandising (that previously played a fairly insignificant role in the fashion system), to high-end designer vintage stores and mass-market fashion stores such as Topshop (as the study highlighted in the introductory chapter). At a market level this study has shown through the finding of 'nostalgic discourses of community' that there has been a shift in perspectives of consuming from a more local marketspace that allows consumers to build relationships with sellers and with each other in a space of 'eclectic homeliness'. In returning to the research findings it was evident that the marketspace also becomes an arena for inclusive sociality and materialising social connections. As the findings show, the collective utilises otherwise unused spaces (empty rooms and closed off areas in commercial spaces) to make material their tribal affiliations and create a transient but shared space. This occupation and reinvigoration of social spaces has important community implications and policy considerations as the pop-up markets were demonstrated as a way to regenerate disused spaces. As Watson (2009) highlights, the markets act as a shared social space and function as sites of social association and inclusion. The importance of this social function was highlighted in all three of the markets of this research study as they play a social role in offering visible forms of social connection occurring in less formalised sites of consumption. Within the sample site of Glasgow, these markets provide an alternative retail destination away from the high street but also as the findings have revealed a space of communal gathering and togetherness. Allowing a space for this communal creativity and regeneration of unused spaces within the city has policy implications

276
for the connections between market organising, local facilities and community building to enhance the viability of markets in towns and cities. These groups were congregating in 'safe spaces' and should be embraced for their sociality and innovation of collective spirit. This provides opportunity for local areas to be developed, regenerated and rebranded in line with this collective valuing of space.§

6.2 Contribution: Framing Value Fluidity

Value appeared as an emergent research theme that emanated from the discussions and findings of practice, space and sociality. Within the consumer research domain the common assumption has been to consider that objects follow a linear trajectory of valuation, consumption and decline, however as the literature review (Türe, 2014; Parsons, 2008; Gregson and Crewe, 2002; Thompson, 1979; Denegri-Knott and Parsons, 2014) and findings of this thesis has shown, objects move through spaces, different contexts and biographies and this disrupts this linear conceptualisation as not accommodating the circularity in the movement of objects and consumer practices (Duffy, 2012c). Throughout the findings chapter value consequences of practice engagement were continually highlighted as this thesis reveals that value in vintage was dynamic and emergent in practice. This study adds to the focus on the tripartite systems of practice (Magaudda, 2011; Arsel and Bean, 2013) explored through the research objectives, which allows for an exploration of value and what consumers actually do with the objects in the pop-up space. Practices have been seen to tie value creation to a specific socio-cultural and spatio-temporal context (as shown in the contribution 'Practice is spatially bound'). This thesis reveals that a turn to practices within CCT allows for a focus beyond the consumption meanings

and to the value of material objects and the everyday doings of people. In this sense, vintage objects become materials that were brought together temporarily, were partially stabilized and then moved on again in the market and this led to a questioning throughout the study as to how value was framed, and the importance of considering *'the social lives of objects'* (Appadurai, 1986:3). This study broadens this perspective and introduces an understanding of value that was formed through a 'spatialized' circuit of practice (Magaudda, 2011: 18). This thesis found that the practice has to be spatialized in order for value to be negotiated and materialised in the marketspace.

This thesis further contributes to the recent attention to value within the CCT dialogue (Arnould, 2014; Venkatesh and Penaloza, 2014; Karababa and Kjeldgaard, 2014) as this study contributes that value was framed in practice in the 'doing' of the consumer marketspace. When seen through this lens, value was understood as fluid and necessitates an understanding of liquidity of value (Bauman, 2000; Bardhi and Eckhardt, 2012). Vintage value does not appear from nowhere, it was generated internally to the market itself by the routine actions of the practitioners (seller and consumer) and by the practices of the market actors (Bourdieu, 1984; Entwistle, 2009). This thesis contributes to an understanding of the fluidity and liquidity (Bauman, 2000) of value by exploring how marketplace consumers and sellers generate value within practice. As Bauman (2000:62) states with regards to 'liquid modernity': *"for the possibilities to remain infinite, none may be allowed to petrify into everlasting reality. They had better stay liquid and fluid and have a 'use-by' date attached, lest they render the remaining opportunities off limits."* This thesis argues that value was ever changing in the vintage marketspace as vintage exists in a

liminal moment whereby value was mobilised by practitioner agency, creativity and skill and this leads to a temporal collective valorisation of 'vintage' (Wenger, 1987). This contribution further develops Bardhi and Eckhardt's (2012) notion of temporality, as they discuss that relationships to possessions are temporary and situational and highlight the liquidity of this consumer practice. This thesis demonstrates that *value fluidity* was seen as temporal and ephemeral, it exists in pockets of time, to both be seized and capitalised on by the practitioner, or for the potential opportunity of discovery to be missed. This contribution represents a shift in how value was viewed within consumer research. From a consumer perspective, value was seen to emerge from what people do, through the distinctions between actors in social pursuit and exchange (Bourdieu 1984, 1985; Arnould, 2014). This thesis contributes to an understanding of value as the vintage context illuminates that value was seen as an enactment of relations, practices and socialities constructed in a specific context that was spatially and temporally bound. This proposal adds to Türe's (2014) understanding of value in which value was "constructed beyond the dyadic relation between objects and consumers" (Türe, 2014:16). The findings illustrate how value was formed in practice, rather than focusing on what kind of value was created. Value in vintage was routinely formed in practice in the 'doing' of vintage through which the social life of the object can begin to be captured (Appadurai, 1986). To conclude this contribution, the findings demonstrated that value in vintage was embedded within wider socio-cultural structures and this adds to the notion that the 'value system' of CCT lacks a clear social domain that impacts market organisation (Venkatesh and Penaloza, 2014:3).

6.3 Methodological Considerations

Coming from the CCT tradition, the literature review established that differences with regards to methodology are encouraged and it does not prioritise any one approach. As explored in the literature review, practice perspectives have gained increased attention within the CCT community (Arsel and Bean, 2013; Magaudda, 2011) as over recent years researchers have strived to situate and contextualise consumer life worlds. This compliments the shift in CCT that calls for a move away from naturalistic research to a research methodology that encompasses the '*cultural* complexity of social action and marketplace activity' (Moisander et al., 2009: 330) and the contextualisation of the consumer experience (Askeggard and Linnet, 2011). However, as yet no agreed methodological approach to the study of practices has emerged. Building on Halkier and Jensen's (2011:103) proposal that the implications of the use of practice theory is to see all types of qualitative data as enactments of social action and to enable the 'construction of analytical generalisations that are not based on methodological individualism. 'This study acknowledges the challenges presented in using a practice approach and the operational difficulties in the analytical translation between applied practice theoretical concepts and operative methodological procedures (Halkier and Jensen, 2011). Martens (2012) advanced the discussion on the methodological implications of using practice theory by exploring how interviewing was both illuminating and obstructing as a method to capture practice and the 'what we do' questions. Martens (2012) questioned the use of language and interviewing when trying to capture practices, her study asserted that they brought out the 'human-to-human interactional concerns of practices' but that methodological creativity is needed moving forward to capture the different

dimensions of practical accomplishments. One aspect of this creativity (Martens, 2012) suggested is through the use of multiple methods of data capturing – such as visual, and/or video. This thesis contributes to this and argues that methodologically this turn to practice has implications in how to capture consumer's lived practices as by their very nature practices are difficult to access, as they are embodied, tacit, material and contextually bound. In this research study using the practice lens and understanding culture as a field of practices, the analytical focus shifted the focus from individual experience, to culture and the collective as a site of social action (Bourdieu, 1984; Shove and Pantzar, 2005). The perspective was thus orientated on the practices through which culture gains meaning and value in the marketspace. In proposing that a practice based approach to markets and marketing elicits that they should be studied '*in the making*' (Araujo et al., 2008: 6), an ethnographic approach to capture this performance of practice was central to this study. Using observations, interviews, field notes and visuals in this ethnographic study allowed for the researcher to follow consumer representations at different moments in the 'circuit of *culture*' (du Gay et al., 1997). Building on the critiques within the CCT domain, this research study used multi-ethnographic methods to capture practices in action. Some methods such as participant diaries or photo elicitation techniques may be further suited to gaining access to practices which respondents found difficult to articulate as they were 'taken for granted practices' and a potential limitation of this study. As an ethnographic researcher, this process has involved self-exploration and reflection throughout. Gaining an insight into the collective was an experience of continual learning and the research methods employed encouraged reflexivity in critically capturing the practices of the marketspace. From a developmental perspective, the

research study increased understanding of others and afforded the opportunity to learn and grow as a researcher.

With regards to the practice theory focus, an increase in scholarly attention has resulted in the discussion of practice within the CCT literature continuing to emerge and receive increased academic attention. This study is situated within the CCT ethnographic tradition as has been used in practice-orientated studies, however as this study has contributed turning to other domains such as Organisational Studies has been necessary to gain an understanding of methodological implications of the practice approach. This study challenges the CCT understanding of capturing practices. As explored in Chapter Three, Simpson (2009) develops a perspective of practice as a transactional social process that involves action and experience, whereby the practice lens was seen as dynamic, emergent and socially agentic. In understanding developing and existing practices within the domain of marketing and consumption, new ways of knowing and methodologically capturing practice are called for by this study (Bajde, 2013). From a meso-level and in line with the theoretical contributions detailed above of considering the space of practice, this thesis argues for a contextual awareness from a methodological perspective and in answering, Thompson et al. (2013:18) who call for CCT to "disentangle the material and discursive webs," argues that the subject has to be placed within the wider social and cultural context within the research study. These points are considered in line with areas of future research and as the researcher has continued to maintain a relationship within the collective it is possible to embrace and explore these areas further.

282

6.4 Areas for Future Research

The following section will outline the areas of future research emanating from this thesis. Suggestions for future research are based on the research contributions presented above and identified limitations of the study.

6.4.1 Space

From a theoretical standpoint as shown from the findings and contributions above, the discussion of practice to date within the CCT field has omitted the active spatial dimension to practice (Arsel and Bean, 2013). This thesis has shown that this contextualising is necessary to fully address the individual and collective nature of practices in the marketspace. Extending works on space, this thesis argues that market contexts such as the vintage marketspace represent a turn to the alternative, a space that was created and manifests the collectives' values. The pop-up format has continued to gain prominence in the UK market and as such a potential limitation of this study is its focus on one context of vintage and the pop-up market format. Future research should address more alternative marketplaces as a means of gaining further understanding into the spatial dimensions of the marketplace. Does the pop-up model offer a future for the consumer as opposed to high street retail or will it continue in the alternative?

6.4.2 Practice and Creativity

The act of vintaging this thesis revealed as practiced into being, with the pop-up markets shown to be a space of *'tribal learning'* (Goulding et al., 2013) and apprenticeship. As with Appadurai's (1986) theory of artifactual biography, each exchange was open to the individual's value perceptions, therefore consumers could

283

remove objects away from and re-insert them into their prescribed paths, through practices of consumer creativity. Türe (2014:3) contends that in light of this perspective, the value of objects is temporally and culturally constructed through trajectories and diversions, and this study further problematizes the interrelations between social relations, creativity and value regimes. As Arnould (2014) depicts, *'human creativity is therefore a cornerstone of value creation'*. As an instance of bricolage, the thesis revealed that vintage was the result of practitioner understanding and knowing, but asks where does the creativity reside in these practices? The turn to the vintage context would suggest a reframing of consumer creativity as an active undertaking by *'unruly bricoleurs'* (Holt, 2002) for whom resistance is constitutive of vintaging but necessitates further questioning as to how this creativity is practiced into being and where the value is located?

6.4.3 Market Formation

This thesis contributes to an understanding of the workings of alternative markets, which other than Sherry's (1990a) seminal discussion of the flea market, has largely been ignored by consumer researchers to date. This thesis contributed to the understanding of the transience and *'value fluidity'* (Bauman, 2000; Bardhi and Eckhardt, 2012) of the marketspace formation for both consumers and sellers, however the area of market formation may potentially illuminate an interesting voice from the practitioners who have long practiced vintage commercially and leads to a questioning of how has their practice of vintage altered as a result of its more widespread appropriation. As Shove (2012) quotes Reckwitz for whom *"individuals feature as the carriers or hosts of a practice. This is a radical departure from more conventional approaches in which understandings, know-how, meanings and*

purposes are taken to be personal attributes. Reckwitz argues that it makes better sense to treat these not as the qualities of an individual but as 'elements and qualities of a practice in which the single individual participates'" (2002: 8). This understanding adds to a problematising of whether their market formation practices shift as a result of vintage success?

6.4.4 Aesthetic Labour

The findings further revealed the skill and work (Denegri-Knott and Molesworth, 2009, 2010; Entwistle, 2009) that goes into both vintage and the transient markets. The nebulous and hard to define aspect of marketplace work should be developed further in line with the understanding of 'aesthetic labour' (Entwistle, 2009) that emanates from this study from both a vendor and consumer perspective. This thesis revealed the construct of the practitioner was grounded in an understanding of this work of the marketspace but could be further developed in line with ideas of categorisation (Slater, 2002a, 2007, 2014) and ordering (Denegri-Knott and Parsons, 2014) of the market.

6.5 Conclusion

To conclude, this thesis has explored how the practices, spaces and socialites intersect in the consumption of vintage. The thesis has addressed four research objectives with regards to theoretical and methodological underpinnings. Firstly it detailed the 'doings' of vintage from a practice perspective and extended the CCT understanding of practice to be spatially and temporally bound. It explored how objects, practices and knowledge interest in the marketspace and performance of 'the alternative,' as practitioner agency combined with aesthetic labour (Entwistle, 2009) moved through networks of objects and social practices. It explored the vintage tribe as a collective consumption space of '*tribal learning*' (Goulding et al., 2013) and the development of a space of the familiar, of '*eclectic homeliness*.' Lastly the thesis contributed to an understanding of the framing of the fluidity of vintage value and how it develops and materialises in practice. The study further highlighted the importance of a turn to a practice perspective methodologically within the CCT landscape and the affordance of future research opportunities. This interpretive research study contributes to discussions on the use of practice theory in CCT research, to explorations of alternative marketplaces with consumer research and the fluidity of value. As well as outlining contributions to both CCT and consumer research, this thesis has presented a plethora of future research areas particularly within the domain of spatial discussions within the practice orientation, the occurrence of market formation and 'liquidity' (Bauman, 2000; Bardhi and Eckhardt, 2012) of value.

It is hoped from this thesis that the reader has gained insight into the collective undertaking of vintage, the impact of the pop-up markets and the value in a turn to this alternative marketspace. Throughout this thesis the question of how vintage was materialised in practice shaped the problematising of the field. This thesis was the result of an initial curiosity into how vintage was materialised and consumed in practice and how this practice influenced not just the object and its subsequent value but the marketspace and the collective. In the opening of the findings section, the researcher field note presented questioned what it was about the markets that marked it as different and what consumers received from this space. Upon re-reading the excerpt and concluding the thesis, the researcher has come to understand the vintage

286

marketspace as an active space, a space that was brought alive in temporal moments through practice and presents opportunities and potential for not just objects, but for individuals and the wider collective. Rather than a space mired in nostalgia for past times, it was an exciting space to trial new aesthetics and consume in a different way.

References

Abercrombie, N., Warde, A. (1988) *Contemporary British Society: A New Introduction to Sociology*. Polity Press: Cambridge, UK.

Abolafia, M.Y. (1998) Markets as Cultures: An Ethnographic Approach, in Callon, M. (ed.) *The Laws of Markets*. Blackwell: Oxford, UK: 69–85.

Agar, M.H. (1996) *The Professional Stranger: An Informal Introduction to Ethnography*, Second Edition, Academic Press, San Diego, USA.

Agnew, J. (1996) Mapping politics: how context counts in electoral geography. *Political Geography*. 15(2), 129-146.

Ahuvia, A.C. (2005) Beyond the Extended Self: Loved Objects and Consumer Narratives. *Journal of Consumer Research*. 32(1), 171-84.

Albinsson, P. A., Perera, B. Y. (2009) From trash to treasure and beyond: the meaning of voluntary disposition. *Journal of Consumer Behaviour*. 8(6), 340–353.

Aldridge, A. (2003) Consumption. Polity Press: UK.

Algesheimer, R., Gurau, C. (2008) Introducing structuration theory in communal consumption behavior research. *Qualitative Market Research: An International* Journal. 11(2), 227-245.

Algesheimer, R., Utpal M. D., Herrmann A. (2005) The social influence of brand community: evidence from European car clubs. *Journal of Marketing*. 69(3), 19-34.

Allen, D.E. (2002) Toward a Theory of Consumer Choice as Sociohistorically Shaped Practical Experience: The Fits-Like-a-Glove (FLAG) Framework. *Journal of Consumer* Research. 28(4), 515–32.

Andersson, P., Aspenberg, K., Kjellberg, H. (2008) The Configuration of Actors in Market Practice. *Marketing Theory*. 8(1), 67–90.

Appadurai, A. (1986) (ed.) *The Social Life of Things: Commodities in Cultural Perspective*. Cambridge University Press: Cambridge, UK.

Araujo, L., Finch, J., Kjellberg, H. (ed.) (2010) *Reconnecting Marketing to Markets*. Oxford University Press, UK.

Araujo, L., Kjellberg, H., Spencer, R. (2008) Market practices and forms: introduction to the special issue. *Marketing Theory*. 8(1), 5-14.

Arnould, E.J. (1989) Toward a broadened theory of preference formation and the diffusion of innovations: cases from Zinder Province, Zaire. *Journal of Consumer Research*. 16(2), 239-66.

Arnould, E.J. (1998) Ethical Concerns in Participant Observation/ Ethnography. *Advances in Consumer Research*. 25, 72-74.

Arnould, E. J. (2008) Service-dominant logic and resource theory. *Journal of the Academy of Marketing Science*, 36(1), 21-24.

Arnould, E.J. (2014) Rudiments of a Value Praxeology. *Marketing Theory*. 14(1), 129-133.

Arnould, E. J., Price, L.L. (1993) River Magic: Extraordinary Experience and the Extended Service Encounter. *Journal of Consumer Research*. 20(1), 24-45.

Arnould, E. J., Wallendorf, M. (1994) Market-orientated ethnography: Interpretation building and marketing strategy formulation. *Journal of Marketing Research*. 31(4), 484-504.

Arnould, E.J., Thompson, C.J. (2007) Consumer Culture Theory (And We Really Mean Theoretics. *Research in Consumer Behavior*. 11, 3-22.

Arnould, E.J., Price L.L, Malshe A. (2006) Toward a cultural resource-based theory of the customer, in *The Service Dominant Logic of Marketing: Dialog, debate and directions*. (ed.) Lusch, R.F., Vargo, S.L. New York: MS. Sharpe: 91-104.

Arnould, E.J., Thompson, C. J. (2005) Reflections and Reviews: Consumer Culture Theory (CCT): Twenty Years of Research. *Journal of Consumer* Culture. 31, 868-882.

Arnould, E.J., Wilk, R.B (1984) Why do the natives wear Adidas? *Advances in Consumer Research*. 11(1), 748-753.

Arnold, R. (2001) *Fashion, Desire and Anxiety: Image and Morality in the Twentieth Century.* I.B. Tauris, London, UK.

Arnold, S. J., Kozinets, R., Handelman, J. (2001) Hometown Ideology and Retailer Legitimation. *Journal of Retailing*. 77(2), 243–71.

Arnold, S.J., Fischer, E. (1994) Hermeneutics and Consumer Research. *Journal of Consumer Research*. 21(1), 55-70.

Arsel, Z., Bean, J. (2013) Taste regimes and market-mediated practice. *Journal of Consumer Research*. 39(5), 899-917.

Arsel, Z., Thompson, C.J. (2011) Demythologizing Consumption Practices: How Consumers Protect Their Field-Dependent Identity Investments from Devaluing Marketplace Myths. *Journal of Consumer Research*. 37(5), 791-806.

Arvidsson, A. (2006) *Brands: Meaning and Value in Media Culture*. Routledge: London, UK.

Arvidsson, A. (2011). Ethics and value in customer co-production. *Marketing Theory*, 11(3), 261-278.

Askegaard, S. (1991) Towards a semiotic structure of cultural identity. In *Marketing and semiotics*, edited by Larsen, H.H, Mick, D.G and Alsted, C., Handelshøjskolens Forlag, Copenhagen, Denmark: 11-25.

Askegaard, S. (2010) Experience economy in the making: Hedonism, play and coolhunting in automotive song lyrics. *Consumption, Markets and Culture*. 13(4), 351-371.

Askegaard, S., Eric J. Arnould, E.J., Kjeldgaard, D. (2005) Postassimilationist Ethnic Consumer Research: Qualifications and Extensions. *Journal of Consumer Research*. 32(1), 160-170.

Askegarrd, S., Linnet, J.T. (2011) Towards an epistemology of consumer culture theory: Phenomenology and the context of context. *Marketing Theory*. 11(4), 381-404.

ASOS Marketplace [online] Available at <u>https://marketplace.asos.com/ [accessed on</u> May 15th 2013.]

Atkinson, P. (1990) *The Ethnographic Imagination: Textual Constructions of Reality*. Routledge: London, UK.

Atkinson, P. (1992) *Understanding ethnographic* texts. Sage publications: Newbury Park California, USA.

Auge, M. (1995) *Non-Places: Introduction to an Anthropology of Supermodernity*. Verso: London, UK.

Babin, B.J., Darden, W. R., Griffin, M. (1994) Work and/or fun: Measuring hedonic and utilitarian shopping value. *Journal of Consumer Research*. 20(4), 644-656.

Badot, O., Bucci, A., Cova, B. (2007) Beyond Marketing Panaceas: In Praise of Societing, in Saren M., Maclaren P., Goulding C., Elliott R., Shankar A., Catterall M. (eds) *Critical Marketing*. Elsevier, Oxford, UK: 85–98.

Bagozzi, R. P., Warshaw, P. R. (1990) Trying to consume. *Journal of Consumer Research*. 17(2), 127-140.

Bagozzi, R.P (1975) Marketing as Exchange: A Theory of Transactions in the Marketplace. *Journal of Marketing*. 39, 32-39.

Bagozzi, R.P. (2000) On the Concept of Intentional Social Action in Consumer Behavior. *Journal of Consumer Research*. 27(3), 388-396.

Bagozzi, R.P., Dholakia U.M. (2002) Intentional Social Action in Virtual Communities. *Journal of Interactive Marketing*. 16(2), 2-21.

Bajde, D. (2013) Consumer culture theory (re)visits actor–network theory: flattening consumption studies. *Marketing Theory*. 13(2), 227-242.

Baranov, D. (2012) *Conceptual Foundations of Social Research Methods, Second Edition,* Paradigm Publishers: London, UK.

Bardhi, F. (2003) Thrill of the Hunt: Thrift Shopping for Pleasure. *Advances in Consumer Research*. 30 (eds.) Keller, P.A and Rook, D.W: 375-376.

Bardhi, F., Arnould, E.J. (2005) Thrift shopping: combining utilitarian thrift and hedonic treat benefits. *Journal of Consumer Behaviour*. 4(4), 223-233.

Bardhi, F., Eckhardt, G.M., Arnould E.J. (2012) Liquid Relationship to Possessions. *Journal of Consumer Research*. 39(3), 510-529.

Bardley, C. (2002) *Wearing Vintage*. Black Dog and Leventhal Publishing: New York, USA.

Baudrillard, J. (1968) The System of Objects. Verso Books.

Baudrillard, J. (1972 [1981]) *For a Critique of the Political Economy of the Sign.* Telos Press Publishing Ltd: St Louis, MO, USA.

Baudrillard, J. (1998) The Consumer Society. Sage: London, UK.

Baudrillard, J. (2000) *The Vital Illusion*. Columbia University Press: New York, USA.

Baudrillard, J. (2006) The System of Objects. Verso: London, UK.

Baurdillard, J. (1993) Symbolic Exchange and Death. Sage Publications Ltd: UK.

Bauman, Z. (1988a) Is there a post-modern sociology? *Theory, Culture and Society*. 5(2), 217-237.

Bauman, Z. (1988b) Sociology and Postmodernity. *The Sociological Review*. 36(4), 790–813.

Bauman, Z. (1990a) Thinking Sociologically. Blackwell: Oxford, UK.

Bauman, Z. (1990b) Searching for a centre that holds. In *Global culture: Nationalism, globalization and modernity. edited* Featherstone, M. Sage: London, UK: 143-69.

Bauman, Z. (1992) Imitations of Postmodernity. Routledge Press: London, UK.

Bauman, Z. (2000) Liquid modernity. Polity, Cambridge.

Beckett, A., Nayak A. (2008) The reflexive consumer. *Marketing Theory*. 8(3), 299-317.

Belk, R.W., Tumbat, G. (2005) The cult of Macintosh. *Consumption, Markets and Culture*. 8(3), 205-17.

Belk, R.W. (1975) Situational Variables and Consumer Behavior. *Journal of Consumer Research*. 2(3), 157-164.

Belk, R.W. (1986) Art Versus Science as Ways of Generating Knowledge About Materialism. *Perspectives on Methodology in Consumer Research*. Brinberg, D. and Lutz, R. (eds). New York: Springer-Verlag: 3-36.

Belk, R.W. (1988) Possessions and the Extended Self. *Journal of Consumer Research*. 15(2), 139-168.

Belk, R.W. (1991a) Collecting in a Consumer Culture, in Belk R. et al. Highways and Buyways: Naturalistic Research from the Consumer Behavior Odyssey, Provo, UT: *Association for Consumer Research*.

Belk, R.W. (1991b) Possessions and a sense of the Past, in Belk R. et al. Highways and Buyways: Naturalistic Research from the Consumer Behavior Odyssey, Provo, UT: *Association for Consumer Research*.

Belk, R.W. (1991c) The History and the Development of the Consumer Behaviour Odyssey, in Belk R. (ed.) Highways and Buyways: Naturalistic Research from the Consumer Behaviour Odyssey, Provo, UT: *Association for Consumer Research*: 1-12.

Belk, R.W. (1992) Moving Possessions: An Analysis Based on Personal Documents from the 1847-1869 Mormon Migration. *Journal of Consumer Research*. 19(3), 339-361.

Belk, R.W. (1995a) Collecting in a Consumer Society. Routledge: London, UK.

Belk, R.W. (1995b) Studies in the New Consumer Behaviour in Miller, D. (ed.). *Acknowledging Consumption*. Routledge: London, UK: 58-95.

Belk, R. W. (1996) Hyperreality and globalization: culture in the age of Ronald McDonald. *Journal of International Consumer Marketing*, 8(3-4), 23-37.

Belk, R.W. (2006) Remembrances of things past: silent voices in collections. *European Advances in Consumer Research*. 7, 392-397.

Belk, R.W. (2010) Sharing. Journal of Consumer Research. 36(5), 715-34.

Belk, R.W. (2013) Visual and projective methods in Asian research. *Qualitative Market Research: An International Journal*. 16(1), 94–107.

Belk, R.W., Costa, J.A. (1998) The Mountain Man Myth: A contemporary consuming fantasy. *Journal of Consumer Research*. 25(3), 218-240.

Belk, R.W., Ger G., Askegaard, S. (2003) The Fire of Desire: A Multi-Sited Enquiry into Consumer Passion. *Journal of Consumer Research*. 30(3), 26-51.

Belk, R.W., Grayson K., Muniz A.M., Schau, H.J. (2011) (ed.) *Research in Consumer Behavior*. Volume 13, Emerald Group Publishing Ltd.

Belk R.W., Tumbat, G. (2005) The Cult of Macintosh. *Consumption Markets and Culture*. 8(3), 205-217.

Belk, R.W., Kozinets, R.V. (2005) Videography in marketing and consumer research. *Qualitative Market Research: An International Journal.* 8(2), 128-141.

Belk, R.W., Kozinets, R.V. (2007) 25 Camcorder society: quality videography in consumer and marketing *research*. *Handbook of qualitative research methods in marketing*. Edward Elgar Publishing Limited.

Belk, R.W., Sherry, J.F., Wallendorf, M. (1988) A Naturalistic Inquiry into Buyer and Seller Behaviour at a Swap Meet. *Journal of Consumer Research*. 14(4), 449-470.

Belk, R.W., Wallendorf, M., Sherry, J., Holbrook M.R S. (1988a) Collectors and Collecting. *Advances in Consumer Research*. 15, 548-553.

Belk R.W., Wallendorf, M., Sherry, J.F.Jr. (1988b) The Sacred and the Profane in Consumer Behaviour: Theodicy on the Odyssey. *Journal of Consumer Research*. 16, 1-38.

Belk, R.W., Wallendorf, M., Sherry, J.F., Holbrook, M.B. (1991) *Collecting in a Consumer Culture*, in Belk R.W, ed., Highways and Buyways: Naturalistic Research from the Consumer Behavior Odyssey, Provo, UT: *Association for Consumer Research*, 178-215.

Bendapudi, N., Leone, R.P. (2003) Psychological implications of customer participation in co-production. *Journal of Marketing*. 67(January), 14-28.

Bernstein, R. J. (2010) The Pragmatic Turn. Polity Press: London, UK.

Beverland, M.B., Farrelly, F.J. (2010) The Quest for Authenticity in Consumption: Consumers' Purposive Choice of Authentic Cues to Shape Experienced Outcomes. *Journal of Consumer Research*. 36(5), 838-856.

Blanchette, A. (2014) Revisiting the "passée": history rewriting in the neo-burlesque community. *Consumption Markets and Culture*. 17(2), 158-184.

Bode, M., Ostergaard, P. (2013) The wild and wacky worlds of consumer oddballs: Analyzing the manifestary context of consumer culture theory. *Marketing Theory*. 13(2), 175-192.

Boltanski, L., Thévenot, L. (1991) De la justification. Paris: Gallimard.

Boltanski, L., Thévenot, L. (2006) *On justification: Economies of worth.* Princeton University Press: USA.

Borgerson, J. (2005) Materiality, Agency and the Constitution of Consuming Subjects: Insights for Consumer Research. *Advances in Consumer Research*. 32, (ed.) Menon, G. and Akshay R.R.: 439-43.

Borgerson, J., Schroeder, J. (2003) Marketing the Retro-Escape of Hawaii, in Brown S., Sherry, J. F. Jr. (ed.) *Time, Space and the Market: Retroscapes Rising*, ME Sharpe, London, UK: 219.

Bouchet, D. (2007) Added Value: The Differences that Count. About the Necessity of Taking Language and Culture into Account as Competitive Factors. *Language at Work Bridging Theory and Practice*. 3(Autumn), 38-44.

Bourdieu, P. (1972[1977]) *Outline of a Theory of Practice*. (Vol. 16). Cambridge University Press, UK.

Bourdieu, P. (2010 [1977])*Outline of a Theory of Practice (translated by Richard Nice)*, Cambridge University Press, Cambridge, UK.

Bourdieu, P. (1984) *Distinction: A Social critique of the judgement of taste,* President and Fellows of Harvard College and Routledge and Kegan Paul Ltd, USA.

Bourdieu, P. (1985) *The Social Space and the Genesis of Groups, Theory and Society,* Springer. 14(6), 723-744.

Bourdieu, P. (1990) The logic of practice. Stanford University Press.

Bourdieu, P. (1993) *The Field of Cultural Production: Essays on Art and Literature*. Polity Press: Cambridge, UK.

Bourdieu, P. (1998) On Television and Journalism. Pluto: London, UK.

Bourdieu, P. (2005) The Political Field, the Social Science Field, and the Journalistic Field, in: Benson, R. and Neveu, E. (ed.) *Bourdieu and the Journalistic Field*, Polity Press, Cambridge, UK: 29-47.

Bourdieu, P., Wacquant, L.J.D. (1992) *An Invitation to Reflexive Sociology*. Polity: Cambridge, UK.

Boyd, H. Jr., Levy, S.J., (1963) New Dimensions in Consumer Analysis. *Harvard Business Review*. 41(November-December): 129-140.

Brace-Govan, J., Binay, I. (2010) Consumption of disposed goods for moral identities: a nexus of organization, place, things and consumers. *Journal of Consumer Behaviour*. 9(1), 69-82.

Brace-Govan, J., Itir, B. (2010) Consumption of disposed goods for moral identities: a nexus of organization, place, things and consumers. *Journal of Consumer Behaviour*. 9(1), 69-82.

Brewer, D. (2000) Ethnography. Open University Press: Buckingham, UK.

Brooks, D. (2000) *Bobos in Paradise: The New Upper Class and How They Got There.* Simon and Schuster: New York, USA.

Brown, S., Kozinets R.V., Sherry J.F. Jr. (2003) Teaching Old Brands New Tricks: Retro Branding and the Art of Brand Revival. *Journal of Marketing*. 67(7), 19–33.

Brown, S., Sherry, J.F. Jr. (ed.) (2003) *Time, Space and the Market: Retroscapes Rising*. ME Sharpe: London, UK.

Brownlie, D., Hewer, P. (2007) Prime beef cuts: culinary images for thinking men, *Consumption, Markets and Culture*. 10(3), 229-250.

Brownlie, D., Hewer, P. (2009) Cultures of Unruly Bricolage: Debadging and the Logic of Resistance. *Advances of Consumer Research*. Volume 36, (ed.) McGill A.L., Shavitt S., Duluth MN: 686-687.

Brownlie, D., Hewer, P. (2011) Articulating consumers through practices of vernacular creativity, *Scandinavian Journal of Management*, 27: 243–253.

Brownlie, D., Hewer, P., Treanor, S. (2007) Sociality in motion: exploring logics of tribal consumption among cruisers, in Cova B., Kozinets R.V, Shankar A. (ed.) *Consumer Tribes*. Butterworth-Heinemann: Oxford, UK, 109-28.

Bryman, A., Bell, E. (2007) *Business Research Methods*. Oxford University Press, Oxford, UK.

Bucci, A., Cova, B. (2007) Beyond Marketing Panaceas: In Praise of Societing, in Saren, M., Maclaren P., Goulding C., Elliott, R., Shankar, A., Catterall, M. (ed.) *Critical Marketing*. Elsevier, Oxford, UK: 85–98.

Burgess, R.G. (1982) *Field Research: A Sourcebook and Field Manual*. Routledge: London, UK.

Butler, J. (1990) Gender Trouble. Routledge: London, UK.

Butler, J. (1997) *The Psychic Life of Power: Theories in Subjection*. Stanford University Press: Stanford, CA, USA.

Callon, M. (ed.) (1998a) The Laws of the Markets, *Sociological Review Monographs*. Basil, Blackwell: Oxford, UK.

Callon, M. (1998b) The embeddedness of economic markets in economics, in The Laws of The Markets, *Sociological Review Monographs*, Basil, Blackwell, Oxford, UK: 1-57.

Campbell, C. (1987) *The Romantic Ethic and the Spirit of Modern Consumerism*. Oxford: Blackwell.

Campbell, C. (1995) The Sociology of Consumption, in Miller D (ed.) *Acknowledging Consumption: A Review of New Studies*. Routledge: London, UK: 96-126.

Campbell, C. (1996) The Meaning of Objects and the Meaning of Actions A Critical Note on the Sociology of Consumption and Theories of Clothing. *Journal of Material Culture*. 1(1), 93-105.

Campbell, C. (2004) I shop therefore I know that I am: the metaphysical basis of modern consumerism. *Elusive Consumption*. 27-44.

Campbell, C. (2005) The Craft Consumer: Culture, Craft and Consumption in a postmodern society. *Journal of Consumer Culture*. 5(1), 23-42.

Campbell, C. (1987) *The Romantic Ethic and the Spirit of Modern Consumerism,* Alcuin Academics: UK.

Canniford, R. (2011a) The Engagement of Customers Beyond Their Expected Roles. *Journal of Strategic Marketing*. 19(7), 591-606.

Canniford, R. (2011b) A Typology of Consumption Communities, in Belk, R.W, Grayson, K., Muñiz, A.M., Schau, H.J. (ed.) *Research in Consumer Behavior* (Research in Consumer Behavior, Volume 13), Emerald Group Publishing Limited: 57-75.

Canniford, R., Karababa, E. (2013) Partly primitive: Discursive constructions of the domestic surfer. *Consumption Markets and Culture*. 16(2), 119-144.

Canniford, R., Shankar, A. (2007) Marketing the savage: Appropriating tribal tropes. *Consumer Tribes*. 35-48.

Canniford, R., Shankar, A. (2013) Purifying practices: How consumers assemble romantic experiences of nature. *Journal of Consumer Research*. 39(5), 1051-1069.

Cappellini, B. (2009) The sacrifice of re-use: the travels of leftovers and family relations. *Journal of Consumer Behaviour*. 8, 365–375

Cappellini, B., Parsons, E. (2012) Practising thrift at dinnertime: Mealtime leftovers, sacrifice and family membership. *The Sociological Review*. 60(2), 121-134.

Carroll, B. A., Ahuvia, A.C. (2006) Some antecedents and outcomes of brand love. *Marketing Letters*. 17(2), 79-89.

Carson, D., Gilmore, A., Perry, C., Gronhaug K. (2001) *Qualitative Marketing Research*. Sage: London, UK.

Castells, M. (1996) *The Information Age, Economy, Society, and Culture*. Vol. 1. Blackwell: Oxford, UK.

Celsi, R., Randall, R., Thomas L. (1993) An Exploration of High-Risk Leisure Consumption Through Skydiving. *Journal of Consumer Research*. 20(June), 1–21.

Cervellon, M.C., Carey, L., Harms, T. (2012) Something old, something used: Determinants of women's purchase of vintage fashion vs. second-hand fashion. *International Journal of Retail and Distribution Management*. 40(12), 956-974.

Chaney, D. (2002) *Cultural Change and Everyday Life*. Palgrave Macmillan: Basingstoke, UK.

Chatzidakis, A., Hibbert, S., Mittusis, D., Smith, A. (2004) Virtue in Consumption? *Journal of Marketing Management*. 20(5-6), 526-543.

Cheetham, F. (2009) Out of control? An ethnographic analysis of the disposal of collectable objects through auction. *Journal of Consumer Behaviour*. 8, 316–326

Cheng, S., Olsen, W., Southerton, D., Warde, A. (2007) The changing practice of eating: evidence from UK time diaries, 1975 and 2001. *The British Journal of Sociology*. 58(1), 39–61.

Cherrier, H. (2009) Disposal and simple living: exploring the circulation of goods and the development of sacred consumption. *Journal of Consumer Behaviour*. 8, 327–339.

Chua, B.H. (1992) Shopping for Women's Fashion in Singapore, in Shields R. (ed.) *Lifestyle Shopping: The Subject of Consumption*. Routledge: London, UK.

Clarke, A., Miller, D. (2002) Fashion and Anxiety. Fashion Theory. 6(2), 191-214.

Clifford, J. (1986) 'Introduction: partial truths', in J. Clifford and G. Marcus (eds), *Writing Culture: The Poetics and Politics of Ethnography*. Berkeley. 1(1), 5–16.

Clifford, J., Marcus G.E. (ed.) (1986) *Writing Culture: The Poetics and Politics of Ethnography*. University of California Press: London, UK.

Cochoy, F. (2007) A sociology of market things: on tending the garden of choices in mass retailing, In Callon et al. [eds.] *Market Devices*. Blackwell: Oxford, UK, 109-129.

Cochrane, K. (2010) (online) Why pop-ups pop up everywhere, The Guardian, (October 12, 2010) Available at:

http://www.guardian.co.uk/lifeandstyle/2010/oct/12/pop-up-temporary-shopsrestaurants [accessed on March 12th 2013].

Coffey, A. (1999) *The Ethnographic Self, Fieldwork and Representations of Identity*. Sage: London, UK.

Coffey, A., Atkinson P. (1996) *Making Sense of Qualitative Data*. Sage Publications Inc: London, UK.

Cohen, M.D. (2007) Reading Dewey: Reflections on the Study of Routine. *Organization Studies*. 28(5), 773-786.

Cornett, J.E. (2010) (online), "What is considered vintage clothing?", available at: <u>www.ehow.com/about_6593593_considered-vintage-clothing_.html</u> (accessed 14 April 2012).

Cova, B. (1997) Community and Consumption: Towards a Definition of the 'Linking Value' of Product or Services. *European Journal of Marketing*. 31(3/4), 297–316.

Cova, B., Cova, V. (2002) Tribal Marketing: The Tribalisation of Society and its Impact on the Conduct of Marketing. *European Journal of Marketing*. 36(5/6), 595–620.

Cova B., Dalli, D. (2009) Working consumers: the next step in marketing theory? *Marketing Theory*. 9(3), 315-339.

Cova, B., Elliott, R. (2008) Everything you always wanted to know about interpretive consumer research but were afraid to ask, *Qualitative Market Research: An International Journal*. 11(2), 121 – 129.

Cova, B., Kozinets, R.V, Shankar, A. (ed.) (2007) *Consumer Tribes*. Butterworth-Heinemann: Oxford, UK.

Cova, V., Cova, B. (2001) Alternatives marketing. Dunod: Paris

Cova, B., Salle, R. (2008) Marketing Solutions in Accordance with the S-D Logic: C oCreating Value with Customer Network Actors. *Industrial Marketing Management*. 7(3), 270-277.

Cova, B., Pace, S. (2006) Brand Community of Convenience Products: New Forms of Customer Empowerment - the Case My Nutella the Community. *European Journal of Marketing*. 40(9/10), 1087-105.

Cova, B., White, T. (2010) Counter-brand and alter-brand communities: the impact of Web 2.0 on tribal marketing approaches. *Journal of Marketing Management*. 26(3/4), 256-270.

Coyne, I.T. (1997) Sampling in Qualiative Research. Purposeful and Theoretical Sampling: Merging or Clear Boundaries? *Journal of Advanced Nursing*. 26(3), 623-630.

Crane, D., Bovone, L. (2006) Approaches to material culture: The sociology of fashion and clothing. *Poetics*. 34(6), 319-333.

Creswell, J.W. (2007) *Qualitative Inquiry and Research Design: Choosing Among Five Approaches*. SAGE Publications.

Crewe, L. (2010) Wear: where? The convergent geographies of architecture and fashion. *Environment and planning A*. 42(9), 2093-2108.

Crewe, L. (2011) Geographies of Retailing and Consumption: The Shopping List Compendium, in A Leyshon, R. Lee, L. McDowell, and P. Sunley (ed.), *The SAGE handbook of economic geography*. 304-22. SAGE Publications Ltd: London, UK.

Crewe, L., Beaverstock J. (1998) Fashioning the city: cultures of consumption in contemporary urban spaces. *Geoforum*. 29(3), 287-308.

Crewe, L., Gregson N. (1998) Tales of the unexpected: exploring car boot sales as marginal spaces of contemporary consumption. *Transactions of the Institute of British Geographers*. 23(1), 39-53.

Crewe, L., Gregson N., Brooks K. (2003) The Discursivities of Difference: Retro retailers and the ambiguities of 'the alternative'. *Journal of Consumer Culture*. 3(1), 61-82.

Crockett, D., Wallendorf M. (2004) The Role of Normative Political Ideology in Consumer Behavior. *Journal of Consumer Research*. 31(3), 511–28.

Dalli, D., Corciolani M. (2008) Collective Forms of Resistance: The Transformative Power of Moderate Communities: Evidence from the Book Crossing Case. *International Journal of Market Research*. 50(6), 757-75.

Dant, T. (2004) The driver-car, Theory, Culture and Society. 21(4/5), 61-79.

Dant, T. (2005) Materiality and Society, Open University Press: Maidenhead, UK.

Dant, T. (2008) The "Pragmatics" of Material Interaction. *Journal of Consumer Culture*. 8(1), 11–33.

Davies, A. Elliott, R. (2006) The evolution of the empowered consumer. *European Journal of Marketing*. 40(9/10), 1106-1121.

Davis, F. (1966) Why all of us may be hippies someday. Society. 5(2), 10-18.

Davis, F. (1979) Yearning for Yesterday: A Sociology of Nostalgia. Free Press: New York, USA.

Davis, F. (1992) Fashion, Culture, Identity. University of Chicago Press: Chicago IL.

Debenedetti, A., Oppewal, H., Arsel, Z. (2014) Place Attachment in Commercial Settings: A Gift Economy Perspective. *Journal of Consumer Research*. 40(5), 904-923.

de Burgh-Woodman, H., Brace-Govan J. (2007) We do not live to buy: Why subcultures are different from brand communities and the meaning for marketing discourse. *International Journal of Sociology and Social Policy*. 27 (5/6), 193-207.

De Certeau, M. (1984a) *The Practice of Everyday Life*. University of California Press: Berkeley, CA, USA.

De Certeau, M. (1984b) 'Walking in the City,' *The Practice of Everyday Life*. University of California Press:Berkeley, CA, USA. 91-110.

De Certeau, M. (1995) Cultura No Plural (a). Papirus Editora.

De Certeau M., Jameson F., Lovitt C. (1980) On the Oppositional Practices of Everyday Life. Social Text. 3, 3-43.

Denzin, N.K., Lincoln, Y.S. (2005) *The SAGE Handbook of Qualitative Research* (3rd edition). Sage Publications, Inc.

Deighton, J. (1992) The Consumption of Performance. *Journal of Consumer Research*. 19(3), 362-372.

Delhaye, C. (2006) The development of consumption culture and the individualization of female identity. *Journal of Consumer Culture*. 6(1), 87-115.

DeLong, M., Heinemann, B., Reiley, K. (2005) Hooked on Vintage! *Fashion Theory*. 9(1), 23-42.

Denegri-Knott, J., Molesworth, M. (2009) 'I'll sell this and I'll buy them that': eBay and the management of possessions as stock. *Journal of Consumer Behaviour*. 8(6), 305-315.

Denegri-Knott, J., Molesworth, M. (2010) Concepts and Practices of Digital Virtual Consumption. *Consumption Markets and Culture*. 13(2), 109–32.

Denegri-Knott, J. Parsons, E. (2014) Disordering Things. *Journal of Consumer Behaviour*. 13(2), 89-98.

Denegri-Knott, J., Zwick, D., Schroeder, J. (2006) Mapping Consumer Power: Integrative Framework for Marketing and Consumer Research, *European Journal of Marketing*. 40(9): 950-71.

Denzin, N.K. (1970) *The Research Act in Sociology: A Theoretical Introduction to Sociological Methods*. Butterworths: London, UK.

Denzin, N.K (1989) *The Research Act in Sociology: A Theoretical Introduction to Sociological Methods*. Prentice Hall: London, UK.

Denzin, N.K. (1997) *Interpretive ethnography: Ethnographic practices for the 21st century*. Sage: Thousand Oaks, CA, USA.

Denzin, N.K., Lincoln, Y.S (ed.) (1998a) *The landscape of qualitative research: Theories and Issue*. Sage Publications: London, UK.

Denzin, N., Lincoln Y. (1998b) Introduction: Entering the field of qualitative research. In N. Denzin and Y. Lincoln (ed.) *Handbook of qualitative research* (1-17), Sage: Thousand Oaks, CA, USA.

Denzin N.K., Lincoln, Y.S. (ed.) (1999) *The Sage Handbook of Qualitative Research* (2nd Edition). Sage Publications: London, UK.

Denzin, N.K, Lincoln, Y.S (ed.) (2004) *The Sage Handbook of Qualitative Research,* Third Edition, Sage Publications, London, UK.

de Valck, K., van Bruggenb, G.H., Wierenga, B. (2009) Virtual communities: A marketing perspective. *Decision Support Systems*. 47(3), 185–203.

Diamond, N., Sherry, J.F., Muniz, Jr A.M., McGrath, M.A., Kozinets, R.V., Borghini, S. (2009) American Girl and the Brand Gestalt: Closing the loop on sociocultural branding research. *Journal of Marketing*. 73(3), 118-134.

Dittmar, H. (1992) *The Social Psychology of Material Possessions*. Harvester Wheatsheaf: Hemel Hempstead, UK.

Douglas, M. (1966) *Purity and Danger-an analysis of the concepts of pollution and taboo.* Routledge and Kegan Paul: London, UK.

Douglas, M. (1973) *Rules and meaning: The anthropology of everyday knowledge*. Penguin Books: UK.

Douglas, M. (1984) *Purity and Danger: An Analysis of the Concept of Pollution and Taboo.* Routledge: London, UK.

Douglas, M., Isherwood, B. (1979) The World of Goods. Allen Lane: London, UK.

Douglas, M., Isherwood, B. (1980) *The World of Goods*. Penguin, Harmondsworth: UK.

Douglas, M. (1986) How Institutions Think. Syracuse University Press.

Dourish, P. (2001) *Where the Action Is: The Foundations of Embodied Interaction*. MIT Press: Cambridge, MA, USA.

du Gay, P. (1997) *Production of Culture/Cultures of Production*. Sage Publications: London.

du Gay, P., Hall, S., James, L., Mackay, H., Negus, K. (1997) *Doing Cultural Studies: The Story of Sony Walkman.* Sage: London.

Dubin T., Berman A.E. (2000) *Vintage Style: Buying and Wearing Classic Vintage Clothes.* Harper Collins, New York, USA.

Dyer, B., Burnsed, K.A., Dyer, C.L. (2006) Ethnicity and household expenditures: furnishings, fashion, and flux? *Family and Consumer Sciences Research Journal*. 35(2), 131-159.

Earley, A. (2014) Connecting contexts: A Badiouian epistemology for consumer culture theory. *Marketing Theory*. 14(1), 73-96.

Eckhardt, G.M., Bengtsson, A. (2010) A Brief History of Branding in China. *Journal of Macromarketing*. 30(3), 210-221.

Elkjær, B., Simpson, B. (2011) Pragmatism: A lived and living philosophy: what can it offer to contemporary organization theory? *Research in the Sociology of Organizations*. 32, 55-84.

Elliott, R. (1996) Discourse Analysis: Exploring Action, Function and Conflict in Social Texts. *Marketing Intelligence and Planning*. 14(6), 65-8.

Elliott, R. (1997) Existential consumption and irrational desire. *European Journal of Marketing*. 31(3/4), 285-296.

Elliott, R., Davies A. (2006) Symbolic brands and authenticity of identity performance. *Brand Culture*. 155-70.

Elliott, R., Jankel-Elliott, N. (2003) Using ethnography in strategic consumer research, *Qualitative Market Research: An International Journal*. 6(4), 215-223.

Emmerson, F., Fretz, R.I., Shaw, L.L (1995) *Writing Ethnographic Fieldnotes*, University of Chicago Press: Chicago, USA.

Entwistle, J. (2000a) *The Fashioned Body, Fashion Dress and Modern Social Theory*. Polity Press: Cambridge, UK.

Entwistle, J. (2000b) *Fashion and the Fleshy Body: Dress as Embodied Practice, Fashion Theory*. 4(3), 323-348.

Entwistle, J. (2009) *The Aesthetic Economy of Fashion: Markets and Values in Clothing and Modelling*. Berg Publishers: Oxford, UK.

Entwistle, J., Rocamora, A. (2006) The Field of Fashion Materialized: A Study of London Fashion Week. *Sociology*. 40(4), 735–51.

Entwistle, J., Slater, D. (2014) Reassembling the cultural: fashion models, brands and the meaning of 'culture' after ANT. *Journal of Cultural Economy*. 7(2), 161-177.

Entwistle, V.A. (2010) Involving service users in qualitative analysis: approaches and assessment. *Health Expectations*. 13, 111–112.

Epp, A.M., Price, L.L. (2010) The Storied Life of Singularized Objects: Forces of Agency and Networked Transformation. *Journal of Consumer Research*. 36(5), 820-837.

Etgar, M. (2008) A Descriptive Model of the Consumer Co-production Process, *Journal of the Academy of Marketing Science*. 36(1), 97-108.

Evans, D. (2012) Beyond the throwaway society: ordinary domestic practice and a sociological approach to household food waste. *Sociology*. 46(1), 41-56.

Evans, D., Campbell, H., Murcott, A. (2013) *Waste matters: new perspectives of food and society*. Sociological Review Monograph Series. Wiley-Blackwell: UK.

Evans, M. (1999) Consumer Behaviour towards Fashion. *European Journal of Marketing*. 23(7), 7-15.

Falk, P., Campbell, C. (1997) *The Shopping Experience*. Sage Publications: London, UK.

Faurschou, G. (1987) Fashion and the cultural logic of postmodernity. *Canadian Journal of Political and Social Theory*. 11(1-2), 68-82.

Featherstone, M. (1990) Global Culture: An Introduction. *Theory, Culture and Society*. 7(1), 1-14.

Featherstone, M. (1991) Consumer Culture and Postmodernism. Sage: London, UK.

Feldman, M.S., Orlikowski, W.J. (2011) Theorizing Practice and Practicing Theory. *Organization Science*. 22(5), 1240–53.

Fernandez, K.V., Lastovicka, J.L. (2011) Making Magic: Fetishes in Contemporary Consumption. *Journal of Consumer Research*. 38(2), 1-23.

Fine, G.A., Deegan, J. (1996) Three principles of serendipity: insight, chance, and discovery in qualitative research. *Qualitative studies in education*. 9(4), 434-447.

Finch, J., Geiger, S. (2010) Positioning and relating: Market boundaries and the slippery identity of the marketing object. *Marketing Theory*. 10(2), 37.

Firat, F.A. (1997) Globalization of fragmentation - A framework for understanding contemporary global markets. *Journal of International Marketing*. 5(2), 77-86.

Firat, F.A. (1999) Rethinking consumption. *Consumption, Markets and Culture*. 3(4), 283-295.

Firat, F.A., Dholakia, N. (1998) Consuming People: From Political Economy to Theatres of Consumption. Routledge: New York, USA.

Firat, F.A., Venkatesh, A. (1995) Liberatory Postmodernism and the Reenchantment of Consumption. *Journal of Consumer Research*. 22(3), 239-267.

Fischer, E., Arnold, S.J. (1990) More Than a Labor of Love: Gender Roles and Christmas Gift Shopping. *Journal of Consumer Research*.17(3), 333–45.

Fischer, E. (2009) Are we there yet? Co-producing succeed and failure in a "consumer-intensive" service context. In Sherry, J. and Fischer, E., *Exploration in consumer culture theory*. Routledge: New York.

Fiske, J. (1989) Understanding Popular Culture. Unwin Hyman: Boston, USA.

Flick, U. (2002) *Qualitative Research - State of the Art*. Social Science Information. 41(1), 5-24.

Fonarow, W. (2006) *Empire of Dirt: The Aesthetics and Rituals of British Indie Music.* Wesleyan University Press: Middletown, CT.

Fortune, J. (2012) Is vintage clothing passé? *The Guardian*, published on 27th January 2012.

Foucault, M. (1970) *The Order of Things: An Archaeology of the Human Sciences*. Pantheon Books.

Foucault, M. (1972) The Archaeology of Knowledge. Tavistock: London, UK.

Foucault, M. (1980) *Power/Knowledge: Selected Interviews and Other Writings* 1972-1977 (ed.) C. Gordon, Harvester Press: Brighton, UK.

Foucault, M. (1984) *Power/Knowledge: Selected Interviews and Other Writings:* 1972-1977, (ed.) C. Gordon. Pantheon: New York, USA.

Foucault, M. (1991) Questions of Method, in *The Foucault Effect: Studies in Governmentality*, (Ed.) G. Burchell, C. Gordon, and P. Miller, Hemel Harvester Wheatsheaf: Hempstead, UK: 73–86.

Foucault, M. (1969) L'arche ologie du savoir, Paris: Gallimard. In: A. Sheridan (Trans.). *The archaeology of knowledge*. Routledge: London, UK.

Fournier, S., Lee, L. (2009) Getting brand communities right. *Harvard Business Review*. 87(4), 105-111.

Frank, T. (1997) *The Conquest of Cool: Business Culture, Counterculture, and the Rise of Hip Consumerism.* University of Chicago Press: Chicago, USA.

Frenzen, J.K., Davis, H.L., (1990) Purchasing Behaviour in Embedded Markets. *Journal of Consumer Research*. 17(1), 1-12.

Friedrichs, J., Kratochwill, F. (2009) On Acting and Knowing: How Pragmatism Can Advance International Relations Research and Methodology. *International Organization*. 63(4), 701-31.

Frith, S. (1996) Music and identity (Chapter 7), *Questions of cultural identity*. SAGE Publications Ltd. 108-27.

Frith, S. (1998) *Performing rites: On the value of popular music*. Harvard University Press: Boston, MA.

Gabriel, Y., Lang, T. (1995) *The Unmanageable Consumer: Contemporary Consumption and its Fragmentation.* SAGE: London, UK.

Gabriel, Y., Lang, T. (2006) The unmanageable consumer. Sage.

Gabriel, Y., Lang, T. (2008) New Faces and New Masks of Today's Consumer. *Journal of Consumer Culture*. 8(3), 321-340.

Geertz, C. (1973) The Interpretation of Cultures. Fontana Press: New York, USA.

Geertz, C. (1975) On the Nature of Anthropological Understanding: Not Extraordinary empathy but readily observable symbolic forms enable the anthropologist to grasp the unarticulated concepts that inform the lives and cultures of other peoples. *American Scientist*. 63(1), 47-53.

Geertz, C. (1983) *Local Knowledge: Further Essays in Interpretive Anthropology*. Basic Books: New York, USA.

Geertz, C. (1984) Distinguished Lecture: Anti Anti-Relativism. *American Anthropologist.* 86(2), pp. 263-278.

Geertz, C. (1988) *Works and Lives, The Anthropologist as Author*. Stanford University Press: Stanford, USA.

Geertz, C. (1994) Thick Description: Toward an Interpretive Theory of Culture, in *Readings in the Philosophy of Social Science*. (ed.) M. Martin, L.C. McIntyre. Massachusetts Institute of Technology. 213-231.

Geiger, S., Kjellberg, H., Spencer, R. (2012) Shaping Exchanges, Building Markets, *Consumption Markets and Culture*. 15(2), 133-147.

Gelder, K., Thornton, S. (1997) The Subcultures Reader. Routledge: London, UK.

Ger, G. (2005) *Reflections on Consumer Culture Research*. Paper presented at ACR Doctoral Symposium. San Antonio: September.

Ger, G., Askegaard, S., Christensen, A. (1999) Experiential nature of product-place images: Image as a narrative. In *Advances in consumer research*, edited by E.J. Arnould and L.M. Scott, 165-69, vol. XXVI. Association for Consumer Research: Provo, UT.

Ger, G., Belk, R.W. (1996a) Cross-Cultural Differences in Materialism. *Journal of Economic Psychology*. 17(1), 55–7.

Ger, G., Belk, R.W. (1996b) I'd like to buy the world a coke: Consumptionscapes of the less affluent world. *Journal of Consumer Policy*. 19(3), 271-304.

Ger, G., Wilk, R. (2005) Religious Material Culture: Morality, Modernity, and Aesthetics. *Advances in Consumer Research*. 32, 80-81.

Gerval, O. (2008) Fashion: Concept to Catwalk. Bloomsbury: London, UK.

Giddens, A. (1979) Central Problems in Social Theory: Action, Structure and Contradiction in Social Analysis. University of California Press: Berkeley, USA.

Giddens, A. (1984) *The Constitution of Society: Outline of the Theory of Structuration.* Polity Press: Cambridge, UK.

Giddens, A. (1990) *The consequences of modernity*. Stanford University Press: Stanford, CA.

Giesler, M. (2006) Consumer Gift Systems. *Journal of Consumer Research*. 33(2), 282-90.

Giesler, M. (2008) Conflict and Compromise: Drama in Marketplace Evolution. *Journal of Consumer Research*. 34(6), 739-753.

Giesler, M. (2012) How Doppelgänger Brand Images Influence the Market Creation Process: Longitudinal Insights from the Rise of Botox Cosmetic. *Journal of Marketing*. 76(6), 55-68.

Gilbert, N. (ed.) (1993) Researching Social Life. Sage: London, UK.

Gladigau, K. (2008) Shop till you drop: youth, distinction and identity in vintage clothing. available at: <u>www.tasa.org.au</u> (accessed 22 February 2011).

Goffman, E. (1959) *The Presentation of Self in Everyday Life*. Doubleday Dell Publishing Group Inc: New York, USA.

Goffman, E. (1986a) Frame Analysis. Northeastern University Press: Boston, MA.

Goffman, E. (1986b) *Stigma: Notes on the Management of Spoiled Identity*. Touchstone: New York, USA.

Golden-Biddle, K., Locke, K. (1993) Appealing work: An investigation of how ethnographic texts convince. *Organization Science*. 4(4), 595-616.

Goss, J. (1993) The magic of the Mall: an analysis of form, function and meaning in the contemporary built environment. *Annals of the Association of American Geographers*. 83(1), 18-47.

Gould, S.J. (2006) Unpacking the Many Faces of Introspective Consciousness: A Metacognitive-Poststructuralist Exercise. In R.W. Belk (ed.) *Handbook of Qualitative Methods in Marketing*. Cheltenham: Edward Elgar. 186-97.

Goulding, C. (1998) Consumer Research, interpretive paradigms and methodological ambiguities. *European Journal of Marketing*. 33(9/10), 859-873.

Goulding, C. (2000) Dance Culture and Consumer Behaviour: Can Marketers Learn Lessons from the 'rave' Generations? *Wolverhampton Business School Conference Proceedings*, Management Research Centre.

Goulding, C. (2002) An Exploratory Study of Age Related Vicarious Nostalgia and Aesthetic Consumption. *Advances in Consumer Research*. 29, 542-546.

Goulding, C. (2003) Corsets, Silk Stockings, and Evening Suits. In Brown, S., Sherry, J.F. Jr. (ed.) *Time, Space and the Market: Retroscapes Rising*. ME Sharpe: London, UK.

Goulding, C. (2005) Grounded theory, ethnography and phenomenology: a comparative analysis of three qualitative strategies for marketing research. *European Journal of Marketing*. 39(3/4), 294-308.

Goulding, C., Follett, J., Saren, M., MacLaren, P. (2004) Process and Meaning in 'Getting a Tattoo'. In NA - *Advances in Consumer Research*, Volume 31, (ed.) Barbara E. Kahn and Mary Frances Luce, Valdosta, GA: Association for Consumer Research. 279-284.

Goulding, C., Saren, M. (2009) Performing Identity: an Analysis of Gender Expressions at the Whitby Goth Festival. *Consumption Markets and Culture*. 12(1), 27-46.

Goulding, C., Saren, M. (2007) Gothic Entrepreneurs: a Study of the Subcultural Commodification Process. In *Consumer Tribes*. (Ed) Cova, B., Kozinets, R.V., Shanker, A. Elsevier: Oxford. 227-242.

Goulding, C., Saren, M., Follett, J. (2004a) Virtue in Darkness: A Subculture of Consumption from the Margins to the Mainstream. *In Proceedings of the Academy of Marketing Conference*. University of Gloucestershire Business School. 6.

Goulding, C., Saren, M., Maclaran, P., Follett, J. (2004b) Into the darkness: Androgyny and gender blurring within the gothic subculture. *In Proceedings of the ACR Gender, Marketing and Consumer Behavior Conference*.

Goulding, C., Shankar, A., Elliott, R., Canniford, R. (2009) The marketplace management of illicit pleasure. *Journal of Consumer Research*. 35(5), 759-771.

Goulding, C., Shankar, A. (2011) Club culture, neo-tribalism and ritualised behaviour. *Annals of Tourism Research*. 38(4), 1435-1453.

Goulding, C., Shankar, A., Canniford, R. (2013) Learning to be Tribal: Facilitating the Formation of Consumer Tribes. *European Journal of Marketing*. 47(5/6), 813-832.

Goulding, C., Shankar, A., Elliot, R. (2002) Working weeks, rave weekends: Identity Fragmentation and the Emergence of New Communities. *Consumption, Markets and Culture*. 5, 261-284.

Goulding C., Shankar A., Elliott R., (2001) Dance Clubs, Rave, and the Consumer Experience: An Exploratory Study of a Sub-cultural phenomenon. *European Advances in Consumer Research*. 5, 203-208.

Graeber, D. (2001) *Toward an Anthropological Theory of Value: The False Coin of Our Own Dreams*. Palgrave: New York, NY.

Graeber, D. (2005) Value: Anthropological Theories of Value, in J.G. Carrier (ed.) *A Handbook of Economic Anthropology*. Edward Elgar: Cheltenham, UK and Northampton, MA, 439–54.

Grainge, P. (2000) Nostalgia and style in retro America: Moods, modes, and media recycling. *Journal of American and Comparative Cultures*. 23(1), 27-34.

Granny Would be Proud Fairs Facebook [online] https://www.facebook.com/GrannyProud?fref=ts

Granny Would be Proud Market Twitter [online] https://twitter.com/GrannyProud

Grayson, K., Johnson, D., Der-Fa, C. (2008) Is Firm Trust Essential in a Trusted Environment? How Trust in the Business Context Influences Customers. *Journal of Marketing Research*. 45(2), 241–56.

Green, K., Randles, S. (2006) *Industrial ecology and spaces of innovation*. Edward Elgar Publishing Ltd: Cheltenham UK.

Gregson, N., Crewe, L. (1997) The bargain, the knowledge and the spectacle: making sense of consumption in the space of the car boot sale. *Environment and Planning D.* 15(1), 87-112.

Gregson, N., Crewe, L. (1998) Tales of the unexpected: exploring car boot sales as marginal spaces of contemporary consumption. *Transactions of the Institute of British Geographers*. 23(1), 39-53.

Gregson, N., Crewe, L. (2002) Second-Hand Cultures. Berg Publishers: Oxford, UK.

Gregson, N., Metcalfe, A., Crewe, L. (2007) Moving things along: the conduits and practice of disinvestment in consumption. *Transactions of the IBG*. 32, 187-200.

Gregson, N., Metcalfe, A., Crewe L. (2009) Practices of object maintenance and repair. *Journal of Consumer Culture*. 9(2), 248–272.

Gronow, J., Warde, A. (2001) Ordinary Consumption. Routledge: London, UK.

Grönroos, C. (2006) Adopting a Service Logic for Marketing. *Marketing Theory*. 6(3), 317-33.

Grönroos C. (2008) Service Logic Revisited: Who Creates Value? And Who Co-Creates? *European Business Review*. 20 (4): 298-314.

Grönroos, C. (2012) Conceptualising Value Co-creation: A Journey to the 1970s and Back to the Future. *Journal of Marketing Management*. 28(13/14). 1520–34.

Gronow, J., Warde, A. (2001) Ordinary Consumption. London: Routledge.

Guba, E.G. (1990) (ed.) The paradigm dialog. Sage.

Guba, E.G., Lincoln, Y.S. (1998) Competing paradigms in qualitative research. In *The Landscape of Qualitative Research* (Denzin N.K., Lincoln Y.S., ed.), Sage: Thousand Oaks, CA: 195–222.

Guiot, D., Roux, D. (2010) A second-hand shoppers' motivation scale: antecedents, consequences and implications for retailers. *Journal of Retailing*. 86(4), 355-71.

Gummerus, J. (2013) Value Creation Processes and Value Outcomes in Marketing Theory: Strangers or Siblings? *Marketing Theory*. 13(1), 19–46.

Gummeson, E. (2008) Extending the Service-Dominant Logic: From Customer Centricity to Balanced Centricity. *Journal of the Academy of Marketing Science*. 36(1), 15–7.

Gummesson E. (1998) Productivity, quality and relationship marketing in service operations. *International Journal of Contemporary Hospitality Management*. 10(1), 4-15.

Halkier, B. (2001) Routinisation or reflexivity? Consumers and normative claims for environmental considerations. In *Ordinary Consumption*, Gronow, J., Warde, A. (ed.). Routledge: London, UK. 25–44.

Halkier, B. (2010) *Consumption Challenged. Food in Medialised Everyday Lives.* Ashgate: Farnham, UK.

Halkier, B., Jensen, I. (2011) Methodological challenges in using practice theory in consumption research: examples from a study on handling nutritional contestations of food consumption. *Journal of Consumer Culture*. 11, 101.

Halkier, B., Katz-Gerro, T., Martens, L. (2011) Applying Practice Theory to the Study of Consumption: Theoretical and Methodological Considerations. *Journal of Consumer Culture*. 11(1), 3–13.

Hall, J. (1992) The Capital(s) of Culture, in *Cultivating Differences: Symbolic Boundaries and the Making of Inequality*. (ed.) Michele Lamont and Marcel Fournier, Chicago: University of Chicago Press. 257–83.

Hall, S. (2004) Who Needs "Identity?" in du Gay, P., Evans, J. and Redman P. (ed.) *Identity: A Reader*. Sage: London. 15-30.

Hall, S., Jefferson, T. (1993) *Resistance through rituals: Youth subcultures in postwar Britain.* Psychology Press: Oxon, UK.

Hamilton, K., Hewer, P. (2009) Salsa magic: An exploratory netnographic analysis of the salsa experience. *Advances in Consumer Research*. 36, 502-508.

Hamilton, K., Hewer, P. (2010) Tribal mattering spaces: Social-networking sites, celebrity affiliations, and tribal innovations. *Journal of Marketing Management*. 26(3/4), 271-289.

Hammersley, M. (1990) *Reading Ethnographic Research: A Critical Guide*, New York: Longman.

Hammersley, M., Atkinson, P. (1995) *Ethnography: Principles in Practice*. Psychology Press.

Hammersley, M. (1995) The Politics of Social Research. Sage: London, UK.

Hammersley, M. (1998) Reading Ethnographic Research (2nd edition). London, UK.

Hammersley, M., Atkinson P., (1983) *Ethnography: Principles in practice*. Tavistock: London.

Hand, M., Shove, E. (2007) Condensing Practices: Ways of Living with a Freezer. *Journal of Consumer Culture*. 7(1), 79-104.

Hand, M., Shove, E., Southerton, D. (2005) Explaining Showering: a Discussion of the Material, Conventional and Temporal Dimensions of Practice. *Sociological Research Online*. 10(2), 22.

Handelman, J.M., Arnold, S.J. (1999) The Role of Marketing Actions with a Social Dimension: Appeals to the Institutional Environment. *Journal of Marketing*. 63(3), 33–48.

Hanna, J. L. (1979) Towards a cross-cultural conceptualization of dance and some correlate considerations. *The performing arts: Music and dance*. 17-45.

Hardy, C., Phillips, N., Clegg, S. (2001) Reflexivity in organization and management theory: A study of the production of the research 'subject'. *Human Relations*. 54, 531-560.

Hargreaves, T. (2011) Practicing Behavior Change: Applying Social Practice Theory to Pro-environmental Behavior Change. *Journal of Consumer Culture*. 11(1), 79–99.

Harrison, N. (2012) [online] Data: UK retail regions ranked by spend, (June 29th, 2012) Available at <u>http://www.retail-week.com/property/data-uk-retail-regions-ranked-by-spend/5038118.article</u> [accessed on May 28th 2014].

Hartmann, B.J., Wiertz, C., Arnould, E.J. (2011) Practice Consumption and Value Creation: Advancing the Practice Theoretical Ontology of Consumption Community, *Advances in Consumer Research*. 39, 519-520.

Hebdige, D. (1997/1979) Subcultures: The Meaning of Style in Gelder K. and Thornton S. (Ed.) *The Subcultures Reader*. Routledge: London, UK.

Hebdige, D. (1979) Subculture: The Meaning of Style. Methuen: London, UK.

Hebdige, D. (1988) *Hiding in the Light: On Images and Things*. Routledge: London, UK.

Hennion, A. (2004) Pragmatics of Taste, in *The Blackwell Companion to the Sociology of Culture,* (ed.) M. Jacobs and N. Hanrahan, Oxford: Blackwell: 131–44.

Herrmann, G. M. (1996) Women's exchange in the US garage sale: Giving Gifts and Creating Community. *Gender and Society*. 10(6), 703-728.

Hetherington, K. (2004) Secondhandedness: consumption, disposal and absent presence. *Environment and Planning D: Society and Space*. 22, 157–173.

Hewer, P., Hamilton, K. (2010) On Emotions and Salsa: some thoughts on dancing to rethink consumers. *Journal of Consumer Behaviour*. 9, 113-125.

Hill, R. (1991) Homeless Women, Special Possessions, and the Meaning of Home: An Ethnographic Case Study. *Journal of Consumer Research*. 18(3), 298-310.

Hirschman, E.C. (1992) The Consciousness of Addiction: Toward a General Theory of Compulsive Consumption. *Journal of Consumer Research*. 19(2), 155-79.

Hirschman, E.C. (1994) Consumers and Their Animal Companions. *Journal of Consumer Research*. 20(4), 616-632.

Hirschman, E.C., Holbrook, M. (1982) Hedonic Consumption: Emerging Concepts, Methods, and Propositions. *Journal of Marketing*. 46 (3), 92–101.
Hirschman, E.C., Stern, B.B. (1998) Consumer behavior and the wayward mind: the influence of mania and depression on consumption. *Advances in Consumer Research*. 25, 421–427.

Hogg, M.K., Maclaran, P. (2008) Rhetorical issues in writing interpretivist consumer research. *Qualitative Market Research: An International Journal*. 11(2), 130-146.

Holbrook, M.B. (1985) Why business is bad for consumer research: the three bears revisited. *Advances in Consumer Research*. 12(1), 145.

Holbrook, M.B. (2006) Consumption experience, customer value, and subjective personal introspection: An illustrative photographic essay. *Journal of Business Research*. 59(6), 714-725.

Holbrook, M.B. (1986) The Role of Emotion in the Consumption Experience: Actions and Reactions in Consumer Behavior. In NA - *Advances in Consumer Research*, Volume 13, (ed.) Richard J. Lutz, Provo, UT: Association for Consumer Research: 666.

Holbrook, M.B. (1999a) Introduction to Consumer Value, in M.B. Holbrook (ed.) *Consumer Value: A Framework for Analysis and Research*, Routledge, London, UK: 1-28.

Holbrook, M.B. (1999b) Higher than the bottom line: Reflections on some recent macromarketing literature. *Journal of Macromarketing*. 19(1), 48-74.

Holbrook, M.B. (1987) What is Consumer Research? *Journal of Consumer Research*. 14(1), 128-132.

Holbrook, M.B., Hirschman, E. (1982) The Experiential Aspects of Consumption: Consumer Fantasies, Feelings, and Fun. *Journal of Consumer Research*. 9(2), 132-140.

Holbrook, M.B., O'Shaughnessy, J. (1988) On the Scientific Status of Consumer Research and the Need for an Interpretive Approach to Studying Consumption Behaviour. *Journal of Consumer Research*. 15(3), 398-402.

Holbrook, M.B., Schindler, R.M. (1991) Echoes of the Dear Departed Past: Some Work in Progress On Nostalgia. *Advances in Consumer Research*. 18(1), 330-333.

Holbrook, M.B., Schindler, R.M., (1994) Age, sex and attitude toward the past as predictors of consumer's aesthetic tastes for cultural products. *Journal of Marketing Research*. 31(3), 412-422.

Holt, D.B. (1995) How Consumers Consume: A Typology of Consumption Practices. *Journal of Consumer Research*. 22(1), 1-16.

Holt, D.B. (1997) Poststructuralist Lifestyle Analysis: Conceptualizing the Social Patterning of Consumption in Postmodernity. *Journal of Consumer Research*. 23(4), 326-350.

Holt, D.B. (1998) Does Cultural Capital Structure American Consumption? *Journal of Consumer Research*. 25(1), 1-25.

Holt, D.B. (2002) Why Do Brands Cause Trouble? A Dialectical Theory of Consumer Culture and Branding. *Journal of Consumer Research*. 29(1), 70-91.

Holt, D.B. (2004) *How Brands Become Icons: The Principles of Cultural Branding*. Harvard Business School Press: Boston, Mass., USA.

Holt, D. B. (2005) How societies desire brands: using cultural theory to explain brand symbolism. *Inside Consumption*. Routledge. 273-291.

Holt, D.B. (2006) Jack Daniel's America: Iconic brands as Ideological Parasites and Proselytizers. *Journal of Consumer Culture*. 6(3), 355–77.

Holt, D.B., Thompson, C.J. (2004) Man-of-Action Heroes: The Pursuit of Heroic Masculinity in Everyday Consumption. *Journal of Consumer Research*. 31(2), 425-40.

Holt, R. (2008) Phenomenology. In: R. Thorpe and R. Holt ed(s). *Sage dictionary of qualitative management research*. Sage: London, UK.

Horne, S., Broadbridge, A. (1995) Charity shops: a classification by merchandise mix. *International Journal of Retail and Distribution Management*. 23(7), 17-23.

Hoskins, J. (1998) *Biographical Objects: How Things Tell the Stories of People's Lives*. Routledge: New York, USA.

Hudson, L.A., Ozanne, J, L. (1988) Alternative Ways of Seeking Knowledge in Consumer Research. *Journal of Consumer Research*. 14(4), 508-521.

Hughes, C. (2007) The equality of social envies. Sociology. 41(2), 347-363.

Hui, A. (2012) Things in motion, things in practices: how mobile practice networks facilitate the travel and use of leisure objects. *Journal of Consumer Culture*. 12(2), 195-215.

Humphrey, C., Hugh-Jones, S. (1992) Introduction: Barter Exchange and Value, in C. Humphrey and S. Hugh-Jones (ed.) *Barter, Exchange and Value: An Anthropological Approach*. Cambridge University Press: Cambridge, UK. 1-20.

Humphreys, A. (2010a) Megamarketing: The Creation of Markets as a Social Process. *Journal of Marketing*. 74(2), 1–19.

Humphreys, A. (2010b) Semiotic Structure and the Legitimation of Consumption Practices: The Case of Casino Gambling. *Journal of Consumer Research*. 37(3), 490-510.

Humphreys, A., Grayson, K. (2008) The Intersecting Roles of Consumer and Producer: A Critical Perspective on Co-Production, Co-Creation and Prosumption. *Sociology Compass.* 2(3), 1-18.

Humphreys, M., Brown, A.D., Hatch, M.J. (2003) Is Ethnography Jazz? *Organisation*. 10(1), 5-31.

Hurdley, R. (2006) Dismantling Mantelpieces: Narrating Identities and Materializing Culture in the Home. *Sociology*. 40(4), 717-733.

Hurdley, R., Dicks, B. (2011) In-between practice: working in the 'thirdspace' of sensory and multimodal methodology. *Qualitative Research*. 11(3), 277 – 292.

Husserl, E. (1946) Phenomenology: its elementary forms.

Hyde, L. (2006) *The Gift: How the Creative Spirit Transforms the World*. Canongate Books Ltd: Edinburgh, UK.

Illouz, E. (2007) *Cold Intimacies: The Making of Emotional Capitalism*. Malden Polity Press: Cambridge, UK.

Illouz, E. (2009) Emotions, imaginations and consumption, *Journal of Consumer Research*. 9(3), 377-413.

Ilmonen, K. (2001) Sociology, Consumption, and Routine, in *Ordinary Consumption,* (ed.) J. Gronow and A. Warde. Routledge: New York, USA.

Iverson, E. (2010) Millennial perspective: vintage fashion and the twice-around economy. Available at: <u>www.newgeography.com</u> (accessed 18 March 2012).

Izberk-Bilgin, E. (2010) An interdisciplinary review of resistance to consumption, some marketing interpretations, and future research suggestions. *Consumption, Markets and Culture*. 13(3), 299-323.

Jaakkola, E. (2011) Unraveling the Practices of "Productization" in Professional Service Firms. *Scandinavian Journal of Management*. 27(2), 221–30.

Jackson, T., Shaw, D. (2008) *Mastering fashion marketing*. Palgrave Macmillan: London, UK.

Jalas, M. (2009) *Making time: reciprocal object relations and the self-legitimizing time of wooden boating.* Berg: Oxford, UK. 203-216.

Jameson, F. (1984) Postmodernism or the cultural logic of late capitalism. *New Left Review*. 146, 53-93.

Jameson, F. (1985) Postmodernism and consumer society. *Postmodern Culture*. 111-25.

Jefferson, T. (1976) Cultural Responses of the Teds: The Defense of Space and Status, in Hall and Jefferson (ed.), *Resistance through Rituals: Youth Subcultures in Post-War Britain*. Hutchinson: London, UK.

Jenss, H. (2004) Dressed in History: Retro Styles and the Construction of Authenticity in Youth Culture. *Fashion Theory*. 8(4), 387-404.

Jenss, H. (2005) Sixties Dress Only! in Palmer Alexandra, Clark Hazel (ed.) *Old Clothes, New Looks: Second-hand Fashion*. Berg Publishers: Oxford, UK.

Johnson, G., Melin, L., Whittington, R. (2003) Micro strategy and strategizing: Towards an activity-based view. *Journal of Management Studies*. 40(1), 3–22.

Johnson, J.M. (2002) In-depth interviewing in Gubrium Jaber F., Holstein James A. (ed) *Handbook of Interviewing Research*. Sage: Thousand Oaks, CA. 103-119.

Jones, J.S., Watt, S. (2010) *Ethnography in social science practice*. Routledge, London, UK.

Jorgensen, D.L. (1989) *Participant observation: A methodology for human studies* (Vol. 15). Sage: Thousand Oaks, CA.

Joy, A.S., Sherry, J.F. (2003) Speaking of art as embodied imagination: a multisensory approach in understanding aesthetic experience. *Journal of Consumer Research*. 28(9), 239–256.

Judy's Vintage Fair - Twitter [online] https://twitter.com/JudyVintageFair

Judy's Vintage Fair [online] <u>http://www.judysvintagefair.co.uk/</u> (accessed on May 26th 2012)

Judy's Vintage Fair Facebook [online] https://www.facebook.com/JudysAffordableVintageFair

Kaeppler, A.L. (1985) Structured movement systems in Tonga. *Society and the Dance*. 92-118.

Kalaignanam, K., Varadarajan, R. (2006) Customers as co-producers. *The service-dominant logic of marketing: dialog, debate, and directions*. 166-79.

Karababa, E., Kjeldgaard, D. (2014) Value in marketing: Toward sociocultural perspectives. *Marketing Theory*. 14(1), 119-127.

Kates, S.M. (2002) The Protean Quality of Subcultural Consumption: An Ethnographic Account of Gay Consumers. *Journal of Consumer Research*. 29(3), 383–99.

Kates, S.M. (2004) The Dynamics of Brand Legitimacy: An Interpretive Study in the Gay Men's Community. *Journal of Consumer Research*. 31(2), 455–65.

Keller, M., Halkier, B. (2014) Positioning consumption: A practice theoretical approach to contested consumption and media discourse. *Marketing Theory*. 14(1), 35-51.

Kellner, D. (1995) *Media Culture: Cultural Studies, Identity and Politics between the Modern and the Post-modern.* Routledge: New York, USA.

Kjeldgaard, D., Askegaard, S. (2006) The glocalization of youth culture: The global youth segment as structures of common difference. *Journal of Consumer Research*. 33(2), 231-247.

Kjellberg, H., Helgesson, C.F. (2006) On the Nature of Markets and Their Practices. *Marketing Theory*. 7(2), 137–62.

Kjellberg, H., Helgesson, C.F. (2010) Political marketing: Multiple values, performativities and modes of engaging. *Journal of Cultural Economy*. 3(2), 279-297.

Klag, M., Langley, A. (2013) Approaching the Conceptual Leap in Qualitative Research. *International Journal of Management Reviews*. 15(2), 149–166.

Koerner, B.I. (2006) Geeks in Toyland. Wired News. http://www.wired.com/news/culture/0,69946-0.html (accessed 05.01.13.).

Kopytoff, I. (1986) The Cultural Biography of Things: commodification as a process, in Appadurai A. (ed.) *The Social Life of Things*. Cambridge University Press: Cambridge, UK. 64-94.

Korchia, M. (2006) The Effects of Brand Associations on Brand Equity, Subjective Knowledge, and Brand Interest. In Proceedings of the *American Marketing Association Conference*. 17.

Korkman, O. (2006) *Customer Value Formation in Practice – A Practice Theoretical Approach*. Hanken School of Economics: Helsinki.

Kotler, P. (1967) *Marketing Management: analysis, planning and control.* Prentice Hall: N.J, USA.

Kotler, P. (1986) The prosumer movement: A new challenge for marketers. *Advances in Consumer Research*. 13(1), 510-513.

Kozinets, R.V. (1997) I want to believe': a netnography of the X-philes' subculture of consumption. In *Advances in Consumer Research*, eds M. Brucks and D. J. MacInnis, Vol. 24, 470–475. Association for Consumer Research: Provo, UT.

Kozinets, R.V. (1998) On Netnography: Initial Reflections on Consumer Research Investigations of Cyberculture. *Advances in Consumer Research*. 25, 366-371.

Kozinets, R.V. (1999) E-Tribalised Marketing? The Strategic Implications of Virtual Communities of Consumption. *European Management Journal*. 17(3), 252-64.

Kozinets, R.V. (2001) Utopian Enterprise: Articulating the Meaning of Star Trek's Culture of Consumption. *Journal of Consumer Research*. 28(1), 67-89.

Kozinets, R.V. (2002) Can Consumers Escape the Market? Emancipatory Illuminations from Burning Man. *Journal of Consumer Research*. 29(1), 20-38.

Kozinets, R.V. (2006) Click to Connect: Netnography and Tribal Advertising. *Journal of Advertising Research*. 46(3), 279-288

Kozinets, R.V. (2007) Inno-tribes: Star Trek as wikimedia. *Consumer Tribes*. 194-211.

Kozinets, R.V. (2008) Technology/Ideology: How ideological fields influence consumers' technology narratives. *Journal of Consumer Research*. 34(6), 865-881.

Kozinets, R.V. (2010) *Netnography: Doing Ethnographic Research Online*. Sage: London, UK.

Kozinets, R.V., Belk, R.W (2006) Videography. *Sage Dictionary of Social Research Methods*. London: 318-320.

Kozinets, R.V., de Valck, K., Wojnicki, A.C., Wilner, S.J. (2010) Networked Narratives: UnderstandingWord-of-Mouth Marketing in Online Communities. *Journal of Marketing*. 74(2), 71–89.

Kozinets, R.V., Handelman, J.M. (2004) Adversaries of Consumption: Consumer Movements, Activism and Ideology. *Journal of Consumer Research*. 31(3), 691-70.

Kozinets, R.V., Hemetsberger, A., Schau, H.J. (2008) The wisdom of crowds: Collective innovation in the age of networked marketing. *Journal of Macromarketing*. 28(4), 339–354. Kozinets, R. V., Sherry, Jr. J. F., Storm, D., Duhachek, A., Nuttavuthisit, K., DeBerry-Spence, B. (2004) Ludic agency and retail spectacle. *Journal of Consumer Research*. 31(3), 658-672.

Krueger, R. A. (1998) *Developing questions for focus groups* (Vol. 3). Sage: Thousand Oaks.

Landen, S.A. (2011) From ethnographic 'self'-discovery to processes of identification, *Qualitative Research*. 11(5), 536 – 549.

Lash, S., Lury, C. (2007) Global Culture Industry. Polity Press.

Lastovicka, J., Fernandez, K. (2005) Three paths to disposition: the movement of meaningful possessions to strangers. *Journal of Consumer Research*. 31(4), 813–823.

Latour, B. (1987) *Science in action: How to follow scientists and engineers through society.* Harvard University Press: USA.

Latour, B. (1991) Technology is society made durable.

Latour, B. (1992) Where are the missing masses? The sociology of a few mundane artefacts, *Shaping Technology*. MIT Press: Cambridge, MA, 225-258.

Latour, B. (2005) *Reassembling the Social: An Introduction to Actor-Network-Theory*. Oxford University Press: New York, USA.

Lave, J., Wenger, E. (1991) *Situated Learning: Legitimate Peripheral Participation*. Cambridge University Press: New York, USA.

Lawrence, T. B., Phillips, N. (2002) Understanding cultural industries. *Journal of Management Inquiry*. 11(4), 430-441.

Leigh, T.W., Peters, C., Shelton, J. (2006) The Consumer Quest for Authenticity: The Multiplicity of Meanings within the MG Subculture of Consumption. *Journal of the Academy of Marketing Science*. 34(4), 481–93.

Levi-Strauss, C. (1966) Toteism. Trans. Needham R, Merlin Press: London, UK.

Levi-Strauss, C. (1967) *Structural Anthropology*. Translated by Claire Jacobson and Brooke Grundfest Schoepf. Doubleday Anchor Books: New York, USA.

Levy, S.J. (1959) 'Symbols for Sale'. *Harvard Business Review*. 37 (July–August), 117–24.

Levy, S.J. (1981) Interpreting Consumer Mythology: A Structural Approach to Consumer Behavior. *Journal of Marketing*. 45(3), 49-61.

Lincoln, Y.S., Guba, E.G. (1985) *Naturalistic Inquiry*. Sage: Beverly Hills. London and New York.

Lincoln, Y.S., Guba, E.G. (2003) Ethics: The failure of positivist science. *Turning points in qualitative research: Tying knots in a handkerchief.* AltaMira Press, CA. 219-238.

Linnet, J.T. (2011) Money can't buy me hygge: Danish middle-class consumption, egalitarianism and the sanctity of inner space. *Social Analysis: Journal of Cultural and Social Practice*. 55(2), 21-44.

Little Birds Market Facebook [online] https://www.facebook.com/littlebirdsmarket

Little Birds Market Twitter [online] https://twitter.com/LittleBirdsChat

Lowenthal, D. (1985) *The past is a foreign country*. Cambridge University Press: Cambridge, UK.

Luedicke, M.K., Thompson, C.J., Giesler, M. (2010) Consumer Identity Work as Moral Protagonism: How Myth and Ideology Animate a Brand-Mediated Moral Conflict. *Journal of Consumer Research*. 36(6), 1016–32.

Lury, C. (1996) Consumer Culture. Rutgers University Press: USA.

Lury, C. (1997) *Consumer Culture*. Polity Press, Blackwell Publishers Ltd: Oxford, UK.

Lusch, R. F., Vargo, S. L. (2006) Service-dominant logic: reactions, reflections and refinements. *Marketing Theory*. 6(3), 281-288.

Lusch, R. F., Vargo, S. L., Tanniru, M. (2010) Service, value networks and learning. *Journal of the Academy of Marketing Science*. 38(1), 19-31.

Maclaran, P., Brown, B. (2005) The Center Cannot Hold: Consuming the Utopian Marketplace. *Journal of Consumer Research*. 32 (2), 311–23.

Maclaran, P., Saren, M., Stern, B., Tadajewski, M., (2009) *The Sage Handbook of Marketing Theory*. Sage Publications: London, UK.

Madden, R. (2010) *Being Ethnographic: A Guide to the Theory and Practice of Ethnography.* SAGE Publications Ltd.

Maffesoli, M. (1988) '*The Time of the Tribes' The Decline of Individualism in Mass Society.* Sage: London, UK.

Maffesoli, M. (1996) *The Time of the Tribes: The Decline of Individualism in Mass Society.* London: Sage.

Maffesoli, M. (2007) 'Tribal Aesthetic', in Cova B., Kozinets R., Shankar A. (ed.) *Consumer Tribes*. Butterworth-Heinemann: Oxford, UK. 27–34.

Magaudda, P. (2011) When materiality 'bites back': Digital music consumption practices in the age of dematerialisation. *Journal of Consumer Culture*. 11(1), 15-36.

Maisel, R. (1974) The flea market as an action scene. *Journal of Contemporary Ethnography*. 2(4), 488-505.

Martens, L. (2012) [online] Practice 'in Talk' and Talk 'as Practice': Dish Washing and the Reach of Language. *Sociological Research Online*. 17(3), 22 [accessed on July 13th 2014]

Marx, K. (1974) Value, Price, Profit, International Publishers, delivered by Karl Marx at two sessions of the General Council of the First International on June 20 and 27, 1865.

Maxwell, J.A. (1998) Designing a qualitative study. In Bickman, L., Rog, D. (Eds.) *Handbook of applied social research methods*. (pp. 69-100). Sage: Thousand Oaks, CA.

Maxwell, J.A. (2005) *Qualitative Research Design: An Interactive Approach*. Sage Publications.

Maycroft, N. (2004) The objectness of everyday life: disburdenment or engagement? *Geoforum.* 35(6), 713–725.

Maycroft, N. (2009) Not moving things along: hoarding, clutter and other ambiguous matter. *Journal of Consumer Behaviour*. 8(6), 354–364.

McAlexander, J.H., Schouten, J.W., Roberts, S.D. (1992) Consumer-behavior in coping strategies for divorce. *Advances in Consumer Research*. 19, 555-556.

McAlexander, J. H., Schouten, J. W. (1998) Brandfests: servicescapes for the cultivation of brand equity. *Servicescapes: The concept of place in contemporary markets*, 377-402.

McAlexander J. H., Schouten, J. W., Koenig, H. F. (2002) Building brand community. *Journal of Marketing*. 66(1), 38-54.

McColl, J., Canning, C., McBride, L., Nobbs, K., Shearer, L. (2013) It's Vintage Darling! An exploration of vintage fashion retailing. *Journal of The Textile Institute*. 104(2), 140-150.

McCracken, B. (2010) *Hipster Christianity: When Church and Cool Collide*. Baker Books: NYC, USA.

McCracken, G. (1986) Culture and consumption: A theoretical account of the structure and movement of the cultural meaning of consumer goods. *Journal of Consumer Research*. 13(1), 71-84.

McCracken, G. (1988a) *Culture and Consumption: New Approaches to the Symbolic Character of Consumer Goods and Activities*. Indiana University Press: Bloomington, IN.

McCracken, G. (1988b) *The Long Interview*. Sage, University Press, Newbury Park, CA.

McCracken, G. (1989) Homeyness: A Cultural Account of One Constellation of Consumer Goods and Meanings, in *Interpretive Consumer Research*, ed. Elizabeth Hirschman, Provo, UT: Association for Consumer Research: 168–83.

McCracken, G. (1998) Consumer goods, gender construction, and a rehabilitated trickle down theory. *Culture and Consumption*. Indiana University Press: Bloomington. 93-103.

McCracken, G. (2005) *Culture and Consumption II: Markets, Meaning, and Brand Management*. Indiana University Press: Bloomington, IN.

McCracken, G. (2008) *Transformations: Identity Construction in Contemporary Culture*. Indiana University Press: Bloomington, IN.

McCracken, G.D., Roth, V. J. (1989) Does clothing have a code? Empirical findings and theoretical implications in the study of clothing as a means of communication. *International Journal of Research in Marketing*. 6(1), 13-33.

McDonald, K. (2004) Oneself as another: from social movement to experience movement. *Current Sociology*. 52(4), 575-593.

McDowell, C. (2000) Fashion Today. Phaiden Press Limited: London, UK.

McGrath, M.A., Sherry, J.F, Heisley, D.D. (1993) An ethnographic study of an urban periodic marketplace: lessons from the Midville Farmer's market. *Journal of Retailing*. 69(3), 280-319.

McRobbie, A. (1988) *Zoot Suits and Second Hand Dresses*. Unwin Hyman: Boston, MA.

McRobbie, A. (1994) Postmodernism and Popular Culture. Routledge: London, UK.

McRobbie, A. (2005) Needs and Norms: Bourdieu and Cultural Studies, *The Uses of Cultural Studies – A Textbook*. Sage Publications: London, UK.

McRobbie, A. (2008) What not to wear and post-feminist symbolic violence, Paper presented at the *Putting Bourdieu to Work Conference*. Manchester University. May: 173-210.

Michael, J. (2013) It's really not hip to be a hipster: Negotiating trends and authenticity in the cultural field. *Journal of Consumer Culture*. 0(0): 1-20.

Mick, D.G (1986) Consumer Research and Semiotics: Exploring the Morphology of Signs, Symbols, and Significance. *Journal of Consumer Research*. 13(2), 196–213.

Mick, D.G. (1988) Schema-Theoretics and Semiotics: Toward More Holistic, Programmatic Research on Marketing Communication. *Semiotica*. 70(1/2), 1-26.

Mick, D.G., DeMoss, M. (1990) Self-Gifts: Phenomenological Insights from Four Contexts. *Journal of Consumer Research*. 17(3), 322–32.

Mick, D.G., Fournier, S. (1998) Paradoxes of Technology: Consumer Cognizance, Emotions, and Coping Strategies. *Journal of Consumer Research*. 25 (2), 123–43.

Miles, M.B., Huberman, M.A (1994) *Qualitative Data Analysis: A Sourcebook of New Methods*. Sage Publications: London, UK.

Miller, D. (1988) Appropriating the state on the council estate. Man. 23(2), 353-372.

Miller, D. (1995) Consumption and the vanguard of history. In *Acknowledging Consumption*. Miller D. (ed.) Routledge:London: 1–57.

Miller, D. (1998a) *A Theory of Shopping*. Cornell University Press: Ithaca, New York, USA.

Miller, D. (1998b) *Material Cultures: Why some things matter*. University of Chicago Press: Chicago, USA.

Miller, D. (2000) 'The Birth of Value' in Jackson, P, Lowe, M, Miller, D and Mort, F., *Commercial Cultures*. Oxford: Berg: 77-83.

Miller, D. (2001) *Home Possessions: Material Culture behind Closed Doors*. Berg: New York, USA.

Miller, D. (2005) Materiality. Duke University Press: London, UK.

Miller, D. (2008a) The Comfort of Things. Polity Press: Cambridge, UK.

Miller, D. (2008b) The Uses of Value. *Geoforum*. 39(3), 1122-1132.

Miller, D. (2010) Stuff. Polity Press: Cambridge, UK.

Mitchell, C., Imrie, B.C. (2011) Consumer tribes: membership, consumption and building loyalty. *Asia Pacific Journal of Marketing and Logistics*. 23(1), 39-56.

Moisander, J., Valtonen, A. (2006) *Qualitative Marketing Methods. A Cultural Approach.* Sage: London, UK.

Moisander, J., Peñaloza, L., Valtonen, A. (2009a) From CCT to CCC: Building consumer culture community, in *Explorations in Consumer Cultural Theory*. Routledge, Taylor and Francis Group: 7-33.

Moisander, J., Valtonen, A., Hirsto, H. (2009b) Personal interviews in cultural consumer research – poststructuralist challenges. *Consumption Markets and Culture*. 12(4), 329–348.

Molander, S. (2011) 'Food, Love and Meta-Practices: A Study of Everyday Dinner Consumption Among Single Mothers', in Belk, R.W., Grayson, K., Muniz, A.M., Schau, H.J. (ed.) *Research in Consumer Behaviour*. Bingley: UK. 77–92.

Molesworth, M., Denegri-Knott, J. (2008) The playfulness of eBay and the implications for business as a game-maker. *Journal of Marcromarketing*. 28(4), 369-380.

Molesworth, M., Jenkins, R., Eccles, S. (2011) Love and Videogames: Negotiating Relationships with Cultural Ideals and Consumer Practices. In Belk, R.W., Grayson, K., Muniz, A.M., Schau, H.J. (ed) *Research in Consumer Behavior*. 13, Emerald Group Publishing Ltd: 145-162.

Morgan, B. (2008) Identity Presentation: The Construction of Identity in Asynchronous Discussion, *Qualitative Social Research*. Volume 9, No. 3, pp. 1-26.

Muniz, A.M., O'Guinn, T.C. (2001) Brand Community. *Journal of Consumer Research*. 27(4), 412-32.

Muniz, Jr. A. M., Schau, H. J. (2005) Religiosity in the abandoned Apple Newton brand community. *Journal of Consumer Research*. 31(4), 737-747.

Murphy, S., Patterson, M. (2011) Motorcycling edgework: A Practice Theory Perspective. *Journal of Marketing Management*. 27(13/14), 1322-1340.

Murray, J. (2002) The Politics of Consumption: A Re-Inquiry on Thompson and Haytko's (1997) Speaking of Fashion. *Journal of Consumer Research*. 29(3), 427-440.

Nadin, S., Cassell, C. (2006) The use of a research diary as a tool for reflexive practice: some reflections from management research. *Qualitative Research in Accounting and Management*. 3(3), 208-217.

Nash, C. (2000) Performativity in practice: some recent work in cultural geography. *Progress in Human Geography*. 24(4): 653-664.

Nelson, M. R., Otnes, C.C. (2005) Exploring cross-cultural ambivalence: a netnography of intercultural wedding message boards. *Journal of Business Research*. 58(1), Longman New York: USA: 89-95.

Nicolini, D. (2012) *Practice Theory, Work, and Organization An Introduction*. Oxford University Press: Oxford, UK.

Nicolini, D., Gherardi, S. Yanow, D. (Eds.). (2003) *Knowing in organizations: A practice-based approach*. ME Sharpe.

Nobbs, K., McColl, J., Shearer, L., Canning, C., McBride, L. (2011) An exploratory study into the strategic significance of visual merchandising: the case of vintage fashion retailing. *In 3rd Global Conference, Fashion: Exploring Critical Issues*. Mansfield College. Oxford, UK. 22-25.

Odulate, F. (2008) *Shopping for Vintage: The Definitive Guide to Vintage Fashion.* Macmillan: New York, USA.

O'Reilly, J. (2009) Key Concepts in Ethnography. Sage London.

O'Reilly, D. (2012) Maffesoli and Consumer Tribes Developing the Theoretical Links. *Marketing Theory*. 12(3), 341-347.

Ortner, S.B (1984) Theory in Anthropology since the Sixties, *Comparative Studies in Society and History*. 26(1), 126-166.

Osterlund, C., Carlile, P. (2005) Relations in Practice: Sorting Through Practice Theories on Knowledge Sharing in Complex Organisations. *The Information Society*. 21, 91-107.

Otnes, C., McGrath, M.A., Lowrey, T.M. (1995) Shopping with consumers: Usage as past, present and future research technique. *Journal of Retailing and Customer Services*. 2(2), 97-110.

Otnes, C., Pleck, E.H. (2003) *Cinderella dreams: The allure of the lavish wedding*. University of California Press: CA, USA.

Ozanne, J.L, Murray, J.B (1991) Uniting Critical Theory and Public Policy to Create the Reflexivity Defiant Consumer. *American Behavioural Scientist*. 38(4), 516-525.

Pahl, R., Pevalin, D.J. (2005) Between family and friends: a longitudinal study of friendship choice. *The British Journal of Sociology*. 56(3), 433-450.

Palmer, A., Clark, H. (ed.) (2005) *Old Clothes, New Looks: Second-Hand Fashion,* Berg Publishers: Oxford, UK.

Parasuraman, A., Grewal, D. (2000) The impact of technology on the quality-valueloyalty chain: a research agenda. *Journal of the Academy of Marketing Science*. 28(1), 168-174.

Parmentier, M.A., Fischer E. (2011) You can't always get what you want: Unsustainable identity projects in the fashion system. *Consumption, Markets and Culture*. 14(1): 7-27.

Parsons, E. (2006) Dealing in Second-hand Goods: Creating meaning and value. *European Advances in Consumer Research*. Vol. 7. (ed.) Karin M. Ekstrom and Helene Brembeck, Gothenburg, Sweden: Association for Consumer Research: 189-194.

Parsons, E. (2007) Dealing in Histories: Durability, Authenticity and Provenance in Markets for Antiques. In S. Borghini, M. McGrath and C. Otnes (ed.) *European Advances in Consumer Research.* 8, 111-112. Duluth, MN: Association for Consumer Research.

Parsons, E. (2008) Thompson's Rubbish Theory: exploring the practices of value creation. *European Advances in Consumer Research*. S., Borghini, MA., McGrath, and C., Otnes. Duluth, MN: Association for Consumer Research: 8, 390-393.

Parsons, E. (2009) 'What Things Do:' Examining Things that 'Matter' in Consumer Research, *Advances in Consumer Research*, Vol. 36, (ed.) A.L. McGill and S. Shavitt, Duluth, MN: Association for Consumer Research: 177-180.

Parsons, E. (2010) Markets, identities and the discourses of antique dealing. *Marketing Theory*. 10(3), 283-298.

Parsons, E., Maclaran, P. (2009) Unpacking disposal: introduction to the special issue. *Journal of Consumer Behaviour*. Special Issue: Unpacking Disposal. 8(6), 301–304.

Patton, M.Q. (2002) *Qualitative Research and the Evaluation Methods*. Sage: Thousand Oaks, CA.

Paulicelli, E., Clark H. (2009) *The Fabric of Cultures: Fashion, Identity and Globalisation*. Routledge: London, UK.

Peñaloza, L. (1994) Atravesando Fronteras/Border Crossings: A Critical Ethnographic Exploration of the Consumer Acculturation of Mexican Immigrants. *Journal of Consumer Research*. 21(1), 32-54.

Peñaloza, L. (2000) The commodification of the American West: Marketers' production of cultural meanings at the trade show. *Journal of Marketing*. 64(4), 82-109.

Peñaloza, L. (2001) Consuming the American West: Animating Cultural Meaning and Memory at a Stock Show and Rodeo. *Journal of Consumer Research*. 28(3), 369–98.

Peñaloza, L. (2011) Just doing it: A visual ethnographic study of spectacular consumption behavior at Nike Town. *Consumption, Markets and Culture*. 2(4), 337-465.

Peñaloza, L., Cayla, J. (2006) Writing pictures/taking fieldnotes: towards a more visual and material ethnographic consumer research in Belk R.W. (ed.) *Handbook of Qualitative Research*. Edward Elgar Publishing: Cheltenham, UK.

Peñaloza, L., Mish J. (2011) The nature and processes of market co-creation in triple bottom line firms: Leveraging insights from consumer culture theory and service dominant logic. *Marketing Theory*. 11(1), 9-34.

Peñaloza, L., Venkatesh A. (2006) Further Evolving the New Dominant Logic of Marketing: From Service and Value to Lean. Consumption, *Marketing Theory*. 6(3), 299–316.

Perren, L., Ram, M. (2004) Case study methods in small business and entrepreneurial research: Mapping boundaries and perspectives. *International Small Business Journal*. 22(1), 83-101.

Petridis, A. (2014) [online] Youth subcultures: what are they now? <u>http://www.theguardian.com/culture/2014/mar/20/youth-subcultures-where-have-they-gone</u> published Thursday 20th March, accessed 21st March 2014.

Pettigrew, S.F. (2000) Ethnography and Grounded Theory: A Happy Marriage? *Advances in Consumer Research*. 27, 256-260.

Pettinger, L. (2004) Brand Culture and Branded Workers: Service Work and Aesthetic Labour in Fashion Retail. *Consumption Markets and Culture*. 7(2), 165-84.

Pickering, A. (1995) *The Mangle of Practice, Time, Agency and Science*. University of Chicago Press: Chicago, IL.

Pickering, A. (1992) *Science as practice and culture*. University of Chicago Press: Chicago, USA.

Pink, S. (2007) Walking with video. Visual Studies. 22(3), 240-252.

Pink, S. (2008) Mobilising Visual Ethnography: Making Routes, Making Place and Making Images. *Forum Qualitative Research*. 9(3), Art 36.

Pink, S. (2009) Doing Sensory Ethnography. SAGE Publications Ltd.

Pongsakornrungsilp, S., Healy, J.C., Bradshaw, A., McDonagh, P. Schroeder, J.E. (2008) 'Left Behind: Local Fans of Global Brands,' in W. E. Kilbourne, and J. D., Mittelstaedt (eds) '*Macromarketing: Systems, Causes, and Consequences'*, 225-37, 33rd Annual Meeting of the Macromarketing Society Conference. Clemson University. South Carolina, June.

Pongsakornrungsilp, S., Schroeder, J.E. (2011) Understanding value co-creation in a co-consuming brand community. *Marketing Theory*. 11(3), 303-324.

Porter, M.E. (1985) *Competitive Advantage: Creating and Sustaining Superior Performance*. Free Press: New York, NY.

Postil, J., Brauchler B. (ed.) (2010) *Theorising Media and Practice*. Berghahn: Oxford, UK.

Postrel, V.I. (2003) *The Substance of Style: How the Rise of Aesthetic Value is Remaking Commerce, Culture, and Consciousness.* Harper Collins: New York, NY.

Pountain, D., Robins, D. (2000) Cool Rules. Reaktion: London, UK.

Prahalad, C.K., Ramaswamy, V. (2000) Co-opting Customer Competence. *Harvard Business Review*. 78 (January/February), 79-87.

Prahalad, C.K., Ramaswamy, V. (2004) Co-creation Unique Value with Customers. *Strategy and Leadership.* 32(3), 4-9.

Prus, R. (1984) Purchasing products for resale: assessing suppliers as partners in trade. *Symbolic Interaction*. 7(2), 249-278.

Prus, R. (1985) Price setting as social activity. Urban Life. 14(1), 59-93.

Prus, R. (1986) It's on sale: An Examination of vendor perspectives, activities and dilemmas. *Canadian Review of Sociology and Anthropology*. 23(1), 72-96.

Rafferty, K. (2009) Lost in Transformation? How Class-Based Emotions Shape Fashion Consumption Practice, Consumers, Commodities and Consumption – a newsletter of the *Consumer Studies Research Network*. 11(1).

Rafferty, K. (2011) Class based emotions and the allure of fashion consumption. *Journal of Consumer Culture*. 11(2), 239-260.

Randles, S., Warde, A. (2006) Consumption: The view from theories of practice. In Green, K. and Randles, S. eds (2006) *Industrial Ecology and Spaces of Innovation*. Cheltenham: Edward Elgar, chapter 10, 220-237.

Reckwitz A. (2002) Toward a theory of social practices: A Development in culturalist theorizing. *European Journal of Social Theory*. 5(2), 243–263.

Reiley, K., DeLong, M. (2011) A Consumer Vision for Sustainable Fashion Practice, *Fashion Practice*. 3(1), 63-84.

Reynolds, S. (2005) *Rip it up and Start Again: Post-Punk 1978-84*. Faber and Faber Inc: London, UK.

Reynolds, S. (2011) *Retromania: Pop Culture's Addiction to Its Own Past.* Faber and Faber Inc: New York, USA.

Richins, M.L. (1994) Valuing Things: The Public and Private Meanings of Possessions. *Journal of Consumer Research*. 21(3), 504–21.

Rindell, A., Korkman, O., Gummerus, J. (2011) The Role of Brand Images in Consumer Practices: uncovering embedded brand strength. *Journal of Product and Brand Management*. 20(6), 440-446.

Ritzer, G. (1993) *The McDonalization of society: An investigation into the changing character of contemporary social life.* Pine Forge Press: Newbury Park, CA.

Ritzer, G. (2010) The McDonaldization of society (6th edition). Pine Forge Press.

Ritzer, G., Goodman, D., Wiedenhoft, W. (2001) Theories of Consumption. In Ritzer, G., Smart, B. (ed.) *Handbook of Social Theory*. Sage: London. 410-27.

Ritzer, G., Jurgenson, N. (2010) Production, consumption and prosumption: the nature of capitalism in the age of the digital prosumer. *Journal of Consumer Culture*. 10(1), 13-36.

Rojek, C. (2011) Pop music, pop culture. Polity.

Rokka, J. (2010) Netnographic inquiry and new translocal sites of the social. *International Journal of Consumer Studies*. 34(4), 381–387.

Rook, D.W. (1985) The Ritual Dimension of Consumer Behavior. *Journal of Consumer Research*. 12(3), 251–64.

Rook, D.W. (1988) Researching consumer fantasy. *Research in Consumer Behavior*. 3, 247-270.

Røpke, I. (2009) Theories of Practice – New Inspiration for Ecological Economic Studies on Consumption. *Ecological Economics*. 68(10), 2490-2497.

Rouse, J. (2006) *Practice Theory, Handbook of the Philosophy of Science*. Elsevier: 15, 499-540.

Roux, D., Guiot, D. (2008) Measuring second-hand shopping motives, antecedents and consequences, *Recherche et Applications en Marketing*. 23(4), 63-94.

Roux, D., Korchia, M. (2006) Am I What I Wear? An Exploratory Study of Symbolic Meanings Associated with Second-hand Clothing. *Advances in Consumer Research*. 33, (ed.) Pechmann, C., Price, L. 29-35.

Ruth, J.A. Otnes, C.C. (2006) *The etiquette of qualitative research: handbook of qualitative research methods in marketing.* Edward Elgar Publishing.

Sánchez-Fernández, R., Ángeles Iniesta-Bonillo, M. (2007) The Concept of Perceived Value: A Systematic Review of the Research. *Marketing Theory*. 7(4), 427–51.

Sandikci, Ö., Ger, G. (2010) Veiling in Style: How Does a Stigmatized Practice Become Fashionable? *Journal of Consumer Research*. 37(1), 15–36.

Sandicki, O., Holt, D. (1998) Malling society: Mall consumption practices and the future of publicpace. In *ServiceScapes: The concept of place in contemporary markets* (305-336) Sherry, J.F., Jr. (ed). Chicago: NTC Business Books.

Sartre, J.P. (1956) *Being and Nothingness*. trans. Hazel E. Barnes, New York: Philosophical Library, 330.

Scaraboto, D., Fischer, E. (2013) Frustrated Fatshionistas: An Institutional Theory Perspective on Consumer Quests for Greater Choice in Mainstream Market. *Journal of Consumer Research*. 39(6), 1234-1257.

Schatzki, T.R. (1996) Social Practices: A Wittgensteinian Approach to Human Activity and the Social. Cambridge University Press: Cambridge.

Schatzki, T.R. (2001) Introduction: Practice Theory. In: Schatzki, T., K. Knorr Cetina and E. von Savigny (ed.) *The Practice Turn in Contemporary Theory*. Routledge: London. 1-14.

Schatzki, T.R. (2002) *The Site of the Social: A Philosophical Account of the Constitution of Social Life and Change*. University Park: Pennsylvania State University Press, USA.

Schatzki, T.R. (2005) The sites of organisations. *Organisation Studies*. 26(3), 465–484.

Schatzki, T.R., Cetina, K.K., Savigny, E. (ed.) (2001) *The Practice Turn in Contemporary Theory*. Routledge: London, UK.

Schau, H., Muñiz, A., Arnould, E.J. (2009) How Brand Community Practices Create Value. *Journal of Marketing*. 73(5), 30-51.

Schiermer, B. (2013) Late-modern hipsters: New tendencies in popular culture. *Acta Sociologica*. 57(2), 167-181.

Schiffman, L.G., Hansen, H., Kanuk, L.L. (2008) *Consumer behaviour: A European outlook.* Pearson Education.

Schouten, J.W., Martin, D.M., McAlexander ,J.H. (2007) The evolution of a subculture of consumption, in Cova, B., Kozinets, R.V., Shankar, A. (ed.) *Consumer Tribes*. Butterworth-Heinemann: Oxford, UK.

Schouten, J.W., Martin, D.M., McAlexander J.H. (2007) The evolution of a subculture of consumption. *Consumer Tribes*. 67-75.

Schouten, J.W., McAlexander J.H. (1995) Subcultures of Consumption: An Ethnography of New Bikers. *Journal of Consumer Research*. 22(1), 43-61.

Schouten, J.W., McAlexander J.H., Koenig H.F. (2007) Transcendent customer experience and brand community. *Journal of the Academy of Marketing Science*. 35(3), 357-368.

Schroeder, J.E. (2002) *Visual Consumption in an Image Economy*. Routledge Interpretive Marketing Research: Abingdon, Oxon.

Schroeder, J.E. (2004) Visual Consumption in the Image Economy. In *Elusive Consumption*. (ed.) Ekstrom, K., Brembeck, H. Berg: Oxford, UK.

Schroeder, J.E., Borgerson, J.L. (2003) Dark desires: Fetishism, ontology and representation in contemporary advertising. *Sex in advertising: Perspectives on the erotic appeal.* 65-89.

Shankar, A., Cherrier, H., Canniford, R. (2006) Consumer Empowerment: A Foucauldain in Interpretation. *European Journal of Marketing*. 40(9/10), 1013-1030.

Shankar, A., Elliott, R., Fitchett, J.A. (2009) Identity, Consumption and Narratives of Socialization. *Marketing Theory*. 9(1), 75-94.

Shankar, A., Elliott, R., Goulding, C. (2001) Understanding Consumption: Contributions from a Narrative Perspective. *Journal of Marketing Management*. 17, 429-53.

Shankar, A., Goulding, C. (2001) Interpretive consumer research: two more contributions to theory and practice. *Qualitative Market Research: An International Journal*. 4(1), 7-16.

Sherry, J.F. Jr. (1983) Gift Giving in Anthropological Perspective, *Journal of Consumer Research*. 10(September): 157–68.

Sherry, J.F Jr. (1990a) A socio-cultural analysis of a Midwestern American Flea-Market. *Journal of Consumer Research*. 17(1), 13-30.

Sherry, J.F. Jr. (1990b) Dealers and dealing in a Periodic Market: Informal Retailing in Ethnographic Perspective. *Journal of Retailing*. 66(2), 174-200.

Sherry, J.F Jr. (1991) Postmodern Alternatives: The Interpretive Turn in Consumer Research. T. Robertson, H. Kassarjian (ed.), *Handbook of Consumer Behavior*. Prentice Hall: Englewood Cliffs, NJ. 548–591.

Sherry, J.F. Jr. (1995) *Contemporary Marketing and Consumer Behavior*. Sage, Thousand Oaks, CA.

Sherry, J.F. Jr. (1998) The Soul of the Company Store: Nike Town Chicago and the Emplaced Brandscape, in *ServiceScapes: The Concept of Place in Contemporary Markets*. ed. Sherry, J.F., Chicago: NTC Business Books: 109–46.

Sherry, J.F. Jr. (2000) Place, Technology, and Representation. *Journal of Consumer Research*. 27(2), 273–278.

Sherry, J.F. Jr. (2004) Culture, Consumption, and Marketing: Retrospect and Prospect. In *Elusive Consumption*, ed. K.M. Ekström and H. Brembeck, Oxford: Berg: 45–64.

Sherry, J.F. Jr. (2009) The Work of Play at American Girl Place. *Social Psychology Quarterly*. 72(3), 199-202.

Sherry J.F. Jr., Fischer E. (2008) (ed.) *Explorations in consumer culture theory*. Routledge: London, UK.

Sherry, J.F. Jr., Fischer, E. (ed.) (2009) *Explorations in Consumer Culture Theory*. Routledge Interpretive Marketing Research, Routledge: London, UK.

Shove, E. (2003a) Comfort, Cleanliness and Convenience: The Social Organization of Normality. Berg: New York, USA.

Shove, E. (2003b) Converging Conventions of Comfort, Cleanliness and Convenience. *Journal of Consumer Policy*. 26(4), 395-418.

Shove, E. (2004) Efficiency and consumption: technology and practice. *Energy and Environment*. 15(6), 1053-1065.

Shove, E. (2005) Changing human behaviour and lifestyle: a challenge for sustainable consumption? *Consumption - Perspectives from ecological economics*. Ropke, I. and Reisch L. (eds.). Cheltenham: Elgar, 111-132.

Shove, E. (2009) Everyday Practice and the Production and Consumption of Time. In *Time, Consumption, and Everyday Life: Practice, Materiality and Culture*. (ed.) Shove, E., Trentmann, F., Wilk, R.R. Berg: New York, USA.

Shove, E., Araujo L. (2010) Consumption, Materiality, and Markets. In *Reconnecting Marketing to Markets*, (ed.) Araujo, L.M., Kjellberg, H., Finch, J. Oxford: Oxford University Press: 13–28.

Shove, E., Hand, M., Ingram, J. (2007) *The Design of Everyday Life*. Berg Publishers: Oxford, UK.

Shove, E., Pantzar, M. (2005) Consumers, Producers and Practices: Understanding the invention and reinvention of Nordic walking. *Journal of Consumer Culture*. 5(1), 43-64.

Shove, E., Pantzar, M., Watson, M. (2012)*The dynamics of social practice: everyday life and how it changes.* Sage: London.

Shove, E., Trentmann, F., Wilk, R. (2009) *Time, Consumption and Everyday Life: Practice, Materiality and Culture.* Berg Publishers: Oxford, UK.

Shove, E., Walker, G. (2010) Governing transitions in the sustainability of everyday life. *Research Policy*. 39(4), 471-476.

Shove, E., Watson, M., Hand, M., Ingram J. (2007) *The Design of Everyday Life*. Berg: Oxford, UK.

Siegle, L. (2011) Why fast fashion is slow death for the planet. *The Observer*, Sunday 8th May 2011.

Silverman, D. (1985) Qualitative Methodology and Sociology. Sage: London, UK.

Silverman, K. (1986) *Fragments of a fashionable discourse*. Indiana University Press: Indianapolis, USA.

Simmel, G. (1950) *The sociology of Georg Simmel* – translated and edited by Kurt H. Wolff. The Free Press: Glencoe, Illinois.

Simmel, G. (1957) Fashion, The American Journal of Sociology. LXII 6: 541-558.

Simmel, G. (1997) The Sociology of Space. In Frisby, D. and Featherstone, M., *Simmel on Culture*. Sage Publications: London, UK. 138 -170

Simmel, G. (2004) *The Philosophy of Money*. D. Frisby (ed) 3rd Edition: Routledge: London and New York.

Simmons, G. (2008) Marketing to postmodern consumers: introducing the internet chameleon. *European Journal of Marketing*. 42(3/4), 299-310.

Simpson, B. (2009) Pragmatism, Mead and the Practice Turn. *Organization Studies*. 30(12), 1329-1347.

Slater, D. (1997) Consumer Culture and Modernity. Polity: London, UK.

Slater, D. (2002a) Capturing Markets from the Economists. In Gay P., Pryke M. (ed.) *Cultural Economy*. London: UK, Sage. 59–77.

Slater, D. (2002b) Markets, materiality and the "new economy", in (ed.) S. Metcalfe and D. Miller. *Market Relations and the Competitive Process*. Manchester University Press: Manchester, UK.

Slater, D. (2014) Ambiguous goods and nebulous things. *Journal of Consumer Behaviour*. 13(2), 99-107.

Slater, D., Miller, D. (2007) Moments and Movements in the Study of Consumer Culture: A discussion between Daniel Miller and Don Slater. *Journal of Consumer Culture*. 7(1), 5-23.

Smith, A. (1937) The wealth of nations (1776). New York: Modern Library, 740.

Solomon, M. (1983) The Role of Products as Social Stimuli: A Symbolic Interactionism Perspective. *Journal of Consumer Research*. 10(3), 319-329.

Solomon, M.R. (1986) The Missing Link: Surrogate Consumers in the Marketing Chain. *Journal of Marketing*. 50(4), 208–18.

Solomon, M.R., Assael, H. (1987) The Forest or The Trees? A Gestalt Approach to Symbolic Consumption. In S.J. Umiker (ed.) *Marketing and Semiotics: New Directions in the Study of Signs for Sale*. Mouton de Gruyter: Berlin. 189-218.

Southerton, D. (2002) Boundaries of 'Us' and 'Them': Class Mobility and Identification in a New Town. *Sociology*. 36(1), 171-93

Spiggle, S. (1994) Analysis and Interpretation of Qualitative Data in Consumer Research. *Journal of Consumer Research*. 21(3), 491-503.

Spradley, J.P. (1979) *The Ethnographic Interview*. Wadsworth Group/Thomas Learning: United States.

Spradley, J.P. (1980) *Participant Observation*. Holt, Rinehart and Winston: New York, USA.

Stafford, B.M. (2007) *Echo objects: The cognitive work of images*. University of Chicago Press: Chicago, USA.

Steele, V. (2006) *Fifty Years of Fashion - New Look to Now*. Yale University Press: New Haven, London, UK.

Stenross, B. (1994) Aesthetics in the Marketplace: Collectors in the Gun Business. *Qualitative Sociology*. 17(1), 29-42.

Stern, B.B. (1995) Consumer Myths: Frye's Taxonomy and the Structural Analysis of Consumption Text. *Journal of Consumer Research*. 22(2), 165-85.

Stewart, S. (2009) [online] Farewell to Glasgow's famous Paddy's market after 200 years, Available at: <u>http://www.dailyrecord.co.uk/news/uk-world-news/farewell-to-glasgows-famous-paddys-market-1023009</u> [accessed on May 29th 2014].

Strasser, S. (1989) *Satisfaction Guaranteed: The Making of the American Mass Market*. Smithsonian Institution Press: Random House, USA.

Strasser, S. (1999) *Waste and Want: A Social History of Trash.* Metropolitan Books: New York, USA.

Strathern, M. (1992) Qualified Value: The Perspective of Gift Exchange, in C. Humphrey and S. Hugh- Jones (ed.) *Barter, Exchange and Value: An Anthropological Approach*. Cambridge University Press: Cambridge, UK. 169–91.

Strauss, A., Corbin J., (2008) *Basics of Qualitative Research: Techniques and Procedures for Developing Grounded Theory (*2nd Edition). Sage Publications: London, UK.

Swain, S. (2011) Modern Vintage. Evening Times. published 4th May 2011.

Szmigin, I., Foxall, G., (2000) Interpretive consumer research: how far have we come? *Qualitative Market Research: An International Journal*. 3(4), 187–197.

Tadajewski, M. (2006) Remembering motivation research: toward an alternative genealogy of interpretive consumer research. *Marketing Theory*. December. 6(4), 429-466.

Tarnanidis, T.K., Owusu-Frimpong, N., Marciniak, R. (2010) Consumer choice: between explicit and implicit reference points. *The Marketing Review*. 10(3), 269-286.

Tarrant, N. (1983) *Collecting Costume: The Care and Display of Clothes and Accessories.* George Allen and Uniwin, London, UK.

The Barras Market [online] Available at <u>http://www.theglasgowbarras.com/</u> [accessed on May 28th 2014].

Thévenot, L. (2006) Convention school. *International Encyclopedia of Economic Sociology*. Routledge: London. 111-115.

Thompson, C.J. (1990) Eureka! And Others Tests of Significance: A New Look at Evaluating Interpretive Research. *Advances in Consumer Research*. 17, 25-30.

Thompson, C.J. (1997) Interpreting Consumers: A Hermeneutical Framework for Deriving Marketing Insights from the Texts of Consumers' Consumption Stories. *Journal of Marketing Research*. 34, 438–55.

Thompson, C.J. (2004) Marketplace Mythology and Discourses of Power. *Journal of Consumer Research*. 31(1), 162–80.

Thompson, C.J., Arnould, E.J., Giesler, M. (2013) Discursivity, difference, and disruption Genealogical reflections on the consumer culture theory heteroglossia. *Marketing Theory*. 13(2), 149-174.

Thompson, C.J., Arsel, Z. (2004) The Starbucks brandscape and consumers'(anticorporate) experiences of glocalization. *Journal of Consumer Research*. 31(3), 631-642.

Thompson, C.J., Gokcen, C. (2007) Countervailing Market Responses to Corporate Co-optation and the Ideological Recruitment of Consumption Communities. *Journal of Consumer Research*. 34 (2), 135–51.

Thompson, C.J., Haytko, D.L. (1997) Speaking of Fashion: Consumers' Uses of Fashion Discourses and the Appropriation of Countervailing Cultural Meanings. *Journal of Consumer Research*. 24(1), 15-42.

Thompson, C.J., Hirschman, E.C. (1995) Understanding the Socialized Body: A Poststructuralist Analysis of Consumers' Self-Conceptions, Body Images, and Self-Care Practices. *Journal of Consumer Research*. 22(2), 139–53.

Thompson, C.J., Locander, W.B., Pollio, H.R. (1989) Putting Consumer Experience Back into Consumer Research: The Philosophy and Method of Existential Phenomenology. *Journal of Consumer Research*. 16(2), 133-146.

Thompson, C.J., Tian K. (2008) Reconstructing the South: How Commercial Myths Compete for Identity Value through the Ideological Shaping of Popular Memories and Countermemories. *Journal of Consumer Research*. 34(5), 595–613.

Thompson, C.J., Troester, M. (2002) Consumer Value Systems in the Age of Postmodern Fragmentation: The Case of the Natural Health Microculture. *Journal of Consumer Research*. 28(4), 550-71.

Thompson, M. (1976) Rubbish Theory. Oxford University Press: Oxford, UK.

Thompson, M. (1979) Rubbish Theory. Oxford University Press: Oxford, UK.

Thompson, M., Ellis, R., Wildavsky, A. (1990) *Cultural Theory*. Boulder: Westview Press.

Thornton, S. (1995) *Club Cultures: Music, Media and Subcultural Capital*. Polity: Cambridge, UK.

Thornton, S. (1996) *Club Cultures: Music, Media, and Subcultural Capital.* University Press of New England: Hanover, NH.

Thrift, N.J. (1996) Spatial formations. Sage: London, UK.

Thrift, N.J. (1997) The still point: resistance, expressive embodiment and dance. In Pile S. and Keith M. (ed.) *Geographies of resistance*. Routledge: London UK. 124–51.

Thrift, N.J. (1999) Steps to an ecology of place. In Massey, D., Allen, J. and Sarre, P., (ed.) *Human Geography Today*. Polity Press: Cambridge. 295–322.

TIME Magazine (1966) [online] London the Swinging City, April 15th 1966, Available at <u>http://content.time.com/time/covers/0,16641,19660415,00.html</u> [accessed on March 18th 2014].

Toffler, A. (1980) *The third wave: The classic study of tomorrow*. Bantam: New York, NY.

Trend Watching.com (2004) [online] Pop-up Retail, Available at http://trendwatching.com/trends/POPUP_RETAIL.htm [accessed on May 15th 2013].

Trend Watching.com (2009) [online] Nowism Why Currency is the New Currency - <u>http://trendwatching.com/trends/nowism/</u> [accessed on May 28th 2014].

Trentmann, F. (2009) Materiality in the future of history: things, practices, and politics. *The Journal of British Studies*. 48(2), 283-307.

Truninger, M. (2011) Cooking with Bimby in a Moment of Recruitment: Exploring Conventions and Practice Perspectives. *Journal of Consumer Culture*. 11(1), 37–59.

Tuan, Y. F. (1977) *Space and place: The perspective of experience*. University of Minnesota Press.

Tungate M. (2008) *Fashion Brands: Branding Style from Armani to Zara*, 2nd Edition. Kogan Press: London, UK.

Türe, M. (2014) Value-in-disposition: Exploring how consumers derive value from disposition of possessions. *Marketing Theory*. 1(14), 53-72.

Turkle, S. (2007) *Evocative Objects: Things We Think With*. MIT Press: Cambridge Massachusetts.

Turner, B.S., Rojek, C. (2000) Decorative Sociology: Towards a Critique of the Cultural Turn. *Sociological Review*. 48, 629–48.

Turner, B.S., Rojek, C. (2001) *Society and Culture: Principles of Scarcity and Solidarity.* Sage: London, UK.

Turner, V. (1995) *The ritual process: Structure and anti-structure*. Transaction Publishers.

Üstüner, T., Holt, D.B. (2007) Dominated Consumer Acculturation: The Social Construction of Poor Migrant Women's Consumer Identity Projects in a Turkish Squatter. *Journal of Consumer Research*. 34(1), 41-56.

Üstüner ,T., Holt, D.B. (2010) Toward a Theory of Status Consumption in Less Industrialized Countries. *Journal of Consumer Research*. 37(1), 37–56.

Üstüner, T., Thompson, C.J. (2012) How marketplace performances produce interdependent status games and contested forms of symbolic capital. *Journal of Consumer Research*. 38(5), 796-814.

Vargo, S.L., Lusch, R.F. (2004) Evolving to a New Dominant Logic for Marketing. *Journal of Marketing*. 68(1),1-17.

Vargo, S.L., Maglio, P.P., Akaka, M.A. (2008) On Value and Value Co-creation: A Service Systems and Service Logic Perspective. *European Management Journal*. 26(3), 145–52.

Veblen, T. (1899) *The theory of the leisure class: An economic study in the evolution of institutions*. Macmillan and Company Limited.

Veenstra, A., Kuipers, G. (2013) It is Not Old-Fashioned, It is Vintage, Vintage Fashion and The Complexities of 21st Century Consumption Practices. *Sociology Compass.* 7(5), 355-365.

Venkatesh, A. (1992) Postmodernism, Consumer Culture and the Society of the Spectacle. *Advances in Consumer Research*. 19, 199-202.

Venkatesh, A. (1995) Ethnoconsumerism: A New Paradigm to Study Cultural and Cross-Cultural Consumer Behavior In Costa J.A., Bamossy G. (ed.) *Marketing in a Multicultural World*. Sage: Thousand Oaks, CA.

Venkatesh, A., Meamber, L.A. (2006) Arts and Aesthetics: Marketing and Cultural Production. *Marketing Theory*. 6(1), 11-39.

Ventakesh, A., Meamber, L.A. (2008) The Aesthetics of Consumption and the Consumer as Aesthetic Subject. *Consumption, Markets and Culture*. 11(1), 45–70.

Venkatesh, A., Peñaloza, L. (2014) The value of value in CCT. *Marketing Theory*. 14(1), 135-138.

Vestiaire Collective [online] Available at <u>http://www.vestiairecollective.com/</u> [accessed on May 15th 2013.]

Vintage by Hemingway [online] <u>http://www.vintagefestival.co.uk/about-page/about-page--2</u> (accessed on June 24th 2013).

Vintage Fashion Guild [online] Granny Takes a Trip, Available at <u>http://vintagefashionguild.org/label-resource/granny-takes-a-trip/</u> [accessed on April 22nd 2014].

Vintage Festival http://www.vintagefestival.co.uk/about-page/about-page--2, [accessed on July 2nd 2013]

Vintage Festival by Hemmingway Design [online] Available at <u>http://www.vintagefestival.co.uk/about-page/about-page--2 [accessed on May 16th 2013].</u>

Visconti, L.M., Sherry, J.F. Jr., Borghini, S., Anderson, L. (2010) Street Art, Sweet Art? Reclaiming the "Public" in Public Place. *Journal of Consumer Research*. 37(3), 511-529.

Von Hippel, E. (2005) Democratizing innovation. MIT Press.

Wallendorf, M., Arnould, E.J. (1988) 'My favourite things:' A cross-cultural inquiry into object attachment, possessiveness and social linkage. *Journal of Consumer Research*. 14(4), 531-47.

Wallendorf, M., Arnould, E.J. (1991) 'We Gather Together': Consumption Rituals of Thanksgiving Day. *Journal of Consumer Research*. 18(1), 13–31.

Wallendorf, M., Belk, R.W. (1989) Assessing Trustworthiness in Naturalistic Consumer Research. In Hirschman E.C (ed.) Interpretive Consumer Research. Provo, UT. *Association for Consumer Research*. 69-84. Wallendorf, M., Brucks, M. (1993) Introspection in Consumer Research: Implementation and Implications. *Journal of Consumer Research*. 20(3), 339–59.

Warde, A. (1996) Afterword: The Future of the Sociology of Consumption. In S.Edgell, K.Hetherington, A.Warde (ed.). *Consumption Matters*. (1996), 302-12.

Warde, A. (1997) Consumption, Food, and Taste: Culinary Antinomies and Commodity Culture. Sage: London, UK.

Warde, A. (2004) Practice and Field: revising Bourdieusian concepts. CRIC Discussion Paper, no 56. Centre for Research on Innovation and Competition. The University of Manchester.

Warde, A. (2005) Consumption and Theories of Practice. *Journal of Consumer Culture*. 5(2), 131-153.

Warde, A. (2007) Does taste still serve power? The fate of distinction in Britain. *Sociologica*. 3, 1-26.

Watson, M., Shove, E. (2008) Product, Competence, Project and Practice: DIY and the dynamics of craft consumption. *Journal of Consumer Culture*. 8(1), 69-89.

Watson, S. (2009) The Magic of the Marketplace: Sociality in a Neglected Public Space. *Urban Studies*. 46(8), 1577-1591.

Watters, J.K., Biernacki, P. (1989) Targeted sampling: options for the study of hidden populations. *Social Problems*. 36(4), 416-430.

Weber, M. (1947) *The Theory of Social and Economic Organization*, translated by A. M. Henderson and Talcott Parsons. Edited with an introduction by T. Parsons. Free Press: New York, USA.

Wenger, E.C. (1987) Artificial Intelligence and Tutoring Systems. Los Altos: Morgan Kaufmann.

Wenger, E.C. (1998) *Communities of Practice: Learning, Meaning and Identity.* Cambridge University Press: Cambridge.

Wenger, E.C. (2000) Communities of Practice and Social Learning Systems. *Organization*. 7(2), 225-46.

Wenger, E.C., Snyder, W.M. (2000) Communities of Practice: The Organisational Frontier. President and Fellows Harvard University College. *Harvard Business Review*. Jan-Feb.

Whittington, R. (2004) Strategy after modernism: recovering practice. *European Management Review*. 1(1), 62-68.

Whittington, R. (2006) Completing the practice turn in strategy research. *Organization Studies*. 27(5), 613-634.

Whittington, R. (2007) Strategy practice and strategy process: family differences and the sociological eye. *Organization Studies*. 28(10), 1575-1586.

Whittington, R., Jarzabkowski, P., Mayer, M., Mounoud, E., Nahapiet, J., Rouleau, L. (2003) Taking strategy seriously: Responsibility and reform for an important social practice. *Journal of Management Inquiry*. 12(4), 396-409.

Wikström, S. (1996) The Customer as Co-producer. *European Journal of Marketing*. 30(4), 6-19.

Williams, C.C., Paddock, C. (2003) Reconciling economic and cultural explanations for participation in alternative consumption spaces. *Geografiska Annaler: Series B, Human Geography*. 85(3), 137-148.

Willis, P. (1978) Profane Culture. Routledge and K. Paul: London, UK.

Willis, P. (1982) The motor-bike and motor-bike culture. *Popular Culture: Past and Present.* 284-93.

Willis, P. (1990) *Moving culture: an enquiry into the cultural activities of young people.* Gulbenkian (Calouste) Foundation: London, UK.

Willis, P. (2000) The Ethnographic Imagination. Polity Press: Cambridge, UK.

Willis, P., Trondman, M. (2000) Manifesto for Ethnography. *Ethnography*. Sage Publications Ltd.

Wittgenstein, L., Winch, P. (1984) *Culture and Value*. University of Chicago Press: Chicago, US.

Wolcott, H.F. (1999) Ethnography: A Way of Seeing. AltaMira Press: Oxford, UK.

Woodruff, R.B., Flint, D.J. (2006) Marketing's service-dominant logic and customer value. *The service-dominant logic of marketing: Dialog, debate, and directions*.183-195.

Woodward, S. (2007) Why Women Wear What They Wear. Berg: Oxford, UK.

Woolgar, S. (1988) Reflexivity is the Ethnographer of the Text, 14-34 in Woolgar S. (ed.) *Knowledge and Reflexivity: New Frontiers in the Sociology of Knowledge*. Sage: London, UK.

Wooten, D.B. (2000) Qualitative Steps toward an Expanded Model of Anxiety in Gift-Giving. *Journal of Consumer Research*. 27(1), 84–95.

Xie, C., Bagozzi, R.P., Troye, S.V. (2008) Trying to Prosume: Toward a Theory of Consumers as Co-creators of Value. *Journal of the Academy of Marketing Science*. 36(1), 109-22.

Yanow, D., Schwartz-Shea, P. (2013) (ed.) *Interpretation and method: Empirical research methods and the interpretive turn.* ME Sharpe.

Yip, N.K.T. (2012) Making Qualitative Decisions from Quantitative Cues: Understanding the Customers' Willingness to Pay. *Journal of Revenue and Pricing Management*.11(5), 562–6.

Zwick, D., Bonsu, S., Darmody, A. (2008) Putting Consumers to Work Co-creation and new marketing govern-mentality. *Journal of Consumer Culture*. 8(2), 163-196.

Zwick, D., Dholakia, N. (2006) The Epistemic Consumption Object and Post-social Consumption: Expanding Consumer-object Theory in Consumer Research. *Consumption, Markets and Culture*. 9(1), 17-43.

Appendix i - Sample Depth interview

Michael 'The Eyedresser' -November 2011

Researcher (**R**) – Thank you very much for giving up your time today.

Michael (M) – Happy to – you have probably got a whole list of questions.

R- It is more a discussion, don't worry. The main thing I have had to do is to try to rein people in.

- Researcher briefs respondent on purpose of interview and that they can ask to stop at any time.

R- We met in the Granny Would be Proud market and then I visited your optician practice. Can I ask to start with how did you get started in selling vintage frames?

M- It wasn't on purpose, I was quietly amassing the vintage collection for my own enjoyment, and it wasn't ever with a thought to selling them. I just liked them for myself. You know I started out in the city centre in the Argyle Arcade? So I had the place called Eyedresser at that point, it wasn't vintage at that stage but I was quietly amassing the vintage collection of my own for my own enjoyment, it wasn't ever with the thought of selling them, just liked them for myself, it was like putting something down for the future, you know, but so emm I think that business suffered from being in the city centre, it is very much better now that I am predominantly working out of the west end, and the Gin in Teacups (city centre vintage market) thing what can I say it highlighted that point. It was an experiment; it wasn't as good as the west end. So there is something inherent about the west end and you cake make your own mind up about that, it is probably student based, but not really, most of my clients are people like yourself you are young, but they are older in spirit, young professionals, newly qualified, whatever got their own businesses or entrepreneurial whatever, they don't tend to be the undergraduate student types in the main. They are probably just too short of money to indulge in luxuries like that maybe often so whatever it is that appeals to people to stay in the west end seems to tie in with this vintage trend.

R- It is interesting that you say that it is quite location based.

M- Well it is has taken me so long to park in the city today. I mean it took me the best part of thirty minutes to park and I don't really come into shop in the city centre often but I always choose not to because of the hassles involved. Which I do think that it must be part and parcel of why there is not more vintage shops or fairs in the city centre, apart from the expense of rent and rates and so forth, emm I mean I couldn't go into the centre, I've looked at premises in the city centre, I couldn't go in - I have looked at premises even in upstairs of offices style places inside old tenement buildings, wee places were you would just have a cupboard in the city centre and the cost of either renting or purchasing would hardly make it worth my while doing it in the first place - that's why I choose to do it on the road now, emm so pop shops that kind of thing, it all works for me. There is something essentially anti entrepreneurial in a small business terms about the city centre which I think is a fault no question about it, the thing is that Glasgow got itself renowned for things like Ingram St, Cruise and all the rest of it. I saw it as a student twenty five years ago and it has just stuck to that image, the American and Japanese tourists like that but I think that the local, indigenous population would rather have row after row of

boutique shops and it would make it an absolute world beater if they did that. I will go on record as saying that (laughs) Glasgow City Council, listen up because the city centre will die with the way it is heading. Let's face it there is nothing that you can buy in the city centre that you couldn't buy on the internet when you think about it now in theory, so people they come in for coffee shops, for social interactions, museums and all the rest of it, that's fine, but would it be wrong to have the whole thing with at least a sprinkling of vintage shops interspersed in amongst all the big guys, you know, all the big guys would benefit as well. I don't think it has to be, you know all the Topman and stuff like that, they would benefit from more people coming into the city centre and I mean don't even start me on out of town shopping centres, I don't know how they started up, I mean that's a complete, there would be a complete impossibility to opening up a small store in a big shopping centre like Braehead or whatever, so I don't know, my thought for the future is that when you know the petrol price goes through you know five pounds a gallon that these places will just become ghost towns, ironically people will flood back into the city centre again. The Internet will have destroyed the big corporations and it will be boutique shops that Glasgow City Council will be trying to persuade to come back here again. It will go back full circle to what it might have been like in the fifties and sixties you know. I hope so. I don't think I will live long enough to see it though (laughs). Sorry it is something I feel passionately about.

R - I kind of wanted to get an idea of where you started and where you got the idea from, moving from personal collecting into marketplace?

M - It was a fluke, you mean vintage in particular? It was a total fluke. One of the girls that works for me, in a Saturday job, you have probably met her, and she

worked as well in the evenings as a waitress in a restaurant in Argyle St, and she just came to one Saturday and said I have heard through the grapevine that they are having a vintage fair in the restaurant that I work in on a Sunday, the restaurant wasn't open so they were using the space, and emm I told them that you had vintage glasses, she must have known this somehow, I don't know I must of told her, but she told them I had a vintage glasses collection and would I want a stall and there is a stall waiting for him if he wants it, tell him he can have it and I went along and gave it a try. And that was that. I was speaking to the Granny Would be Proud people the other day, Elena and Adam, and ehh I hadn't actually appreciated this but I was actually at the first one, I was there from the beginning, that one was the first one I was invited to, I didn't realise that at the time I thought it had been going for a while, months. So that's it I've been there since the beginning.

R – That is quite special, there can't be many sellers that are still there that have been there since the start?

M – No there is only the three of us that have been there since the start and funnily enough those people are kind of mainstays if you know what I mean, they are emm people who make a living out of selling vintage or whatever, jewellery whatever, so they are not - they weren't just people who had a stall as a hobby or had a stall because they fancied it on a particular weekend - they were people who actually make a living out of doing this so to be honest with you that's...I think about this all the time, if she hadn't come to me that day, if she hadn't been at work that day, would I have found out about that? I personally don't think because I wasn't linked with Facebook at the time or anything like that, I wasn't involved in any social networking, to this day two years on, I probably would never have heard of it. Which is an interesting thing to think about, if someone like me, who is so interested in vintage would probably still not know about it. That always quizzes me, so are we missing a huge sector of the audience of people like me who want to know about it but don't have the means of finding out? Cause there is no presence, no shop presence, no obvious outward presence, apart from social networking sites and all the rest of it. So now ironically that has drawn me into social networking and that has now significantly improved the fortunes of my business but that's, somewhere in that question why would I not have known about it, might be the answer to the question why it is not a bigger phenomenon that it is even just now. Do you know what I mean? I might be the reason why shops work so well, because of their constant presence and constant reminder that they are there.

R - I think that is an interesting point. The Granny Would be Proud market always seems quite fleeting on a Sunday afternoon, by a Monday there is no sign that it was there or will be there again.

M - I think that's, so is it thinking deeply about that, is it that that makes it work? Is it that slight exclusivity that makes the people that do go to it more excited about it and less mass market? I mean if it was more mass market I wouldn't be, ironically so into it. You know. It appals me when people like Specsavers get involved in the selling of vintage looking glasses or marketing using vintage images, you know retro glasses, because they have got no right, absolutely no right to be involved in that market at all, they have no interest in spectacles or the aesthetic, they have no interest, it's all money for shareholders and owners. There is no passion behind it, they just jump in, kill it dead in six months and move onto the next thing. I mean no sooner were they into remanufacturing retro 1950s and 60s glasses, then all of a

347

sudden they announced that the 80s trend was the next big thing. And you are thinking you know you are jumping around here, it is not based on anything. That's an interesting thing from my point of view, no one ever talks about fashion or fads or trends or anything like that, people come to and they genuinely want a pair of glasses, not they want a pair because my friend's got a pair or I want a pair because I want to look like such and such on the tele or whatever, they seem to nine times out of ten because they genuinely want to get away from the mainstream, they want to get away from all of this what everybody else is wearing. With vintage it is not that they are moving to something, they are moving away from something and trying to be themselves and trying to be more, it's not disingenuous, they are kind of, in my opinion doing it for the right reasons to create a look for themselves that is honest to them not just copy fashion for the sake of it. Which I think is very healthy you know.

R – Can you tell me about your customers?

M – They are usually enthusiastic and style driven, they care. Often again, they are just looking for the pair of glasses that suits them. They are not sure - what you have to understand is that spectacles are a very, very specialised industry and I notice this when I tour around, when I go to other cities, and I go into the vintage shops and ask 'do you have any vintage glasses for sale' and they say 'oh no we don't deal with that Sir.' Almost in hush tones as if to say oh no that's not an area of the market we don't have any expertise in so we don't dabble in it, as if there is some mystique about it, so they deal with everything else, they deal with all sorts of other clothing and accessories and so forth, jewellery and watches but as far as glasses are concerned that's like 'no that's something outside our sphere. 'I mean it's good for me, and I think the public are the same with vintage it is almost like a black art, you
know they do, emm they do like to ask my advice and leave it up to me a lot of the time, as if they don't trust themselves to make up their own mind. There are one or two obvious exceptions to that rule that know exactly what they are looking for, I mean they have a particular image in mind and they've got the thing set up for themselves, they are very image conscious and they are very much aware of what suits them and what they look good in, the colours, I love dealing with that too. But I'd say that 75% of the time, in fact that's quite often my opening gambit, 'what era or what look are you trying to create' and people say 'oh I don't really know, just something that suits me'.

R- It seems like quite a personal chat with them in a way.

M – Yes I mean it is quite complex and I need to get to know them a bit. I mean what they associate with, what history they like. In terms of history of the pieces, yeah I think on that basis as well, on a similar related note to that, it is very, very often because they (glasses) didn't have a CE marking or anything like that back then, back until about the 90s they didn't have any of this stuff so. So there was no cataloguing, or dating or identifying markings on these glasses so it wasn't necessary, a lot of them were hand finished, they just had a stamp on them and that was the only way to identify them, the identifying markings would have been on a backing card which was lost in most cases and is now lost to the system, so but I have to say that it is quite often impossible for me to tell what year a frame is from because since people started wearing glasses there has always been this backwards forwards cross over so there might be a trend for five or ten years, or maybe less than that back then trends lasted a shorter time, a season maybe six months miniskirts may have lasted for six months, then box skirts after that and then miniskirts again and

then trousers would come in, there was this recycling of the fashion. Now, in my opinion anything goes, the whole market is open to everyone and everyone can choose all styles, or history they can choose vintage or retro copies, originals, they can go into Topshop and buy the current trends whatever they want to do. But it is more, it is not to do with 'oh I'll have a mini skirt cos this year that what's on trend,' yeah? And what would have happened back then is a look say in the late 50s may have fallen by the wayside for five years, this is what I am thinking with the whole of my collection and then they might have resurrected it in the early 60s or 70s. So they had at various points in time retro revivals of old fashioned styles and after all in the 1970s, 50s glasses would have been fifteen years ago, it was long enough that some people would have forgotten about them in the first place and young people would have come on board and liked them the second time round, you know. Maybe that has happened two or three times, you know it is very, very hard for me to tell the age of a pair of glasses from the look of it and I am beginning to realise more and more that it is irrelevant to vintage. It is actually how it suits you that matters and even valuing the glasses I've stopped so much valuing them on the age of them, it is not like antique value, it is not that, it is intrinsic the quality, rarity and the look of them. Do you know what I mean? It is the, how unique or individual they are or whatever, how irreplaceable they are not whether or not they are fifty or forty or twenty years old, are you with me?

R- That's an interesting idea about how irreplaceable they are.

M- I mean for example I have five or six of Alexander McQueen's original first catalogue range and the instant he passed away, I just took them away, literally took them out, polished them up, put them away, thinking who's to know they have incredible historical value in fifteen years' time and that will be a real talking point. It is not the sort of thing I should just sell and have someone wear it just now and wear it to destruction and chuck it in the bin, that's clearly something of intrinsic historical value in the grand scheme of things. Emm so that was a rare occasion where I could actually put away something that would become collectable in the future, normally I am buying things that I know are collectable and are keeping them to make them even more collectable but it is not often that something, you know it would be like having an original Mary Quant dress or something from the 60s. Who would have thought to actually keep one of those in its wrapper and stick it in a drawer and never wear it at the time, so that you could pull it out in fifty years' time for your daughter, you know what I mean. I'm sure lots of people did that kind of thing, so that was my wee attempt to just to try that you know, future items of historical significance you know. But I think it is all, it is hard to see that we could ever go back to that seasonal trend thing again now, don't you think. You are a woman you are probably much more in touch with that than me, magazines and those kinds of things, I don't read those kinds of things that tell me what's in one month or another. So it does seem to me that when I was a student and you would pick up Cosmopolitan and it would say, this summer's hot fashion and everyone would have it or everyone would be trying to copy it, they would go to Topshop and get the cheaper copy or the original designer name or you know and the streets would be awash with people wearing the same sort of thing like long coats one season and short coats the next, hoods no hoods, trousers, skirts whatever and that was in my cynical view the market's way of leading people into changing their wardrobe every six months or year clearly, it is all gone bang a wee bit now. I think companies like

Topshop are now latching onto vintage trends, it is almost like they are back tracking to try to keep up with this mass market idea that everybody wants everything at the same time, it's all you would think it would be very confusing but I think that the individual knows what they want and they've got enough avenues now to go down and source these things, vintage fairs, high street shops, markets, eBay there are plenty of ways to find what you are looking for.

R – Can you tell me more about that point around vintage and high street retailers?

M – They have a habit of mixing up retro with vintage. In other words as long as it is based on a vintage look, I mean I am very, very strict with myself about referring to vintage. My definition of vintage is something that is actually of the era that it is from, if you know what I mean by that, so in other words, something aged in some way. Retro is a copy of a vintage look and when I am selling glasses I am very careful to define the two as separate and make sure that people understand the difference and what they are buying, I can't tell you the number of times in the market that I have had people in saying 'I bought this pair of vintage glasses over on eBay the other day, what do you think of them, can you put lenses in them' and I look at them and they are some £2.99 frame from China or a copy which Topshop would sell for ± 12 or something and you don't know how to tell them that it is not a vintage frame, cause they say 'yes it is it said it in the advert'- it is a copy of a vintage frame and not a very good one at that and I've stopped arguing with people about it now cause people get very 'hoighty toighty' about it because people are protective of their purchases, so you can't really contradict people but at the end of the day you, I find that annoying when you sell the real thing, it is stopping trade and devaluing what I do. But going back to what I was saying earlier on, that doesn't

seem to be my audience anymore, people when they come to me and are buying vintage more often than nought, I've started doing good quality retro as well now, because the vintage glasses are a finite market or a finite supply I've got to try and do something to plug the gaps now, there are various gaps needing plugged and I will bring in decent copies or retro styles, copies of that original look just to try and satisfy that market but almost in a way to try and stop that from people going and getting the £2.99 spectacles from eBay which are a bit of a waste of time, but emm so that's not really my customer, my customers are not are more and more not that type of person now, they come to me because they know what I do, all of that doesn't need explained anymore I can get straight to the point and do the eyedressing thing you know.

R- That is an interesting point that you make about the finite nature of vintage. Do you find it harder to source stock now?

M- Almost impossible, I can tell you that since I started doing this two years ago I have possibly had a dozen frames come my way, so I am on a dwindling supply here and ehh if I carry on like this for five years I will have nothing left as simple as that. Which you have to philosophical about it. But as I saw on Flog it yesterday afternoon that the chap said vintage is the ultimate form of recycling or antiques in his case, it was antique silver milk jug that he was selling and I mean silver doesn't corrode or tarnish or age. I mean it does tarnish but it doesn't destroy itself with time, it is there a little piece that someone has crafted out of a piece of silver, that has been passed around from person to person down the centuries and it will never be binned so it is a fantastic form of recycling taking that piece of metal and changed it into something that people will keep and covet and that they will keep it in their

possession, polishing it away and all the rest of it. So it is the same with the glasses, so what will happen presumably over time is that as people don't destroy their glasses, say no one ever sat on their glasses ever again, they all remained in the pool as it were these glasses would come back to me eventually but I would have to pay a higher market value for them and that's what is pushing the price all the time, so maybe in twenty years' time I might get my hands on some lovely fifties glasses but by that time they will be seventy or eighty years old and they will be ten times more expensive and I will have to mark them up slightly higher and sell them on again, and it will just be those and such as those who can actually afford to get them and that is an odd dynamic it becomes very, what will happen when these glasses are still in the system and they become antiques it is bound to change the audience somehow.

R- That is an interesting concept of how they will age.

M- yeah, like on TV last night, I was watching a bit of TV for some reason last night, they were talking about foraging for food, mushrooms berries etc., the same basic idea, the unique forces of the recession that people think about foraging and thrifting and all the rest of it or would they be doing it anyway just now? It is a question I quite often ask myself, you know, say the recession ended tomorrow and all of a sudden magically there was money back in the system, would people suddenly start going back to Cruise, you know I don't think so. I think it is a change this time, I think it is more linked with recycling subconsciously in a lot of people's minds like mine rather than price or market forces or anything like that, it pleases me when you go to Granny Would be Proud that a lot of what is ending up out of landfill and all the rest of it, you know. Because someone has taken the care to make those things in

the first place, it seems a shame to throw them in the bin. What do you think; I will throw it back at you?

R- I think you mention some interesting points. I think that in some ways there is a slight move away from conspicuous consumption and people are more conscious of things and where they go.

M- Exactly. Right now, I've got a pair of Ralph Lauren trousers on that I bought in TK Maxx, why did I buy them in TK Maxx and not go into the Ralph Lauren store and pay £120? It is because my experience of buying designer glasses tells me that Ralph Lauren, is an organisation, it is an industry, it is not Ralph Lauren sitting at a cutting room in the back of his workshop or whatever cutting out trousers and selling them to me for a £120 a throw, it is all horse trading is what I call it. If I could afford to buy young designers, you know like people out of the art school who made a pair of jeans or whatever fantastic that's what I am talking about buying quality something that someone has put their sweat and blood into it shows in the final product you know. And that is presumably how the big names started out today, but as it happens I know that the Ralph Lauren glasses come out of a head office in Italy which sells probably 50% of the world's designer glasses from the one head office, none of these designer names are lined in real life other than through spectacles, the people at Ralph Lauren will never even have seen them, they use the name under licence to sell glasses and they ram the price up because of it. This is a fact it is not an urban myth, it is a bit of an urban myth that people have thought that you pay for the name, this is absolutely the truth with glasses, you pay for the name, there is no reason to think that that factory is any different than the one that Specsavers glasses from. And what Specsavers try to do is develop, develop their brand, they are the 8th

biggest advertisers in the UK and it shows they are everywhere, talking of semantics it doesn't have the same cache as Ralph Lauren, the name Specsavers but they have made, it is like Tesco, it is acceptable to talk about in common circles, you are no longer a cheapskate to go to Specsavers to get your glasses, so they have achieved what they set out to do. I totally think quality is what it is all about, the Ralph Lauren, Christian Dior glasses from the 1970s that I have seen are way beyond today's quality they are classics, they are absolutely beautiful frames, I mean they are lovely, I just love them, I just, you can feel the difference, they look different there is a lustre about them, there is something built at the time and it was still there today, they don't have that mass market shadow hanging over them, there is still that exclusivity, that we about them based on an excitement because they are just lovely. It was just holding this thing in my hand, maybe the Christian Dior design house rubber stamped the design and though oh that represents our brand properly or we will let you use that one but it fits in with what they were trying to achieve. So at the moment I've got those Ralph Lauren trousers on from TK Maxx, I don't feel I am being ripped off if I get them there, I get them for what I think they are worth, I don't think that they are any better quality than what I would get out of Topman or whatever. It just so happens that I like the way TK Maxx do their shops, it is unobtrusive, you just go in, no one annoys you, you can shop around, try them on, do what you want, it is easy shopping. But at the same time I have a Tesco cardigan on that my friend got me a couple of years ago, it must have cost me about £2.99 with his club card vouchers you know and I just think it is nothing to do with names or vintage or who designed it but it is just whether it suits you, is it comfortable, do you enjoy wearing it. I suppose can you afford it comes into it, so in other words you've

got to be able to afford it in the first place, do you love it so much that you would go and spend £120 on it, which in the past you might have done. In the past I've done it, I've went into Cruise and bought Jeffrey West shoes at £150 a throw, you just felt like sex on legs, I mean it totally changed your whole opinion of yourself, it just elevated your thoughts about yourself for a wee while. And then it's fabulous but I wouldn't do it every day you know, but at the same time I've got a vintage shirt on from Armstrong's in Edinburgh, another place I like shopping. Again you are left to your own devices, you can rummage around and all the rest of it, I think I probably bought it there - I have vintage shirts that I bought in the Victorian Village when I was a student. I mean I've done all of this the first time round so the funny thing was there wasn't a big audience at the time, I really, really thought I was an oddball back then, I wore collarless shirts which I had bought Edwardian ones or Georgian ones that would be maybe early twentieth century, stud collared shirts and all the rest of it to work and things like that with bow ties and all the rest of it, people kind of thought that you were a bit odd, you stood out from the crowd, I mean that wasn't my intention for doing it but now that's not the case, you feel as if you are part of a bigger club now and it is really, really quite nice to think now that at my age, I can revisit this with the same clothes I was wearing twenty years ago and people think it less strange now than they did back then.

R- Why do you think that is?

M- I think young people have become more open minded, this was what I was thinking about in the car on the way over that I was going to tell you. I think social networking has a lot to do with it, when I was 26 for example, say you had a particular thing for vintage clothes, you would maybe meet other people who liked

vintage clothes at the vintage clothes shop but the chances of hitting it off with them or going for a pint with them were pretty slim, that random chance meeting people is very, very unusual even to this day. But because now people can express their interests up front on Facebook and places like that, there is more interaction with people, with groups that are like-minded, and you don't necessarily have to meet them, just knowing that they are out there, I think maybe gives people that maybe don't have the absolute confidence, which I didn't have at the time, I was very, very shy, it's almost like you needed a leg up and I'm not saying that's right but it is maybe in the Scots nature we are quite shy, there's a reserved nature about the Scots, I think nowadays at least knowing there are people out there who are like yourself, you don't have to hob knobbing with them on a Friday night or whatever it maybe just is that group psychology of human beings that it is nice to know you are not alone. There are true individuals and there are people, the peacock type people, who just want to look as individual as they can and they have extreme true confidence just to be, and I admire them but I have never been one of those. I think lots of young students sitting there are in their bedsits, plugging away on the computer, chatting to people online going' I'm going to Granny Would be Proud today and all the rest of it' just I think it gives people that little edge of confidence to go out and strut stuff and be proud of the fact that they do enjoy looking or appearing to look the way they look. You know, it sounds very image conscious and all the rest of it but are we not all to a certain extent? Are we not all, I mean people say 'oh I don't care how I look or what I wear' and you think well how do you choose what you wore today then, there must be some thought going into the process behind it.

M- Maybe grunge was a bringing of 'the couldn't care less attitude' to the mainstream and it made it acceptable, acceptable to be grungy but then in itself it becomes a fashion trend.

M - Let me put it to you, the weather is absolutely dismal right, if ever there was a day you would like to stay in your bed this was the one and for the first time that I can remember in a long time, and bear in mind that we are at the bottom of a pit of a recession which is deeper than I've ever seen in my entire life, even I would have to admit to say that this is the worst I have ever seen and I had to park on the top floor of the St Enoch's car park today, not second or third, it's still five weeks to go until Christmas so what does that tell you, people are still in and buying stuff, simple as that. It's not window shopping, they are not going around with their brolly just enjoying looking in the windows surely, they are not all in coffee shops and socialising, they are hardened shoppers. Consumer culture is not dead, absolutely the opposite we are right in, we are in a consumer culture I think that's it, that's what drives it.

R- Can you expand on that? Why do you think that is?

M- I think people like you and me can intellectualise this but when you really get down to it, the ordinary Joe Blogs on the street is still just buying stuff. You know just wantonly going out, spending their money, it is like going out, it is like a drug do you not think, it is a way of relieving the boredom or the depression of their life you know to a certain extent, I mean that's a bit overarching but at the end of the day I would have to say that it is a pretty dismal time at the moment with one thing or another isn't it, things are pretty dismal round about us and I think unless you have the intellectual capacity round about us, you could get into a bit of a rut. And the way to dig yourself out of a rut easily is just to shop you know and I think that the corporatism just panders to that to people's need to shop, it is very cynical. It's all aspirational, yeah. Absolutely it is umm, you often think what would life be like if we had had a this summer have been like if it had been an absolute scorcher, just glorious that say like they have in Australia or California or places like that, how much different would your life be, you know, if that brightness was around you all the time, that does make you feel more like being ostentatious, you dress up a bit more, you know on holiday, you play it up a bit more, you know the Hollywood lifestyle, so maybe in our climate here...it's quite interesting, I wonder if that's, if there is some link between that and the vintage thing, I mean the very Granny Would be Proud, now I know that is just one vintage fair but the very essence of that is wearing something that your granny would have, might have been proud of you to wear, it does have a slight sort of dour aspect to it doesn't it and yet it doesn't work out that way when you see people buying at the fairs, it really is quite uplifting stuff, you know it's not the everyday kind of granny clothes from back then, or the everyday garb of back then, people are not buying the grey suits and all the rest of it, they are buying the good stuff, the old designer stuff or the bespoke stuff or things that were brightly coloured or were ostentatious at the time, do you not think, the best of, people are going back and selecting the best of what was available at the time. It's a good point actually, it goes back to what you were saying about me being in a position to select what I sell on now, I probably do unconsciously pick out what is more intrinsically interesting from my collection, maybe favour it, is it glib to say that, but I think that probably is the case that everyone would do that. Having said

that one of my biggest sellers are just the NHS glasses from the 1970s, because they were incredibly well designed I mean there was no, there's no contradicting this, they suited everybody that puts them on, so some, I don't know who the designer was I will need to try and find that out, but whoever it was that designed those glasses really had a handle on the human face they knew, they had a real kind of eyedressing bent as it were.

R- Can you tell me more about selecting the items to sell?

M- Put it this way, I am now someone who has kicked around in the west-end most of his life, the first time was when I was a student at Glasgow University when I was in my early twenties or late teens, and then again now. In the middle period, I went away from Glasgow and didn't have much to do with it. Back then in my uni days, I was just anonymous, just invisible, no one gave a damn what I was doing, who I was or whatever, now all these years later, I come back and I've only been back for two years and the only two times I have went up and down Byres Road shopping over the summer there, I've literally been stopped two or three times by people that know me in the space of ten minutes. It is quite nice you know, it is all positive stuff you know it is all good stuff, I get really good vibes behind the whole thing so that's nice. So there is no doubt about it there is a community thing going on, I think this is what will keep the thing going, no matter what happens to the vintage that undercurrent of a vintage trend at the moment which has sparked up because of Hollywood stars and everything, I mean I saw Specsavers are jumping on the bandwagon that's not going to last obviously whereas a year ago that would have worried me, I'm not concerned in the slightest now, I think it would be a good thing if that disappeared, it would get us back to a hard-core of people who are actually into vintage and they

will always be into it you know, but I think what is being talked about by myself and a few of the people around me, what happens next and I think it is probably is based around quality, quality will always sell, it might not have the vintage tag on it, it might not have to have the vintage tag on it, it could be young designers, it could be in other words, it could develop into us moving into what will become vintage in the future. Which is all about quality, something won't last, people won't enjoy wearing it if it falls to bits, it will end up in the bin, it will be put into a charity shop and it will never exist beyond this moment but if people covet something and they keep it in their wardrobe for twenty years and they pass it onto their daughter or whatever and then they put it into a vintage fair to sell or they become a vintage seller or whatever then that thing perpetuates through time again, so if corporate companies like companies that sell Ralph Lauren glasses are not making tomorrow's vintage. who is? So depressingly, it might be the eighties, nineties and noughties were not a lot was being made that was of any intrinsic value, there was a lot of Chinese products, a lot of products were being mass marketed, mass produced, cheapened, profit margins were being raised trying to cut the cost of things in the first place, all of those endless aisle after aisle of stuff, in what are these places called, New Look and places like that you know massive great big warehouses of clothes, none of it is of any intrinsic value, that's a bit of a blanket but I think I would say 95% percent of it is of no inherent value what so ever to history so we are going to have a gap there, maybe what's going to happen now because we have had this reaction against corporatism and against people, with people valuing more what they spend their money on, people are going to start saving up and buying that bespoke piece, coveting it, cherishing it and preserving it for a future vintage, you know I mean like

our children. And I think probably that's where my heart is lying at the moment is to try to find a way in parallel with the vintage is to bring in I just thought of this today, but what is going to be future vintage, I mean it's not that I think that at the moment, something that is going to be able to be sold to people who appreciate the value like you do or I do and they don't just want to go into another optician and buy what is effectively stuff that is of no intrinsic design or value what so ever it is just mass market garbage really, you know, what a shame that we missed all those years, you know. My example of Alexander McQueen, that is an example of somebody who tried to make an effort to bring it back to where it was, for that I have kept his glasses to one side, I think that is a very, very... I think his glasses that he made, only the first catalogue because the second catalogue he had sold his name onto one of the big manufacturing companies, so the first catalogue had his name stamped on it and he had a hand in designing them I believe, I still have the first catalogue which is quite nice as well. I'm sure I am not the only one that has got them because they were still sold in thousands, they weren't just dozens, it wasn't haute couture or anything like that, they were meant for the mass market but they didn't have that valueless feel to them that things seem to have nowadays. It just bores me when people say 'what do you think about the new Armani glasses', they are interchangeable, they come out of the same design studio, and the same plant manufacturer and there are questions about whether that is now in China, and they may be, some of the best glasses in the world come out of China and Japan and places like that, but it is not what you spend £250 on spectacle frame for, do you know what I mean? You buy it because you think it is an Italian designer frame out of an Italian design house, endorsed by Ralph Lauren himself or his minions, you know what I mean? And because of that I

wouldn't buy one new because I would as if I was being cheated because I know the truth, just the same way as I wouldn't walk into a Ralph Lauren store in Ingram St, I just wouldn't humour it at all, I would always know that there was a back story that I wouldn't agree with, it's just the way of it. You can probably tell, I feel very strongly about this, almost every buying decision I make now, I do think about this inherent value thing, is what I am buying, am I making the right decision buying this thing, you know this object, is it just being mass marketed for the sake of it, or has somebody breathed something into this, is there something, life in it from somewhere, either design life or passion or something, colour or whatever it is, has someone actually thought about this? Or is it just orange is this year's colour, let's make 50 million of those, you know and then all of a sudden, magically it is this year's colour because everyone is wearing it and then they get sick of it, do you know what I mean? Anyway, sorry...

R- You spoke about your buying decisions; could you tell me more about that?

M- I suppose I am trying to read trends and judging what will sell, where the interest is at a moment in time. But I do pack my stock based on how I feel that day, I couldn't tell you what influences me really, I just follow my gut. In general, for me, it is not a nostalgia thing, not in the main, if anything the thing that puts people off a particular pair of glasses is because they remember someone wearing it at the time, so ironically if they remember their granny wearing a frame like that, they won't wear it themselves. That's quite often an excuse that I get, is that 'I have to avoid that one because'... so you go into a different era, you go into an era before that or after that, you know. So no, my client base is probably between 18 and 35 predominantly and I don't know if they have a great sense of nostalgia at that point. I am trying to

ask myself the question what is nostalgia? Do I have a sense of nostalgia? I do yes, I definitely do, but it is based on my own experience, I can now remember far enough back to be nostalgic about it, you know. And I probably do moan on about it quite a lot, 'I remember things used to be better', but I didn't probably do that when I was 26 because I didn't have a point of reference far enough back to get nostalgic about it, so I don't think it is so important. A lot of older people just won't wear vintage glasses as everyone round about them was wearing them at the time or they remember not liking wearing glasses back in the fifties and sixties themselves or when they were at school, they hated it. It is a very interesting point, speaking about my own profession, I can't talk about fashion in general, there was a real stigma against glasses and that wouldn't have been the case with shirts or blouses, ordinary items of fashion that you either liked them or you didn't like them, but you didn't have a stigma against shirts you know, you have got to remember when I started in optics it was the case, almost without fail, that every child under ten who came into the practice and was prescribed glasses and needed to wear them, went out of the practice crying because of that, now it is exactly the opposite, so if someone comes in and they don't need glasses they are going 'mummy, mummy, can I have them anyway?' or if their brother or sister do they want to wear them too, wearing them for the sake of it you know. So that's flipped on its head in twenty years you know, if you speak to people and back in the fifties and sixties you know, people absolutely detested them because presumably they made little mini versions of the adult ones, you know little cat's eyes or whatever was about, heavy set little frames, and ehh, or wee round ones you know like what has become known as the John Lennon style and people would hate that, so quite a lot of the time older people than me, don't buy

vintage glasses because they have this stigma ingrained into them from childhood, they hated them at the time and they can't get past that, you know. So that makes me wonder were these glasses ahead of their time because how come we love them so much now, you know it's not just the people that wear them that love them, other people do admit, if you wore a pair tomorrow people do comment and compliment you on them. Whereas when I was your age it would have been sniggering and nasty comments if I had worn them to university back then and all the rest of it. More or less just laughing at you behind your back, I mean I wore designer glasses at the time, contemporary designer glasses at the time, and I would get comments on the street, predominantly negative comments not ever really positive comments. You would be on the train and people would be looking at you going, 'he looks like a weirdo', it might be hard to appreciate but some of the glasses I wore back then were incredibly off the wall, even by today's standards, I pull them out now and think that would be rich for my blood now, even back then it was just way off the wall, I mean that was the 1980s remember, that was a time of experimental fashion. It was a fabulous time, I think if anything that is my favourite decade of my own experience, it was a fabulous time, so much joy behind everything, it was experimental, colour for the first time was brought right to the fore see when you think about it, at the time we all laughed but looking back on it now, I just remember it being a really light, airy, joyful, happy decade, anything went you know. There was no question about it and that came out in the design of things, particularly glasses, there is no doubt about it. It is hard to imagine that we could ever get that back again, I know we could go back to an 80s trend but I am talking about could you imagine coming out the other side of this recession and everybody just having a ball again like that? It is hard to

imagine, I mean I suppose it could happen. I hope it does. The problem is, this is nothing to do with what you are doing but let's just say, I think the problem is well globalisation is kind of the problem in an overarching sense in that everybody knows too much now. I think when people lived their lives more locally it wasn't really important what was happening round the corner, as long as your were happy within yourself and having a good time, do you know what I mean, looking after your own family or doing what needs to be done in your own corner as it were. I think now maybe people are just constantly trying to pull themselves up all the time and are constantly just getting battered down by what is happening round about them. It is all so much, there is always some bad news somewhere in the world and it is just no matter how hard you try to admit it, people must be effected by that, you know. When you are constantly bombarded with stories about war and all the rest of it, it is not healthy, I know that no matter the politics and the reasoning behind these wars are, and whatever is happening with regards natural disasters and all the rest of it and that kind of stuff (long pause). I don't know it makes me think of the phrase or the saying 'charity begins at home' I was never quite sure what that means but there is something in that somewhere, in other words if you don't look after your own corner, you are not really there to look after anyone else. It is the same wise global events and wars and stuff like that, there's not much point in fighting those wars if there is not hope back home if you know what I mean. But that is a side-line issue, sorry. I digress.

R- Going in a different direction. What do you think about the idea of the effort people make with vintage?

M- In some ways it is the same argument as with someone who wears all designer clothes, there is not much point unless people know that they are designer clothes. Hmmm, I would tend to disagree with that. I think that was exactly what was wrong when I was at university. So if we are talking two decades ago here, I would be your kind of age, I can't remember what was going on at the time or why there was a particular thought for me to wear vintage clothes but there were second-hand clothing shops in the city centre which aren't there now and I used to go across to Edinburgh especially and all the rest of it, to seek these things out, but why, why did I get involved in that? (Pause) I think that round about, there was no question about it, people were shopping at Cruise and making sure that they forgot to remove the label on the collar and things like that, I mean that might be an exaggeration but you know what I mean. You knew that they had went out and spent all that money on that, that was a very 80s/90s thing, that was a very kind of a Thatcheristic, you liked to wear your money kind of thing. There is no doubt about it that was all very false at the time. That annoyed me but looking back on it, I honestly can't remember what persuaded me to start wearing vintage clothes, other than I desired to do it so I just did it. Maybe it was because it was smarter maybe I liked the quality, I think from my own point of view back then, there probably was a genuine reason behind it, everything going on round about me was false and I didn't like that. I would say now, and I have said to quite a few people, is that I don't see that so much round about me, I think I see more of myself in the people, a genuine desire to wear what they are wearing for their own personal reasons. What would you get from that, you would get a sense of self or something or a sense of your own identity, I am not 100% sure, you would have to interview a few of them to see what is going on in

their head. There is less falseness around nowadays I think, less wearing things because of what they are and more wearing things because you want to.

R- I suppose as you say it is hard to explain the rationale.

M- The 80s and 90s were just unbelievably false decades, I liked the spirit of the 80s but there was no doubt about it that people were lying all over the place – to themselves and other people. So it was a kind of a right feeling but based on a wrong ideal at the time. You know what magic it would be if you could have captured that time and bring it to young people nowadays and give them an economy and a culture that would drive that forward. You know I think that would be the future. That would be happy days for everyone, so people could have either an affordable corporate culture, where by people would buy what they wanted to wear and display themselves how they wanted to, either by reducing the price of things or by increasing the amount of second-hand that we use or whatever. There is a balance to be created there, in other words not this constant shadow of designer shops dictating to us what should or should not be in fashion. Forget them, they should become more bespoke for people who want it, less domineering in a way. For that you would have to, I think advertising is the big problem in our society, I think it is the devil in all of this. Only the large corporate companies can afford to advertise and their messages are essentially false. It is self-serving on their part, so it goes back to what I said about how I would have known about Granny Would be Proud if someone hadn't pointed it out to me. They not only wouldn't really be able to afford to advertise but also where would they advertise as it is a transient thing, so it is not like you would be using the yellow pages. It happens every couple of weeks it is not that you spend £3000 on an advert to tell people about something that only happens once in a while.

So in some ways it is really relevant to advertise it that is why social networking has become so valuable to them. So converse to that, somewhere like the Ralph Lauren shop on Ingram St or Cruise or wherever, they need to spend about 10% of their turnover on advertising in order to stay afloat and that 10% is added directly on to the cost of the clothing. So it is almost like a self-fulfilling prophecy, the very fact that they are there means that they have to keep telling people that they are there, so because they do that it in turn keeps the cost of the product high. It is all a bit false of you ask me, I don't think that we need it. The world doesn't need designer clothes. But what it does need is things to be created of value. I think this mixture of advertising, designer clothing, big names and all the rest of it; it is feeding corporatism to the detriment of everything else. And everything else would probably be far more prevalent if it wasn't for that constant pressure from the corporate side to keep things down. If people stopped shopping in these shops tomorrow they would disappear without a trace.

R- can you tell me about what you think about the people who make the decision to shop for second-hand and/or vintage?

M- Probably to a certain extent, everyone except the lower classes has hijacked the vintage or technically the second-hand trend a bit. Charity shops are now all posh and merchandised, second-hand places are now fairs instead of jumble sales and bring and buy sales, the prices have elevated. If you can make people want something, then you can charge more for it. So now, I don't think Paddy's market is there anymore, it is gone, so that it where people would have gone in the past. But there would have been a stigma against that to the extent that I would probably have felt odd shopping at Paddy's market when I was a student. I would have went to Flip

or Armstrong's in Edinburgh, or Victorian Village or Saratoga Trunk or these places that were about at the time. Probably rather than go to Paddy's market and probably for snobbish reasons. I probably missed a whole lot of good stuff because of that thinking about it now. Maybe I preferred it to be cleaned up and presented properly to me with a price ticket on it and all the rest of it. That was probably just the culture of the time. You have to remember that I was heavily indoctrinated in that 80s culture because that is when I grew up. That was when I started to have money of my own and it was an impossible thing to avoid. I mean you talk about getting things like a party dress for your Christmas night out but back then everything just piled on top of you, it was almost a social stigma not to buy things you know back in those days. It was about throwing money around, you showed people how much you earned by what you wore and what type of car you drove or house you owned and things like that. I don't think young people have reacted against that but it isn't a backlash, so now young people will rent houses rather than buy them, they choose to go by public transport rather than drive or owning a car. I don't think that is a backlash against what I grew up with; I think that it is a consequence of the time. So they don't know, its borderline ignorance, well it is not even ignorance, they don't have the knowledge to backlash against me or the way that I grew up. So they just have evolved into this society where there is this money around, there is more competition for jobs; it's harder probably much harder now than when I left university. So they are doing what is the only available plan to them. It is not through any politics or anything and because of that they are buying second-hand not buying second-hand because they are anti-designer or anti-corporate or anything like that. It just seems like the natural thing to do, 'I will go to the charity shop and buy

something because I don't have as much money.' It is not even a comparison, they don't have enough money to go and shop in Cruise. Maybe they are valuing those purchases more, putting outfits together and accessorising, making a bigger effort because there is inherently more value in those things that they are buying because they are using what is not very much of their hard earned money to do it and they are appreciating it more. We didn't appreciate things as much in the eighties, you took it for granted that it was your right as you could afford it. It wasn't that you though oh this is an item that I should keep good but I think with the vintage purchases people are putting that in, they are buying such and such and thinking I will get that to go with it, and I will really enjoy wearing it because it really has been a labour of love to try and put the thing together in the first place. So maybe this trend, which isn't a trend, is born of the cultural and economic times we live in and the only thing that could endanger that is in the future if some kind of good times came back again. With people having this blow out – that was what the eighties were – this big blow out - it was just a reaction to years and years of austerity. All of a sudden you want to wear bright colours and have flashing lights, posh cars and parties, huge telephones, everything was big, there was a lot of other cultures about, now I think the Americans are looking to us more to a certain extent. Yeah I think it is exciting, not like in the eighties but it is a very hopeful time just now and to a certain extent just now, talking about Granny Would be Proud, people talk the way my granny used to talk they will make do and mend, they will go home and darn something that is ripped or whatever, whereas for twenty years now if you said you were going to do that people would have laughed in your face, just buy a new one, this throw away culture.

R- I think it is interesting that some of these skills, such as sewing are really becoming more prevalent again?

M- I've let things pass by, I have ripped trousers or damaged things and I have really missed them, before people may have sewn new cuffs onto a shirt or whatever and it would have done another couple of years with it you know, now you would have just chucked it in the bin and been ready for a new one. But you would miss them like old friends. If you wore the soles through on shoes, you need to go get another pair and you could never find the same ones again, looking for the same spirit of what was there in the original pair. I mean for a while if I found something that I liked I would go into the shop and I would buy three of them. But then you found actually this is really boring and you get fed up and end up giving them to the charity shop or something. You can't win! It is all about that cache. If you have three of something it is no longer special. If I could mass-produce vintage glasses, those chosen pair wouldn't feel as special to you. What is that? Is it snobbery? Who cares, whatever it is that gives you enjoyment out of wearing your glasses is fine by me. But trying to put into word what that thing is that that is about, maybe it is just a confidence or something I don't know. There is something about it; it brings you something that is a wee bit special to you. Therefore you feel more confident because of that.

R- I think there is always something special about feeling you have found something treasured and special.

M- I have a 1972 bay window VW campervan, which is at the moment getting renovated and it is prized by me. Apparently the chap says that he is going to make it a showstopper for me. So that is cherished, that counts doesn't it, 40 years old? As far as clothes are concerned I have a couple of things tucked away, pristine 1940s shirts, I haven't even brought them out recently cause it is almost like I know they are so good, that I don't even want to wear them, every time I think about wearing them I just look at them. I just love them. I just love the actual items... I have a really nice shirt that I got which I love so much, it is worn in, it is not in anyway pristine, but every time I put it on I think will I wear it today or will I save it because every day that I don't wear it is another day that it is going to last and it is getting a bit thread bare and all the rest of it. It is an old Tutol gent's shirt, do you know the brand? It is what we used to always wear to school in the seventies. It was a mini gent's shirt, white starched collar, you would have bought one in Frasers with a set, so this is one of these chaps, it is just a classic shirt and just one of those finds, and I think I got that in Edinburgh. It is funny, I still go to Edinburgh, and I have more luck in Edinburgh than I do in Glasgow despite the fact of what has been going on here. It is very hard to come by gent's clothes. I've got some of the original what everyone refers to as a granddad shirt, the shirt that becomes collarless and then you can button on a collar, I have at least one of those in its original wrapper – literally the way it came in its greaseproof brown paper with the collar manufacturer's info on it. I have never worn it. What else? I have a huge vintage record collection -60s, 70s, 80s vintage records, LPs which I love, and I still collect them to be honest till this day. As far as the glasses are concerned, I have glasses that would melt your heart; I mean literally that I just think 'oh.' I mean real kind of anorak material that I think I would never sell that in a million years I mean absolutely dreamy glasses, the things that I would if it ever became appropriate to do so, maybe I would have a glass cabinet in my own practice or if I ever have a second-hand shop or something

like that, I would just have a display of these things out almost like a museum or an archive or whatever - this is how it should be done you know? All these new designers take a look at this and see if you can emulate this past.

R- Have you always-collected things?

M- I am a hoarder yeah. I hoard things; I didn't collect things to start with. It is only recently that I have realised how much that people covet the kind of things that I just happen to hoard. But I have hoarded because I have that typical Cancerian trait of never throwing anything out, I have just always done it. I have a house full of stuff, which I didn't collect through any will, I just couldn't get rid of it, I didn't want to let it go. So here I am, I am old enough now that some of these things do mean things to other people. So I am an accidental collector. I have loads of old stuff, maybe I like old things, old books as well, I have a big collection of old trade books to do with optometry. One or two bits of old furniture as well. I love the things that I have got. It is not materialistic. Materialism the definition of that and the sort of commercial sense is what we were talking about of wearing designer clothes and not being happy if people don't realise that they are designer. That is materialism, buying things for the sake of it. I think that you can have material possessions but they are an extension of yourself almost, they are part of your personality. Which I think is a different thing all together.

R- I think it is so personal in that way and the way people describe their possessions.

M- Well it inspires me to pull out some of these glasses. I mean I love all the glasses that I sell but there are just one or two pieces that I just don't know - it is almost like a sexual feeling towards them. It is a strange thing, it is larger than life considering it

is just an item. But it inspires me because just the time when you are flagging slightly or you are thinking to yourself, you know it is hard going against the corporate machine, you know it is hard battling against this constant threat of being put out of business by people who don't have any intrinsic interest in it. Pulling these things out just makes me think, you don't have a clue what it is all about, the corporate types don't have a clue that the answer is somewhere on this product, it is where we came from and it is the reason that the industry still stands until this day. If no one had enjoyed wearing these things in the past or enjoyed wearing them then we wouldn't be here to tell the story, everyone would get contact lenses or laser eye surgery; people are constantly harking onto us about these things. I mean it is not all fashion; there is the medical aspect too. It is nice to have an antidote to that - that somehow embodied in that is the fact that a lot of people care about the items and industry.

R- It is an interesting marriage behind the science and the aesthetics?

M- There is a lot behind it. The aesthetics is always the bit that has attracted me to be honest, I mean the origins of Eyedresser, well that was like 12 years ago that I set up the Eyedresser in the city centre, but I mean I had come up with concept maybe 10 years before that. I was having a boring afternoon and I was doodling names for my own potential business and Eyedresser was one of those, I still have that bit of paper till this day. I had a logo and everything just put it all down. It was well ahead of its time as a concept, I mean of dressing your eyes, loads of people have latched onto it now. But I was just swimming against the tide in those days, getting people just to wear glasses. So I suppose the industry has changed, I was really enthusiastic about it but not all were and just sold the same frame over and over again. It was such a refreshing difference now I would say just even in the past 5 years it has changed. It is very freeing now that people are biting my hand off to come and get frames, and they want me to dress their eyes. This is fantastic, I mean my 'friends of Eyedresser' album on Facebook is my recognition of that that people are so enthusiastic and proud of the glasses I am giving them that they are willing to go online with their own pictures, that is really significant to me as it is a big deal. They are better than model pictures or anything; they are inspiring like looking at the vintage spectacles. I mean with vintage, something is working here, it is not a downward trend at the moment, there is something uncanny about despite how deep the recession gets and how much people talk about this possible ebbing in the vintage trend, the actual high street trend disappearing and it going back to like a niche market, I mean I am not getting any quieter. I mean at Granny Would be Proud it is exactly the opposite. Something is working there, I mean the social networking aspect and also there is no doubt that Granny Would be Proud are right behind me I mean it is literally they ask me to tell them if I don't want to come to a particular fair. I mean everyone else has to request a specific date or whatever. So it must work both ways and I am not stepping on anyone else's toes. I went away down to England last week, to Manchester I had a cruise around the northern quarter, there was not a pair of specs to be had.

R- That's funny that there was none. I mean I don't see a lot at any of the vintage fairs or shops?

M- I mean I stumbled across a Judy's fair in Cambridge when I was there a few week ago, it was in a church and there was nothing, nothing special, it was at the fringes of what I do, I did get some good records though, what I did notice there though was going back to the vintage thing was boxes of things like keys, a £1 each and I was

thinking I have loads of those in the house myself maybe I am sitting on a gold mine. Who would have thought that people actually want to buy old keys, like door keys and wardrobe door keys and the old keys and I was thinking is this getting a bit ridiculous now. Is this getting a bit like the jumble sale again? Are we just taking this too far? What do you think?

R- When I have seen old photos for sale, I always think that is a bit odd.

M- Old photos are a bit creepy but I must admit that I collect old postcards. I have always just collected all the old postcards that I get. But I have been keeping it for 10-15 years, I haven't look in it but there must be things in there that in a few years' time must be of some intrinsic cultural value. I don't know, it's an odd one that, see for me collecting postcards the value in that for me is collecting the postcards that you get. So it is about you, you look back on them. Just collecting them for the sake of it seems a bit odd to me. I mean I don't buy records because I think that they are inherently valuable or will be in the future, I buy the record to listen to it. If it happens to be inherently valuable in the future, that's nice, but it is not the reason why you buy it in the first place. It is coming back to that old thing, like designer clothes in the 1980s I think underpinning all of this. I think maybe the thing that you need to take away is, more than anything else that I have said, is that there is some shifting of culture of young people, I don't know enough of them to comment on this on a real psychological basis or whatever but my feeling is that people are more like me and the way that I have always been than at any time in my life. I feel more in tune with what is going on round about me, it is almost like I wish I could actually go back and personally go back to being 26 just to see what that feels like to be in amongst all these people that think the same way as me. To experience what it would

feel like cause as I said I was absolutely swimming against the tide. When I was 26 I was totally and utterly alien to the thinking of other people round about me. My profession in optics was surrounded by; it was all about people who couldn't wait to get out to buy their first Porsche or whatever. It seemed to me like there was more to life than to go out and spend £33,000 on a car that was going to be worth £24,000 the day after you bought it. Whatever floats your boat you know, it was all comparing the number of bedrooms in your houses and stuff, that seems to have disappeared a wee bit now, so maybe there are, maybe I don't mix in those circles anymore, maybe I deflect them. Do you meet people like that?

R- I suppose times have moved on but there must still be an undercurrent of that.

M- Perhaps it is more kept to those cliques now. They keep it to their group now, whereas before it was the norm and if you didn't follow it you were an oddball. Yeah definitely so I think something has happened here, I think there is still that upper layer, that upper strata, not as in class but as in financially secure who know about that. Where I stay they are building yacht marinas like no bodies business. I mean you can't buy a yacht for less than £25,000 so who has money to do this? Never mind the £3000 yacht marina fees and stuff like that. And the having time to go away and sail round the... or whatever, that used to me much more prevalent back in like the 80s. I used to mix with people that did that kind of thing, now I don't see it anymore, they are not broadcasting the fact that, I suppose there is a social stigma against that or something. I think it seems a bit too try hard in a way. Vintage and the return to prizing objects, almost seems like a counter to that.

R- I suppose it is that kind of social knowledge or being seen as trying too hard, like with cooking you don't want to be seen as trying too hard or having put too much effort in.

M- Yeah, yeah, yeah.

R- No one wants to be seen as caring too much or showy.

M- I mean it is a bit like when you went to your granny's house just to say hello it was always like get the kettle on, a bit of cake, no don't go to any bother, do you want any dinner while you're here, it won't be any bother at all. You are quite right. It is always underplayed; I think that is Scots hospitality for you that nothing is too much trouble you know what I mean. Underplaying it all the time, it is quite nice, buy yeah, I mean you yourself you always felt more special because of it. I mean it got to the point where in amongst my circles, you would be down at the pub and the first question people would ask would be how much did you pay for that then. Not what's the engine size or what made you choose that colour? It was like a badge of honour, I mean oh yes it was $\pm 25,000$ but I managed to hammer the deal down 5% or he gave me free mats or whatever, it was almost like yeah I mean I spend a lot of money on it but I am quite hardnosed myself and I did manage to hammer him down a bit which that makes me a good business man as well, so that is just like a double whammy: not only can I afford to buy it but also I am a good business man so I hammered it down and I got the free mats and that and the next thing. But there was nothing particularly special about that brand new car that you have got. There are no scratches or dents in the paint work that make it feel homely or smell inside that makes it feel like yours or anything like that. I mean we are thinking about buying a

new house, and for the first time in our lives we are thinking about buying a new house – not a new house, like a ten years old one, as compared to the 150-year-old one we live in at the moment. I mean this is a huge, huge question, I am thinking about this all the time, could I live with myself buying a house that is only ten years old? It just has no personality stamped on it of mine, nothing about it that says anything about me but for an easy life we could do what everyone else does and have less maintenance and a safer place for Abigail to be and all the rest of it. It is kind of ripping me apart this question. It is just not my style.

R- how much of yourself do you compromise?

M- Precisely, exactly or can you make up for it in other ways. Could I immerse myself in vintage clothes or whatever so that it didn't matter so much the four walls that I stay in. cause after all it doesn't really matter what four walls you stay in. ironically it doesn't matter so much in the long run so why should I care so much about my old house, it would make more sense to live in a new house. Less upkeep so I could get on with the other part of my life, which I enjoy. Socialising or whatever. So I am really asking myself all these questions just now. What's behind this, what is it that makes me tick? Why is this so important at all? Does it really matter – there are lots of people that can't afford to buy a house just now, so maybe I should be thankful that I can even do it at all. It is nice to have choices, but it can be a bit confusing.

R – I live in an old tenement so I can understand.

M- What you have you love and it is so hard to give it up. I would dread the thought of going back and starting over again now, it was scary enough back then. It's a

nightmare. Jeez, I didn't realise the time so maybe we should wind up. I just get going and don't stop or anything, you shouldn't let me keep going.

M- one other thing before we finish up, The Granny Would be Proud guys and I we would like to progress things, developments and so forth and they want me to be a part of it. The will is there to, even if it doesn't have this vintage tag on it, I think quality will be the buy word into the future so maybe it will develop from this into something inherently lasting that can last in spite of all the fads and all the rest of it. So that even if the vintage current doesn't last, the spirit lives on in how people shop and view their belongings. There is a feeling in me just now that this is very much how I felt when I started Eyedresser off 12 years ago. It is just that nobody was behind me at the time. There wasn't this social networking thing, remember Facebook didn't exist back then, nobody had even considered the possibility. It has that same excitement, then it was a personal excitement, now it is it is a lot to do with trust after doing this vintage selling for two years people trust me, they trust that I am not out to fleece them, I can make the lenses up to the required standards. A lot of trust needs to be built up before people will come to you. Especially with vintage and particularly when it comes to glasses. It is inherently something important to people, it helps them see so I like that bit to almost go without saying. I've got past the point where you know you want them; we know you need your lenses in them so they can concentrate on the fun bit, the good bit, getting themselves looking the way that they want to look. That's the fun bit for me too; the rest of it is all nuts and bolts. The mechanics of it I could do in my sleep so but it has taken 25 years for me to get to that point. I mean I just happened to be in the right place at the right time for once in my life. I think I am owed this one; just a wee moment, if it all ends tomorrow then I

had my fun. If you didn't have the motivation within you in the first place then it would die, it would die a natural death. So I mean say tomorrow all of a sudden everyone wants only the new next big thing and it wasn't old stuff, say they wanted space age or whatever, then I think that if you love your product as much as I do then you will be able to bring your expertise and your passion into whatever happens in the future if you know what I mean. It can move with you or you can move with it. I don't think it is false to say that. I suppose that the challenge is just keeping smiles on people's faces. I think it is exciting.

R- Thank you so much for your time today. It is great to be able to start to talk to people about their passions.

M- It is all so fresh, it must be hard to find written materials, it is so of the moment, the history but it has to be contemporary consumerism, it is so live, it is different today than tomorrow. I need two of me cause it is all so interesting. No matter what I try to do, it sounds a bit arrogant, but it isn't right if someone else does it, I mean people come to you because of you; they want it done because of you, more often than nought people say can I not just see Michael. So I got to the point where I do what I need to do myself and I realised that I will never be a millionaire that way but I don't care, as long as I am having fun as I go along that is all that matters. I find it easy to sleep at night because I put my rubber stamp on everything that I do and I am willing to stand by that you know. It is a nice position to be in, well it is a funny backward forward age thing, it comes from having done it for so long. If I had one regret it would be that I would have like to have enjoyed this feeling back then when I was your age you know and still have the rest to look forward to. Rather than getting to this point and all of a sudden being here but what just happened to that

twenty years there. You wonder if there is a way to skip that or if it is just you have to do the apprenticeship and leap through the hoops along the way. If I could bring one thing to this, it is personal but for me people have been critical or disparaging of what you do and sometimes people not encouraging you and whatever on my path, you have to have this belief in yourself to keep that spark going and you can't let that spark even dwindle. Don't get to the point where you have to resurrect it, keep it alive because of whatever you have got does matter, it is going to matter to enough people and I wish I had known that at the time. There are enough people in the world that are going to be into what you are doing or saying or whatever to make it worthwhile and it might not always seem like. So rather than letting things get you down, shrug it off be proactive, rather than dig yourself out of holes all the time. I wish I had known that, I have a feeling that I am on the right path and I will get there in the end. Don't know what is it, human nature to trip people up and dwell on the negative.

M- I mean you get the feeling that there are all these subgroups and things that still exist but it is just that you get the feeling that all these different strata in society are still there it is just that you find yourself funnelled into particular areas in your own particular interest sphere. Is it possible that in a world of 7 billion people, everything just exists at the same time as everything else? And there are enough people to buffer each other from each other person's groups. In other words there are 100,000 people into vintage in the UK and 100,000 into Harley Davidson, I mean in a country of 60 million people it is quite possible that those two groups are never going to bump into each other. Maybe bump into each other on the odd occasion, they can exist entirely independently of each other and almost think that the world revolves around about

them. I often sit here and wonder about that, it is possible we think that the world is all about vintage cause we are in it and the rest of the world is out there thinking what the hell are you talking about, it is only a tiny little corner of nothing there. They are getting on with their own lives and it has no bearing on this what so ever. But it doesn't seem that way, it seems that when you put on the telly people are wearing vintage clothes or glasses or when you are reading a magazine they are talking about it as a trend and when you are down the pub people are talking about clothes and more people talking about charity shops. It seems to me to be deeper than just the niche that we are in. as if it is bigger than that, bigger than Harley Davidson biking, that is an incredibly specialised market, this, vintage seems to be crossing borders and infiltrating lots of people's lives in different ways and in that way it is more culturally significant. I constantly just stop myself and think are my feet on the ground here, it feels like the world is revolving round about me cause that's the way it feels or is there something physically going on here...who knows! It will be interesting for you finding out.